# **UNITED STATES** SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

T.	$\cap$	D	<b>T</b>	O	$\mathbf{I}Z$
r	U	ĸ	M	0-	·V

**CURRENT REPORT PURSUANT TO SECTION 13 OR 15(d)** OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest event reported): May 4, 2021

# THE BEAUTY HEALTH COMPANY

(Exact name of registrant as specified in its charter)

001-39565	85-1908962
(Commission	(IRS Employer
File Number)	Identification No.)

2165 Spring Street Long Beach, CA (Address of principal executive offices)

90806 (Zip Code)

(800) 603-4996 (Registrant's telephone number, including area code)

(Former nar	Not Applicable ne or former address, if changed since last	report)	
Check the appropriate box below if the Form 8-K is intended provisions:	l to simultaneously satisfy the filing	obligation of the registrant under any of the following	
Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)			
$\square$ Soliciting material pursuant to Rule 14a-12 under the I	Exchange Act (17 CFR 240.14a-12)		
☐ Pre-commencement communications pursuant to Rule	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))		
☐ Pre-commencement communications pursuant to Rule	13e-4(c) under the Exchange Act (1	7 CFR 240.13e-4(c))	
Securities registered pursuant to Section 12(b) of the Act:			
Title of each class	Trading Symbol(s)	Name of each exchange on which registered	
Class A Common Stock, par value \$0.0001 per share	SKIN	The Nasdaq Stock Market LLC	
Warrants, each exercisable for one share of Class A Common Stock at a price of \$11.50	SKINW	The Nasdaq Stock Market LLC	
Indicate by check mark whether the registrant is an emerging chapter) or Rule 12b-2 of the Securities Exchange Act of 193	1 7	405 of the Securities Act of 1933 (§ 230.405 of this	
		Emerging growth company	

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. □

#### **Introductory Note**

On May 4, 2021 (the "Closing Date"), the registrant consummated the previously announced business combination pursuant to that certain Agreement and Plan of Merger, dated December 8, 2020, by and among Vesper Healthcare Acquisition Corp. ("Vesper Healthcare"), Hydrate Merger Sub I, Inc. ("Merger Sub I"), Hydrate Merger Sub II, LLC ("Merger Sub II"), LCP Edge Intermediate, Inc., the indirect parent of Edge Systems LLC d/b/a The HydraFacial Company ("HydraFacial"), and LCP Edge Holdco, LLC ("LCP," and, in its capacity as the stockholders' representative, the "Stockholders' Representative") (the "Merger Agreement"), which provided for: (a) the merger of Merger Sub I with and into HydraFacial, with HydraFacial continuing as the surviving corporation (the "First Merger"), and (b) immediately following the First Merger and as part of the same overall transaction as the First Merger, the merger of HydraFacial with and into Merger Sub II, with Merger Sub II continuing as the surviving entity (the "Second Merger" and, together with the First Merger, the "Mergers" and, together with the other transactions contemplated by the Merger Agreement, the "Business Combination"). As a result of the First Merger, the registrant owns 100% of the outstanding common stock of HydraFacial and each share of common stock and preferred stock of HydraFacial has been cancelled and converted into the right to receive a portion of the consideration payable in connection with the Mergers. As a result of the Second Merger, the registrant owns 100% of the outstanding interests in Merger Sub II. In connection with the closing of the Business Combination (the "Closing"), the registrant owns, directly or indirectly, 100% of the stock of HydraFacial and its subsidiaries and the stockholders of HydraFacial as of immediately prior to the effective time of the First Merger (the "HydraFacial Stockholders") hold a portion of the Class A Common Stock, par value \$0.0001 per share, of the registrant (the "Class A Stock").

In connection with the Closing, the registrant changed its name from "Vesper Healthcare Acquisition Corp." to "The Beauty Health Company." Unless the context otherwise requires, in this Current Report on Form 8-K, the "registrant" and the "Company" refer to Vesper Healthcare Acquisition Corp. prior to the Closing and to the combined company and its subsidiaries following the Closing and "HydraFacial" refers to the business of LCP Edge Intermediate, Inc. and its subsidiaries prior to the Closing and the business of the combined company and its subsidiaries following the Closing.

This Current Report on Form 8-K omits certain information that would otherwise be provided under Item 2.01, including the Management's Discussion and Analysis of Financial Condition and Results of Operations of the Company for the year ended December 31, 2020 (the "Company MD&A"), and financial statements that would otherwise be provided under (i) Item 9.01(a), including (a) the audited financial statements of LCP Edge Intermediate, Inc. as of and for the year ended December 31, 2020 (the "HydraFacial Financials") and (b) the audited financial statements of the Company and its subsidiaries as of and for the year ended December 31, 2020 (the "Company Financials") and (ii) Item 9.01(b), including the unaudited pro forma condensed combined statement of operations of the Company and HydraFacial for the year ended December 31, 2020 and the unaudited pro forma condensed combined balance sheet of the Company and HydraFacial as of December 31, 2020 (the "Pro Formas").

The Company expects to file the Company MD&A, the Company Financials, the HydraFacial Financials and the Pro Formas with an amendment to this Current Report on Form 8-K as soon as practicable. As further described in Item 4.02 herein, the Company's management and the Audit Committee of the Company's board of directors (the "Audit Committee") concluded that, in light of the Staff Statement (as defined in Item 4.02 herein), it is appropriate to restate the Company's previously issued audited financial statements as of December 31, 2020 and for the period from July 8, 2020 (inception) through December 31, 2020. The Company intends to file an amendment to its Annual Report on Form 10-K for the fiscal year ended December 31, 2020, which will include the restated audited financial statements of the Company as of December 31, 2020 and for the period from July 8, 2020 (inception) through December 31, 2020, and intends to file an amendment to this Current Report on Form 8-K, which will include the Company MD&A, the Company Financials, the HydraFacial Financials and the Pro Formas as soon as practicable thereafter.

#### Item 1.01 Entry into a Material Definitive Agreement.

#### **Registration Rights Agreement**

On the Closing Date, pursuant to the terms of the Merger Agreement, the Company entered into that certain Amended and Restated Registration Rights Agreement (the "Registration Rights Agreement") with BLS Investor Group LLC (the "Sponsor") and the HydraFacial Stockholders.

Pursuant to the terms of the Registration Rights Agreement, (i) any outstanding share of Class A Stock or any other equity security (including the warrants held by the Sponsor that were issued to the Sponsor on October 2, 2020, each of which is exercisable for one share of Class A Stock (the "Private Placement Warrants") and including shares of Class A Stock issued or issuable upon the exercise of any other equity security) of the Company held by a the Sponsor or the HydraFacial Stockholders (together, the "Restricted Stockholders") as of the date of the Registration Rights Agreement or thereafter acquired by a Restricted Stockholder (including the shares of Class A Stock issued upon conversion of the Class B Common Stock, par value \$0.0001 per share, of the registrant (the "Class B Stock" and, together with the Class A Stock, the "Common Stock") and upon exercise of any Private Placement Warrants) and shares of Class A Stock issued or issuable as earn-out shares to the HydraFacial Stockholders and (ii) any other equity security of the Company issued or issuable with respect to any such share of Common Stock by way of a stock dividend or stock split or in connection with a combination of shares, recapitalization, merger, consolidation or other reorganization or otherwise will be entitled to registration rights.

The Registration Rights Agreement provides that the Company will, within 60 days after the consummation of the transactions contemplated by the Merger Agreement, file with the Securities and Exchange Commission (the "SEC") a shelf registration statement registering the resale of the shares of Common Stock held by the Restricted Stockholders and will use its reasonable best efforts to have such registration statement declared effective as soon as practicable after the filing thereof, but in no event later than 60 days following the filing deadline. The HydraFacial Stockholders are entitled to make up to an aggregate of two demands for registration, excluding short form demands, that the Company register shares of Common Stock held by these parties. In addition, the Restricted Stockholders have certain "piggy-back" registration rights. The Company will bear the expenses incurred in connection with the filing of any registration statements filed pursuant to the terms of the Registration Rights Agreement. The Company and the Restricted Stockholders agree in the Registration Rights Agreement to provide customary indemnification in connection with any offerings of Common Stock effected pursuant to the terms of the Registration Rights Agreement.

The foregoing description of the Registration Rights Agreement is not complete and is qualified in its entirety by reference to the complete text of the Registration Rights Agreement, a copy of which is filed as Exhibit 10.2 hereto and is incorporated herein by reference.

#### **Investor Rights Agreement**

On the Closing Date, pursuant to the terms of the Merger Agreement, the Company and LCP entered into that certain Investor Rights Agreement (the "Investor Rights Agreement"). Pursuant to the Investor Rights Agreement, LCP will have the right to designate a number of directors for appointment or election to the Company's board of directors as follows: (i) one director for so long as LCP holds at least 10% of the outstanding Class A Stock, (ii) two directors for so long as LCP holds at least 15% of the outstanding Class A Stock, and (iii) three directors for so long as LCP holds at least 40% of the outstanding Class A Stock. Pursuant to the Investor Rights Agreement, for so long as LCP holds at least 10% of the outstanding Class A Stock, LCP will be entitled to have at least one of its designees represented on the compensation committee and nominating committee and corporate governance committee of the Company's board of directors.

The foregoing description of the Investor Rights Agreement is not complete and is qualified in its entirety by reference to the complete text of the Investor Rights Agreement, a copy of which is filed as Exhibit 10.3 hereto and is incorporated herein by reference.

### Amended and Restated Management Services Agreement

On the Closing Date, the Company, its subsidiary, Edge Systems LLC, and the Linden Manager (as defined below) entered into an Amended and Restated Management Services Agreement pursuant to which the Linden Manager may continue to provide advisory services at the request of the Company related to mergers and acquisitions. As consideration for such services, the Company will pay a fee, equal to 1% of enterprise value, to the Linden Manager upon the consummation of any such transaction.

The foregoing description of the Amended and Restated Management Services Agreement is not complete and is qualified in its entirety by reference to the complete text of the Amended and Restated Management Services Agreement, a copy of which is filed as Exhibit 10.14 hereto and is incorporated herein by reference.

#### **Indemnity Agreements**

In connection with the Closing, the Company entered into indemnity agreements with each of its directors and executive officers and certain other officers of the Company. Each indemnity agreement provides for indemnification and advancement by the Company of certain expenses and costs relating to claims, suits or proceedings arising from service to the Company or, at its request, service to other entities, as officers or directors to the maximum extent permitted by applicable law. The foregoing description of the indemnity agreements does not purport to be complete and is qualified in its entirety by the terms and conditions of the indemnity agreements, a form of which is attached hereto as Exhibit 10.13 and is incorporated herein by reference.

#### Item 2.01 Completion of Acquisition or Disposition of Assets.

The disclosure set forth in the "Introductory Note" above is incorporated into this Item 2.01 by reference. On April 29, 2021, the Business Combination was approved by the Company's stockholders at a special meeting thereof (the "Special Meeting"), held in lieu of the 2021 annual meeting of the Company's stockholders.

Pursuant to the terms of the Merger Agreement and customary adjustments set forth therein, the aggregate merger consideration paid to the HydraFacial Stockholders in connection with the Business Combination was approximately \$975,000,000 less HydraFacial's net indebtedness as of the Closing Date, and subject to further adjustments for transaction expenses, and net working capital relative to a target. The merger consideration included both cash consideration and consideration in the form of newly issued Class A Stock. The aggregate cash consideration paid to the HydraFacial Stockholders at the Closing was approximately \$368 million, consisting of the Company's cash and cash equivalents as of the Closing (including proceeds of \$350 million from the Company's private placement of an aggregate of 35,000,000 shares of Class A Stock (the "Private Placement") with a limited number of accredited investors (as defined by Rule 501 of Regulation D) without any form of general solicitation or general advertising pursuant to Section 4(a)(2) of the Securities Act of 1933, as amended (the "Securities Act"), and approximately \$433 million of cash available to the Company from the trust account that held the proceeds from the Company's initial public offering (the "Trust Account")) after giving effect to income and franchise taxes payable in respect of interest income earned in the Trust Account and redemptions that were elected by the Company's public stockholders, *minus* approximately \$224 million used to repay HydraFacial's outstanding indebtedness at the Closing, *minus* approximately \$94 million of transaction expenses of HydraFacial and the Company, *minus* \$100 million. The remainder of the consideration paid to the HydraFacial Stockholders consisted of 35,501,743 newly issued shares of Class A Stock (the "Stock Consideration").

The foregoing consideration paid to the HydraFacial Stockholders may be further increased by up to \$75.0 million payable as earn-out shares of Class A Stock upon the completion of certain acquisitions within one year of Closing pursuant to the terms of the Merger Agreement.

All outstanding shares of Class B Stock were automatically converted into shares of Class A Stock on a one-for-one basis at the Closing and will continue to be subject to the transfer restrictions applicable to such shares of Class B Stock.

The material terms and conditions of the Merger Agreement are described in greater detail in the section of the Company's definitive proxy statement filed with the SEC on April 7, 2021 (the "Proxy Statement") entitled "Proposal No. 1 – Approval of the Business Combination – The Merger Agreement" beginning on page 132, which information is incorporated herein by reference.

#### FORM 10 INFORMATION

Prior to the Closing, the Company was a shell company (as defined in Rule 12b-2 of the Securities Exchange Act of 1934, as amended (the "Exchange Act")) with no operations, formed as a vehicle to effect a business combination with one or more operating businesses. After the Closing, the Company became a holding company whose only assets consist of equity interests in HydraFacial.

#### CAUTIONARY NOTE REGARDING FORWARD-LOOKING STATEMENTS

This Current Report on Form 8-K, including the information incorporated herein by reference, contains "forward-looking statements" within the meaning of Section 27A of the Securities Act and Section 21E of the Securities Exchange Act of 1934, as amended. These forward-looking statements relate to expectations for future financial performance, business strategies or expectations for the Company's business. Specifically, forward-looking statements may include statements relating to:

• the benefits of the Business Combination;

- the Company's ability to maintain its listing on the Nasdaq Stock Market ("Nasdaq") following the Business Combination;
- the Company's ability to raise financing or complete acquisitions in the future;
- the Company's success in retaining or recruiting, or changes required in, its officers, key employees or directors following the Business Combination;
- the future financial performance of the Company following the Business Combination;
- intense competition and competitive pressures from other companies in the industry in which the Company will operate;
- the business, operations and financial performance of the Company, including market conditions and global and economic factors beyond the Company's control;
- the impact of COVID-19 and related changes in base interest rates and significant market volatility on the Company's business, our industry and the global economy;
- · the effect of legal, tax and regulatory changes; and
- other statements preceded by, followed by or that include the words "anticipate," "believe," "continue," "could," "estimate," "expect," "intends," "may," "might," "plan," "possible," "potential," "predict," "project," "should," "would" and similar expressions.

These forward-looking statements are based on information available as of the date of this Current Report on Form 8-K and management's current expectations, forecasts and assumptions, and involve a number of judgments, risks and uncertainties. Accordingly, forward-looking statements should not be relied upon as representing the Company's views as of any subsequent date. The Company does not undertake any obligation to update forward-looking statements to reflect events or circumstances after the date they were made, whether as a result of new information, future events or otherwise, except as may be required under applicable securities laws.

As a result of a number of known and unknown risks and uncertainties, the Company's actual results or performance may be materially different from those expressed or implied by these forward-looking statements. Some factors that could cause actual results to differ include:

- the inability to maintain the Company's listing on Nasdaq following the Business Combination;
- the risk that the Business Combination disrupts current plans and operations as a result of the announcement and consummation of the transactions:
- the ability to recognize the anticipated benefits of the Business Combination, which may be affected by, among other things, competition and the ability of the combined business to grow and manage growth profitably;
- costs related to the Business Combination;
- the outcome of any legal proceedings that may be instituted against the Company following consummation of the Business Combination;
- changes in applicable laws or regulations;
- the possibility that the Company may be adversely affected by other economic, business and/or competitive factors;
- the impact of the continuing COVID-19 pandemic on the Company's business; and
- other risks and uncertainties indicated or incorporated by reference in this Current Report on Form 8-K, including those set forth in the section of the Proxy Statement entitled "Risk Factors" beginning on page 58.

#### **Business and Properties**

The information set forth in the section of the Proxy Statement entitled "Information About HydraFacial" beginning on page 213, including the information regarding the properties used in the business included in the subsection thereof entitled "—Properties" on page 220, and in the section of the Proxy Statement entitled "Information About the Company" beginning on page 198 is incorporated herein by reference.

The Company's principal executive office is located at 2165 Spring Street, Long Beach, CA 90806.

#### **Risk Factors**

The information set forth in the section of the Proxy Statement entitled "Risk Factors" beginning on page 58 is incorporated herein by reference.

#### Selected Consolidated Historical Financial and Other Information

The information set forth in the section of the Proxy Statement entitled "Selected Historical Consolidated Financial Information of HydraFacial" beginning on page 46 is incorporated herein by reference.

The unaudited pro forma condensed combined balance sheet and statements of operations as of December 31, 2020 and for the year ended December 31, 2020 will be filed as Exhibit 99.4 to an amendment to this Current Report on Form 8-K as described in the Introductory Note to this Current Report on Form 8-K.

# Management's Discussion and Analysis of Financial Condition and Results of Operations and Quantitative and Qualitative Disclosures About Market Risk

The information set forth in the section of the Proxy Statement entitled "HydraFacial's Management's Discussion and Analysis of Financial Condition and Results of Operations" beginning on page 223, including the information in the subsection thereof entitled "—Quantitative and Qualitative Disclosures About Market Risks" beginning on page 239, is incorporated herein by reference. Management's Discussion and Analysis of Financial Condition and Results of Operations for The Beauty Health Company for the year ended December 31, 2020 will be filed as Exhibit 99.2 to an amendment to this Current Report on Form 8-K as described in the Introductory Note to this Current Report on Form 8-K.

#### Security Ownership of Certain Beneficial Owners and Management

The following table sets forth information known to the Company regarding beneficial ownership of shares of the Company's common stock as of the Closing Date by:

- each person known by the Company to be the beneficial owner of more than 5% of the Company's outstanding common stock;
- each of the Company's named executive officers and directors; and
- all executive officers and directors as a group.

Beneficial ownership is determined according to the rules of the SEC, which generally provide that a person has beneficial ownership of a security if he, she or it possesses sole or shared voting or investment power over that security, including options, warrants and certain other derivative securities that are currently exercisable or will become exercisable within 60 days.

The percentage of beneficial ownership is based on 125,329,053 shares of Company common stock issued and outstanding as of the Closing Date, which calculation includes all shares of Class A Stock issued and outstanding as of the Closing Date, the only outstanding class of the Company's common stock following the Business Combination. All shares of Class B Stock were converted into shares of Class A Stock or cancelled in connection with the Closing.

Unless otherwise indicated, the business address of each of the entities, directors and executives in this table is 2165 Spring Street, Long Beach, CA 90806. Unless otherwise indicated and subject to community property laws and similar laws, the Company believes that all parties named in the table below have sole voting and investment power with respect to all shares of common stock beneficially owned by them.

Name and Address of Beneficial Owners	Number of Shares	Ownership Percentage (%)
BLS Investor Group LLC (1)(2)	11,500,000	9.2
LCP Edge Holdco, LLC (3)	33,356,338	26.6
Brenton L. Saunders (1)(2)	11,500,000	9.2
Clinton E. Carnell	_	_
Michael D. Capellas	_	_
Dr. Julius Few	_	_
Michelle Kerrick	_	_
Brian Miller	_	_
Doug Schillinger	_	_
Liyuan Woo	_	_
All directors and executive officers as a group (8 individuals)	11,500,000	9.2

- (1) BLS Investor Group LLC is the record holder of the shares reported herein. Each of Brenton L. Saunders, Michael D. Capellas and Dr. Julius Few are among the members of BLS Investor Group LLC. Brenton L. Saunders is the sole managing member of BLS Investor Group LLC. Mr. Saunders has sole voting and investment discretion with respect to the common stock held of record by BLS Investor Group LLC. Mr. Capellas and Mr. Few disclaim any beneficial ownership of any shares held by BLS Investor Group LLC except to the extent of his or her ultimate pecuniary interest therein.
- (2) The business address of each of BLS Investor Group LLC and Brenton L. Saunders is 1819 West Avenue, Bay 2, Miami Beach, FL 33139.
- (3) The shares held by LCP Edge Holdco LLC ("LCP Edge Holdco") may be deemed to be beneficially owned by Linden Capital III LLC ("Linden Capital III"), the general partner of Linden Manager III LP ("Linden Manager"). Linden Manager is the general partner of both Linden Capital Partners III LP ("Linden Capital Partners III-A"), which are the controlling unitholders of LCP Edge Holdco. As the members of a limited partner committee of Linden Capital III that has the power to vote or dispose of the shares directly held by LCP Edge Holdco, Brian Miller and Anthony Davis may be deemed to have shared voting and investment power over such shares. Each of Linden Capital III, Linden Manager, Linden Capital Partners III, Linden Capital Partners III-A, Mr. Miller and Mr. Davis hereby disclaim any beneficial ownership of any shares held by LCP Edge Holdco except to the extent of any pecuniary interest therein. The address for the Linden entities and persons is 150 North Riverside Plaza, Suite 5100, Chicago, Illinois 60606.

### **Directors and Executive Officers**

Information with respect to the Company's directors and executive officers immediately after the Closing, including biographical information regarding these individuals, is set forth in the Proxy Statement in the section entitled "Management After the Business Combination" beginning on page 250, which information is incorporated herein by reference.

Each of Ms. Michelle Kerrick and Messrs. Brenton L. Saunders, Clinton E. Carnell, Michael D. Capellas, Dr. Julius Few, Doug Schillinger and Brian Miller were elected by the Company's stockholders at the Special Meeting to serve as directors of the Company, effective upon consummation of the Business Combination, at which time the size of the board was seven members. Ms. Kerrick and Mr. Carnell were elected to serve as Class I directors with a term expiring at the Company's 2022 annual meeting of stockholders; Messrs. Capellas, Few and Miller were elected to serve as Class II directors with a term expiring at the Company's 2023 annual meeting of stockholders; and Messrs. Saunders and Schillinger were elected to serve as Class III directors with a term expiring at the Company's 2024 annual meeting of stockholders. Mr. Saunders was appointed to serve as the executive chairman of the board of directors.

Ms. Kerrick and Messrs. Capellas and Schillinger will serve as members of the Audit Committee, with Ms. Kerrick serving as its chairman. Messrs. Schillinger and Miller will serve as members of the compensation committee, with Mr. Schillinger serving as its chairman. Information with respect to the Audit Committee and compensation committee is set forth in the Proxy Statement in the section entitled "Information About the Company – Management – Committees of the Board of Directors" beginning on page 251, which information is incorporated herein by reference.

In addition, on May 4, 2021, the board of directors formed the nominating and corporate governance committee. Ms. Kerrick and Messrs. Capellas, Few and Miller will serve as members of the nominating and corporate governance committee, with Mr. Capellas serving as its chairman. The purpose of the nominating and corporate governance committee is to identify individuals qualified to serve as directors of the Company and on committees of the Company's board of directors, to recommend to the Company's board of directors the director nominees for election at the next annual meeting of shareholders, to advise the Company's board of directors with respect to the composition of the board of directors, procedures and committees, to develop and recommend to the Company's board of directors a set of corporate governance guidelines applicable to the Company and to oversee the evaluation of the Company's board of directors and the Company's management. The nominating and corporate governance committee has a written charter that sets forth the committee's purpose and responsibilities, which, in addition to the items listed above, include:

- identifying, recruiting and, if appropriate, interviewing candidates to fill positions on the Company's board of directors, including persons suggested by shareholders or others;
- reviewing the background and qualifications of individuals being considered as director candidates;
- recommending to the Company's board of directors the director nominees for election by the Company's shareholders or appointment by the Company's board of directors;
- reviewing the suitability for continued service as a director of each member of the board of directors when his or her term expires and in certain other circumstances;
- reviewing annually with the Company's board of directors the composition of the Company's board of directors as a whole and to
  recommend, if necessary, measures to be taken so that the Company's board of directors reflect the appropriate balance of knowledge,
  experience, skills, expertise and diversity required for the Company's board of directors as a whole and contains at least the minimum
  number of independent directors required by Nasdaq;
- monitoring the functioning of the committees of the Company's board of directors and to make recommendations for any changes;
- reviewing annually committee size, membership and composition, including chairpersonships, and recommended any changes to the Company's board of directors for approval;
- developing and recommending to the Company's board of directors a set of corporate governance guidelines for the Company;
- review periodically, and at least annually, the corporate governance guidelines adopted by the Company's board of directors to assure that they are appropriate for the Company; and
- evaluating its performance and submitting any recommended changes to the board for its consideration.

The nominating and corporate governance committee has the authority to retain advisors as the committee deems appropriate.

In connection with the consummation of the Business Combination, on the Closing Date, Clinton E. Carnell was appointed to serve as the Company's Chief Executive Officer, Liyuan Woo was appointed to serve as the Company's Chief Financial Officer and Daniel Watson was appointed to serve as the Company's Executive Vice President Americas Sales.

In connection with the Closing, each of the Company's executive officers prior to the Closing resigned from his or her respective position as an executive officer of the Company, in each case effective as of the effective time of the First Merger.

#### Compensation Committee Interlocks and Insider Participation

None of the Company's executive officers currently serve, or in the past year have served, as members of the board of directors or compensation committee of any entity that has one or more executive officers serving on the Company's board of directors.

#### **Executive Compensation**

The compensation of the Company's named executive officers before the consummation of the Business Combination is described in the Proxy Statement in the section entitled "Executive Compensation" beginning on page 244, which information is incorporated herein by reference.

On April 29, 2021, the stockholders of the Company approved The Beauty Health Company 2021 Incentive Award Plan (the "2021 Plan"), which become effective upon the Closing. The material terms of the 2021 Plan are described in the Proxy Statement in the section entitled "Proposal No. 6 – The Incentive Award Plan Proposal" beginning on page 187, which information is incorporated herein by reference.

#### **Director Compensation**

In connection with the Business Combination, the Company adopted a new board of directors compensation program which is designed to provide competitive compensation necessary to attract and retain high quality non-employee directors and to encourage ownership of Company stock to further align their interests with those of our stockholders. The new program will provide the following compensation for non-employee directors going forward:

- an annual cash retainer of \$45,000 for each non-employee director;
- an annual cash retainer of \$10,000 for the chair of the audit committee, \$7,500 for the chair of the compensation committee and \$5,000 for the chair of the nominating and corporate governance committee;
- an annual cash retainer of \$10,000 for each member of the audit committee; \$7,500 for each member of the compensation committee and \$5,000 for each member of the nominating and corporate governance committee;
- an annual cash retainer of \$25,000 for the lead director, if applicable; and
- an equity retainer with grant date fair value of \$135,000 upon such director's election to office, payable in the form of restricted stock units, granted in connection with each annual meeting of stockholders that vests on the earlier of the one-year anniversary of the grant and the next annual meeting of stockholders, subject to the director's continuous service. The first equity retainer grant (with grant date fair value of \$135,000) is expected to be made on or promptly following the filing of the Registration Statement on Form S-8 with respect to shares reserved under the 2021 Plan, and vest at the first regularly scheduled annual meeting of stockholders of the Company, subject to the director's continuous service.

All cash retainers will be payable quarterly in arrears.

Under the 2021 Plan, in a single fiscal year, a non-employee director may not be granted awards for such individual's service on the board of directors of the Company having a value that, together with cash fees paid or other compensation provided to such individual for service on the board of directors of the Company, exceed \$500,000.

As executive chairman of the board of directors, Mr. Saunders will not be eligible to receive the abovementioned retainers. Further, Mr. Saunders has declined to receive any cash compensation for services rendered to the Company, and will instead receive compensation solely in the form of equity awards. In connection with his appointment as executive chairman, Mr. Saunders received an initial grant of stock options to purchase 1,860,000 shares of Company common stock, vesting over four years, with 25% of the shares vesting on each of the first four anniversaries of the Closing Date, and will receive a grant of performance-based restricted stock units covering 60,000 shares of Company common stock effective upon the filing of the Registration Statement on Form S-8 with respect to shares reserved under the 2021 Plan. The award of performance-based restricted stock units may be earned over a four-year performance period based on the achievement of performance goals related to the Company's stock price and Mr. Saunders' continued employment with the Company through the end of the performance period.

#### Certain Relationships and Related Transactions

The information set forth in the section of the Proxy Statement entitled "Certain Relationships and Related Transactions" beginning on page 273 and the information set forth under the headings "Registration Rights Agreement" and "Investor Rights Agreement" in Item 1.01 of this Current Report on Form 8-K is incorporated herein by reference.

### Director Independence

Nasdaq listing standards require that a majority of the members of the board of directors be independent. An "independent director" is defined generally as a person other than an officer or employee of a company or its subsidiaries or any other individual having a relationship which, in the opinion of the board of directors of such company, would interfere with the director's exercise of independent judgment in carrying out the responsibilities of a director.

The Company currently has five "independent directors" as defined in the Nasdaq listing standards and applicable SEC rules and as determined by the board of directors using its business judgment: Ms. Kerrick and Messrs. Capellas, Few, Miller and Schillinger.

#### Legal Proceedings

Information about legal proceedings is set forth in the section of the Proxy Statement entitled "Information About HydraFacial – Legal Proceedings" beginning on page 222, which information is incorporated herein by reference.

#### Market Price of and Dividends on the Registrant's Common Equity and Related Stockholder Matters

Information about the market price, number of stockholders and dividends for the Company's securities is set forth in the section of the Proxy Statement entitled "Price Range of Securities and Dividends" on page 279, which information is incorporated herein by reference. Additional information regarding holders of the Company's securities is set forth under "Description of the Company's Securities" below.

Following the Closing, on May 6, 2021, the Class A Stock and publicly traded warrants were listed on Nasdaq under the symbols "SKIN" and "SKINW," respectively. The warrants may be delisted from Nasdaq if there is not a sufficient number of round lot holders within 15 days of the consummation of the Business Combination and if delisted, may be quoted on the OTC Bulletin Board or OTC Pink, an inter-dealer automated quotation system for equity securities that is not a national securities exchange. The public units of the Company automatically separated into the component securities upon consummation of the Business Combination and, as a result, no longer trade as a separate security.

#### **Recent Sales of Unregistered Securities**

Information regarding unregistered sales of the Company's securities is set forth in: Part II, Item 5 of the Company's Annual Report on Form 10-K filed with the SEC on March 18, 2021 and Item 3.02 of the Company's Current Report on Form 8-K filed with the SEC on December 9, 2020.

The description of the Stock Consideration under Item 2.01 of this Current Report on Form 8-K is incorporated herein by reference. The issuances of the shares of the Class A Stock issued as Stock Consideration and in the Private Placement were not registered under the Securities Act in reliance on the exemption from registration provided by Section 4(a)(2) of the Securities Act and/or Regulation D promulgated thereunder as a transaction by an issuer not involving a public offering without any form of general solicitation or general advertising.

#### Description of the Company's Securities

Information regarding the Class A Stock and the Company's warrants is included in the section of the Proxy Statement entitled "Description of Securities" beginning on page 254, which information is incorporated herein by reference.

The Company has authorized 321,000,000 shares of capital stock, consisting of (a) 320,000,000 shares of Class A Stock and (b) 1,000,000 shares of preferred stock, par value \$0.0001 per share. The outstanding shares of the Company's common stock are fully paid and non-assessable. As of the Closing Date, there were 125,329,053 shares of Class A Stock outstanding held of record by approximately 42 stockholders, no shares of preferred stock outstanding, and warrants to purchase 24,666,638 shares of Class A Stock outstanding held of record by approximately 2 holders. Such holder numbers do not include Depository Trust Company participants or beneficial owners holding shares through nominee names.

#### **Indemnification of Directors and Officers**

The information set forth in Item 1.01 of this Current Report on Form 8-K is incorporated herein by reference.

Further information about the indemnification of the Company's directors and officers is set forth in the section of the Proxy Statement entitled "Information About the Company – Limitation on Liability and Indemnification of Officers and Directors" on page 206, which information is incorporated herein by reference.

### Changes in and Disagreements with Accountants on Accounting and Financial Disclosure

Not applicable.

#### Financial Statements, Supplementary Data and Exhibits

The information set forth in Item 9.01 of this Current Report on Form 8-K is incorporated herein by reference.

#### Item 3.02 Unregistered Sales of Equity Securities.

The description of the Stock Consideration set forth in Item 2.01 of this Current Report on Form 8-K is incorporated herein by reference. The issuances of the shares of Class A Stock issued as Stock Consideration and in the Private Placement were not registered under the Securities Act in reliance on the exemption from registration provided by Section 4(a)(2) of the Securities Act and/or Regulation D promulgated thereunder as a transaction by an issuer not involving a public offering without any form of general solicitation or general advertising.

#### Item 3.03 Material Modification to Rights of Security Holders.

On the Closing Date, the Company filed the Second Amended and Restated Certificate of Incorporation of the Company (the "A&R Certificate") with the Secretary of State of the State of Delaware. The material terms of the A&R Certificate and the general effect upon the rights of holders of the Company's capital stock are described in the sections of the Proxy Statement entitled "Proposal No. 3 – Approval of the Second Amended and Restated Certificate of Incorporation" and "Proposal No. 4 – Approval of Certain Governance Provisions in the Second Amended and Restated Certificate of Incorporation" beginning on pages 177 and 181, respectively, which information is incorporated herein by reference. A copy of the A&R Certificate is filed as Exhibit 3.1 to this Current Report on Form 8-K and is incorporated herein by reference.

In addition, upon the Closing, pursuant to the terms of the Merger Agreement, the Company amended and restated its bylaws. A copy of the Company's Amended and Restated Bylaws is filed as Exhibit 3.2 to this Current Report on Form 8-K and is incorporated herein by reference.

#### Item 4.02 Non-Reliance on Previously Issued Financial Statements or a Related Audit Report or Completed Interim Review.

On April 12, 2021, the Staff of the SEC released the Staff Statement on Accounting and Reporting Considerations for Warrants Issued by Special Purpose Acquisition Companies ("SPACs") (the "Staff Statement"). The Staff Statement sets forth the conclusion of the SEC's Office of the Chief Accountant that certain provisions included in the warrant agreements entered into by many SPACs, including the Company, require such warrants to be accounted for as liabilities measured at fair value, rather than as equity securities, with changes in fair value during each financial reporting period reported in earnings. The Company has previously classified its private placement warrants and public warrants as equity (for a full description of the Company's private placement warrants and public warrants (collectively, the "warrants"), refer to the registration statement on Form S-1 (File No. 333-248717), filed in connection with the Company's initial public offering, declared effective by the SEC on September 30, 2020.

As previously disclosed in the Company's Current Report on Form 8-K filed with the SEC on April 26, 2021, as amended on April 27, 2021, the Company undertook an evaluation of the Staff Statement with respect to the Company's accounting treatment of the warrants and determined on a preliminary basis that such guidance would result in the warrants being classified as liabilities on the balance sheet with the mark to market change in fair value reflected in the statement of operations.

On May 9, 2021, the Company's management and the Audit Committee concluded that, in light of the Staff Statement, it is appropriate to restate the Company's previously issued audited financial statements as of December 31, 2020 and for the period from July 8, 2020 (inception) through December 31, 2020 (the "Affected Periods"). Considering such restatement, such audited financial statements should no longer be relied upon. Similarly, any previously furnished or filed reports including the audited balance sheet filed with the SEC on Form 8-K on October 8, 2020, related earnings releases, investor presentations or similar communications of the Company describing the Company's financial results for the Affected Periods should no longer be relied upon.

The Company intends to file an amendment to its Annual Report on Form 10-K for the fiscal year ended December 31, 2020, which will include the restated financials of the Company as of December 31, 2020 and for the period from July 8, 2020 (inception) through December 31, 2020 (the "Amended 10-K"). The adjustments to the financial statement items for the Affected Periods will be set forth in the financial statements included in the Amended 10-K, including further describing the restatement and its impact on previously reported amounts.

The Audit Committee and the Company's management have discussed the matters disclosed pursuant to this Item 4.02 with Marcum LLP, the Company's independent registered public accounting firm.

#### Item 5.01 Changes in Control of the Registrant.

The information set forth in the "Introductory Note" and in Item 2.01 of this Current Report on Form 8-K is incorporated herein by reference.

Item 5.02 Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers; Compensatory Arrangements of Certain Officers.

#### **Directors and Executive Officers**

Information with respect to the Company's directors and executive officers before and after the consummation of the Business Combination is set forth in the Proxy Statement in the sections entitled "Information About the Company – Management – Directors and Officers" beginning on page 200 and "Management After the Business Combination – Management and Board of Directors" beginning on page 250 and "Proposal No. 5 – Election of Directors to the Board of Directors" beginning on page 184, which are incorporated herein by reference.

The information regarding the Company's officers and directors set forth under the headings "Directors and Executive Officers" and "Executive Compensation" in Item 2.01 of this Current Report on Form 8-K is incorporated herein by reference.

#### Chief Executive Officer and Chief Financial Officer Employment Agreements

The Company has entered into an employment agreement with Clinton E. Carnell to serve as the Chief Executive Officer of the Company (the "CEO Agreement") and with Liyuan Woo to serve as the Chief Financial Officer of the Company (the "CFO Agreement" and, together with the CEO Agreement, the "Executive Agreements"), both effective as of and contingent upon the consummation of the Business Combination. The Executive Agreements provide for an annual compensation package consisting of a base salary of \$675,000 and \$415,000 for Mr. Carnell and Ms. Woo, respectively, a target bonus opportunity of 100% and 60% of base salary for Mr. Carnell and Ms. Woo, respectively, and eligibility for annual long-term incentive awards pursuant to the 2021 Plan beginning in 2022, with the form of such award and the value of such award determined by the compensation committee of the Company's board of directors.

Pursuant to the Executive Agreements, Mr. Carnell and Ms. Woo will receive the following one-time equity awards, pursuant to the 2021 Plan: (1) an award of stock options to purchase 3,100,000 shares of Company common stock and 744,000 shares of Company common stock, for Mr. Carnell and Ms. Woo, respectively, vesting over four years, with 25% of the shares vesting on each of the first four anniversaries of the Closing Date, subject to Mr. Carnell and Ms. Woo's continued employment with the Company through the applicable vesting date (the "Executive Option Awards") and (2) an award of performance-based restricted stock units covering 100,000 shares of Company common stock and 50,000 shares of Company common stock for Mr. Carnell and Ms. Woo, respectively, to be effective upon the filing of the Registration Statement on Form S-8 with respect to shares reserved under the 2021 Plan, which may be earned over a four-year performance period based on achievement of performance goals related to the Company's stock price and Mr. Carnell and Ms. Woo's continued employment with the Company through the end of the performance period (the "PBRSU Awards").

If Mr. Carnell or Ms. Woo's employment is terminated without cause (as defined in the Executive Agreements) or if Mr. Carnell or Ms. Woo resign for good reason (as defined in the Executive Agreements), Mr. Carnell and Ms. Woo will be entitled to the following: (1) cash severance equal to 18 months base salary, (2) a prorated target bonus, and (3) reimbursement of the employer portion of COBRA premium payments for 18 months (the "Executive Severance Benefits"), in each case subject to Mr. Carnell and Ms. Woo's timely execution of an irrevocable release of claims against the Company (the "Release Requirement"). If Mr. Carnell or Ms. Woo's employment is terminated without cause or if Mr. Carnell or Ms. Woo resign for good reason in connection with a change in control (as defined in the Executive Agreements), Mr. Carnell and Ms. Woo will be entitled to the Executive Severance Benefits, along with a cash payment equal to one and one-half (1.5) times Mr. Carnell's or Ms. Woo's target bonus, respectively, in each case, subject to the Release Requirement. The Executive Agreements also contain standard restrictive covenants and confidentiality and invention assignment provisions, in an agreement attached thereto.

The foregoing summary description of the Executive Agreements does not purport to be complete and is qualified in its entirety by the full text of the Executive Agreements, which are attached hereto as Exhibit 10.6 and Exhibit 10.7 to this Current Report on Form 8-K.

The foregoing summary description of the Executive Option Awards and the PBRSU Awards does not purport to be complete and is qualified in its entirety by the full text of the forms of award agreements for such awards, which are attached hereto as Exhibit 10.9 and Exhibit 10.11 to this Current Report on Form 8-K.

#### **Executive Vice President of Americas Sales Offer Letter**

The Company has entered into an offer letter agreement with Daniel Watson to serve as the Executive Vice President of Americas Sales of the Company, effective as of and contingent upon the consummation of the Business Combination (the "Offer Letter"). The Offer Letter provides for an annual compensation package consisting of a base salary of \$371,197, a target bonus opportunity of 60% of base salary and eligibility for annual long-term incentive awards, pursuant to the 2021 Plan beginning in 2022, with the form of such award and the value of such award determined by the compensation committee of the Company's board of directors. Additionally, pursuant to the Offer Letter, Mr. Watson will be designated as a "Tier One" participant in the Company's Executive Severance Plan (the "Severance Plan"), as described in further detail below. The Offer Letter also contains standard restrictive covenants and confidentiality and invention assignment provisions, in an agreement attached thereto.

Pursuant to the Offer Letter, Mr. Watson will receive a one-time equity award of stock options to purchase 310,000 shares of Company common stock, pursuant to the 2021 Plan, vesting over four years, with 25% of the shares vesting on each of the first four anniversaries of the Closing Date, subject to Mr. Watson's continued employment with the Company through the applicable vesting date (the "Option Award").

The foregoing summary description of the Offer Letter does not purport to be complete and is qualified in its entirety by the full text of the Offer Letter, which is attached hereto as Exhibit 10.8 to this Current Report on Form 8-K.

The foregoing summary description of the Option Award does not purport to be complete and is qualified in its entirety by the full text of the form of award agreement for such award, which is attached hereto as Exhibit 10.10 to this Current Report on Form 8-K.

#### Company Executive Severance Plan

In connection with the consummation of the Business Combination, the Company's board of directors approved the Severance Plan which entitles eligible participants to receive certain severance payments and benefits upon a qualifying termination (as defined in the Severance Plan). Under the terms of the Severance Plan, upon a qualifying termination of Mr. Watson's employment, Mr. Watson will be entitled to the following: (1) cash severance equal to 12 months base salary, (2) a prorated target bonus, and (3) reimbursement of the employer portion of COBRA premium payments for 12 months (the "Severance Benefits"), in each case subject to the Release Requirement. Additionally, under the terms of the Severance Plan, upon a qualifying termination of Mr. Watson's employment in connection with a change in control (as defined in the Severance Plan), Mr. Watson will be entitled to the Severance Benefits, along with a cash severance payment equal to 100% of his target bonus, subject to the Release Requirement.

The foregoing summary description of the Severance Plan does not purport to be complete and is qualified in its entirety by the full text of the Severance Plan, which is attached hereto as Exhibit 10.12 to this Current Report on Form 8-K.

#### **Director Compensation**

The information set forth under the heading "Director Compensation" in Item 2.01 of this Current Report on Form 8-K is incorporated herein by reference.

#### Item 5.03 Amendments to Articles of Incorporation or Bylaws; Change in Fiscal Year.

The information set forth in Item 3.03 of this Current Report on Form 8-K is incorporated herein by reference.

#### Item 5.06 Change in Shell Company Status.

As a result of the Business Combination, which fulfilled the definition of an "initial business combination" as required by the Company's Amended and Restated Certificate of Incorporation, the Company ceased to be a shell company upon the Closing. The material terms of the Business Combination are described in the section of the Proxy Statement entitled "Proposal No. 1 – Approval of the Business Combination" beginning on page 132, which information is incorporated herein by reference.

### Item 9.01 Financial Statements and Exhibits.

- (a) Financial statements of businesses acquired
- 1. The audited consolidated financial statements of LCP Edge Intermediate, Inc. as of December 31, 2020 and 2019, and for the years ended December 31, 2020, 2019 and 2018 will be filed as Exhibit 99.3 to an amendment to this Current Report on Form 8-K as described in the Introductory Note to this Current Report on Form 8-K.
- 2. The audited consolidated financial statements of The Beauty Health Company as of December 31, 2020 and for the year ended December 31, 2020 will be filed as Exhibit 99.1 to an amendment to this Current Report on Form 8-K as described in the Introductory Note to this Current Report on Form 8-K.

## (b) Pro Forma Financial Information

1. The unaudited pro forma condensed combined balance sheet and statements of operations as of December 31, 2020 and for the year ended December 31, 2020 will be filed as Exhibit 99.4 to an amendment to this Current Report on Form 8-K as described in the Introductory Note to this Current Report on Form 8-K.

(c) Exhibits

# EXHIBIT INDEX

Exhibit Number	Description
2.1+	Agreement and Plan of Merger, dated as of December 8, 2020, by and among Vesper Healthcare Acquisition Corp., Hydrate Merger Sub I, Inc., Hydrate Merger Sub II, LLC, LCP Edge Intermediate, Inc. and LCP Edge Holdco, LLC, in its capacity as the Stockholders' Representative (filed as Exhibit 2.1 to the Current Report on Form 8-K of the Company on December 9, 2020 and incorporated herein by reference).
3.1	Second Amended and Restated Certificate of Incorporation of The Beauty Health Company.
3.2	Amended and Restated Bylaws of The Beauty Health Company.
4.1	Specimen Class A Common Stock Certificate (filed as Exhibit 4.2 to the Registration Statement on Form S-1 (File No. 333-248717) of the Company on September 21, 2020 and incorporated herein by reference).
4.2	Specimen Warrant Certificate (filed as Exhibit 4.3 to the Registration Statement on Form S-1 (File No. 333-248717) of the Company on September 21, 2020 and incorporated herein by reference).
4.3	Warrant Agreement, dated September 29, 2020, between the Company and Continental Stock Transfer & Trust Company, as warrant agent (filed as Exhibit 4.1 to the Current Report on Form 8-K of the Company on October 5, 2020 and incorporated herein by reference).
10.1	Form of Subscription Agreement (filed as Exhibit 10.1 to the Current Report on Form 8-K of the Company on December 9, 2020 and incorporated herein by reference).
10.2	Amended and Restated Registration Rights Agreement dated as of May 4, 2021, by and among the Company, BLS Investor Group LLC and the stockholders of LCP Edge Intermediate, Inc.
10.3	Investor Rights Agreement dated as of May 4, 2021, by and between the Company and LCP Edge Holdco, LLC.
10.4#	The Beauty Health Company 2021 Incentive Award Plan (filed as Exhibit 10.1 to the Current Report on Form 8-K of the Company on April 30, 2021 and incorporated herein by reference).
10.5#	The Beauty Health Company 2021 Employee Stock Purchase Plan (filed as Exhibit 10.2 to the Current Report on Form 8-K of the Company on April 30, 2021 and incorporated herein by reference).
10.6#	Employment Agreement, dated as of May 4, 2021, between Clinton E. Carnell, Edge Systems LLC d/b/a The HydraFacial Company and The Beauty Health Company.
10.7#	Employment Agreement, dated as of May 4, 2021, between Liyuan Woo, Edge Systems LLC d/b/a The HydraFacial Company and The Beauty Health Company.
10.8#	Offer Letter dated as of April 29, 2021, between Daniel Watson, Edge Systems LLC d/b/a The HydraFacial Company and The Beauty Health Company.
10.9#	Form of Stock Option Award Agreement (CEO and CFO).
10.10#	Form of Stock Option Award Agreement Form (Non-CEO and CFO).
10.11#	Form of Performance-Based Restricted Stock Unit Agreement.
10.12#	The Beauty Health Company Executive Severance Plan.
10.13#	Form of Indemnity Agreement.
10.14	Amended and Restated Management Services Agreement dated as of May 4, 2021, by and among Linden Manager III LP, Edge Systems LLC d/b/a The HydraFacial Company and The Beauty Health Company.
21.1	Subsidiaries of the Registrant.
99.1*	Audited Consolidated Financial Statements of The Beauty Health Company and its subsidiaries as of December 31, 2020 for the year ended December 31, 2020.
99.2*	Management's Discussion and Analysis of Financial Condition and Results of Operations for The Beauty Health Company for the year ended December 31, 2020.

Exhibit Number	Description
99.3*	Audited Consolidated Financial Statements of LCP Edge Intermediate, Inc. as of December 31, 2020 and for the year ended December 31, 2020.
99.4*	Unaudited Pro Forma Condensed Combined Financial Statements of The Beauty Health Company and its subsidiaries as of December 31, 2020 and for the year ended December 31, 2020.

- The Company agrees to furnish supplementally to the SEC a copy of any omitted schedule or exhibit upon the request of the SEC in accordance with Item 601(b)(2) of Regulation S-K.

  Management contract or compensatory plan or arrangement.
- To be filed by amendment.

### **SIGNATURE**

Pursuant to the requirements of the Securities Exchange Act of 1934, as amended, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Dated: May 10, 2021

# **The Beauty Health Company**

By: /s/Clinton E. Carnell
Name: Clinton E. Carnell
Title: Chief Executive Officer

17

### SECOND AMENDED AND RESTATED CERTIFICATE OF INCORPORATION OF VESPER HEALTHCARE ACQUISITION CORP.

May 4, 2021

Vesper Healthcare Acquisition Corp., a corporation organized and existing under the laws of the State of Delaware (the "Corporation"), DOES HEREBY CERTIFY AS FOLLOWS:

- 1. The name of the Corporation is "Vesper Healthcare Acquisition Corp." The original certificate of incorporation of the Corporation was filed with the Secretary of State of the State of Delaware on July 8, 2020. The Corporation filed an amended and restated certificate of incorporation with the Secretary of State of the State of Delaware on September 28, 2020 (the "First Amended and Restated Certificate").
- 2. This Second Amended and Restated Certificate of Incorporation (this "Second Amended and Restated Certificate") was duly adopted by the Board of Directors of the Corporation (the "Board") and the stockholders of the Corporation in accordance with Sections 228, 242 and 245 of the General Corporation Law of the State of Delaware (as amended from time to time, the "DGCL").
- 3. This Second Amended and Restated Certificate restates, integrates and amends the provisions of the First Amended and Restated Certificate. Certain capitalized terms used in this Second Amended and Restated Certificate are defined where appropriate herein.
  - 4. The text of the First Amended and Restated Certificate is hereby restated and amended in its entirety to read as follows:

#### ARTICLE I NAME

The name of the corporation is The Beauty Health Company (the "Corporation").

#### ARTICLE II PURPOSE

The purpose of the Corporation is to engage in any lawful act or activity for which corporations may be organized under the DGCL. In addition to the powers and privileges conferred upon the Corporation by law and those incidental thereto, the Corporation shall possess and may exercise all the powers and privileges that are necessary or convenient to the conduct, promotion or attainment of the business or purposes of the Corporation.

#### ARTICLE III REGISTERED AGENT

The address of the Corporation's registered office in the State of Delaware is 251 Little Falls Drive, in the City of Wilmington, County of New Castle, State of Delaware, 19808. The name of its registered agent at such address is Corporation Service Company.

# ARTICLE IV CAPITALIZATION

- Section 4.1 Authorized Capital Stock. Subject to Section 4.2, the total number of shares of all classes of capital stock, each with a par value of \$0.0001 per share, which the Corporation is authorized to issue is 321,000,000, consisting of (a) 320,000,000 shares of common stock (the "Common Stock"), including two separate classes of common stock consisting of (i) 300,000,000 shares of Class A Common Stock (the "Class B Common Stock"); and (ii) 20,000,000 shares of preferred stock (the "Preferred Stock").
- Section 4.2 Class B Common Stock. Following the filing of this Second Amended and Restated Certificate with the Secretary of State of the State of Delaware and immediately prior to the Corporation's consummation of any business combination, each share of Class B Common Stock outstanding immediately prior to the filing of this Second Amended and Restated Certificate shall automatically be converted into one share of Class A Common Stock without any action on the part of any person, including the Corporation, and concurrently with such conversion, the number of authorized shares of Class B Common Stock shall be reduced to zero.
- Section 4.3 Preferred Stock. The Preferred Stock may be issued from time to time in one or more series. The Board is hereby expressly authorized to provide for the issuance of shares of the Preferred Stock in one or more series and to establish from time to time the number of shares to be included in each such series and to fix the voting rights, if any, designations, powers, preferences and relative, participating, optional, special and other rights, if any, of each such series and any qualifications, limitations and restrictions thereof, as shall be stated in the resolution or resolutions adopted by the Board providing for the issuance of such series and included in a certificate of designation (a "**Preferred Stock Designation**") filed pursuant to the DGCL, and the Board is hereby expressly vested with the authority to the full extent now or hereafter provided by law to adopt any such resolution or resolutions.

#### Section 4.4 Common Stock.

(a) Except as otherwise required by law or this Second Amended and Restated Certificate (including any Preferred Stock Designation), at any annual or special meeting of the stockholders of the Corporation, the holders of the Common Stock shall have the exclusive right to vote for the election of directors and on all other matters properly submitted to a vote of the stockholders. The holders of shares of Common Stock shall be entitled to one vote for each such share on each matter properly submitted to the stockholders on which the holders of the Common Stock are entitled to vote. Notwithstanding the foregoing, except as otherwise required by law or this Second Amended and Restated Certificate (including any Preferred Stock Designation), the

holders of the Common Stock shall not be entitled to vote on any amendment to this Second Amended and Restated Certificate (including any amendment to any Preferred Stock Designation) that relates solely to the terms of one or more outstanding series of the Preferred Stock if the holders of such affected series are entitled, either separately or together with the holders of one or more other such series, to vote thereon pursuant to this Second Amended and Restated Certificate (including any Preferred Stock Designation) or the DGCL.

- (b) Subject to the rights, if any, of the holders of any outstanding series of the Preferred Stock, the holders of the Common Stock shall be entitled to receive such dividends and other distributions (payable in cash, property or capital stock of the Corporation) when, as and if declared thereon by the Board from time to time out of any assets or funds of the Corporation legally available therefor, and shall share equally on a per share basis in such dividends and distributions.
- (c) Subject to the rights, if any, of the holders of any outstanding series of the Preferred Stock, in the event of any voluntary or involuntary liquidation, dissolution or winding-up of the Corporation, after payment or provision for payment of the debts and other liabilities of the Corporation, the holders of the Common Stock shall be entitled to receive all the remaining assets of the Corporation available for distribution to its stockholders, ratably in proportion to the number of shares of the Common Stock held by them.
- Section 4.5 Rights and Options. The Corporation has the authority to create and issue rights, warrants and options entitling the holders thereof to purchase shares of any class or series of the Corporation's capital stock or other securities of the Corporation, and such rights, warrants and options shall be evidenced by instrument(s) approved by the Board. The Board is empowered to set the exercise price, duration, times for exercise and other terms and conditions of such rights, warrants or options; provided, however, that the consideration to be received for any shares of capital stock subject thereto may not be less than the par value thereof.
- Section 4.6 No Class Vote on Changes in Authorized Number of Shares of Stock. Subject to the rights of the holders of any outstanding series of Preferred Stock, the number of authorized shares of any class or classes of stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of at least a majority of the voting power of the stock entitled to vote thereon irrespective of the provisions of Section 242(b)(2) of the DGCL.

#### ARTICLE V BOARD OF DIRECTORS

Section 5.1 Board Powers. The business and affairs of the Corporation shall be managed by, or under the direction of, the Board. In addition to the powers and authority expressly conferred upon the Board by statute, this Second Amended and Restated Certificate or the Amended and Restated By Laws of the Corporation (the "*Bylaws*"), the Board is hereby empowered to exercise all such powers and do all such acts and things as may be exercised or done by the Corporation, subject, nevertheless, to the provisions of the DGCL and this Second Amended and Restated Certificate.

#### Section 5.2 Number, Election and Term.

- (a) The number of directors of the Corporation shall be fixed from time to time exclusively by the Board pursuant to a resolution adopted by a majority of the Board.
- (b) Subject to Section 5.5 hereof, the Board shall be divided into three classes, as nearly equal in number as possible, and designated Class I, Class II and Class III. The Board is authorized to assign members of the Board already in office to Class I, Class II or Class III. The term of the initial Class I Directors shall expire at the first annual meeting of the stockholders of the Corporation following the effectiveness of this Second Amended and Restated Certificate; the term of the initial Class II Directors shall expire at the second annual meeting of the stockholders of the Corporation following the effectiveness of this Second Amended and Restated Certificate; and the term of the initial Class III Directors shall expire at the third annual meeting of the stockholders of the Corporation following the effectiveness of this Second Amended and Restated Certificate. At each succeeding annual meeting of the stockholders of the Corporation, beginning with the first annual meeting of the stockholders of the Corporation following the effectiveness of this Second Amended and Restated Certificate, each of the successors elected to replace the class of directors whose term expires at that annual meeting shall be elected for a three-year term or until the election and qualification of their respective successors in office, subject to their earlier death, resignation or removal. Subject to Section 5.5 hereof, if the number of directors is changed, any increase or decrease shall be apportioned by the Board among the classes so as to maintain the number of directors in each class as nearly equal as possible, but in no case shall a decrease in the number of directors shorten the term of any incumbent director. Subject to the rights of the holders of one or more series of Preferred Stock, voting separately by class or series, to elect directors pursuant to the terms of one or more series of Preferred Stock, the election of directors shall be determined by a plurality of the votes cast by the stockholders present in person or represented by proxy at the meeting and entitled to vote thereon. The Board is hereby expressly authorized, by resolution or resolutions thereof, to assign members of the Board already in office to the aforesaid classes at the time this Second Amended and Restated Certificate (and therefore such classification) becomes effective in accordance with the DGCL.
- (c) Subject to Section 5.5 hereof, a director shall hold office until the annual meeting for the year in which his or her term expires and until his or her successor has been elected and qualified, subject, however, to such director's earlier death, resignation, retirement, disqualification or removal.
- (d) Unless and except to the extent that the Bylaws shall so require, the election of directors need not be by written ballot. The holders of shares of Common Stock shall not have cumulative voting rights.
- Section 5.3 Newly Created Directorships and Vacancies. Subject to Section 5.5 hereof, newly created directorships resulting from an increase in the number of directors and any vacancies on the Board resulting from death, resignation, retirement, disqualification, removal or other cause may be filled solely and exclusively by a majority vote of the remaining directors then in office, even if less than a quorum, or by a sole remaining director (and not by stockholders), and any director so chosen shall hold office for the remainder of the full term of the class of directors to which the new directorship was added or in which the vacancy occurred and until his or her successor has been elected and qualified, subject, however, to such director's earlier death, resignation, retirement, disqualification or removal.

<u>Section 5.4</u> <u>Removal</u>. Subject to <u>Section 5.5</u> hereof, any or all of the directors may be removed from office at any time, but only for cause and only by the affirmative vote of holders of a majority of the voting power of all then outstanding shares of capital stock of the Corporation entitled to vote generally in the election of directors, voting together as a single class.

Section 5.5 Preferred Stock – Directors. Notwithstanding any other provision of this Article V, and except as otherwise required by law, whenever the holders of one or more series of the Preferred Stock shall have the right, voting separately by class or series, to elect one or more directors, the term of office, the filling of vacancies, the removal from office and other features of such directorships shall be governed by the terms of such series of the Preferred Stock as set forth in this Second Amended and Restated Certificate (including any Preferred Stock Designation) and such directors shall not be included in any of the classes created pursuant to this Article V unless expressly provided by such terms.

#### ARTICLE VI BYLAWS

In furtherance and not in limitation of the powers conferred by law, subject to any limitations contained in this Second Amended and Restated Certificate, the Board is expressly authorized to make, alter and repeal the Bylaws, but any Bylaws adopted by the Board may be adopted, amended, altered or repealed by the stockholders entitled to vote thereon; provided, however, that in addition to any vote of the holders of any class or series of capital stock of the Corporation required by law or by this Second Amended and Restated Certificate (including any Preferred Stock Designation), the affirmative vote of the holders of at least a majority of the voting power of all then outstanding shares of capital stock of the Corporation entitled to vote generally in the election of directors, voting together as a single class, shall be required for the stockholders to adopt, amend, alter or repeal the Bylaws; and provided, further, however, that no Bylaws hereafter adopted by the stockholders shall invalidate any prior act of the Board that would have been valid if such Bylaws had not been adopted.

# ARTICLE VII MEETINGS OF STOCKHOLDERS; ACTION BY WRITTEN CONSENT

<u>Section 7.1</u> <u>Meetings</u>. Subject to the rights, if any, of the holders of any outstanding series of the Preferred Stock, and to the requirements of applicable law, special meetings of stockholders of the Corporation may be called only by the Chairman of the Board, Chief Executive Officer of the Corporation, or the Board pursuant to a resolution adopted by a majority of the Board, and the ability of the stockholders to call a special meeting is hereby specifically denied.

<u>Section 7.2</u> <u>Advance Notice</u>. Advance notice of stockholder nominations for the election of directors and of business to be brought by stockholders before any meeting of the stockholders of the Corporation shall be given in the manner provided in the Bylaws.

<u>Section 7.3</u> <u>Action by Written Consent</u>. Any action required or permitted to be taken by the stockholders of the Corporation must be effected by a duly called annual or special meeting of such stockholders and may not be effected by written consent of the stockholders.

#### ARTICLE VIII LIMITED LIABILITY; INDEMNIFICATION

Section 8.1 <u>Limitation of Director Liability.</u> A director of the Corporation shall not be personally liable to the Corporation or its stockholders for monetary damages for breach of fiduciary duty as a director, except to the extent such exemption from liability or limitation thereof is not permitted under the DGCL as the same exists or may hereafter be amended. Any amendment, modification or repeal of the foregoing sentence shall not adversely affect any right or protection of a director of the Corporation hereunder in respect of any act or omission occurring prior to the time of such amendment, modification or repeal.

#### Section 8.2 Indemnification and Advancement of Expenses.

(a) To the fullest extent permitted by Delaware law, as the same exists or may hereafter be amended, the Corporation shall indemnify, defend, advance expenses and hold harmless each person who is or was made a party or is threatened to be made a party to or is otherwise involved in any threatened, pending or completed action, suit, investigation, arbitration or proceeding, whether civil, criminal, administrative or investigative, including an action by or in the right of the Corporation to procure a judgment in its favor (each, a "proceeding"), by reason of the fact that he or she, or a person for whom he or she is the legal representative, is or was a director or officer of the Corporation or, while a director or officer of the Corporation, is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation or of a partnership, joint venture, trust, other enterprise or nonprofit entity, including service with respect to an employee benefit plan (an "indemnitee"), whether the basis of such proceeding is alleged action in an official capacity as a director, officer, employee or agent, or in any other capacity while serving as a director, officer, employee or agent, against all liability and loss suffered and expenses (including, without limitation, attorneys' fees and disbursements, judgments, fines, ERISA excise taxes, damages, claims and penalties and amounts paid in settlement) reasonably incurred by such indemnitee in connection with such proceeding. The Corporation shall to the fullest extent not prohibited by applicable law pay the expenses (including attorneys' fees) incurred by an indemnitee in defending or otherwise participating in any proceeding in advance of its final disposition (including by making any payment directly to the applicable third parties if requested by the indemnitee); provided, however, that, to the extent required by applicable law, such payment of expenses in advance of the final disposition of the proceeding shall be made only upon receipt of an undertaking, by or on behalf of the indemnitee, to repay all amounts so advanced if it shall ultimately be determined that the indemnitee is not entitled to be indemnified under this Section 8.2 or otherwise. The rights to indemnification and advancement of expenses conferred by this Section 8.2 shall be contract rights and such rights shall continue as to an indemnitee who has ceased to be a director, officer, employee or agent and shall inure to the benefit of his or her heirs, executors and administrators. Notwithstanding the foregoing provisions of this Section 8.2(a), except for proceedings to enforce rights to indemnification and advancement of expenses (which are, for the avoidance of doubt, indemnified proceedings and expenses), the Corporation shall indemnify and advance expenses to an indemnitee in connection with a proceeding (or part thereof) initiated by such indemnitee only if such proceeding (or part thereof) was, or is, authorized by the Board.

- (b) The rights to indemnification and advancement of expenses conferred on any indemnitee by this <u>Section 8.2</u> shall not be exclusive of any other rights that any indemnitee may have or hereafter acquire under law, this Second Amended and Restated Certificate, the Bylaws, an agreement, vote of stockholders or disinterested directors, or otherwise.
- (c) Any repeal or amendment of this Section 8.2 by the stockholders of the Corporation or by changes in law, or the adoption of any other provision of this Second Amended and Restated Certificate inconsistent with this Section 8.2, shall, unless otherwise required by law, be prospective only (except to the extent such amendment or change in law permits the Corporation to provide broader indemnification rights on a retroactive basis than permitted prior thereto), and shall not in any way diminish or adversely affect any right or protection existing at the time of such repeal or amendment or adoption of such inconsistent provision in respect of any proceeding (regardless of when such proceeding is first threatened, commenced or completed) arising out of, or related to, any act or omission occurring prior to such repeal or amendment or adoption of such inconsistent provision. If the DGCL is amended after the effectiveness of this Second Amended and Restated Certificate to authorize corporation action further eliminating or limiting the personal liability of directors, then the liability of a director of the Corporation shall be eliminated or limited to the fullest extent permitted by the DGCL, as so amended.
- (d) This <u>Section 8.2</u> shall not limit the right of the Corporation, to the extent and in the manner authorized or permitted by law, to indemnify and to advance expenses to persons other than indemnitees.

# ARTICLE IX CORPORATE OPPORTUNITY

Section 9.1 The doctrine of corporate opportunity, or any other analogous doctrine, shall not apply with respect to the Corporation or any of its officers or directors in circumstances where the application of any such doctrine to a corporate opportunity would conflict with any fiduciary duties or contractual obligations they may have as of the date of this Second Amended and Restated Certificate or in the future. In addition to the foregoing, the doctrine of corporate opportunity shall not apply to any other corporate opportunity with respect to any of the directors or officers of the Corporation unless such corporate opportunity is offered to such person solely in his or her capacity as a director or officer of the Corporation and such opportunity is one the Corporation is legally and contractually permitted to undertake and would otherwise be reasonable for the Corporation to pursue.

<u>Section 9.2</u> Without limiting the foregoing, to the extent permitted by applicable law, each of BLS Investor Group LLC, Linden Manager III LP, DW Management Services, L.L.C. and the investment funds affiliated with the foregoing and their respective successors and Affiliates (as defined in <u>Section 10.3</u>) (other than the Corporation and its subsidiaries) and all of their respective partners, principals, directors, officers, members, managers, equity holders and/or employees, including any of the foregoing who serve as officers or directors of the Corporation (each, an "*Exempted Person*") shall not have any fiduciary duty to refrain from engaging directly

or indirectly in the same or similar business activities or lines of business as the Corporation or any of its subsidiaries, except as otherwise expressly provided in any agreement entered into between the Corporation and such Exempted Person. To the fullest extent permitted by applicable law, the Corporation, on behalf of itself and its subsidiaries, renounces any interest or expectancy of the Corporation and its subsidiaries in, or in being offered an opportunity to participate in, business opportunities that are from time to time available to the Exempted Persons, even if the opportunity is one that the Corporation or its subsidiaries might reasonably be deemed to have pursued or had the ability or desire to pursue if granted the opportunity to do so, and each such Exempted Person shall have no duty to communicate or offer such business opportunity to the Corporation (and there shall be no restriction on the Exempted Persons using the general knowledge and understanding of the industry in which the Corporation operates which it has gained as an Exempted Person in considering and pursuing such opportunities or in making investment, voting, monitoring, governance or other decisions relating to other entities or securities) and, to the fullest extent permitted by applicable law, shall not be liable to the Corporation or any of its subsidiaries or stockholders for breach of any fiduciary or other duty, as a director or officer or otherwise, by reason of the fact that such Exempted Person pursues or acquires such business opportunity, directs such business opportunity to another person or fails to present such business opportunity, or information regarding such business opportunity, to the Corporation or its subsidiaries, or uses such knowledge and understanding in the manner described herein, in each case, except as otherwise expressly provided in any agreement entered into between the Corporation and such Exempted Person. In addition to and notwithstanding the foregoing, a corporate opportunity shall not be deemed to belong to the Corporation if it is a business opportunity that the Corporation is not financially able or contractually permitted or legally able to undertake, or that is, from its nature, not in the line of the Corporation's business or is of no practical advantage to it or that is one in which the Corporation has no interest or reasonable expectancy. Any person or entity purchasing or otherwise acquiring any interest in any shares of stock of the Corporation shall be deemed to have notice of the provisions of this Article IX.

Section 9.3 Neither the alteration, amendment, addition to or repeal of this Article IX, nor the adoption of any provision of this Second Amended and Restated Certificate (including any Preferred Stock Designation) inconsistent with this Article IX, shall eliminate or reduce the effect of this Article IX in respect of any business opportunity first identified or any other matter occurring, or any cause of action, suit or claim that, but for this Article IX, would accrue or arise, prior to such alteration, amendment, addition, repeal or adoption. This Article IX shall not limit any protections or defenses available to, or indemnification or advancement rights of, any director or officer of the Corporation under this Second Amended and Restated Certificate, the Bylaws or applicable law.

# ARTICLE X BUSINESS COMBINATIONS

Section 10.1 Opt Out of DGCL 203. The Corporation expressly elects not to be governed by Section 203 of the DGCL.

<u>Section 10.2</u> <u>Limitations on Business Combinations</u>. Notwithstanding the foregoing, the Corporation shall not engage in any business combination, at any point in time at which the Common Stock is registered under Section 12(b) or 12(g) of the Exchange Act with any interested stockholder for a period of three (3) years following the time that such stockholder became an interested stockholder, unless:

- (a) prior to such time, the Board approved either the business combination or the transaction which resulted in the stockholder becoming an interested stockholder;
- (b) upon consummation of the transaction which resulted in the stockholder becoming an interested stockholder, the interested stockholder owned at least 85% of the voting stock of the Corporation outstanding at the time the transaction commenced, excluding for purposes of determining the voting stock outstanding (but not the outstanding voting stock owned by the interested stockholder) those shares owned by: (i) persons who are directors and also officers; or (ii) employee stock plans in which employee participants do not have the right to determine confidentially whether shares held subject to the plan will be tendered in a tender or exchange offer; or
- (c) at or subsequent to such time, the business combination is approved by the Board and authorized at an annual or special meeting of stockholders, and not by written consent, by the affirmative vote of at least two thirds of the outstanding voting stock of the Corporation which is not owned by the interested stockholder.

#### <u>Section 10.3</u> <u>Definitions</u>. For purposes of this <u>Article X</u>, the term:

- (a) "Affiliate" means, with respect to any person, any other person that controls, is controlled by, or is under common control with such person.
- (b) "associate," when used to indicate a relationship with any person, means: (i) any corporation, partnership, unincorporated association or other entity of which such person is a director, officer or partner or is, directly or indirectly, the owner of 20% or more of any class of voting stock; (ii) any trust or other estate in which such person has at least a 20% beneficial interest or as to which such person serves as trustee or in a similar fiduciary capacity; and (iii) any relative or spouse of such person, or any relative of such spouse, who has the same residence as such person.
  - (c) "business combination," when used in reference to the Corporation and any interested stockholder of the Corporation, means:
- (i) any merger or consolidation of the Corporation or any direct or indirect majority-owned subsidiary of the Corporation: (A) with the interested stockholder; or (B) with any other corporation, partnership, unincorporated association or other entity if the merger or consolidation is caused by the interested stockholder and as a result of such merger or consolidation <u>Section 10.2</u> is not applicable to the surviving entity;
- (ii) any sale, lease, exchange, mortgage, pledge, transfer or other disposition (in one transaction or a series of transactions), except proportionately as a stockholder of the Corporation, to or with the interested stockholder, whether as part of a dissolution or otherwise, of assets of the Corporation or of any direct or indirect majority-owned subsidiary of the Corporation which assets have an aggregate market value equal to 10% or more of either the aggregate market value of all the assets of the Corporation determined on a consolidated basis or the aggregate market value of all the outstanding stock of the Corporation;

- (iii) any transaction which results in the issuance or transfer by the Corporation or by any direct or indirect majority-owned subsidiary of the Corporation of any stock of the Corporation or of such subsidiary to the interested stockholder, except: (A) pursuant to the exercise, exchange or conversion of securities exercisable for, exchangeable for or convertible into stock of the Corporation or any such subsidiary which securities were outstanding prior to the time that the interested stockholder became such; (B) pursuant to a merger under Section 251(g) of the DGCL; (C) pursuant to a dividend or distribution paid or made, or the exercise, exchange or conversion of securities exercisable for, exchangeable for or convertible into stock of the Corporation or any such subsidiary which security is distributed, pro rata to all holders of a class or series of stock of the Corporation subsequent to the time the interested stockholder became such; (D) pursuant to an exchange offer by the Corporation to purchase stock made on the same terms to all holders of said stock; or (E) any issuance or transfer of stock by the Corporation; provided, however, that in no case under items (C) (E) of this subsection (iii) shall there be an increase in the interested stockholder's proportionate share of the stock of any class or series of the Corporation or of the voting stock of the Corporation (except as a result of immaterial changes due to fractional share adjustments);
- (iv) any transaction involving the Corporation or any direct or indirect majority-owned subsidiary of the Corporation which has the effect, directly or indirectly, of increasing the proportionate share of the stock of any class or series, or securities convertible into the stock of any class or series, of the Corporation or of any such subsidiary which is owned by the interested stockholder, except as a result of immaterial changes due to fractional share adjustments or as a result of any purchase or redemption of any shares of stock not caused, directly or indirectly, by the interested stockholder; or
- (v) any receipt by the interested stockholder of the benefit, directly or indirectly (except proportionately as a stockholder of the Corporation), of any loans, advances, guarantees, pledges, or other financial benefits (other than those expressly permitted in subsections (i)-(iv) above) provided by or through the Corporation or any direct or indirect majority-owned subsidiary.
- (d) "control," including the terms "controlling," "controlled by" and "under common control with," means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a person, whether through the ownership of voting stock, by contract, or otherwise. A person who is the owner of 20% or more of the outstanding voting stock of the Corporation, partnership, unincorporated association or other entity shall be presumed to have control of such entity, in the absence of proof by a preponderance of the evidence to the contrary. Notwithstanding the foregoing, a presumption of control shall not apply where such person holds voting stock, in good faith and not for the purpose of circumventing this Article X, as an agent, bank, broker, nominee, custodian or trustee for one or more owners who do not individually or as a group have control of such entity.

- (e) "Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations thereunder (as so amended and inclusive of such rules and regulations).
- (f) "interested stockholder" means any person (other than the Corporation or any direct or indirect majority-owned subsidiary of the Corporation) that: (i) is the owner of 15% or more of the outstanding voting stock of the Corporation; or (ii) is an Affiliate or associate of the Corporation and was the owner of 15% or more of the outstanding voting stock of the Corporation at any time within the three (3) year period immediately prior to the date on which it is sought to be determined whether such person is an interested stockholder; or (iii) an Affiliate or associate of any such person described in clauses (i) and (ii); provided, however, that the term "interested stockholder" shall not include: (A) the Sponsor Holders or their transferees; or (B) any person whose ownership of shares in excess of the 15% limitation set forth herein is the result of any action taken solely by the Corporation; provided, that such person specified in this clause (B) shall be an interested stockholder if thereafter such person acquires additional shares of voting stock of the Corporation, except as a result of further corporate action not caused, directly or indirectly, by such person. For the purpose of determining whether a person is an interested stockholder, the voting stock of the Corporation deemed to be outstanding shall include stock deemed to be owned by the person through application of the definition of "owner" below but shall not include any other unissued stock of the Corporation which may be issuable pursuant to any agreement, arrangement or understanding, or upon exercise of conversion rights, warrants or options, or otherwise.
- (g) "owner," including the terms "own" and "owned," when used with respect to any stock, means a person that individually or with or through any of its Affiliates or associates:
  - (i) beneficially owns such stock, directly or indirectly; or
- (ii) has: (A) the right to acquire such stock (whether such right is exercisable immediately or only after the passage of time) pursuant to any agreement, arrangement or understanding, or upon the exercise of conversion rights, exchange rights, warrants or options, or otherwise; provided, however, that a person shall not be deemed the owner of stock tendered pursuant to a tender or exchange offer made by such person or any of such person's Affiliates or associates until such tendered stock is accepted for purchase or exchange; or (B) the right to vote such stock pursuant to any agreement, arrangement or understanding; provided, however, that a person shall not be deemed the owner of any stock because of such person's right to vote such stock if the agreement, arrangement or understanding to vote such stock arises solely from a revocable proxy or consent given in response to a proxy or consent solicitation made to 10 or more persons; or
- (iii) has any agreement, arrangement or understanding for the purpose of acquiring, holding, voting (except voting pursuant to a revocable proxy or consent as described in item (B) of subsection (ii) above), or disposing of such stock with any other person that beneficially owns, or whose Affiliates or associates beneficially own, directly or indirectly, such stock.

- (h) "person" means any individual, corporation, partnership, unincorporated association or other entity.
- (i) "Sponsor Holders" means the equityholders of BLS Investor Group LLC and their respective successors and Affiliates.
- (j) "stock" means, with respect to any corporation, capital stock and, with respect to any other entity, any equity interest.
- (k) "voting stock" means stock of any class or series entitled to vote generally in the election of directors.

# ARTICLE XI AMENDMENT OF AMENDED AND RESTATED CERTIFICATE OF INCORPORATION

The Corporation reserves the right at any time and from time to time to amend, alter, change or repeal any provision contained in this Second Amended and Restated Certificate (including any Preferred Stock Designation), in the manner now or hereafter prescribed by this Second Amended and Restated Certificate and the DGCL, and, except as set forth in <a href="Article VIII">Article VIII</a>, all rights, preferences and privileges herein conferred upon stockholders, directors or any other persons by and pursuant to this Second Amended and Restated Certificate in its present form or as hereafter amended are granted subject to the right reserved in this <a href="Article XI">Article XI</a>. Notwithstanding anything to the contrary contained in this Second Amended and Restated Certificate, and notwithstanding that a lesser percentage may be permitted from time to time by applicable law, no provision of <a href="Article V, Section 7.1">Article V, Section 7.1</a>, Section 7.3,

<a href="Article VIII">Article IX</a>, Article X and this <a href="Article XI">Article XI</a> may be altered, amended or repealed in any respect, nor may any provision or bylaw inconsistent therewith be adopted, unless, in addition to any other vote required by this Second Amended and Restated Certificate or otherwise required by law, such alteration, amendment, repeal or adoption is approved by the affirmative vote of the holders of at least two thirds of the voting power of the outstanding shares of capital stock of the Corporation entitled to vote generally in the election of directors, voting together as a single class.

# ARTICLE XII EXCLUSIVE FORUM

Section 12.1 Unless the Corporation consents in writing to the selection of an alternative forum, (a) the Court of Chancery of the State of Delaware (or, in the event that the Court of Chancery does not have jurisdiction, the federal district court for the District of Delaware or, in the event that the federal district court for the District of Delaware does not have jurisdiction, other state courts of the State of Delaware) shall, to the fullest extent permitted by law, be the sole and exclusive forum for (i) any derivative action, suit or proceeding brought on behalf of the Corporation, (ii) any action, suit or proceeding asserting a claim of breach of a fiduciary duty owed by any director, officer, employee or stockholder of the Corporation to the Corporation or to the Corporation's stockholders, (iii) any action, suit or proceeding asserting a claim against the Corporation, its directors, officers or employees arising pursuant to any provision of the DGCL or the Bylaws or this Second Amended and Restated Certificate (as they may be amended and/or

restated from time to time) or (iv) any action, suit or proceeding asserting a claim against the Corporation, its directors, officers or employees governed by the internal affairs doctrine and, if brought outside of Delaware, the stockholder bringing the suit will be deemed to have consented to service of process on such stockholder's counsel; and (b) the federal district courts of the United States of America shall, to the fullest extent permitted by law, be the exclusive forum for the resolution of any complaint asserting a cause of action arising under the Securities Act of 1933, as amended, or the rules and regulations promulgated thereunder. If any action the subject matter of which is within the scope of clause (a) of the immediately preceding sentence is filed in a court other than a court located within the State of Delaware (a "Foreign Action") in the name of any stockholder, such stockholder shall be deemed to have consented to (x) the personal jurisdiction of the state and federal courts located within the State of Delaware in connection with any action brought in any such court to enforce the provisions of clause (a) of the immediately preceding sentence and (y) having service of process made upon such stockholder in any such action by service upon such stockholder's counsel in the Foreign Action as agent for such stockholder.

<u>Section 12.2</u> Any person or entity purchasing or otherwise acquiring any interest in any security of the Corporation shall be deemed to have notice of and consented to this <u>Article XII</u>. Notwithstanding the foregoing, the provisions of this <u>Article XII</u> shall not apply to suits brought to enforce any liability or duty created by the Exchange Act (as defined in <u>Section 10.3</u>) or any other claim for which the federal courts of the United States have exclusive jurisdiction.

### ARTICLE XIII SEVERABILITY

If any provision or provisions of this Second Amended and Restated Certificate (including any Preferred Stock Designation relating to any series of Preferred Stock) shall be held to be invalid, illegal or unenforceable as applied to any person or entity or circumstance for any reason whatsoever, then, to the fullest extent permitted by law: (i) the validity, legality and enforceability of such provision or provisions in any other circumstance and of the remaining provisions of this Second Amended and Restated Certificate (including, without limitation, any Preferred Stock Designation relating to any series of Preferred Stock and each portion of any paragraph of this Second Amended and Restated Certificate or Preferred Stock Designation containing any such provision or provisions held to be invalid, illegal or unenforceable that is not itself held to be invalid, illegal or unenforceable) and the application of such provision to other persons or entities and circumstances shall not in any way be affected or impaired thereby and (ii) to the fullest extent possible, the provisions of this Second Amended and Restated Certificate (including, without limitation, any Preferred Stock Designation relating to any series of Preferred Stock and each such portion of any paragraph of this Second Amended and Restated Certificate or Preferred Stock Designation containing any such provision or provisions held to be invalid, illegal or unenforceable) shall be construed so as to permit the Corporation to protect its directors, officers, employees and agents from personal liability in respect of their good faith service or for the benefit of the Corporation to the fullest extent permitted by law. Any person or entity purchasing or otherwise acquiring any interest in any security of the Corporation shall be deemed to have notice of and consented to this Article XIII.

IN WITNESS WHEREOF, Vesper Healthcare Acquisition Corp. has caused this Second Amended and Restated Certificate to be duly executed and acknowledged in its name and on its behalf by an authorized officer as of the date first set forth above.

## VESPER HEALTHCARE ACQUISITION CORP.

By: /s/ Brenton L. Saunders

Name: Brenton L. Saunders Title: Chairman of the Board

[Signature Page to Second Amended and Restated Certificate of Incorporation]

# AMENDED AND RESTATED BY LAWS OF THE BEAUTY HEALTH COMPANY

#### ARTICLE I

#### **OFFICES**

**Section 1.1 Registered Office**. The registered office of The Beauty Health Company (the "*Corporation*") within the State of Delaware shall be located at either: (a) the principal place of business of the Corporation in the State of Delaware; or (b) the office of the corporation or individual acting as the Corporation's registered agent in Delaware.

**Section 1.2** Additional Offices. The Corporation may, in addition to its registered office in the State of Delaware, have such other offices and places of business, both within and outside the State of Delaware, as the Board of Directors of the Corporation (the "Board") may from time to time determine or as the business and affairs of the Corporation may require.

#### ARTICLE II

#### STOCKHOLDERS MEETINGS

- **Section 2.1 Annual Meetings**. The annual meeting of stockholders shall be held at such place, either within or without the State of Delaware, and time and on such date as shall be determined by the Board and stated in the notice of the meeting; <u>provided</u> that the Board may in its sole discretion determine that the meeting shall not be held at any place, but may instead be held solely by means of remote communication pursuant to <u>Section 9.5(a)</u>. At each annual meeting, the stockholders entitled to vote on such matters shall elect those directors of the Corporation to fill any term of a directorship that expires on the date of such annual meeting and may transact any other business as may properly be brought before the meeting.
- **Section 2.2 Special Meetings.** Subject to the rights of the holders of any outstanding series of the preferred stock of the Corporation (the "*Preferred Stock*"), and to the requirements of applicable law, special meetings of stockholders, for any purpose or purposes, may be called only by the Chairman of the Board, Chief Executive Officer, or the Board pursuant to a resolution adopted by a majority of the Board, and may not be called by any other person. Special meetings of stockholders shall be held at such place, either within or without the State of Delaware, and at such time and on such date as shall be determined by the Board and stated in the Corporation's notice of the meeting; <u>provided</u> that the Board may in its sole discretion determine that the meeting shall not be held at any place, but may instead be held solely by means of remote communication pursuant to Section 9.5(a).
- **Section 2.3 Notices.** Written notice of each stockholders meeting stating the place, if any, date, and time of the meeting, and the means of remote communication, if any, by which stockholders and proxy holders may be deemed to be present in person and vote at such meeting and the record date for determining the stockholders entitled to vote at the meeting, if such date is different from the record date for determining stockholders entitled to notice of the meeting, shall

be given in the manner permitted by Section 9.3 to each stockholder entitled to vote thereat as of the record date for determining the stockholders entitled to notice of the meeting, by the Corporation not less than 10 nor more than 60 days before the date of the meeting unless otherwise required by the General Corporation Law of the State of Delaware (the "*DGCL*"). If said notice is for a stockholders meeting other than an annual meeting, it shall in addition state the purpose or purposes for which the meeting is called, and the business transacted at such meeting shall be limited to the matters so stated in the Corporation's notice of meeting (or any supplement thereto). Any meeting of stockholders as to which notice has been given may be postponed, and any meeting of stockholders as to which notice has been given may be cancelled, by the Board upon public announcement (as defined in Section 2.7(c)) given before the date previously scheduled for such meeting.

Section 2.4 Quorum. Except as otherwise provided by applicable law, the Corporation's Second Amended and Restated Certificate of Incorporation, as the same may be amended or restated from time to time (the "Certificate of Incorporation") or these By Laws, the presence, in person or by proxy, at a stockholders meeting of the holders of shares of outstanding capital stock of the Corporation representing a majority of the voting power of all outstanding shares of capital stock of the Corporation entitled to vote at such meeting shall constitute a quorum for the transaction of business at such meeting, except that when specified business is to be voted on by a class or series of stock voting as a class, the holders of shares representing a majority of the voting power of the outstanding shares of such class or series shall constitute a quorum of such class or series for the transaction of such business. If a quorum shall not be present or represented by proxy at any meeting of the stockholders of the Corporation, the chairman of the meeting may adjourn the meeting from time to time in the manner provided in Section 2.6 until a quorum shall attend. The stockholders present at a duly convened meeting may continue to transact business until adjournment, notwithstanding the withdrawal of enough stockholders to leave less than a quorum. Shares of its own stock belonging to the Corporation or to another corporation, if a majority of the voting power of the shares entitled to vote in the election of directors of such other corporation is held, directly or indirectly, by the Corporation, shall neither be entitled to vote nor be counted for quorum purposes; provided, however, that the foregoing shall not limit the right of the Corporation or any such other corporation to vote shares held by it in a fiduciary capacity.

# Section 2.5 Voting of Shares.

(a) <u>Voting Lists</u>. The Secretary of the Corporation (the "*Secretary*") shall prepare, or shall cause the officer or agent who has charge of the stock ledger of the Corporation to prepare, at least 10 days before every meeting of stockholders, a complete list of the stockholders of record entitled to vote at such meeting and showing the address and the number and class of shares registered in the name of each stockholder. Nothing contained in this <u>Section 2.5(a)</u> shall require the Corporation to include electronic mail addresses or other electronic contact information on such list. Such list shall be open to the examination of any stockholder, for any purpose germane to the meeting, during ordinary business hours for a period of at least 10 days prior to the meeting: (i) on a reasonably accessible electronic network; <u>provided</u> that the information required to gain access to such list is provided with the notice of the meeting; or (ii) during ordinary business hours, at the principal place of business of the Corporation. In the event that the Corporation determines to make the list available on an electronic network, the Corporation may take reasonable steps to ensure that such information is available only to stockholders of the

Corporation. If the meeting is to be held at a place, then the list shall be produced and kept at the time and place of the meeting during the whole time thereof, and may be inspected by any stockholder who is present. If a meeting of stockholders is to be held solely by means of remote communication as permitted by Section 9.5(a), the list shall be open to the examination of any stockholder during the whole time of the meeting on a reasonably accessible electronic network, and the information required to access such list shall be provided with the notice of meeting. The stock ledger shall be the only evidence as to who are the stockholders entitled to examine the list required by this Section 2.5(a) or to vote in person or by proxy at any meeting of stockholders.

- (b) Manner of Voting. At any stockholders meeting, every stockholder entitled to vote may vote in person or by proxy. If authorized by the Board, the voting by stockholders or proxy holders at any meeting conducted by remote communication may be effected by a ballot submitted by electronic transmission (as defined in Section 9.3); provided that any such electronic transmission must either set forth or be submitted with information from which the Corporation can determine that the electronic transmission was authorized by the stockholder or proxy holder. The Board, in its discretion, or the chairman of the meeting of stockholders, in such person's discretion, may require that any votes cast at such meeting shall be cast by written ballot.
- (c) <u>Proxies</u>. Each stockholder entitled to vote at a meeting of stockholders or to express consent or dissent to corporate action in writing without a meeting may authorize another person or persons to act for such stockholder by proxy, but no such proxy shall be voted or acted upon after three years from its date, unless the proxy provides for a longer period. Proxies need not be filed with the Secretary until the meeting is called to order, but shall be filed with the Secretary before being voted. Without limiting the manner in which a stockholder may authorize another person or persons to act for such stockholder as proxy, either of the following shall constitute a valid means by which a stockholder may grant such authority. No stockholder shall have cumulative voting rights.
- (i) A stockholder may execute a writing authorizing another person or persons to act for such stockholder as proxy. Execution may be accomplished by the stockholder or such stockholder's authorized officer, director, employee or agent signing such writing or causing such person's signature to be affixed to such writing by any reasonable means, including, but not limited to, by facsimile signature.
- (ii) A stockholder may authorize another person or persons to act for such stockholder as proxy by transmitting or authorizing the transmission of an electronic transmission to the person who will be the holder of the proxy or to a proxy solicitation firm, proxy support service organization or like agent duly authorized by the person who will be the holder of the proxy to receive such transmission; provided that any such electronic transmission must either set forth or be submitted with information from which it can be determined that the electronic transmission was authorized by the stockholder. Any copy, facsimile telecommunication or other reliable reproduction of the writing or transmission authorizing another person or persons to act as proxy for a stockholder may be substituted or used in lieu of the original writing or transmission for any and all purposes for which the original writing or transmission could be used; provided that such copy, facsimile telecommunication or other reproduction shall be a complete reproduction of the entire original writing or transmission.

- (d) Required Vote. Subject to the rights of the holders of one or more series of the Preferred Stock, voting separately by class or series, to elect directors pursuant to the terms of one or more series of Preferred Stock, at all meetings of stockholders at which a quorum is present, the election of directors shall be determined by a plurality of the votes cast by the stockholders present in person or represented by proxy at the meeting and entitled to vote thereon. All other matters presented to the stockholders at a meeting at which a quorum is present shall be determined by the vote of a majority of the votes cast by the stockholders present in person or represented by proxy at the meeting and entitled to vote thereon, unless the matter is one upon which, by applicable law, the Certificate of Incorporation, these By Laws or applicable stock exchange rules, a different vote is required, in which case such provision shall govern and control the decision of such matter.
- (e) Inspectors of Election. The Board may, and shall if required by law, in advance of any meeting of stockholders, appoint one or more persons as inspectors of election, who may be employees of the Corporation or otherwise serve the Corporation in other capacities, to act at such meeting of stockholders or any adjournment thereof and to make a written report thereof. The Board may appoint one or more persons as alternate inspectors to replace any inspector who fails to act. If no inspectors of election or alternates are appointed by the Board, the chairman of the meeting shall appoint one or more inspectors to act at the meeting. Each inspector, before discharging his or her duties, shall take and sign an oath faithfully to execute the duties of inspector with strict impartiality and according to the best of his or her ability. The inspectors shall ascertain and report the number of outstanding shares and the voting power of each; determine the number of shares present in person or represented by proxy at the meeting and the validity of proxies and ballots; count all votes and ballots and report the results; determine and retain for a reasonable period a record of the disposition of any challenges made to any determination by the inspectors; and certify their determination of the number of shares represented at the meeting and their count of all votes and ballots. No person who is a candidate for an office at an election may serve as an inspector at such election. Each report of an inspector shall be in writing and signed by the inspector or by a majority of them if there is more than one inspector acting at such meeting. If there is more than one inspector, the report of a majority shall be the report of the inspectors.

Section 2.6 Adjournments. Any meeting of stockholders, annual or special, may be adjourned by the chairman of the meeting, from time to time, whether or not there is a quorum, to reconvene at the same or some other place. Notice need not be given of any such adjourned meeting if the date, time, and place, if any, thereof, and the means of remote communication, if any, by which stockholders and proxy holders may be deemed to be present in person and vote at such adjourned meeting are announced at the meeting at which the adjournment is taken. At the adjourned meeting the stockholders, or the holders of any class or series of stock entitled to vote separately as a class, as the case may be, may transact any business that might have been transacted at the original meeting. If the adjournment is for more than 30 days, notice of the adjourned meeting shall be given to each stockholder of record entitled to vote at the meeting. If after the adjournment a new record date for stockholders entitled to vote is fixed for the adjourned meeting, the Board shall fix a new record date for notice of such adjourned meeting in accordance with Section 9.2, and shall give notice of the adjourned meeting to each stockholder of record entitled to vote at such adjourned meeting as of the record date for notice of such adjourned meeting.

#### **Section 2.7** Advance Notice for Business.

- (a) Annual Meetings of Stockholders. No business may be transacted at an annual meeting of stockholders, other than business that is either: (i) specified in the Corporation's notice of meeting (or any supplement thereto) given by or at the direction of the Board; (ii) otherwise properly brought before the annual meeting by or at the direction of the Board; or (iii) otherwise properly brought before the annual meeting by any stockholder of the Corporation: (A) who is a stockholder of record entitled to vote at such annual meeting on the date of the giving of the notice provided for in this Section 2.7(a) and on the record date for the determination of stockholders entitled to vote at such annual meeting; and (B) who complies with the notice procedures set forth in this Section 2.7(a). Notwithstanding anything in this Section 2.7(a) to the contrary, only persons nominated for election as a director to fill any term of a directorship that expires on the date of the annual meeting pursuant to Section 3.2 will be considered for election at such meeting.
- (i) In addition to any other applicable requirements, for business (other than nominations) to be properly brought before an annual meeting by a stockholder, such stockholder must have given timely notice thereof in proper written form to the Secretary and such business must otherwise be a proper matter for stockholder action. Subject to Section 2.7(a)(iii), a stockholder's notice to the Secretary with respect to such business, to be timely, must be received by the Secretary at the principal executive offices of the Corporation not later than the close of business on the 90th day nor earlier than the opening of business on the 120th day before the anniversary date of the immediately preceding annual meeting of stockholders; provided, however, that in the event that the annual meeting is more than 30 days before or more than 60 days after such anniversary date, notice by the stockholder to be timely must be so delivered not earlier than the close of business on the 120th day before the meeting and not later than the close of business on the 90th day before the meeting, or if the first public announcement of the date of the meeting is less than 100 days prior to the date of the meeting, the close of business on the 10th day following the day on which public announcement of the date of the annual meeting is first made by the Corporation. The public announcement of an adjournment or postponement of an annual meeting shall not commence a new time period (or extend any time period) for the giving of a stockholder's notice as described in this Section 2.7(a). In addition, to be considered timely, a stockholder's notice shall further be updated and supplemented, if necessary, so that the information provided or required to be provided in such notice shall be true and correct as of the record date for the meeting and as of the date that is ten (10) business days prior to the meeting or any adjournment or postponement thereof, and such update and supplement shall be delivered to the Secretary at the principal executive offices of the Corporation not later than five (5) business days after the record date for the meeting in the case of the update and supplement required to be made as of the record date, and not later than eight (8) business days prior to the date for the meeting or any adjournment or postponement thereof in the case of the update and supplement required to be made as of ten (10) business days prior to the meeting or any adjournment or postponement thereof. For the avoidance of doubt, the obligation to update and supplement as set forth in this paragraph or any other Section of these By Laws shall not limit the Corporation's rights with respect to any deficiencies in any notice provided by a stockholder, extend any applicable deadlines hereunder or enable or be deemed to permit a stockholder who has previously submitted notice hereunder to amend or update any proposal or to submit any new proposal, including by changing or adding nominees, matters, business and or resolutions proposed to be brought before a meeting of the stockholders.

(ii) To be in proper written form, a stockholder's notice to the Secretary with respect to any business (other than nominations) must set forth as to each such matter such stockholder proposes to bring before the annual meeting: (A) a brief description of the business desired to be brought before the annual meeting, the text of the proposal or business (including the text of any resolutions proposed for consideration and in the event such business includes a proposal to amend these By Laws, the language of the proposed amendment) and the reasons for conducting such business at the annual meeting; (B) the name and record address of such stockholder, as they appear on the Corporation's books, and the name and address of the beneficial owner, if any, on whose behalf the proposal is made and their respective affiliates or associates or others acting in concert therewith: (C) the class or series and number of shares of capital stock of the Corporation that are owned beneficially and of record by such stockholder and by the beneficial owner, if any, on whose behalf the proposal is made and their respective affiliates or associates or others acting in concert therewith; (D) a description of all arrangements or understandings between such stockholder and the beneficial owner, if any, on whose behalf the proposal is made and any other person or persons (including their names) in connection with the proposal of such business by such stockholder; (E) any material interest of such stockholder and the beneficial owner, if any, on whose behalf the proposal is made in such business; (F) a representation that such stockholder (or a qualified representative of such stockholder) intends to appear in person or by proxy at the annual meeting to bring such business before the meeting; (G) any option, warrant, convertible security, stock appreciation right, or similar right with an exercise or conversion privilege or a settlement payment or mechanism at a price related to any class or series of shares of the Corporation or with a value derived in whole or in part from the value of any class or series of shares of the Corporation, or any derivative or synthetic arrangement having the characteristics of a long position in any class or series of shares of the Corporation, or any contract, derivative, swap or other transaction or series of transactions designed to produce economic benefits and risks that correspond substantially to the ownership of any class or series of shares of the Corporation, including due to the fact that the value of such contract, derivative, swap or other transaction or series of transactions is determined by reference to the price, value or volatility of any class or series of shares of the Corporation, whether or not such instrument, contract or right shall be subject to settlement in the underlying class or series of shares of the Corporation, through the delivery of cash or other property, or otherwise, and without regard to whether the stockholder of record, the beneficial owner, if any, or any affiliates or associates or others acting in concert therewith, may have entered into transactions that hedge or mitigate the economic effect of such instrument, contract or right, or any other direct or indirect opportunity to profit or share in any profit derived from any increase or decrease in the value of shares of the Corporation (any of the foregoing, a "Derivative Instrument") directly or indirectly owned beneficially by such stockholder, the beneficial owner, if any, or any affiliates or associates or others acting in concert therewith; (H) any proxy, contract, arrangement, understanding, or relationship pursuant to which such stockholder, such beneficial owner or any of their respective affiliates or associates or others acting in concert therewith has any right to vote any class or series of shares of the Corporation; (I) any agreement, arrangement, understanding, relationship or otherwise, including any repurchase or similar so-called "stock borrowing" agreement or arrangement, involving such stockholder, such beneficial owner or any of their respective affiliates or associates or others acting in concert therewith, directly or indirectly, the purpose or effect of which is to mitigate loss to,

reduce the economic risk (of ownership or otherwise) of any class or series of the shares of the Corporation by, manage the risk of share price changes for, or increase or decrease the voting power of, such stockholder, such beneficial owner or any of their respective affiliates or associates or others acting in concert therewith with respect to any class or series of the shares of the Corporation, or which provides, directly or indirectly, the opportunity to profit or share in any profit derived from any decrease in the price or value of any class or series of the shares of the Corporation (any of the foregoing, a "Short Interest"); (J) any rights to dividends on the shares of the Corporation owned beneficially by such stockholder, such beneficial owner or any of their respective affiliates or associates or others acting in concert therewith that are separated or separable from the underlying shares of the Corporation: (K) any proportionate interest in shares of the Corporation or Derivative Instruments held, directly or indirectly, by a general or limited partnership in which such stockholder, such beneficial owner or any of their respective affiliates or associates or others acting in concert therewith is a general partner or, directly or indirectly, beneficially owns an interest in a general partner of such general or limited partnership; (L) any performance-related fees (other than an asset-based fee) that such stockholder, such beneficial owner or any of their respective affiliates or associates or others acting in concert therewith is entitled to based on any increase or decrease in the value of shares of the Corporation or Derivative Instruments, if any, including without limitation any such interests held by members of the immediate family sharing the same household of such stockholder, such beneficial owner and their respective affiliates or associates or others acting in concert therewith; (M) any significant equity interests or any Derivative Instruments or Short Interests in any principal competitor of the Corporation held by such stockholder, such beneficial owner or any of their respective affiliates or associates or others acting in concert therewith; (N) any direct or indirect interest of such stockholder, such beneficial owner and their respective affiliates or associates or others acting in concert therewith in any contract with the Corporation, any affiliate of the Corporation or any principal competitor of the Corporation (including, in any such case, any employment agreement, collective bargaining agreement or consulting agreement); (O) all information that would be required to be set forth in a Schedule 13D filed pursuant to Rule 13d-1(a) or an amendment pursuant to Rule 13d-2(a) if such a statement were required to be filed under the Securities Exchange Act of 1934, as amended (the "Exchange Act") and the rules and regulations promulgated thereunder by such stockholder, such beneficial owner and their respective affiliates or associates or others acting in concert therewith, if any; and (P) any other information relating to such stockholder, such beneficial owner or any of their respective affiliates or associates or others acting in concert therewith, if any, that would be required to be disclosed in a proxy statement and form or proxy or other filings required to be made in connection with solicitations of proxies for, as applicable, the proposal and/or for the election of directors in a contested election pursuant to Section 14 of the Exchange Act and the rules and regulations promulgated thereunder.

(iii) The foregoing notice requirements of this Section 2.7(a) shall be deemed satisfied by a stockholder as to any proposal (other than nominations) if the stockholder has notified the Corporation of such stockholder's intention to present such proposal at an annual meeting in compliance with Rule 14a-8 (or any successor thereof) of the Exchange Act and such stockholder has complied with the requirements of such Rule for inclusion of such proposal in a proxy statement prepared by the Corporation to solicit proxies for such annual meeting. No business shall be conducted at the annual meeting of stockholders except business brought before the annual meeting in accordance with the procedures set forth in this Section 2.7(a); provided, however, that once business has been properly brought before the annual meeting in accordance

with such procedures, nothing in this Section 2.7(a) shall be deemed to preclude discussion by any stockholder of any such business. If the Board or the chairman of the annual meeting determines that any stockholder proposal was not made in accordance with the provisions of this Section 2.7(a) or that the information provided in a stockholder's notice does not satisfy the information requirements of this Section 2.7(a), such proposal shall not be presented for action at the annual meeting. Notwithstanding the foregoing provisions of this Section 2.7(a), if the stockholder (or a qualified representative of the stockholder) does not appear at the annual meeting of stockholders of the Corporation to present the proposed business, such proposed business shall not be transacted, notwithstanding that proxies in respect of such matter may have been received by the Corporation.

- (iv) In addition to the provisions of this <u>Section 2.7(a)</u>, a stockholder shall also comply with all applicable requirements of the Exchange Act and the rules and regulations thereunder with respect to the matters set forth herein. Nothing in this <u>Section 2.7(a)</u> shall be deemed to affect any rights of stockholders to request inclusion of proposals in the Corporation's proxy statement pursuant to Rule 14a-8 under the Exchange Act.
- (b) <u>Special Meetings of Stockholders</u>. Only such business shall be conducted at a special meeting of stockholders as shall have been brought before the meeting pursuant to the Corporation's notice of meeting. Nominations of persons for election to the Board may be made at a special meeting of stockholders at which directors are to be elected pursuant to the Corporation's notice of meeting only pursuant to <u>Section 3.2</u>.
- (c) <u>Public Announcement</u>. For purposes of these By Laws, "*public announcement*" shall mean disclosure in a press release reported by the Dow Jones News Service, Associated Press or comparable national news service or in a document publicly filed by the Corporation with the Securities and Exchange Commission pursuant to Sections 13, 14 or 15(d) of the Exchange Act (or any successor thereto).
- Section 2.8 Conduct of Meetings. The chairman of each annual and special meeting of stockholders shall be the Chairman of the Board or, in the absence (or inability or refusal to act) of the Chairman of the Board, the Chief Executive Officer (if he or she shall be a director) or, in the absence (or inability or refusal to act) of the Chief Executive Officer or if the Chief Executive Officer is not a director, the President (if he or she shall be a director) or, in the absence (or inability or refusal to act) of the President or if the President is not a director, such other person as shall be appointed by the Board. The date and time of the opening and the closing of the polls for each matter upon which the stockholders will vote at a meeting shall be announced at the meeting by the chairman of the meeting. The Board may adopt such rules and regulations for the conduct of the meeting of stockholders as it shall deem appropriate. Except to the extent inconsistent with these By Laws or such rules and regulations as adopted by the Board, the chairman of any meeting of stockholders shall have the right and authority to convene and to adjourn the meeting, to prescribe such rules, regulations and procedures and to do all such acts as, in the judgment of such chairman, are appropriate for the proper conduct of the meeting. Such rules, regulations or procedures, whether adopted by the Board or prescribed by the chairman of the meeting, may include, without limitation, the following: (a) the establishment of an agenda or order of business for the meeting; (b) rules and procedures for maintaining order at the meeting and the safety of those present; (c) limitations on attendance at or participation in the meeting to stockholders of

record of the Corporation, their duly authorized and constituted proxies or such other persons as the chairman of the meeting shall determine; (d) restrictions on entry to the meeting after the time fixed for the commencement thereof; and (e) limitations on the time allotted to questions or comments by participants. Unless and to the extent determined by the Board or the chairman of the meeting, meetings of stockholders shall not be required to be held in accordance with the rules of parliamentary procedure. The secretary of each annual and special meeting of stockholders shall be the Secretary or, in the absence (or inability or refusal to act) of the Secretary, an Assistant Secretary so appointed to act by the chairman of the meeting. In the absence (or inability or refusal to act) of the Secretary and all Assistant Secretaries, the chairman of the meeting may appoint any person to act as secretary of the meeting.

**Section 2.9 Consents in Lieu of Meeting**. Any action required or permitted to be taken by the stockholders of the Corporation must be effected by a duly called annual or special meeting of such stockholders and may not be effected by written consent of the stockholders.

# ARTICLE III

# **DIRECTORS**

**Section 3.1 Powers; Number**. The business and affairs of the Corporation shall be managed by or under the direction of the Board, which may exercise all such powers of the Corporation and do all such lawful acts and things as are not by statute or by the Certificate of Incorporation or by these By Laws required to be exercised or done by the stockholders. Directors need not be stockholders or residents of the State of Delaware. Subject to the Certificate of Incorporation, the number of directors shall be fixed exclusively by resolution of the Board.

#### Section 3.2 Advance Notice for Nomination of Directors.

- (a) Only persons who are nominated in accordance with the following procedures or, with respect to LCP Edge Holdco, LLC ("LCP") and its affiliates, the Investor Rights Agreement between the Corporation and LCP, dated as of May 4, 2021, shall be eligible for election as directors of the Corporation, except as may be otherwise provided by the terms of one or more series of Preferred Stock with respect to the rights of holders of one or more series of Preferred Stock to elect directors. Nominations of persons for election to the Board at any annual meeting of stockholders, or at any special meeting of stockholders called for the purpose of electing directors as set forth in the Corporation's notice of such special meeting, may be made: (i) by or at the direction of the Board; or (ii) by any stockholder of the Corporation: (A) who is a stockholder of record entitled to vote in the election of directors on the date of the giving of the notice provided for in this Section 3.2 and on the record date for the determination of stockholders entitled to vote at such meeting; and (B) who complies with the notice procedures set forth in this Section 3.2 (including the completed and signed questionnaire, representation and agreement required by Section 3.2(h) of these By Laws).
- (b) In addition to any other applicable requirements, for a nomination to be made by a stockholder, such stockholder must have given timely notice thereof in proper written form to the Secretary. To be timely, a stockholder's notice to the Secretary must be received by the Secretary at the principal executive offices of the Corporation: (i) in the case of an annual

meeting, not later than the close of business on the 90th day nor earlier than the close of business on the 120th day before the anniversary date of the immediately preceding annual meeting of stockholders; provided, however, that in the event that the annual meeting is more than 30 days before or more than 60 days after such anniversary date, notice by the stockholder to be timely must be so received not earlier than the close of business on the 120th day before the meeting and not later than the close of business on the 90th day before the meeting or, if the first public announcement of the date of the meeting is less than 100 days prior to the date of the meeting, the close of business on the 10th day following the day on which public announcement of the date of the annual meeting was first made by the Corporation; and (ii) in the case of a special meeting of stockholders called for the purpose of electing directors, not later than the close of business on the 10th day following the day on which public announcement of the date of the special meeting is first made by the Corporation. In no event shall the public announcement of an adjournment or postponement of an annual meeting or special meeting commence a new time period (or extend any time period) for the giving of a stockholder's notice as described in this Section 3.2. In addition, to be considered timely, a stockholder's notice shall further be updated and supplemented, if necessary, so that the information provided or required to be provided in such notice shall be true and correct as of the record date for the meeting and as of the date that is ten (10) business days prior to the meeting or any adjournment or postponement thereof, and such update and supplement shall be delivered to the Secretary at the principal executive offices of the Corporation not later than five (5) business days after the record date for the meeting in the case of the update and supplement required to be made as of the record date, and not later than eight (8) business days prior to the date for the meeting or any adjournment or postponement thereof in the case of the update and supplement required to be made as of ten (10) business days prior to the meeting or any adjournment or postponement thereof. For the avoidance of doubt, the obligation to update and supplement as set forth in this paragraph or any other Section of these By Laws shall not limit the Corporation's rights with respect to any deficiencies in any notice provided by a stockholder, extend any applicable deadlines hereunder or enable or be deemed to permit a stockholder who has previously submitted notice hereunder to amend or update any proposal or to submit any new proposal, including by changing or adding nominees, matters, business and or resolutions proposed to be brought before a meeting of the stockholders.

- (c) Notwithstanding anything in paragraph (b) to the contrary, in the event that the number of directors to be elected to the Board at an annual meeting is greater than the number of directors whose terms expire on the date of the annual meeting and there is no public announcement by the Corporation naming all of the nominees for the additional directors to be elected or specifying the size of the increased Board before the close of business on the 100th day prior to the anniversary date of the immediately preceding annual meeting of stockholders, a stockholder's notice required by this <u>Section 3.2</u> shall also be considered timely, but only with respect to nominees for the additional directorships created by such increase that are to be filled by election at such annual meeting, if it shall be received by the Secretary at the principal executive offices of the Corporation not later than the close of business on the 10th day following the date on which such public announcement was first made by the Corporation.
- (d) To be in proper written form, a stockholder's notice to the Secretary must set forth: (i) as to each person whom the stockholder proposes to nominate for election as a director: (A) the name, age, business address and residence address of the person; (B) the principal occupation or employment of the person; (C) the class or series and number of shares of capital

stock of the Corporation, if any, that are owned beneficially or of record by the person; (D) any other information relating to the person that would be required to be disclosed in a proxy statement or other filings required to be made in connection with solicitations of proxies for election of directors pursuant to Section 14 of the Exchange Act and the rules and regulations promulgated thereunder; and (E) a description of all direct and indirect compensation and other material monetary agreements, arrangements and understandings during the past three years, and any other material relationships, between or among such stockholder and beneficial owner, if any, and their respective affiliates and associates, or others acting in concert therewith, on the one hand, and each proposed nominee, and his or her respective affiliates and associates, or others acting in concert therewith, on the other hand, including, without limitation all information that would be required to be disclosed pursuant to Rule 404 promulgated under Regulation S-K if the stockholder making the nomination and any beneficial owner on whose behalf the nomination is made, if any, or any affiliate or associate thereof or person acting in concert therewith, were the "registrant" for purposes of such rule and the nominee were a director or executive officer of such registrant; and (ii) as to the stockholder giving the notice: (A) the name and record address of such stockholder, as they appear on the Corporation's books, and the name and address of the beneficial owner, if any, on whose behalf the nomination is made, and their respective affiliates or associates or others acting in concert therewith; (B) the class or series and number of shares of capital stock of the Corporation that are owned beneficially and of record by such stockholder and the beneficial owner, if any, on whose behalf the nomination is made, and their respective affiliates or associates or others acting in concert therewith; (C) a description of all arrangements or understandings relating to the nomination to be made by such stockholder among such stockholder, the beneficial owner, if any, on whose behalf the nomination is made, each proposed nominee and any other person or persons (including their names); (D) a representation that such stockholder (or a qualified representative of such stockholder) intends to appear in person or by proxy at the meeting to nominate the persons named in its notice; (E) any Derivative Instrument directly or indirectly owned beneficially by such stockholder, the beneficial owner, if any, or any affiliates or associates or others acting in concert therewith; (F) any proxy, contract, arrangement, understanding, or relationship pursuant to which such stockholder, such beneficial owner or any of their respective affiliates or associates or others acting in concert therewith has any right to vote any class or series of shares of the Corporation; (G) any Short Interest; (H) any rights to dividends on the shares of the Corporation owned beneficially by such stockholder, such beneficial owner or any of their respective affiliates or associates or others acting in concert therewith that are separated or separable from the underlying shares of the Corporation; (I) any proportionate interest in shares of the Corporation or Derivative Instruments held, directly or indirectly, by a general or limited partnership in which such stockholder, such beneficial owner or any of their respective affiliates or associates or others acting in concert therewith is a general partner or, directly or indirectly, beneficially owns an interest in a general partner of such general or limited partnership; (J) any performance-related fees (other than an asset-based fee) that such stockholder, such beneficial owner or any of their respective affiliates or associates or others acting in concert therewith is entitled to based on any increase or decrease in the value of shares of the Corporation or Derivative Instruments, if any, including without limitation any such interests held by members of the immediate family sharing the same household of such stockholder, such beneficial owner and their respective affiliates or associates or others acting in concert therewith; (K) any significant equity interests or any Derivative Instruments or Short Interests in any principal competitor of the Corporation held by such stockholder, such beneficial owner or any of their respective affiliates

or associates or others acting in concert therewith; (L) any direct or indirect interest of such stockholder, such beneficial owner and their respective affiliates or associates or others acting in concert therewith in any contract with the Corporation, any affiliate of the Corporation or any principal competitor of the Corporation (including, in any such case, any employment agreement, collective bargaining agreement or consulting agreement); (M) all information that would be required to be set forth in a Schedule 13D filed pursuant to Rule 13d-1(a) or an amendment pursuant to Rule 13d-2(a) if such a statement were required to be filed under the Exchange Act and the rules and regulations promulgated thereunder by such stockholder, such beneficial owner and their respective affiliates or associates or others acting in concert therewith, if any; and (N) any other information relating to such stockholder and the beneficial owner, if any, on whose behalf the nomination is made that would be required to be disclosed in a proxy statement or other filings required to be made in connection with solicitations of proxies for election of directors pursuant to Section 14 of the Exchange Act and the rules and regulations promulgated thereunder. Such notice must be accompanied by a written consent of each proposed nominee to being named as a nominee and to serve as a director if elected.

- (e) With respect to each individual, if any, whom the stockholder proposes to nominate for election or reelection to the Board of Directors, a stockholder's notice must, in addition to the matters set forth in paragraphs (d) above, also include a completed and signed questionnaire, representation and agreement required by Section 3.2(h) of these By Laws. The Corporation may require any proposed nominee to furnish such other information as may reasonably be required by the Corporation to determine the eligibility of such proposed nominee to serve as an independent director of the Corporation or that could be material to a reasonable stockholder's understanding of the independence, or lack thereof, of such nominee. Notwithstanding anything to the contrary, only persons who are nominated in accordance with the procedures set forth in these By Laws, including without limitation this Section 3.2, shall be eligible for election as directors.
- (f) If the Board or the chairman of the meeting of stockholders determines that any nomination was not made in accordance with the provisions of this Section 3.2 or that the information provided in a stockholder's notice does not satisfy the information requirements of this Section 3.2, then such nomination shall not be considered at the meeting in question. Notwithstanding the foregoing provisions of this Section 3.2, if the stockholder (or a qualified representative of the stockholder) does not appear at the meeting of stockholders of the Corporation to present the nomination, such nomination shall be disregarded, notwithstanding that proxies in respect of such nomination may have been received by the Corporation.
- (g) In addition to the provisions of this <u>Section 3.2</u>, a stockholder shall also comply with all of the applicable requirements of the Exchange Act and the rules and regulations thereunder with respect to the matters set forth herein. Nothing in this <u>Section 3.2</u> shall be deemed to affect any rights of the holders of Preferred Stock to elect directors pursuant to the Certificate of Incorporation.
- (h) To be eligible to be a nominee of any stockholder for election or reelection as a director of the Corporation, a person must deliver (in accordance with the time periods prescribed for delivery of notice under <u>Section 3.2</u> of these By Laws) to the Secretary at the principal executive offices of the Corporation a written questionnaire with respect to the

background and qualification of such individual and the background of any other person or entity on whose behalf, directly or indirectly, the nomination is being made (which questionnaire shall be provided by the Secretary upon written request), and a written representation and agreement (in the form provided by the Secretary upon written request) that such individual (A) is not and will not become a party to (1) any agreement, arrangement or understanding with, and has not given any commitment or assurance to, any person or entity as to how such person, if elected as a director of the Corporation, will act or vote on any issue or question (a "Voting Commitment") that has not been disclosed to the Corporation, and (2) any Voting Commitment that could limit or interfere with such individual's ability to comply, if elected as a director of the corporation, with such individual's fiduciary duties under applicable law, (B) is not and will not become a party to any agreement, arrangement or understanding with any person or entity other than the Corporation with respect to any direct or indirect compensation, reimbursement or indemnification in connection with service or action as a director that has not been disclosed therein, (C) in such individual's personal capacity and on behalf of any person or entity on whose behalf, directly or indirectly, the nomination is being made, would be in compliance, if elected as a director of the Corporation, and will comply, with all applicable corporate governance, conflict of interest, confidentiality and stock ownership and trading policies and guidelines of the Corporation publicly disclosed from time to time and (D) consents to being named as a nominee in the Corporation's proxy statement pursuant to Rule 14a-4(d) under the Exchange Act and any associated proxy card of the Corporation and agrees to serve if elected as a director.

- **Section 3.3 Compensation.** Unless otherwise restricted by the Certificate of Incorporation or these By Laws, the Board shall have the authority to fix the compensation of directors, including for service on a committee of the Board, and may be paid either a fixed sum for attendance at each meeting of the Board or other compensation as director. The directors may be reimbursed their expenses, if any, of attendance at each meeting of the Board. No such payment shall preclude any director from serving the Corporation in any other capacity and receiving compensation therefor. Members of committees of the Board may be allowed like compensation and reimbursement of expenses for service on the committee.
- **Section 3.4 Newly Created Directorships and Vacancies**. Unless otherwise provided by the Certificate of Incorporation, newly created directorships resulting from an increase in the number of directors and any vacancies on the Board resulting from death, resignation, retirement, disqualification, removal or other cause shall be filled solely by a majority vote of the remaining directors then in office, even if less than a quorum, or by a sole remaining director (and not by stockholders), and any director so chosen shall hold office for the remainder of the full term of the class of directors to which the new directorship was added or in which the vacancy occurred and until his or her successor has been elected and qualified, subject, however, to such director's earlier death, resignation, retirement, disqualification or removal.
- **Section 3.5 Chairman of the Board.** The Board shall, from time to time by vote of the Board, elect a Chairman of the Board. The Chairman of the Board shall preside when present at all meetings of the stockholders and the Board. The Chairman of the Board shall have general supervision and control of the acquisition activities of the Corporation subject to the ultimate authority of the Board, and shall be responsible for the execution of the policies of the Board with respect to such matters. In the absence (or inability or refusal to act) of the Chairman of the Board, the Chief Executive Officer (if he or she shall be a director) shall preside when present at all

meetings of the stockholders and the Board. The powers and duties of the Chairman of the Board shall not include supervision or control of the preparation of the financial statements of the Corporation (other than through participation as a member of the Board). The position of Chairman of the Board and Chief Executive Officer may be held by the same person.

#### ARTICLE IV

#### BOARD MEETINGS

**Section 4.1 Annual Meetings.** The Board shall meet as soon as practicable after the adjournment of each annual stockholders meeting at the place of the annual stockholders meeting unless the Board shall fix another time and place and give notice thereof in the manner required herein for special meetings of the Board. No notice to the directors shall be necessary to legally convene this meeting, except as provided in this <u>Section 4.1</u>.

**Section 4.2 Regular Meetings**. Regularly scheduled, periodic meetings of the Board may be held without notice at such times, dates and places (within or without the State of Delaware) as shall from time to time be determined by the Board.

Section 4.3 Special Meetings. Special meetings of the Board: (a) may be called by the Chairman of the Board or President; and (b) shall be called by the Chairman of the Board, President or Secretary on the written request of at least a majority of directors then in office, or the sole director, as the case may be, and shall be held at such time, date and place (within or without the State of Delaware) as may be determined by the person calling the meeting or, if called upon the request of directors or the sole director, as specified in such written request. Notice of each special meeting of the Board shall be given, as provided in Section 9.3, to each director: (i) at least 24 hours before the meeting if such notice given personally or by telephone or written notice given by hand delivery or by means of a form of electronic transmission and delivery; (ii) at least two days before the meeting if such notice is sent by a nationally recognized overnight delivery service; and (iii) at least five days before the meeting if such notice is sent through the United States mail. If the Secretary shall fail or refuse to give such notice, then the notice may be given by the officer who called the meeting or the directors who requested the meeting. Any and all business that may be transacted at a regular meeting of the Board may be transacted at a special meeting. Except as may be otherwise expressly provided by applicable law, the Certificate of Incorporation, or these By Laws, neither the business to be transacted at, nor the purpose of, any special meeting need be specified in the notice or waiver of notice of such meeting. A special meeting may be held at any time without notice if all the directors are present or if those not present waive notice of the meeting in accordance with Section 9.4.

**Section 4.4 Quorum; Required Vote.** A majority of the Board shall constitute a quorum for the transaction of business at any meeting of the Board, and the act of a majority of the directors present at any meeting at which there is a quorum shall be the act of the Board, except as may be otherwise specifically provided by applicable law, the Certificate of Incorporation or these By Laws. If a quorum shall not be present at any meeting, a majority of the directors present may adjourn the meeting from time to time, without notice other than announcement at the meeting, until a quorum is present.

**Section 4.5 Consent In Lieu of Meeting.** Unless otherwise restricted by the Certificate of Incorporation or these By Laws, any action required or permitted to be taken at any meeting of the Board or any committee thereof may be taken without a meeting if all members of the Board or committee, as the case may be, consent thereto in writing or by electronic transmission, and the writing or writings or electronic transmission or transmissions (or paper reproductions thereof) are filed with the minutes of proceedings of the Board or committee. Such filing shall be in paper form if the minutes are maintained in paper form and shall be in electronic form if the minutes are maintained in electronic form.

Section 4.6 Organization. The chairman of each meeting of the Board shall be the Chairman of the Board or, in the absence (or inability or refusal to act) of the Chairman of the Board, the Chief Executive Officer (if he or she shall be a director) or, in the absence (or inability or refusal to act) of the Chief Executive Officer or if the Chief Executive Officer is not a director, the President (if he or she shall be a director) or in the absence (or inability or refusal to act) of the President or if the President is not a director, a chairman elected from the directors present. The Secretary shall act as secretary of all meetings of the Board. In the absence (or inability or refusal to act) of the Secretary, an Assistant Secretary shall perform the duties of the Secretary at such meeting. In the absence (or inability or refusal to act) of the Secretary and all Assistant Secretaries, the chairman of the meeting may appoint any person to act as secretary of the meeting.

#### ARTICLE V

# **COMMITTEES OF DIRECTORS**

- **Section 5.1 Establishment.** The Board may by resolution of the Board designate one or more committees, each committee to consist of one or more of the directors of the Corporation. Each committee shall keep regular minutes of its meetings and report the same to the Board when required by the resolution designating such committee. The Board shall have the power at any time to fill vacancies in, to change the membership of, or to dissolve any such committee.
- **Section 5.2 Available Powers**. Any committee established pursuant to <u>Section 5.1</u> hereof, to the extent permitted by applicable law and by resolution of the Board, shall have and may exercise all of the powers and authority of the Board in the management of the business and affairs of the Corporation, and may authorize the seal of the Corporation to be affixed to all papers that may require it.
- **Section 5.3 Alternate Members**. The Board may designate one or more directors as alternate members of any committee, who may replace any absent or disqualified member at any meeting of such committee. In the absence or disqualification of a member of the committee, the member or members thereof present at any meeting and not disqualified from voting, whether or not he, she or they constitute a quorum, may unanimously appoint another member of the Board to act at the meeting in place of any such absent or disqualified member.
- **Section 5.4 Procedures.** Unless the Board otherwise provides, the time, date, place, if any, and notice of meetings of a committee shall be determined by such committee. At meetings of a committee, a majority of the number of members of the committee (but not including any alternate member, unless such alternate member has replaced any absent or disqualified member

at the time of, or in connection with, such meeting) shall constitute a quorum for the transaction of business. The act of a majority of the members present at any meeting at which a quorum is present shall be the act of the committee, except as otherwise specifically provided by applicable law, the Certificate of Incorporation, these By Laws or the Board. If a quorum is not present at a meeting of a committee, the members present may adjourn the meeting from time to time, without notice other than an announcement at the meeting, until a quorum is present. Unless the Board otherwise provides and except as provided in these By Laws, each committee designated by the Board may make, alter, amend and repeal rules for the conduct of its business. In the absence of such rules each committee shall conduct its business in the same manner as the Board is authorized to conduct its business pursuant to Article III and Article IV of these By Laws.

### ARTICLE VI

#### **OFFICERS**

- Section 6.1 Officers. The officers of the Corporation elected by the Board shall be a Chief Executive Officer, a President, a Chief Financial Officer, a Secretary and such other officers (including without limitation, Vice Presidents, Assistant Secretaries and a Treasurer) as the Board from time to time may determine. Officers elected by the Board shall each have such powers and duties as generally pertain to their respective offices, subject to the specific provisions of this Article VI. Such officers shall also have such powers and duties as from time to time may be conferred by the Board. The Chief Executive Officer or President may also appoint such other officers (including without limitation one or more Vice Presidents and Controllers) as may be necessary or desirable for the conduct of the business of the Corporation. Such other officers shall have such powers and duties and shall hold their offices for such terms as may be provided in these By Laws or as may be prescribed by the Board or, if such officer has been appointed by the Chief Executive Officer or President, as may be prescribed by the appointing officer.
- (a) <u>Chief Executive Officer</u>. The Chief Executive Officer shall be the chief executive officer of the Corporation, shall have general supervision of the affairs of the Corporation and general control of all of its business subject to the ultimate authority of the Board, and shall be responsible for the execution of the policies of the Board with respect to such matters, except to the extent any such powers and duties have been prescribed to the Chairman of the Board pursuant to <u>Section 3.5</u> above. In the absence (or inability or refusal to act) of the Chairman of the Board, the Chief Executive Officer (if he or she shall be a director) shall preside when present at all meetings of the stockholders and the Board. The position of Chief Executive Officer and President may be held by the same person.
- (b) <u>President</u>. The President shall make recommendations to the Chief Executive Officer on all operational matters that would normally be reserved for the final executive responsibility of the Chief Executive Officer. In the absence (or inability or refusal to act) of the Chairman of the Board and Chief Executive Officer, the President (if he or she shall be a director) shall preside when present at all meetings of the stockholders and the Board. The President shall also perform such duties and have such powers as shall be designated by the Board. The position of President and Chief Executive Officer may be held by the same person.

(c) <u>Vice Presidents</u>. In the absence (or inability or refusal to act) of the President, the Vice President (or in the event there be more than one Vice President, the Vice Presidents in the order designated by the Board) shall perform the duties and have the powers of the President. Any one or more of the Vice Presidents may be given an additional designation of rank or function.

#### (d) Secretary.

- (i) The Secretary shall attend all meetings of the stockholders, the Board and (as required) committees of the Board and shall record the proceedings of such meetings in books to be kept for that purpose. The Secretary shall give, or cause to be given, notice of all meetings of the stockholders and special meetings of the Board and shall perform such other duties as may be prescribed by the Board, the Chairman of the Board, Chief Executive Officer or President. The Secretary shall have custody of the corporate seal of the Corporation and the Secretary, or any Assistant Secretary, shall have authority to affix the same to any instrument requiring it, and when so affixed, it may be attested by his or her signature or by the signature of such Assistant Secretary. The Board may give general authority to any other officer to affix the seal of the Corporation and to attest the affixing thereof by his or her signature.
- (ii) The Secretary shall keep, or cause to be kept, at the principal executive office of the Corporation or at the office of the Corporation's transfer agent or registrar, if one has been appointed, a stock ledger, or duplicate stock ledger, showing the names of the stockholders and their addresses, the number and classes of shares held by each and, with respect to certificated shares, the number and date of certificates issued for the same and the number and date of certificates cancelled.
- (e) <u>Assistant Secretaries</u>. The Assistant Secretary or, if there be more than one, the Assistant Secretaries in the order determined by the Board shall, in the absence (or inability or refusal to act) of the Secretary, perform the duties and have the powers of the Secretary.
- (f) <u>Chief Financial Officer</u>. The Chief Financial Officer shall perform all duties commonly incident to that office (including, without limitation, the care and custody of the funds and securities of the Corporation, which from time to time may come into the Chief Financial Officer's hands and the deposit of the funds of the Corporation in such banks or trust companies as the Board, the Chief Executive Officer or the President may authorize).
- (g) <u>Treasurer</u>. The Treasurer shall, in the absence (or inability or refusal to act) of the Chief Financial Officer, perform the duties and exercise the powers of the Chief Financial Officer.
- **Section 6.2 Term of Office; Removal; Vacancies**. The elected officers of the Corporation shall be appointed by the Board and shall hold office until their successors are duly elected and qualified by the Board or until their earlier death, resignation, retirement, disqualification, or removal from office. Any officer may be removed, with or without cause, at any time by the Board. Any officer appointed by the Chief Executive Officer or President may also be removed, with or without cause, by the Chief Executive Officer or President, as the case may be, unless the Board otherwise provides. Any vacancy occurring in any elected office of the

Corporation may be filled by the Board. Any vacancy occurring in any office appointed by the Chief Executive Officer or President may be filled by the Chief Executive Officer, or President, as the case may be, unless the Board then determines that such office shall thereupon be elected by the Board, in which case the Board shall elect such officer.

**Section 6.3 Other Officers**. The Board may delegate the power to appoint such other officers and agents, and may also remove such officers and agents or delegate the power to remove same, as it shall from time to time deem necessary or desirable.

**Section 6.4 Multiple Officeholders; Stockholder and Director Officers**. Any number of offices may be held by the same person unless the Certificate of Incorporation or these By Laws otherwise provide. Officers need not be stockholders or residents of the State of Delaware.

#### **ARTICLE VII**

#### **SHARES**

**Section 7.1 Certificated and Uncertificated Shares.** The shares of the Corporation may be certificated or uncertificated, subject to the sole discretion of the Board and the requirements of the DGCL.

**Section 7.2 Multiple Classes of Stock**. If the Corporation shall be authorized to issue more than one class of stock or more than one series of any class, the Corporation shall: (a) cause the powers, designations, preferences and relative, participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences and/or rights to be set forth in full or summarized on the face or back of any certificate that the Corporation issues to represent shares of such class or series of stock; or (b) in the case of uncertificated shares, within a reasonable time after the issuance or transfer of such shares, send to the registered owner thereof a written notice containing the information required to be set forth on certificates as specified in clause (a) above; <u>provided</u>, <u>however</u>, that, except as otherwise provided by applicable law, in lieu of the foregoing requirements, there may be set forth on the face or back of such certificate or, in the case of uncertificated shares, on such written notice a statement that the Corporation will furnish without charge to each stockholder who so requests the powers, designations, preferences and relative, participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences or rights.

**Section 7.3 Signatures**. Each certificate representing capital stock of the Corporation shall be signed by or in the name of the Corporation by: (a) the Chairman of the Board, Chief Executive Officer, the President or a Vice President; and (b) the Treasurer, an Assistant Treasurer, the Secretary or an Assistant Secretary of the Corporation. Any or all of the signatures on the certificate may be a facsimile. In case any officer, transfer agent or registrar who has signed or whose facsimile signature has been placed upon a certificate shall have ceased to be such officer, transfer agent or registrar before such certificate is issued, such certificate may be issued by the Corporation with the same effect as if such person were such officer, transfer agent or registrar on the date of issue.

#### **Section 7.4** Consideration and Payment for Shares.

- (a) Subject to applicable law and the Certificate of Incorporation, shares of stock may be issued for such consideration, having in the case of shares with par value a value not less than the par value thereof, and to such persons, as determined from time to time by the Board. The consideration may consist of any tangible or intangible property or any benefit to the Corporation, including cash, promissory notes, services performed, contracts for services to be performed or other securities, or any combination thereof.
- (b) Subject to applicable law and the Certificate of Incorporation, shares may not be issued until the full amount of the consideration has been paid, unless upon the face or back of each certificate issued to represent any partly paid shares of capital stock or upon the books and records of the Corporation in the case of partly paid uncertificated shares, there shall have been set forth the total amount of the consideration to be paid therefor and the amount paid thereon up to and including the time said certificate representing certificated shares or said uncertificated shares are issued.

# Section 7.5 Lost, Destroyed or Wrongfully Taken Certificates.

- (a) If an owner of a certificate representing shares claims that such certificate has been lost, destroyed or wrongfully taken, the Corporation shall issue a new certificate representing such shares or such shares in uncertificated form if the owner: (i) requests such a new certificate before the Corporation has notice that the certificate representing such shares has been acquired by a protected purchaser; (ii) if requested by the Corporation, delivers to the Corporation a bond sufficient to indemnify the Corporation against any claim that may be made against the Corporation on account of the alleged loss, wrongful taking or destruction of such certificate or the issuance of such new certificate or uncertificated shares; and (iii) satisfies other reasonable requirements imposed by the Corporation.
- (b) If a certificate representing shares has been lost, apparently destroyed or wrongfully taken, and the owner fails to notify the Corporation of that fact within a reasonable time after the owner has notice of such loss, apparent destruction or wrongful taking and the Corporation registers a transfer of such shares before receiving notification, the owner shall be precluded from asserting against the Corporation any claim for registering such transfer or a claim to a new certificate representing such shares or such shares in uncertificated form.

# Section 7.6 Transfer of Stock.

- (a) If a certificate representing shares of the Corporation is presented to the Corporation with an endorsement requesting the registration of transfer of such shares or an instruction is presented to the Corporation requesting the registration of transfer of uncertificated shares, the Corporation shall register the transfer as requested if:
- (i) in the case of certificated shares, the certificate representing such shares has been surrendered; (A) with respect to certificated shares, the endorsement is made by the person specified by the certificate as entitled to such shares; (B) with respect to uncertificated shares, an instruction is made by the registered owner of such uncertificated shares; or (C) with respect to certificated shares or uncertificated shares, the endorsement or instruction is made by any other appropriate person or by an agent who has actual authority to act on behalf of the appropriate person;

- (ii) the Corporation has received a guarantee of signature of the person signing such endorsement or instruction or such other reasonable assurance that the endorsement or instruction is genuine and authorized as the Corporation may request;
- (iii) the transfer does not violate any restriction on transfer imposed by the Corporation that is enforceable in accordance with Section 7.8(a); and
  - (iv) such other conditions for such transfer as shall be provided for under applicable law have been satisfied.
- (b) Whenever any transfer of shares shall be made for collateral security and not absolutely, the Corporation shall so record such fact in the entry of transfer if, when the certificate for such shares is presented to the Corporation for transfer or, if such shares are uncertificated, when the instruction for registration of transfer thereof is presented to the Corporation, both the transferor and transferee request the Corporation to do so.
- **Section 7.7 Registered Stockholders**. Before due presentment for registration of transfer of a certificate representing shares of the Corporation or of an instruction requesting registration of transfer of uncertificated shares, the Corporation may treat the registered owner as the person exclusively entitled to inspect for any proper purpose the stock ledger and the other books and records of the Corporation, vote such shares, receive dividends or notifications with respect to such shares and otherwise exercise all the rights and powers of the owner of such shares, except that a person who is the beneficial owner of such shares (if held in a voting trust or by a nominee on behalf of such person) may, upon providing documentary evidence of beneficial ownership of such shares and satisfying such other conditions as are provided under applicable law, may also so inspect the books and records of the Corporation.

# Section 7.8 Effect of the Corporation's Restriction on Transfer.

- (a) A written restriction on the transfer or registration of transfer of shares of the Corporation or on the amount of shares of the Corporation that may be owned by any person or group of persons, if permitted by the DGCL and noted conspicuously on the certificate representing such shares or, in the case of uncertificated shares, contained in a notice, offering circular or prospectus sent by the Corporation to the registered owner of such shares within a reasonable time prior to or after the issuance or transfer of such shares, may be enforced against the holder of such shares or any successor or transferee of the holder including an executor, administrator, trustee, guardian or other fiduciary entrusted with like responsibility for the person or estate of the holder.
- (b) A restriction imposed by the Corporation on the transfer or the registration of shares of the Corporation or on the amount of shares of the Corporation that may be owned by any person or group of persons, even if otherwise lawful, is ineffective against a person without actual knowledge of such restriction unless: (i) the shares are certificated and such restriction is noted conspicuously on the certificate; or (ii) the shares are uncertificated and such restriction was contained in a notice, offering circular or prospectus sent by the Corporation to the registered owner of such shares within a reasonable time prior to or after the issuance or transfer of such shares.

**Section 7.9 Regulations**. The Board shall have power and authority to make such additional rules and regulations, subject to any applicable requirement of law, as the Board may deem necessary and appropriate with respect to the issue, transfer or registration of transfer of shares of stock or certificates representing shares. The Board may appoint one or more transfer agents or registrars and may require for the validity thereof that certificates representing shares bear the signature of any transfer agent or registrar so appointed.

#### ARTICLE VIII

#### INDEMNIFICATION

Section 8.1 Right to Indemnification. To the fullest extent permitted by applicable law, as the same exists or may hereafter be amended, the Corporation shall indemnify and hold harmless each person who was or is made a party or is threatened to be made a party to or is otherwise involved in any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (a "proceeding"), by reason of the fact that he or she is or was a director or officer of the Corporation or, while a director or officer of the Corporation, is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation or of a partnership, joint venture, trust, other enterprise or nonprofit entity, including service with respect to an employee benefit plan (an "Indemnitee"), whether the basis of such proceeding is alleged action in an official capacity as a director, officer, employee or agent, or in any other capacity while serving as a director, officer, employee or agent, against all liability and loss suffered and expenses (including, without limitation, attorneys' fees, judgments, fines, ERISA excise taxes and penalties and amounts paid in settlement) reasonably incurred by such Indemnitee in connection with such proceeding; provided, however, that, except as provided in Section 8.3 with respect to proceedings to enforce rights to indemnification, the Corporation shall indemnify an Indemnitee in connection with a proceeding (or part thereof) initiated by such Indemnitee only if such proceeding (or part thereof) was authorized by the Board.

Section 8.2 Right to Advancement of Expenses. In addition to the right to indemnification conferred in Section 8.1, an Indemnitee shall also have the right to be paid by the Corporation to the fullest extent not prohibited by applicable law the expenses (including, without limitation, attorneys' fees) incurred in defending or otherwise participating in any such proceeding in advance of its final disposition (hereinafter an "advancement of expenses"); provided, however, that, if the DGCL requires, an advancement of expenses incurred by an Indemnitee in his or her capacity as a director or officer of the Corporation (and not in any other capacity in which service was or is rendered by such Indemnitee, including, without limitation, service to an employee benefit plan) shall be made only upon the Corporation's receipt of an undertaking (hereinafter an "undertaking"), by or on behalf of such Indemnitee, to repay all amounts so advanced if it shall ultimately be determined that such Indemnitee is not entitled to be indemnified under this Article VIII or otherwise.

**Section 8.3 Right of Indemnitee to Bring Suit.** If a claim under Section 8.1 or Section 8.2 is not paid in full by the Corporation within 60 days after a written claim therefor has been received by the Corporation, except in the case of a claim for an advancement of expenses, in which case the applicable period shall be 20 days, the Indemnitee may at any time thereafter bring suit against the Corporation to recover the unpaid amount of the claim. If successful in whole or in part in any such suit, or in a suit brought by the Corporation to recover an advancement of expenses pursuant to the terms of an undertaking, the Indemnitee shall also be entitled to be paid the expense of prosecuting or defending such suit. In (a) any suit brought by the Indemnitee to enforce a right to indemnification hereunder (but not in a suit brought by an Indemnitee to enforce a right to an advancement of expenses) it shall be a defense that, and (b) any suit brought by the Corporation to recover an advancement of expenses pursuant to the terms of an undertaking, the Corporation shall be entitled to recover such expenses upon a final judicial decision from which there is no further right to appeal (hereinafter a "final adjudication") that, the Indemnitee has not met any applicable standard for indemnification set forth in the DGCL. Neither the failure of the Corporation (including its directors who are not parties to such action, a committee of such directors, independent legal counsel, or its stockholders) to have made a determination prior to the commencement of such suit that indemnification of the Indemnitee is proper in the circumstances because the Indemnitee has met the applicable standard of conduct set forth in the DGCL, nor an actual determination by the Corporation (including a determination by its directors who are not parties to such action, a committee of such directors, independent legal counsel, or its stockholders) that the Indemnitee has not met such applicable standard of conduct, shall create a presumption that the Indemnitee has not met the applicable standard of conduct or, in the case of such a suit brought by the Indemnitee, shall be a defense to such suit. In any suit brought by the Indemnitee to enforce a right to indemnification or to an advancement of expenses hereunder, or by the Corporation to recover an advancement of expenses pursuant to the terms of an undertaking, the burden of proving that the Indemnitee is not entitled to be indemnified, or to such advancement of expenses, under this Article VIII or otherwise shall be on the Corporation.

**Section 8.4 Non-Exclusivity of Rights.** The rights provided to any Indemnitee pursuant to this <u>Article VIII</u> shall not be exclusive of any other right, which such Indemnitee may have or hereafter acquire under applicable law, the Certificate of Incorporation, these By Laws, an agreement, a vote of stockholders or disinterested directors, or otherwise.

**Section 8.5 Insurance**. The Corporation may maintain insurance, at its expense, to protect itself and/or any director, officer, employee or agent of the Corporation or another corporation, partnership, joint venture, trust or other enterprise against any expense, liability or loss, whether or not the Corporation would have the power to indemnify such person against such expense, liability or loss under the DGCL.

**Section 8.6 Indemnification of Other Persons.** This <u>Article VIII</u> shall not limit the right of the Corporation to the extent and in the manner authorized or permitted by law to indemnify and to advance expenses to persons other than Indemnitees. Without limiting the foregoing, the Corporation may, to the extent authorized from time to time by the Board, grant rights to indemnification and to the advancement of expenses to any employee or agent of the Corporation and to any other person who is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation or of a partnership, joint venture, trust or other enterprise, including service with respect to an employee benefit plan, to the fullest extent of the provisions of this Article VIII with respect to the indemnification and advancement of expenses of Indemnitees under this Article VIII.

**Section 8.7 Amendments**. Any repeal or amendment of this <u>Article VIII</u> by the Board or the stockholders of the Corporation or by changes in applicable law, or the adoption of any other provision of these By Laws inconsistent with this <u>Article VIII</u>, will, to the extent permitted by applicable law, be prospective only (except to the extent such amendment or change in applicable law permits the Corporation to provide broader indemnification rights to Indemnitees on a retroactive basis than permitted prior thereto), and will not in any way diminish or adversely affect any right or protection existing hereunder in respect of any act or omission occurring prior to such repeal or amendment or adoption of such inconsistent provision; <u>provided</u>, <u>however</u>, that amendments or repeals of this <u>Article VIII</u> shall require the affirmative vote of the stockholders holding at least 66.7% of the voting power of all outstanding shares of capital stock of the Corporation.

**Section 8.8 Certain Definitions.** For purposes of this <u>Article VIII</u>: (a) references to "other enterprise" shall include any employee benefit plan; (b) references to "fines" shall include any excise taxes assessed on a person with respect to an employee benefit plan; (c) references to "serving at the request of the Corporation" shall include any service that imposes duties on, or involves services by, a person with respect to any employee benefit plan, its participants, or beneficiaries; and (d) a person who acted in good faith and in a manner such person reasonably believed to be in the interest of the participants and beneficiaries of an employee benefit plan shall be deemed to have acted in a manner "not opposed to the best interest of the Corporation" for purposes of Section 145 of the DGCL.

**Section 8.9 Contract Rights**. The rights provided to Indemnitees pursuant to this <u>Article VIII</u> shall be contract rights and such rights shall continue as to an Indemnitee who has ceased to be a director, officer, agent or employee and shall inure to the benefit of the Indemnitee's heirs, executors and administrators.

**Section 8.10 Severability**. If any provision or provisions of this <u>Article VIII</u> shall be held to be invalid, illegal or unenforceable for any reason whatsoever: (a) the validity, legality and enforceability of the remaining provisions of this <u>Article VIII</u> shall not in any way be affected or impaired thereby; and (b) to the fullest extent possible, the provisions of this <u>Article VIII</u> (including, without limitation, each such portion of this <u>Article VIII</u> containing any such provision held to be invalid, illegal or unenforceable) shall be construed so as to give effect to the intent manifested by the provision held invalid, illegal or unenforceable.

# ARTICLE IX

#### **MISCELLANEOUS**

**Section 9.1 Place of Meetings**. If the place of any meeting of stockholders, the Board or committee of the Board for which notice is required under these By Laws is not designated in the notice of such meeting, such meeting shall be held at the principal business office of the Corporation; <a href="provided">provided</a>, <a href="however">however</a>, if the Board has, in its sole discretion, determined that a meeting shall not be held at any place, but instead shall be held by means of remote communication pursuant to <a href="Section 9.5">Section 9.5</a> hereof, then such meeting shall not be held at any place.

#### Section 9.2 Fixing Record Dates.

- (a) In order that the Corporation may determine the stockholders entitled to notice of or to vote at any meeting of stockholders or any adjournment thereof, the Board may fix a record date, which shall not precede the date upon which the resolution fixing the record date is adopted by the Board, and which record date shall not be more than 60 nor less than 10 days before the date of such meeting. If the Board so fixes a date, such date shall also be the record date for determining the stockholders entitled to vote at such meeting unless the Board determines, at the time it fixes such record date, that a later date on or before the date of the meeting shall be the date for making such determination. If no record date is fixed by the Board, the record date for determining stockholders entitled to notice of or to vote at a meeting of stockholders shall be at the close of business on the business day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the business day next preceding the day on which the meeting is held. A determination of stockholders of record entitled to notice of or to vote at a meeting of stockholders shall apply to any adjournment of the meeting; provided, however, that the Board may fix a new record date for the adjourned meeting, and in such case shall also fix as the record date for stockholders entitled to notice of such adjourned meeting the same or an earlier date as that fixed for determination of stockholders entitled to vote in accordance with the foregoing provisions of this Section 9.2(a) at the adjourned meeting.
- (b) In order that the Corporation may determine the stockholders entitled to receive payment of any dividend or other distribution or allotment of any rights or the stockholders entitled to exercise any rights in respect of any change, conversion or exchange of stock, or for the purpose of any other lawful action, the Board may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted, and which record date shall be not more than 60 days prior to such action. If no record date is fixed, the record date for determining stockholders for any such purpose shall be at the close of business on the day on which the Board adopts the resolution relating thereto.

# Section 9.3 Means of Giving Notice.

(a) Notice to Directors. Whenever under applicable law, the Certificate of Incorporation or these By Laws notice is required to be given to any director, such notice may be given: (i) in writing and sent either by hand delivery, through the United States mail, or by a nationally recognized delivery service; (ii) by means of facsimile telecommunication or other form of electronic transmission; or (iii) by oral notice given personally or by telephone. A notice to a director will be deemed given as follows: (A) if given by hand delivery, orally, or by telephone, when actually received by the director; (B) if sent through the United States mail, when deposited in the United States mail, with postage and fees thereon prepaid, addressed to the director at the director's address appearing on the records of the Corporation; (C) if sent for by a nationally recognized delivery service, when deposited with such service, with fees thereon prepaid, addressed to the director at the director's address appearing on the records of the Corporation; (D) if sent by facsimile telecommunication, when sent to the facsimile transmission number for such director appearing on the records of the Corporation; (E) if sent by electronic mail, when sent to the electronic mail address for such director appearing on the records of the Corporation; or (F) if sent by any other form of electronic transmission, when sent to the address, location or number (as applicable) for such director appearing on the records of the Corporation.

- (b) Notice to Stockholders. Whenever under applicable law, the Certificate of Incorporation or these By Laws notice is required to be given to any stockholder, such notice may be given: (i) in writing and sent either by hand delivery, through the United States mail, or by a nationally recognized delivery service; or (ii) by means of a form of electronic transmission consented to by the stockholder, to the extent permitted by, and subject to the conditions set forth in Section 232 of the DGCL. A notice to a stockholder shall be deemed given as follows: (A) if given by hand delivery, when actually received by the stockholder; (B) if sent through the United States mail, when deposited in the United States mail, with postage and fees thereon prepaid, addressed to the stockholder at the stockholder's address appearing on the stock ledger of the Corporation; (C) if sent for delivery by a nationally recognized delivery service, when deposited with such service, with fees thereon prepaid, addressed to the stockholder at the stockholder's address appearing on the stock ledger of the Corporation; and (D) if given by a form of electronic transmission consented to by the stockholder to whom the notice is given and otherwise meeting the requirements set forth above: (1) if by facsimile transmission, when directed to a number at which the stockholder has consented to receive notice; (2) if by electronic mail, when directed to an electronic mail address at which the stockholder has consented to receive notice; (3) if by a posting on an electronic network together with separate notice to the stockholder of such specified posting, upon the later of: (x) such posting; and (y) the giving of such separate notice; and (4) if by any other form of electronic transmission, when directed to the stockholder. A stockholder may revoke such stockholder's consent to receiving notice by means of electronic communication by giving written notice of such revocation to the Corporation. Any such consent shall be deemed revoked if: (x) the Corporation is unable to deliver by electronic transmission two consecutive notices given by the Corporation in accordance with such consent; and (y) such inability becomes known to the Secretary or an Assistant Secretary or to the Corporation's transfer agent, or other person responsible for the giving of notice; provided, however, the inadvertent failure to treat such inability as a revocation shall not invalidate any meeting or other action.
- (c) <u>Electronic Transmission</u>. "*Electronic transmission*" means any form of communication, not directly involving the physical transmission of paper, that creates a record that may be retained, retrieved and reviewed by a recipient thereof, and that may be directly reproduced in paper form by such a recipient through an automated process, including but not limited to transmission by telex, facsimile telecommunication, electronic mail, telegram and cablegram.
- (d) Notice to Stockholders Sharing Same Address. Without limiting the manner by which notice otherwise may be given effectively by the Corporation to stockholders, any notice to stockholders given by the Corporation under any provision of the DGCL, the Certificate of Incorporation or these By Laws shall be effective if given by a single written notice to stockholders who share an address if consented to by the stockholders at that address to whom such notice is given. A stockholder may revoke such stockholder's consent by delivering written notice of such revocation to the Corporation. Any stockholder who fails to object in writing to the Corporation within 60 days of having been given written notice by the Corporation of its intention to send such a single written notice shall be deemed to have consented to receiving such single written notice.
- (e) Exceptions to Notice Requirements. Whenever notice is required to be given, under the DGCL, the Certificate of Incorporation or these By Laws, to any person with whom communication is unlawful, the giving of such notice to such person shall not be required

and there shall be no duty to apply to any governmental authority or agency for a license or permit to give such notice to such person. Any action or meeting that shall be taken or held without notice to any such person with whom communication is unlawful shall have the same force and effect as if such notice had been duly given. In the event that the action taken by the Corporation is such as to require the filing of a certificate with the Secretary of State of Delaware, the certificate shall state, if such is the fact and if notice is required, that notice was given to all persons entitled to receive notice except such persons with whom communication is unlawful.

Whenever notice is required to be given by the Corporation, under any provision of the DGCL, the Certificate of Incorporation or these By Laws, to any stockholder to whom: (1) notice of two consecutive annual meetings of stockholders and all notices of stockholder meetings to such stockholder during the period between such two consecutive annual meetings; or (2) all, and at least two payments (if sent by first-class mail) of dividends or interest on securities during a 12-month period, have been mailed addressed to such stockholder at such stockholder's address as shown on the records of the Corporation and have been returned undeliverable, the giving of such notice to such stockholder shall not be required. Any action or meeting that shall be taken or held without notice to such stockholder shall have the same force and effect as if such notice had been duly given. If any such stockholder shall deliver to the Corporation a written notice setting forth such stockholder's then current address, the requirement that notice be given to such stockholder shall be reinstated. In the event that the action taken by the Corporation is such as to require the filing of a certificate with the Secretary of State of Delaware, the certificate need not state that notice was not given to persons to whom notice was not required to be given pursuant to Section 230(b) of the DGCL. The exception in subsection (1) of the first sentence of this paragraph to the requirement that notice be given shall not be applicable to any notice returned as undeliverable if the notice was given by electronic transmission.

**Section 9.4** Waiver of Notice. Whenever any notice is required to be given under applicable law, the Certificate of Incorporation, or these By Laws, a written waiver of such notice, signed by the person or persons entitled to said notice, or a waiver by electronic transmission by the person entitled to said notice, whether before or after the time stated therein, shall be deemed equivalent to such required notice. All such waivers shall be kept with the books of the Corporation. Attendance at a meeting shall constitute a waiver of notice of such meeting, except where a person attends for the express purpose of objecting to the transaction of any business on the ground that the meeting was not lawfully called or convened.

# Section 9.5 Meeting Attendance via Remote Communication Equipment.

- (a) <u>Stockholder Meetings</u>. If authorized by the Board in its sole discretion, and subject to such guidelines and procedures as the Board may adopt, stockholders entitled to vote at such meeting and proxy holders not physically present at a meeting of stockholders may, by means of remote communication:
  - (i) participate in a meeting of stockholders; and
- (ii) be deemed present in person and vote at a meeting of stockholders, whether such meeting is to be held at a designated place or solely by means of remote communication; provided that: (A) the Corporation shall implement reasonable measures to verify

that each person deemed present and permitted to vote at the meeting by means of remote communication is a stockholder or proxy holder; (B) the Corporation shall implement reasonable measures to provide such stockholders and proxy holders a reasonable opportunity to participate in the meeting and, if entitled to vote, to vote on matters submitted to the applicable stockholders, including an opportunity to read or hear the proceedings of the meeting substantially concurrently with such proceedings; and (C) if any stockholder or proxy holder votes or takes other action at the meeting by means of remote communication, a record of such votes or other action shall be maintained by the Corporation.

- (b) <u>Board Meetings</u>. Unless otherwise restricted by applicable law, the Certificate of Incorporation or these By Laws, members of the Board or any committee thereof may participate in a meeting of the Board or any committee thereof by means of conference telephone or other communications equipment by means of which all persons participating in the meeting can hear each other. Such participation in a meeting shall constitute presence in person at the meeting, except where a person participates in the meeting for the express purpose of objecting to the transaction of any business on the ground that the meeting was not lawfully called or convened.
- **Section 9.6 Dividends**. The Board may from time to time declare, and the Corporation may pay, dividends (payable in cash, property or shares of the Corporation's capital stock) on the Corporation's outstanding shares of capital stock, subject to applicable law and the Certificate of Incorporation.
- **Section 9.7 Reserves.** The Board may set apart out of the funds of the Corporation available for dividends a reserve or reserves for any proper purpose and may abolish any such reserve.
- Section 9.8 Contracts and Negotiable Instruments. Except as otherwise provided by applicable law, the Certificate of Incorporation or these By Laws, any contract, bond, deed, lease, mortgage or other instrument may be executed and delivered in the name and on behalf of the Corporation by such officer or officers or other employee or employees of the Corporation as the Board may from time to time authorize. Such authority may be general or confined to specific instances as the Board may determine. The Chairman of the Board, the Chief Executive Officer, the President, the Chief Financial Officer, the Treasurer or any Vice President may execute and deliver any contract, bond, deed, lease, mortgage or other instrument in the name and on behalf of the Corporation. Subject to any restrictions imposed by the Board, the Chairman of the Board, Chief Executive Officer, President, the Chief Financial Officer, the Treasurer or any Vice President may delegate powers to execute and deliver any contract, bond, deed, lease, mortgage or other instrument in the name and on behalf of the Corporation to other officers or employees of the Corporation under such person's supervision and authority, it being understood, however, that any such delegation of power shall not relieve such officer of responsibility with respect to the exercise of such delegated power.

**Section 9.9 Fiscal Year**. The fiscal year of the Corporation shall be fixed by the Board.

- **Section 9.10 Seal.** The Board may adopt a corporate seal, which shall be in such form as the Board determines. The seal may be used by causing it or a facsimile thereof to be impressed, affixed or otherwise reproduced.
- **Section 9.11 Books and Records**. The books and records of the Corporation may be kept within or outside the State of Delaware at such place or places as may from time to time be designated by the Board.
- **Section 9.12 Resignation**. Any director, committee member or officer may resign by giving notice thereof in writing or by electronic transmission to the Chairman of the Board, the Chief Executive Officer, the President or the Secretary. The resignation shall take effect at the time it is delivered unless the resignation specifies a later effective date or an effective date determined upon the happening of an event or events. Unless otherwise specified therein, the acceptance of such resignation shall not be necessary to make it effective.
- Section 9.13 Surety Bonds. Such officers, employees and agents of the Corporation (if any) as the Chairman of the Board, Chief Executive Officer, President or the Board may direct, from time to time, shall be bonded for the faithful performance of their duties and for the restoration to the Corporation, in case of their death, resignation, retirement, disqualification or removal from office, of all books, papers, vouchers, money and other property of whatever kind in their possession or under their control belonging to the Corporation, in such amounts and by such surety companies as the Chairman of the Board, Chief Executive Officer, President or the Board may determine. The premiums on such bonds shall be paid by the Corporation and the bonds so furnished shall be in the custody of the Secretary.
- Section 9.14 Securities of Other Corporations. Powers of attorney, proxies, waivers of notice of meeting, consents in writing and other instruments relating to securities owned by the Corporation may be executed in the name of and on behalf of the Corporation by the Chairman of the Board, Chief Executive Officer, President, any Vice President or any officers authorized by the Board. Any such officer, may, in the name of and on behalf of the Corporation, take all such action as any such officer may deem advisable to vote in person or by proxy at any meeting of security holders of any corporation in which the Corporation may own securities, or to consent in writing, in the name of the Corporation as such holder, to any action by such corporation, and at any such meeting or with respect to any such consent shall possess and may exercise any and all rights and power incident to the ownership of such securities and which, as the owner thereof, the Corporation might have exercised and possessed. The Board may from time to time confer like powers upon any other person or persons.
- **Section 9.15** Amendments. The Board shall have the power to adopt, amend, alter or repeal the By Laws. The affirmative vote of a majority of the Board shall be required to adopt, amend, alter or repeal the By Laws. The By Laws also may be adopted, amended, altered or repealed by the stockholders; <u>provided</u>, <u>however</u>, that in addition to any vote of the holders of any class or series of capital stock of the Corporation required by applicable law or the Certificate of Incorporation, the affirmative vote of the holders of at least a majority of the voting power (except as otherwise provided in <u>Section 8.7</u>) of all outstanding shares of capital stock of the Corporation entitled to vote generally in the election of directors, voting together as a single class, shall be required for the stockholders to adopt, amend, alter or repeal the By Laws.

#### AMENDED AND RESTATED REGISTRATION RIGHTS AGREEMENT

THIS AMENDED AND RESTATED REGISTRATION RIGHTS AGREEMENT (this "Agreement"), dated as of May 4, 2021, is made and entered into by and among The Beauty Health Company, a Delaware corporation (the "Company"), BLS Investor Group LLC, a Delaware limited liability company (the "Sponsor"), the undersigned parties listed as Existing Holders on the signature pages hereto (each such party, together with the Sponsor and any person or entity deemed an "Existing Holder" who is a Permitted Transferee (as defined below) of an Existing Holder and hereafter becomes a party to this Agreement pursuant to Section 5.2 of this Agreement, an "Existing Holder" and, collectively, the "Existing Holders") and the undersigned parties listed as New Holders on the signature pages hereto (each such party, together with any Person deemed a "New Holder" who is a Permitted Transferee of a New Holder and hereafter becomes a party to this Agreement pursuant to Section 5.2 of this Agreement, a "New Holder" and, collectively, the "New Holders"). Capitalized terms used but not otherwise defined in this Agreement shall have the meaning ascribed to such terms in the Merger Agreement (as defined below).

#### RECITALS

**WHEREAS**, on September 29, 2020, the Company and the Sponsor entered into that certain Registration Rights Agreement (the "*Existing Registration Rights Agreement*"), pursuant to which the Company granted the Existing Holders certain registration rights with respect to certain securities of the Company;

**WHEREAS**, the Company has entered into that certain Agreement and Plan of Merger (the "*Merger Agreement*"), dated as of December 8, 2020, by and among the Company, Hydrate Merger Sub, Inc., a Delaware corporation, LCP Edge Intermediate, Inc., a Delaware corporation, and LCP Edge Holdco, LLC, a Delaware limited liability company;

**WHEREAS**, upon the closing of the transactions contemplated by the Merger Agreement and subject to the terms and conditions set forth therein, the Existing Holders and New Holders will hold shares of Class A common stock, par value \$0.0001 per share, of the Company ("*Class A Common Stock*"), in each case, in such amounts and subject to such terms and conditions as set forth in the Merger Agreement;

**WHEREAS**, in connection with the transactions contemplated by the Merger Agreement, the Company is conducting a private placement of its Class A Common Stock (the "*PIPE Investment*") pursuant to the terms of one or more Subscription Agreements;

**WHEREAS**, certain New Holders may receive additional shares of Class A Common Stock (the "*Earnout Shares*") pursuant to certain provisions in the Merger Agreement;

**WHEREAS**, pursuant to Section 5.5 of the Existing Registration Rights Agreement, the provisions, covenants and conditions set forth therein may be amended or modified upon the written consent of the Company and the holders of a majority-in-interest of the "Registrable Securities" (as such term is defined in the Existing Registration Rights Agreement) at the time in question; and

**WHEREAS**, the Company and Sponsor desire to amend and restate the Existing Registration Rights Agreement in order to provide the Existing Holders and the New Holders certain registration rights with respect to certain securities of the Company, as set forth in this Agreement.

**NOW, THEREFORE**, in consideration of the representations, covenants and agreements contained herein, and certain other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

# ARTICLE I DEFINITIONS

1.1 <u>Definitions</u>. The terms defined in this <u>Article I</u> shall, for all purposes of this Agreement, have the respective meanings set forth below:

"Adverse Disclosure" shall mean any public disclosure of material non-public information, which disclosure, in the good-faith determination of the Company, after consultation with counsel to the Company, (i) would be required to be made in any Registration Statement or Prospectus in order for the applicable Registration Statement or Prospectus not to contain any Misstatement, (ii) would not be required to be made at such time if the Registration Statement were not being filed, and (iii) the Company has a bona fide business purpose for not making such information public.

"Affiliate" means, with respect to any Person, any other Person who, directly or indirectly, controls, is controlled by, or is under direct or indirect common control with, such Person, and, in the case of an individual, also includes any member of such individual's Immediate Family; provided that the Company and its subsidiaries will not be deemed to be Affiliates of any holder of Registrable Securities. As used in this definition, "control," (including, with its correlative meanings, "controlling," "controlled by" and "under common control") shall mean possession, directly or indirectly, of power to direct or cause the direction of the management and policies of a Person, directly or indirectly, whether through ownership of voting securities or partnership or other ownership interests, by contract or otherwise.

"Agreement" shall have the meaning given in the Preamble.

"*Block Trade*" means an offering or sale of Registrable Securities by any Holder on a block trade or underwritten basis (whether firm commitment or otherwise) effected pursuant to a Registration Statement without substantial marketing efforts prior to pricing, including, without limitation, a same day trade, overnight trade or similar transaction.

"Board" shall mean the Board of Directors of the Company.

"Business Day" shall mean a day that is not a Saturday or Sunday or a day on which banks in New York City are authorized or requested by law to close.

"Class A Common Stock" shall have the meaning given in the Recitals hereto.

- "Class B Common Stock" shall mean Class B common stock, par value \$0.0001 per share, of the Company.
- "Closing Date" shall mean the date of the consummation of the transactions contemplated by the Merger Agreement.
- "Commission" shall mean the Securities and Exchange Commission.
- "Company" shall have the meaning given in the Preamble.
- "Company Shelf Takedown Notice" shall have the meaning given in subsection 2.1.3.
- "Demand Registration" shall have the meaning given in subsection 2.2.1.
- "Demanding Holders" shall mean, as applicable, (a) the Existing Holders of at least a majority in interest of the then-outstanding number of Registrable Securities held by the Existing Holders or (b) the New Holders of at least a majority in interest of the then-outstanding number of Registrable Securities held by the New Holders.
  - "Earnout Shares" shall have the meaning given in the Recitals hereto.
  - "Effectiveness Deadline" shall have the meaning given in subsection 2.1.1.
  - "Exchange Act" shall mean the Securities Exchange Act of 1934, as it may be amended from time to time.
  - "Existing Registration Rights Agreement" shall have the meaning given in the Recitals hereto.
  - "Form S-3 Shelf" shall have the meaning given in subsection 2.1.2.
- "Founder Shares" shall mean all shares of Class B Common Stock that are issued and outstanding as of the date hereof and all shares of Class A Common Stock issued upon conversion thereof.
- "Founder Shares Lock-up Period" shall mean, with respect to the Founder Shares held by the Existing Holders or their Permitted Transferees, the period ending on the earlier of (A) one year after the date hereof, (B) the first date that the closing price of the Class A Common Stock equals or exceeds \$12.00 per share (as adjusted for stock splits, stock dividends, reorganizations, recapitalizations and the like) for any twenty (20) trading days within any thirty (30)-trading-day period commencing at least one hundred and fifty (150) days after the date hereof, and (C) the date on which the Company completes a liquidation, merger, capital stock exchange, reorganization or other similar transaction that results in all of the Company's stockholders having the right to exchange their shares of Class A Common Stock for cash, securities or other property.
- "*Holders*" shall mean the Existing Holders and the New Holders and any person or entity who hereafter becomes a party to this Agreement pursuant to <u>Section 5.2</u>.

- "Immediate Family" shall mean any child, stepchild, grandchild, parent, stepparent, grandparent, spouse, sibling, mother-in-law, father-in-law, son-in-law, daughter-in-law, brother-in-law, sister-in-law and shall include adoptive relationships.
- "Insider Letter" shall mean that certain letter agreement, dated as of September 29, 2020, by and among the Company, the Sponsor and each of the Company's officers, directors and director nominees.
  - "Maximum Number of Securities" shall have the meaning given in subsection 2.2.4.
  - "Merger Agreement" shall have the meaning given in the Recitals hereto.
  - "Minimum Amount" shall have the meaning given in subsection 2.1.3.
- "*Misstatement*" shall mean an untrue statement of a material fact or an omission to state a material fact required to be stated in a Registration Statement or Prospectus, or necessary to make the statements in a Registration Statement or Prospectus (in the case of any Prospectus, in the light of the circumstances under which they were made) not misleading.
  - "Nasdaq" shall have the meaning given in subsection 3.1.4.
  - "New Holder(s)" shall have the meaning given in the Preamble.
  - "Participating Holder" shall have the meaning given in subsection 2.6.
- "Permitted Transferees" shall mean (a) with respect to an Existing Holder, any Person to whom a Holder of Registrable Securities is permitted to transfer such Registrable Securities prior to the expiration of the Founder Shares Lock-up Period, Private Placement Lock-Up Period or any other lock-up period, as the case may be, under the Insider Letter, the Private Placement Warrants Purchase Agreement, this Agreement and any other applicable agreement between such Existing Holder and the Company, and to any transferee thereafter; and (b) with respect to a New Holder, (i) in the case of an individual (1) by gift to a member of the individual's Immediate Family, to a trust, the beneficiary of which is a member of the individual's Immediate Family or an Affiliate of such Person, or to a charitable organization, (2) by virtue of laws of descent and distribution upon death of the individual and (3) pursuant to a qualified domestic relations order or (ii) in the case of an entity, (1) by distribution to such entity's members, partners, stockholders or equityholders, (2) to any of such entity's Affiliates or to any fund or other entity controlled or managed by such entity or any of its Affiliates, or to investment manager or investment advisor and (3) to a nominee or custodian of a person or entity to whom a disposition or transfer would be permissible under clause (b) of this definition, provided that such transferee to which a transfer is being made pursuant to clause (a) or (b) above, if not a Holder, enters into a written agreement with the Company agreeing to be bound by the restrictions, including restrictions specific to certain holders, herein.
  - "Person" shall mean any individual, corporation, partnership, unincorporated association or other entity.
  - "Piggyback Registration" shall have the meaning given in subsection 2.3.1.

"Private Placement Lock-up Period" shall mean, with respect to Private Placement Warrants that are held by the initial purchasers of such Private Placement Warrants or their Permitted Transferees, the Private Placement Warrants and shares of Class A Common Stock issuable upon the exercise or conversion of the Private Placement Warrants, and that are held by the initial purchasers of the Private Placement Warrants or their Permitted Transferees, the period ending thirty (30) days after the date hereof.

"*Private Placement Warrants*" shall mean the warrants to purchase shares of Class A Common Stock purchased by the Sponsor pursuant to the Private Placement Warrants Purchase Agreement.

"Private Placement Warrants Purchase Agreement" shall mean that certain Private Placement Warrants Purchase Agreement by and between the Company and the Sponsor, dated as of September 29, 2020.

"Pro Rata" shall have the meaning given in subsection 2.2.4.

"*Prospectus*" shall mean the prospectus included in any Registration Statement, as supplemented by any and all prospectus supplements and as amended by any and all post-effective amendments and including all material incorporated by reference in such prospectus.

"Registrable Security" shall mean (a) the Founder Shares and the shares of Class A Common Stock issued or issuable upon the conversion of the Founder Shares, (b) the Private Placement Warrants (including any shares of Class A Common Stock issued or issuable upon the exercise of the Private Placement Warrants), (c) any issued and outstanding shares of Class A Common Stock or any other equity security (including the shares of Class A Common Stock issued or issuable upon the exercise of any other equity security) of the Company held by a Holder as of the date of this Agreement, (d) any equity securities (including the shares of Class A Common Stock issued or issuable upon the exercise of any such equity security) of the Company issuable upon conversion of any working capital loans in an amount up to \$1,500,000 made to the Company by a Holder (including the Working Capital Warrants and shares of Class A Common Stock issued or issuable upon the exercise of the Working Capital Warrants), (e) any outstanding shares of Class A Common Stock or any other equity security of the Company held by a New Holder issued in connection with the transactions contemplated by the Merger Agreement (including any Earnout Shares), (f) any other equity securities (including shares of Class A Common Stock) of the Company acquired by a New Holder at a time that they otherwise hold Registrable Securities and (g) any other equity security of the Company issued or issuable with respect to any such share of Class A Common Stock by way of a stock dividend or stock split or in connection with a combination of shares, recapitalization, merger, consolidation or reorganization, subject in each case to the expiration of any applicable lock-up periods; provided, however, that, as to any particular Registrable Security, such securities shall cease to be Registrable Securities when: (A) a Registration Statement with respect to the sale of such securities shall have become effective under the Securities Act and such securities shall have been sold, transferred, disposed of or exchanged in accordance with such Registration Statement; (B) such securities shall have been otherwise transferred, new certificates for such securities not bearing a legend restricting further transfer shall have been delivered by the Company and subsequent public distribution of such securities shall not require registration under the Securities Act; (C) such securities shall have ceased to be

outstanding; (D) such securities are eligible for transfer without registration pursuant to Rule 144 promulgated under the Securities Act (or any successor rule promulgated thereafter by the Commission) without volume or other restrictions or limitations; or (E) such securities have been sold to, or through, a broker, dealer or underwriter in a public distribution or other public securities transaction.

"*Registration*" shall mean a registration effected by preparing and filing a registration statement or similar document in compliance with the requirements of the Securities Act, and the applicable rules and regulations promulgated thereunder, and such registration statement becoming effective.

"Registration Expenses" shall mean the out-of-pocket expenses of a Registration, including, without limitation, the following:

- (a) all registration and filing fees (including fees with respect to filings required to be made with the Financial Industry Regulatory Authority, Inc.) and any securities exchange on which the Class A Common Stock is then listed;
- (b) fees and expenses of compliance with securities or blue sky laws (including reasonable fees and disbursements of counsel for the Underwriters in connection with blue sky qualifications of Registrable Securities);
  - (c) printing, messenger, telephone and delivery expenses;
  - (d) reasonable fees and disbursements of counsel for the Company;
- (e) reasonable fees and disbursements of all independent registered public accountants of the Company incurred specifically in connection with such Registration; and
  - (f) reasonable fees and expenses of one (1) legal counsel selected by the New Holders, not to exceed \$50,000.

"*Registration Statement*" shall mean any registration statement that covers the Registrable Securities pursuant to the provisions of this Agreement, including the Prospectus included in such registration statement, amendments (including post-effective amendments) and supplements to such registration statement, and all exhibits to and all material incorporated by reference in such registration statement.

"Requesting Holder" shall have the meaning given in subsection 2.2.1.

"Securities Act" shall mean the Securities Act of 1933, as amended, and the rules and regulations of the Commission promulgated thereafter, all as the same shall be in effect from time to time.

"Shelf Take Down Notice" shall have the meaning given in subsection 2.1.3.

"Shelf Underwritten Offering" shall mean an underwritten offering that is registered pursuant to a shelf registration statement, including a Block Trade.

"Sponsor" shall have the meaning given in the Preamble.

"Subscription Agreements" shall means those certain subscription agreements dated December 8, 2020 by and between the Company and certain subscribers to shares of Class A Common Stock.

"*Underwriter*" shall mean a securities dealer who purchases any Registrable Securities as principal in an Underwritten Offering and not as part of such dealer's market-making activities.

"*Underwritten Registration*" or "*Underwritten Offering*" shall mean a Registration in which securities of the Company are sold to an Underwriter in a firm commitment underwriting for distribution to the public.

"*Working Capital Warrants*" shall mean the warrants to purchase shares of Class A Common Stock, if any, that are converted from loans made to the Company of up to \$1,500,000 by the Sponsor or an affiliate of the Sponsor or certain of the Company's officers and directors from time to time.

# ARTICLE II REGISTRATIONS

# 2.1 Shelf Registration.

2.1.1 Shelf Registration. The Company shall, as soon as practicable, but in any event within sixty (60) days after the Closing Date (the "Filing Deadline"), file a Registration Statement under the Securities Act (the "Initial Shelf") to permit the public resale of all the Registrable Securities held by the Holders from time to time as permitted by Rule 415 under the Securities Act (or any successor or similar provision adopted by the Commission then in effect) on the terms and conditions specified in this subsection 2.1.1 and shall use its commercially reasonable efforts to cause such Initial Shelf to be declared effective as soon as practicable after the filing thereof, but in no event later than sixty (60) days following the Filing Deadline (the "Effectiveness Deadline"); provided that the Effectiveness Deadline shall be extended to ninety (90) days after the filing deadline if the Initial Shelf is reviewed by, and receives comments from, the Commission. The Initial Shelf filed with the Commission pursuant to this subsection 2.1.1 shall be on Form S-3 or, if Form S-3 is not then available to the Company, on Form S-1 or such other form of registration statement as is then available to effect a registration for resale of such Registrable Securities, covering such Registrable Securities, and shall contain a Prospectus in such form as to permit any Holder to sell such Registrable Securities pursuant to Rule 415 under the Securities Act (or any successor or similar provision adopted by the Commission then in effect) at any time beginning on the effective date for such Initial Shelf. The Initial Shelf shall provide for the resale pursuant to any method or combination of methods legally available to, and requested by, the Holders. The Company shall use its commercially reasonable efforts to cause the Initial Shelf to remain effective, and to be supplemented and amended to the extent necessary to ensure that the Initial Shelf is available or, if not available, that another Registration Statement is available, for the

Registrable Securities have ceased to be Registrable Securities. As soon as practicable following the effective date of the Initial Shelf, but in any event within five (5) Business Days of such date, the Company shall notify the Holders of the effectiveness of such the Initial Shelf.

- 2.1.2 <u>Form of Registration</u>. If the Company files the Initial Shelf on Form S-3 (a "*Form S-3 Shelf*") and thereafter the Company becomes ineligible to use Form S-3 for secondary sales, the Company shall use its commercially reasonable efforts to file the Initial Shelf on Form S-1 as promptly as practicable to replace the shelf registration statement that is on Form S-3 and have the Initial Shelf declared effective as promptly as practicable and to cause such Initial Shelf to remain effective, and to be supplemented and amended to the extent necessary to ensure that such Initial Shelf is available or, if not available, that another Registration Statement is available, for the resale of all the Registrable Securities held by the Holders until all such Registrable Securities have ceased to be Registrable Securities.
- 2.1.3 Underwritten Shelf Takedowns. At any time and from time to time following the effectiveness of the Initial Shelf, any Holder may request to sell all or a portion of their Registrable Securities in a Shelf Underwritten Offering; provided that such Holder(s) reasonably expects to sell Registrable Securities yielding aggregate gross proceeds in excess of \$50,000,000 from such Shelf Underwritten Offering (such amount of Registrable Securities, the "Minimum Amount"). Notwithstanding the foregoing, the New Holders may request to sell their Registrable Securities in a Shelf Underwritten Offering yielding less than the Minimum Amount to the extent that such request comprises all of the remaining Registrable Securities held by such New Holder. All requests for a Shelf Underwritten Offering shall be made by giving written notice to the Company (the "Shelf Take Down Notice"). Each Shelf Take Down Notice shall specify the approximate number of Registrable Securities proposed to be sold in the Shelf Underwritten Offering and the expected price range (net of underwriting discounts and commissions) of such Shelf Underwritten Offering. Except with respect to any Registrable Securities distributed by the Sponsor to its members following the expiration of the Founder Shares Lock-up Period or the Private Placement Lock-up Period, as applicable, within five (5) days after receipt of any Shelf Take Down Notice, the Company shall give written notice of such requested Shelf Underwritten Offering to all other Holders of Registrable Securities (the "Company Shelf Takedown Notice") and, subject to the provisions of subsection 2.2.4, shall include in such Shelf Underwritten Offering all Registrable Securities with respect to which the Company has received written requests for inclusion therein, within five (5) days after sending the Company Shelf Takedown Notice. The Company shall enter into an underwriting agreement in a form as is customary in Underwritten Offerings of securities by the Company with the managing Underwriter or Underwriters selected by the Holders after consultation with the Company and shall take all such other reasonable actions as are requested by the managing Underwriter or Underwriters in order to expedite or facilitate the disposition of such Registrable Securities. In connection with any Shelf Underwritten Offering contemplated by this subsection 2.1.3, subject to Section 3.3 and Article IV, the underwriting agreement into which each Holder and the Company shall enter shall contain such representations, covenants, indemnities and other rights and obligations of the Company and the selling stockholders as are customary in underwritten offerings of securities by the Company.
- 2.1.4 At least ten (10) Business Days prior to the first anticipated filing date of a Registration Statement pursuant to this Article II, the Company shall use reasonable efforts to notify each Holder in writing (which may be by email) of the information reasonably necessary

about the Holder to include such Holder's Registrable Securities in such Registration Statement. Notwithstanding anything else in this Agreement, the Company shall not be obligated to include such Holder's Registrable Securities to the extent the Company has not received such information, and received any other reasonably requested agreements or certificates, on or prior to the third (3<sup>rd</sup>) Business Day prior to the first anticipated filing date of a Registration Statement pursuant to this <u>Article II</u>.

#### 2.2 Demand Registration.

2.2.1 Request for Registration. Subject to the provisions of subsection 2.2.5 and Sections 2.4 and 3.4 hereof, and provided that the Company does not have an effective Registration Statement pursuant to subsection 2.1.1, outstanding covering Registrable Securities, following the expiration of the Founder Shares Lock-up Period, the Private Placement Lock-up Period or any other lock-up period, as the case may be, a Demanding Holder may make a written demand for Registration of all or part of their Registrable Securities, which written demand shall describe the amount and type of securities to be included in such Registration and the intended method(s) of distribution thereof (such written demand a "Demand Registration"). The Company shall, within two (2) days of the Company's receipt of the Demand Registration, notify, in writing all other Holders of Registrable Securities (other than a Demand Registration with respect to any Registrable Securities to be distributed by the Sponsor to its members following the expiration of the Founder Shares Lock-up Period, Private Placement Lock-up Period or any other lock-up period, as the case may be) of such demand, and each Holder of Registrable Securities who thereafter wishes to include all or a portion of such Holder's Registrable Securities in a Registration pursuant to a Demand Registration (each such Holder that includes all or a portion of such Holder's Registrable Securities in such Registration, a "Requesting Holder") shall so notify the Company, in writing, within five (5) days after the receipt by the Holder of the notice from the Company. Upon receipt by the Company of any such written notification from a Requesting Holder(s) to the Company, subject to subsection 2.2.4 below, such Requesting Holder(s) shall be entitled to have their Registrable Securities included in a Registration pursuant to a Demand Registration and the Company shall effect, as soon thereafter as practicable, but not more than sixty (60) days immediately after the Company's receipt of the Demand Registration, the Registration of all Registrable Securities requested by the Demanding Holders and Requesting Holders pursuant to such Demand Registration. Under no circumstances shall the Company be obligated to effect more than (A) an aggregate of two (2) Registrations pursuant to a Demand Registration initiated by the Existing Holders and (B) an aggregate of five (5) Registrations pursuant to a Demand Registration initiated by the New Holders, in each case under this <u>subsection 2.2.1</u> with respect to any or all Registrable Securities. Notwithstanding the foregoing, (i) the Company shall not be required to give effect to a Demand Registration from a Demanding Holder if the Company has registered Registrable Securities pursuant to a Demand Registration from such Demanding Holder in the preceding one hundred and eighty (180) days, or (ii) the Company's obligations with respect to any Demand Registration shall be deemed satisfied so long as the Registration Statement filed pursuant to subsection 2.1.1 includes all of such Demanding Holder's Registrable Securities and is effective.

2.2.2 <u>Effective Registration</u>. Notwithstanding the provisions of <u>subsection 2.2.1</u> above or any other part of this Agreement, a Registration pursuant to a Demand Registration shall not count as a Registration unless and until (a) the Registration Statement filed with the

Commission with respect to a Registration pursuant to a Demand Registration has been declared effective by the Commission, (b) the Company has complied with all of its obligations under this Agreement with respect thereto and (c) at least 75% of the Registrable Securities requested by the Requesting Holders to be registered on behalf of the Requesting Holders in such Registration pursuant to a Demand Registration have been sold. If, after such Registration Statement has been declared effective, an offering of Registrable Securities in a Registration pursuant to a Demand Registration is subsequently interfered with by any stop order or injunction of the Commission, federal or state court or any other governmental agency, the Registration Statement with respect to such Registration shall be deemed not to have been declared effective, unless and until, (i) such stop order or injunction is removed, rescinded or otherwise terminated, and (ii) a majority-in-interest of the Demanding Holders initiating such Demand Registration thereafter affirmatively elect to continue with such Registration and accordingly notify the Company in writing, but in no event later than five (5) days, of such election. The Company shall not be obligated or required to file another Registration Statement until the Registration Statement that has been previously filed with respect to a Registration pursuant to a Demand Registration by the same Demanding Holder becomes effective or is subsequently terminated.

- 2.2.3 <u>Underwritten Offering</u>. Subject to the provisions of <u>subsection 2.2.4</u> and <u>Sections 2.4</u> and <u>3.4</u> hereof, if a majority-in-interest of the Demanding Holders so advise the Company as part of their Demand Registration that the offering of the Registrable Securities pursuant to such Demand Registration shall be in the form of an Underwritten Offering, then the right of such Demanding Holder or Requesting Holder (if any) to include its Registrable Securities in such Registration shall be conditioned upon such Holder's participation in such Underwritten Offering and the inclusion of such Holder's Registrable Securities in such Underwritten Offering to the extent provided herein. All such Holders proposing to distribute their Registrable Securities through an Underwritten Offering under this <u>subsection 2.2.3</u>, subject to <u>Section 3.3</u> and <u>Article IV</u>, shall enter into an underwriting agreement in customary form with the Company and the Underwriter(s) selected for such Underwritten Offering by the majority-in-interest of the Demanding Holders initiating the Demand Registration, which Underwriter(s) shall be reasonably satisfactory to the Company.
- 2.2.4 Reduction of Underwritten Offering. If a Demand Registration is to be an Underwritten Offering and the managing Underwriter or Underwriters, in good faith, advises the Company, the Demanding Holders and the Requesting Holders (if any) in writing that, in its opinion, the dollar amount or number of Registrable Securities that the Demanding Holders and the Requesting Holders (if any) desire to sell, taken together with all other Class A Common Stock or other equity securities that the Company desires to sell for its own account and the Class A Common Stock, if any, as to which a Registration has been requested pursuant to separate written contractual piggy-back registration rights held by any other stockholders of the Company who desire to sell, exceeds the maximum dollar amount or maximum number of equity securities that can be sold in such Underwritten Offering without adversely affecting the proposed offering price, the timing, the distribution method, or the probability of success of such offering (such maximum dollar amount or maximum number of such securities, as applicable, the "Maximum Number of Securities"), then the Company shall include in such Underwritten Offering, as follows:
  - (a) first, the Registrable Securities of the Demanding Holders and the Requesting Holders (if any) (pro rata based on the respective number of Registrable Securities

that each Demanding Holder and Requesting Holder (if any) has requested be included in such Underwritten Registration and the aggregate number of Registrable Securities that the Demanding Holders and Requesting Holders have requested be included in such Underwritten Registration (such proportion is referred to herein as "*Pro Rata*")) that can be sold without exceeding the Maximum Number of Securities;

- (b) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing <u>clause (a)</u>, the Registrable Securities of Holders (Pro Rata, based on the respective number of Registrable Securities that each Holder has so requested) exercising their rights to register their Registrable Securities pursuant to <u>subsection 2.3.1</u> hereof, without exceeding the Maximum Number of Securities,
- (c) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing <u>clauses (a) and (b)</u>, the Class A Common Stock or other equity securities that the Company desires to sell for its own account, which can be sold without exceeding the Maximum Number of Securities; and
- (d) fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing <u>clauses (a)</u>, <u>(b)</u> and <u>(c)</u>, the Class A Common Stock or other equity securities of other persons or entities that the Company is obligated to register in a Registration pursuant to separate written contractual arrangements with such persons and that can be sold without exceeding the Maximum Number of Securities.
- 2.2.5 <u>Demand Registration Withdrawal</u>. A Demanding Holder or a Requesting Holder shall have the right to withdraw all or a portion of its Registrable Securities included in a Demand Registration pursuant to <u>subsection 2.2.1</u> or a Shelf Underwritten Offering pursuant to <u>subsection 2.1.3</u> for any or no reason whatsoever upon written notification to the Company and the Underwriter or Underwriters (if any) of its intention to so withdraw at any time prior to (a) in the case of a Demand Registration not involving an Underwritten Offering, the effectiveness of the applicable Registration Statement or (b) in the case of any Demand Registration involving an Underwritten Offering or any Shelf Underwritten Offering, prior to the pricing of such Underwritten Offering or Shelf Underwritten Offering; <u>provided</u>, <u>however</u>, that upon withdrawal by a majority-in-interest of the Demanding Holders initiating a Demand Registration (or in the case of a Shelf Underwritten Offering, withdrawal of an amount of Registrable Securities included by the Holders in such Shelf Underwritten Offering, in their capacity as Demanding Holders, being less than the Minimum Amount), the Company shall cease all efforts to secure effectiveness of the applicable Registration Statement or complete the Underwritten Offering, as applicable. For the avoidance of doubt, any Demand Registration withdrawn pursuant to this <u>subsection 2.2.5</u> shall be counted toward the aggregate number of Demand Registrations the Company is obligated to effect pursuant to <u>subsection 2.2.1</u>. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with a Registration pursuant to a Demand Registration or a Shelf Underwritten Offering prior to its withdrawal under this <u>subsection 2.2.5</u>.

#### 2.3 Piggyback Registration.

- 2.3.1 Piggyback Rights. If the Company proposes to file a Registration Statement under the Securities Act with respect to an offering of equity securities, or securities or other obligations exercisable or exchangeable for, or convertible into equity securities, for its own account or for the account of stockholders of the Company (or by the Company and by the stockholders of the Company including, without limitation, pursuant to Section 2.2 hereof), other than a Registration Statement (a) filed in connection with any employee stock option or other benefit plan, (b) for an exchange offer or offering of securities solely to the Company's existing stockholders, (c) for an offering of debt that is convertible into equity securities of the Company, for a dividend reinvestment plan, (d) for any issuances of securities in connection with a transaction involving a merger, consolidation, sale, exchange, issuance, transfer, reorganization or other extraordinary transaction between the Company or any of its Affiliates and any third party, or (e) filed pursuant to subsection 2.1.1, then the Company shall give written notice of such proposed filing to all of the Holders of Registrable Securities (excluding the Sponsor with respect to any Registrable Securities distributed by the Sponsor to its members following the expiration of the Founder Shares Lock-up Period or the Private Placement Lock-up Period, as applicable) as soon as practicable but not less than twenty (20) days before the anticipated filing date of such Registration Statement, which notice shall (i) describe the amount and type of securities to be included in such offering, the intended method(s) of distribution (including whether such registration will be pursuant to a shelf registration statement), and the proposed price and name of the proposed managing Underwriter or Underwriters, if any, in such offering, (ii) describe such Holders' rights under this Section 2.3, and (iii) offer to all of the Holders of Registrable Securities the opportunity to register the sale of such number of Registrable Securities as such Holders may request in writing within five (5) days after receipt of such written notice (such Registration, a "Piggyback Registration"). The Company shall, in good faith, cause such Registrable Securities identified in a Holder's response noticed described in the foregoing sentence to be included in such Piggyback Registration and shall use its commercially reasonable efforts to cause the managing Underwriter or Underwriters of a proposed Underwritten Offering, if any, to permit the Registrable Securities requested by the Holders pursuant to this subsection 2.3.1 to be included in a Piggyback Registration on the same terms and conditions as any similar securities of the Company or Company stockholder(s) for whose account the Registration Statement is to be filed included in such Registration and to permit the sale or other disposition of such Registrable Securities in accordance with the intended method(s) of distribution thereof. All such Holders proposing to distribute their Registrable Securities through an Underwritten Offering under this subsection 2.3.1, subject to Section 3.3 and Article IV, shall enter into an underwriting agreement in customary form with the Underwriter(s) selected for such Underwritten Offering by the Company.
- 2.3.2 <u>Reduction of Piggyback Registration</u>. If a Piggyback Registration is to be an Underwritten Offering and the managing Underwriter or Underwriters, in good faith, advises the Company and the Holders of Registrable Securities participating in the Piggyback Registration in writing that, in its opinion, the dollar amount or number of the Class A Common Stock that the Company desires to sell, taken together with (a) the Class A Common Stock, if any, as to which Registration has been demanded pursuant to separate written contractual arrangements with persons or entities other than the Holders of Registrable Securities hereunder, (b) the Registrable Securities as to which registration has been requested pursuant to <u>Section 2.3</u> hereof, and (c) the

Class A Common Stock, if any, as to which Registration has been requested pursuant to separate written contractual piggy-back registration rights of other stockholders of the Company, exceeds the Maximum Number of Securities, then:

2.3.2.1 if the Registration is undertaken for the Company's account, the Company shall include in any such Registration (a) first, the Class A Common Stock or other equity securities that the Company desires to sell for its own account, which can be sold without exceeding the Maximum Number of Securities; (b) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (a), the Registrable Securities of Holders exercising their rights to register their Registrable Securities pursuant to subsection 2.3.1 hereof, Pro Rata, which can be sold without exceeding the Maximum Number of Securities; and (c) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (a) and (b), the Class A Common Stock, if any, as to which Registration has been requested pursuant to written contractual piggy-back registration rights of other stockholders of the Company, which can be sold without exceeding the Maximum Number of Securities; and

2.3.2.2 if the Registration is pursuant to a request by persons or entities other than the Holders of Registrable Securities, then the Company shall include in any such Registration (a) first, the Class A Common Stock or other equity securities, if any, of such requesting persons or entities, other than the Holders of Registrable Securities, which can be sold without exceeding the Maximum Number of Securities; (b) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (a), the Registrable Securities of Holders exercising their rights to register their Registrable Securities pursuant to subsection 2.3.1 hereof, Pro Rata, which can be sold without exceeding the Maximum Number of Securities; (c) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (a) and (b), the Class A Common Stock or other equity securities that the Company desires to sell for its own account, which can be sold without exceeding the Maximum Number of Securities; and (d) fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (a), (b) and (c), the Class A Common Stock or other equity securities for the account of other persons or entities that the Company is obligated to register pursuant to separate written contractual arrangements with such persons or entities, which can be sold without exceeding the Maximum Number of Securities.

2.3.3 <u>Piggyback Registration Withdrawal</u>. Any Holder of Registrable Securities shall have the right to withdraw all or any portion of its Registrable Securities in a Piggyback Registration for any or no reason whatsoever upon written notification to the Company and the Underwriter or Underwriters (if any) of his, her or its intention to withdraw such Registrable Securities from such Piggyback Registration prior to (a) in the case of a Piggyback Registration not involving an Underwritten Offering or Shelf Underwritten Offering, the effectiveness of the applicable Registration Statement or (b) in the case of any Piggyback Registration involving an Underwritten Offering or any Shelf Underwritten Offering, prior to the pricing of such Underwritten Offering or Shelf Underwritten Offering. The Company (whether on its own good-faith determination or as the result of a request for withdrawal by persons pursuant to separate written contractual obligations) may withdraw a Registration Statement filed with the Commission in connection with a Piggyback Registration at any time prior to the effectiveness of such Registration Statement. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with the Piggyback Registration prior to its withdrawal under this subsection 2.3.3.

- 2.3.4 <u>Unlimited Piggyback Registration Rights</u>. For purposes of clarity, any Registration effected pursuant to <u>Section 2.3</u> hereof shall not be counted as a Registration pursuant to a Demand Registration effected under <u>Section 2.2</u> hereof or a Shelf Underwritten Offering effected under <u>subsection 2.1.3</u>.
- 2.4 Restrictions on Registration Rights. If (a) during the period starting with the date sixty (60) days prior to the Company's good-faith estimate of the date of the filing of, and ending on a date one hundred and twenty (120) days after the effective date of, a Company initiated Registration and provided that the Company has delivered written notice to the Holders prior to receipt of a Demand Registration pursuant to subsection 2.2.1 and it continues to actively employ, in good faith, all reasonable efforts to cause the applicable Registration Statement to become effective; (b) the Holders have requested an Underwritten Registration and the Company and the Holders are unable to obtain the commitment of underwriters to firmly underwrite the offer; or (c) in the good-faith judgment of the Board such Registration would be seriously detrimental to the Company and the Board concludes as a result that it is essential to defer the filing of such Registration Statement at such time, then in each case the Company shall furnish to such Holders a certificate signed by the Chairman of the Board (or, if applicable, any Co-Chairman of the Board) stating that in the good-faith judgment of the Board it would be seriously detrimental to the Company for such Registration Statement to be filed in the near future and that it is therefore essential to defer the filing of such Registration Statement. In such event, the Company shall have the right to defer such filing for a period of not more than sixty (60) days; provided, however, that the Company shall not defer its obligation in this manner more than once in any twelve (12)-month period.
- 2.5 <u>Block Trades</u>. Notwithstanding any other provision of this Agreement, but subject to <u>Sections 2.4 and 3.4</u>, if the New Holders desire to effect a Block Trade, then notwithstanding any other time periods in this <u>Article II</u>, the New Holders shall provide written notice to the Company at least two (2) business days prior to the date such Block Trade is anticipated to commence. If requested by the New Holders, the Company will promptly notify other Holders of such Block Trade and such notified Holders may elect whether or not to participate no later than the next Business Day (i.e., one (1) Business Day prior to the day such offering is to commence) (unless a longer period is agreed to by the New Holders), and the Company will as expeditiously as possible use its best efforts to facilitate such Block Trade (which may close as early as two (2) Business Days after the date it commences). Notwithstanding anything to the contrary in this Agreement, no Holder (other than a New Holder) will be permitted to participate in an Block Trade without the consent of the New Holders. Any Holder's request to participate in a Block Trade shall be binding on such Holder.

#### 2.6 Lock-up Periods.

2.6.1 Notwithstanding anything to the contrary contained in this Agreement, except with respect to transfers to a Permitted Transferee, each Existing Holder agrees not to sell, transfer or otherwise dispose of, including any sale pursuant to Rule 144, any Founder Shares during the Founder Shares Lock-Up Period or Private Placement Warrants during the Private

Placement Lock-Up Period; provided that, except with respect to transfers to a Permitted Transferee, the Sponsor shall not sell, transfer or otherwise dispose of shares of Class A Common Stock, Class B Common Stock or any other equity security of the Company or any security convertible into or exchangeable or exercisable for any equity security of the Company during the period beginning on the date of this Agreement and ending on the first anniversary of the date of this Agreement, other than (a) as distributions to limited partners or members of the Sponsor; and (b) by virtue of the laws of the State of Delaware or of the Sponsor's organizational documents upon liquidation or dissolution of the Sponsor, in each case immediately following which Brent Saunders and Manisha Narasimhan shall collectively hold a majority of the Founder Shares then outstanding; provided, further, that in the event of any such distribution, liquidation or dissolution that results in Brent Saunders and/or Manisha Narasimhan becoming Existing Holders, except with respect to transfers to their Permitted Transferees, such person(s) (and their Permitted Transferees to whom they transfer) shall not sell, transfer or otherwise dispose of shares of Class A Common Stock, Class B Common Stock or any other equity security of the Company or any security convertible into or exchangeable or exercisable for any equity security of the Company during the period beginning on the date of this Agreement and ending on the first anniversary of the date of this Agreement.

2.6.2 Each Holder participating in a Registration (each, a "Participating Holder") agrees, to the extent requested in writing by a managing Underwriter, if any, of any Underwritten Offering hereunder, not to sell, transfer or otherwise dispose of, including any sale pursuant to Rule 144, any shares of Common Stock, or any other equity security of the Company or any security convertible into or exchangeable or exercisable for any equity security of the Company other than as part of such underwritten public offering during the time period reasonably requested by the managing underwriter, not to exceed 30 days from the date such Underwritten Offering is priced. Notwithstanding the foregoing, (i) no Participating Holder shall be required to agree to any such restrictions unless each other Participating Holder is also required to agree to such restrictions, (ii) the managing Underwriter (if any) may waive such restrictions in its reasonable discretion upon the written request of a Holder, subject to the terms set forth in any written lock-up agreement with respect thereto, and (iii) any waiver or release of such lock-up, holdback or similar agreement shall be on a consistent basis among the Participating Holders.

# ARTICLE III COMPANY PROCEDURES

- 3.1 <u>General Procedures</u>. If the Company is required to effect the Registration of Registrable Securities, the Company shall use its commercially reasonable efforts to effect such Registration to permit the sale of such Registrable Securities in accordance with the intended plan of distribution thereof. When effective, the Registration Statements filed pursuant to this Agreement (including the documents incorporated therein by reference) will comply as to form in all material respects with all applicable requirements of the Securities Act and the Exchange Act and will not contain a Misstatement. In connection with effecting a Registration of Registrable Securities pursuant to this Agreement, the Company shall, as expeditiously as possible:
- 3.1.1 prepare and file with the Commission as soon as practicable a Registration Statement with respect to such Registrable Securities and, except as otherwise set forth herein, use its commercially reasonable efforts to cause such Registration Statement to become effective and remain effective until all Registrable Securities covered by such Registration Statement have been sold;

- 3.1.2 prepare and file with the Commission such amendments and post-effective amendments to the Registration Statement, and such supplements to the Prospectus, as may be reasonably requested by a majority in interest of the applicable Holders of Registrable Securities registered on such Registration Statement or any Underwriter of Registrable Securities or as may be required by the rules, regulations or instructions applicable to the registration form used by the Company or by the Securities Act or rules and regulations thereunder to keep the Registration Statement effective until all Registrable Securities covered by such Registration Statement are sold in accordance with the intended plan of distribution set forth in such Registration Statement or supplement to the Prospectus;
- 3.1.3 prior to filing a Registration Statement or Prospectus, or any amendment or supplement thereto, furnish without charge to the Underwriters, if any, and the Holders of Registrable Securities included in such Registration, and such Holders' legal counsel, copies of such Registration Statement as proposed to be filed, each amendment and supplement to such Registration Statement (in each case including all exhibits thereto and documents incorporated by reference therein), the Prospectus included in such Registration Statement (including each preliminary Prospectus), and such other documents as the Underwriters and the Holders of Registrable Securities included in such Registration or the legal counsel for any such Holders may request in order to facilitate the disposition of the Registrable Securities owned by such Holders;
- 3.1.4 prior to any public offering of Registrable Securities, but in any case no later than the effective date of the applicable Registration Statement, use its commercially reasonable efforts to (a) register or qualify the Registrable Securities covered by the Registration Statement under such securities or "blue sky" laws of such jurisdictions in the United States as the Holders of Registrable Securities included in such Registration Statement (in light of their intended plan of distribution) may request and to keep such registration or qualification in effect for so long as such Registration Statement remains in effect, and (b) take such action necessary to cause such Registrable Securities covered by the Registration Statement to be registered with or approved by such other governmental authorities or securities exchanges, including the applicable Nasdaq Stock Market ("Nasdaq"), as may be necessary by virtue of the business and operations of the Company or otherwise and do any and all other acts and things that may be necessary or advisable, in each case, to enable the Holders of Registrable Securities included in such Registration Statement to consummate the disposition of such Registrable Securities in such jurisdictions; provided, however, that the Company shall not be required to qualify generally to do business in any jurisdiction where it would not otherwise be required to qualify or take any action to which it would be subject to general service of process or taxation in any such jurisdiction where it is not then otherwise so subject;
- 3.1.5 cause all such Registrable Securities to be listed on each securities exchange or automated quotation system on which similar securities issued by the Company are then listed no later than the effective date of such Registration Statement;
- 3.1.6 provide a transfer agent or warrant agent, as applicable, and registrar for all such Registrable Securities no later than the effective date of such Registration Statement;

- 3.1.7 advise each seller of such Registrable Securities, promptly after it shall receive notice or obtain knowledge thereof, of any request by the Commission that the Company amend or supplement such Registration Statement or Prospectus or the issuance of any stop order by the Commission suspending the effectiveness of such Registration Statement or Prospectus or the initiation or threatening of any proceeding for such purpose and promptly use its commercially reasonable efforts to amend or supplement such Registration Statement or Prospectus or prevent the issuance of any stop order or to obtain its withdrawal if such stop order should be issued, as applicable;
- 3.1.8 at least five (5) business days (or, in the case of a Block Trade, at least one (1) day) prior to the filing of any Registration Statement or Prospectus or any amendment or supplement to such Registration Statement or Prospectus or any document that is to be incorporated by reference into such Registration Statement or Prospectus, furnish a copy thereof to each seller of such Registrable Securities or its counsel, including, without limitation, providing copies promptly upon receipt of any comment letters received with respect to any such Registration Statement or Prospectus;
- 3.1.9 notify the Holders at any time when a Prospectus relating to such Registration Statement is required to be delivered under the Securities Act, of the happening of any event as a result of which the Prospectus included in such Registration Statement, as then in effect, includes a Misstatement, and then to correct such Misstatement or include such information as is necessary to comply with law, in each case as set forth in Section 3.4 hereof;
- 3.1.10 permit a representative of the New Holders, the Underwriters, if any, and any attorney or accountant retained by such Holders or Underwriter to participate, at each such person's own expense, in the preparation of any Registration Statement and cause the Company's officers, directors and employees to supply all information reasonably requested by any such representative, Underwriter, attorney or accountant in connection with the Registration; provided, however, that, if requested by the Company, such representatives or Underwriters enter into a confidentiality agreement, in form and substance reasonably satisfactory to the Company, prior to the release or disclosure of any such information;
- 3.1.11 obtain a "comfort" letter (including a bring-down letter dated as of the date the Registrable Securities are delivered for sale pursuant to such Registration) from the Company's independent registered public accountants in the event of an Underwritten Offering that the participating Holders may rely on, in customary form and covering such matters of the type customarily covered by "comfort" letters as the managing Underwriter may reasonably request, and reasonably satisfactory to a majority-in-interest of the participating Holders and any Underwriter;
- 3.1.12 on the date the Registrable Securities are delivered for sale pursuant to such Registration, obtain an opinion and negative assurance letter, dated such date, of counsel representing the Company for the purposes of such Registration, addressed to the Holders, the placement agent or sales agent, if any, and the Underwriters, if any, covering such legal matters with respect to the Registration in respect of which such opinion is being given as the Holders, placement agent, sales agent, or Underwriter may reasonably request and as are customarily included in such opinions and negative assurance letters, and reasonably satisfactory to a majority in interest of the participating Holders and any Underwriter;

- 3.1.13 in the event of any Underwritten Offering, enter into and perform its obligations under an underwriting agreement, in usual and customary form, with the managing Underwriter of such offering;
- 3.1.14 otherwise use its commercially reasonable efforts to comply with all applicable rules and regulations of the Commission, and to make available to its security holders, as soon as reasonably practicable, an earnings statement covering the period of at least twelve (12) months, beginning with the first day of the Company's first full calendar quarter after the effective date of the Registration Statement which satisfies the provisions of Section 11(a) of the Securities Act and the rules and regulations thereunder, including Rule 158 thereunder (or any successor rule promulgated thereafter by the Commission);
- 3.1.15 if the Registration involves the Registration of Registrable Securities involving gross proceeds in excess of \$50,000,000, use its reasonable efforts to make available senior executives of the Company to participate in customary "road show" presentations that may be reasonably requested by the Underwriter in any Underwritten Offering; and
- 3.1.16 otherwise, in good faith, cooperate reasonably with, and take such customary actions as may reasonably be requested by the Holders in connection with such Registration.
- 3.2 <u>Registration Expenses</u>. Except as otherwise provided herein, the Registration Expenses of all Registrations shall be borne by the Company. It is acknowledged by the Holders that the Holders shall bear all incremental selling expenses relating to the sale of Registrable Securities, such as Underwriters' commissions and discounts, brokerage fees, Underwriter marketing costs and, other than as set forth in the definition of "Registration Expenses," all reasonable fees and expenses of any legal counsel representing the Holders.
- 3.3 <u>Participation in Underwritten Offerings.</u> No Person may participate in any Underwritten Offering for equity securities of the Company pursuant to a Registration initiated by the Company hereunder unless such Person (a) agrees to sell such Person's securities on the basis provided in any underwriting arrangements approved by the Company and (b) completes and executes all customary questionnaires, powers of attorney, indemnities, lock-up agreements, underwriting agreements and other customary documents as may be reasonably required under the terms of such underwriting arrangements.
- 3.4 <u>Suspension of Sales; Adverse Disclosure</u>. Upon receipt of written notice from the Company that a Registration Statement or Prospectus contains a Misstatement, or in the opinion of counsel for the Company it is necessary to supplement or amend such Prospectus to comply with law, each of the Holders shall forthwith discontinue disposition of Registrable Securities until it has received copies of a supplemented or amended Prospectus correcting the Misstatement or including the information counsel for the Company instructs is necessary to comply with law (it being understood that the Company hereby covenants to prepare and file such supplement or amendment as soon as practicable after the time of such notice such that the Registration Statement

or Prospectus, as so amended or supplemented, as applicable, will not include a Misstatement and complies with law), or until it is advised in writing by the Company that the use of the Prospectus may be resumed. If the filing, initial effectiveness or continued use of a Registration Statement in respect of any Registration at any time would require the Company to make an Adverse Disclosure or would require the inclusion in such Registration Statement of financial statements that are unavailable to the Company for reasons beyond the Company's control, the Company may, upon giving prompt written notice of such action to the Holders, delay the filing or initial effectiveness of, or suspend use of, such Registration Statement for the shortest period of time required to resolve such issue, but in no event more than sixty (60) consecutive days, determined in good faith by the Board to be necessary for such purpose; provided that the Company shall not defer its obligations in this manner more than twice during any twelve (12)-month period pursuant to Section 2.4 hereof. In the event the Company exercises its rights under the preceding sentence, the Holders agree to suspend, immediately upon their receipt of the notice referred to above, their use of the Prospectus relating to any Registration in connection with any sale or offer to sell Registrable Securities. The Company shall immediately notify the Holders of the expiration of any period during which it exercised its rights under this Section 3.4. The Holders agree that, except as required by applicable law, the Holders shall treat as confidential the receipt of written notice from the Company under this Section 3.4 (provided that in no event shall such notice contain any material nonpublic information of the Company) and shall not disclose or use the information contained in such written notice without the prior written consent of the Company until such time as the information contained therein is or becomes public, other than as a result of disclosure by a holder

- 3.5 <u>Covenants of the Company.</u> As long as any Holder shall own Registrable Securities, the Company, at all times while it shall be a reporting company under the Exchange Act, covenants to file timely (or obtain extensions in respect thereof and file within the applicable grace period) all reports required to be filed by the Company after the date hereof pursuant to Sections 13(a) or 15(d) of the Exchange Act and to promptly furnish the Holders with true and complete copies of all such filings. The Company further covenants that it shall take such further action as any Holder may reasonably request, all to the extent required from time to time to enable such Holder to sell shares of Class A Common Stock held by such Holder without registration under the Securities Act within the limitation of the exemptions provided by Rule 144 promulgated under the Securities Act (or any successor rule promulgated thereafter by the Commission), including providing any legal opinions.
- 3.6 <u>Information</u>. The Holders shall provide such information as may reasonably be requested by the Company, or the managing Underwriter, if any, in connection with the preparation of any Registration Statement, including amendments and supplements thereto, in order to effect the Registration of any Registrable Securities under the Securities Act pursuant to <u>Article 2</u> and in connection with the Company's obligation to comply with federal and applicable state securities laws.

# ARTICLE IV INDEMNIFICATION AND CONTRIBUTION

#### 4.1 <u>Indemnification</u>.

- 4.1.1 The Company agrees to indemnify, to the extent permitted by law, each Holder of Registrable Securities, its officers and directors and agents and each Person who controls such Holder (within the meaning of the Securities Act) against all losses, claims, damages, liabilities and expenses (including, without limitation, reasonable attorneys' fees) resulting from any Misstatement or alleged Misstatement, except insofar as the same are contained in any information furnished in writing to the Company by such Holder expressly for use therein. The Company shall indemnify the Underwriters, their officers and directors and agents and each Person who controls such Underwriters (within the meaning of the Securities Act) to the same extent as provided in the foregoing with respect to the indemnification of the Holder.
- 4.1.2 In connection with any Registration Statement in which a Holder of Registrable Securities is participating, such Holder shall furnish to the Company in writing such information and affidavits as the Company reasonably requests for use in connection with any such Registration Statement or Prospectus and, to the extent permitted by law, shall indemnify the Company, its directors and officers and agents and each Person who controls the Company (within the meaning of the Securities Act) against any losses, claims, damages, liabilities and expenses (including without limitation reasonable attorneys' fees) resulting from any Misstatement or alleged Misstatement, but only to the extent that such Misstatement or alleged Misstatement is contained in any information or affidavit so furnished in writing by such Holder expressly for use therein; provided, however, that the obligation to indemnify shall be several, not joint and several, among such Holders of Registrable Securities, and the liability of each such Holder of Registrable Securities shall be in proportion to and limited to the net proceeds received by such Holder from the sale of Registrable Securities in such offering giving rise to such liability. The Holders of Registrable Securities shall indemnify the Underwriters, their officers, directors and each Person who controls such Underwriters (within the meaning of the Securities Act) to the same extent as provided in the foregoing with respect to indemnification of the Company.
- 4.1.3 Any Person entitled to indemnification herein shall (i) give prompt written notice to the indemnifying party of any claim with respect to which it seeks indemnification (provided that the failure to give prompt notice shall not impair any Person's right to indemnification hereunder to the extent such failure has not materially prejudiced the indemnifying party) and (ii) unless in such indemnified party's reasonable judgment a conflict of interest between such indemnified and indemnifying parties may exist with respect to such claim, permit such indemnifying party to assume the defense of such claim with counsel reasonably satisfactory to the indemnified party. If such defense is assumed, the indemnifying party shall not be subject to any liability for any settlement made by the indemnified party without its consent (but such consent shall not be unreasonably withheld). An indemnifying party who is not entitled to, or elects not to, assume the defense of a claim shall not be obligated to pay the fees and expenses of more than one counsel (plus local counsel) for all parties indemnified by such indemnifying party with respect to such claim, unless in the reasonable judgment of any indemnified party a conflict of interest may exist between such indemnified party and any other of such indemnified parties with respect to such claim. No indemnifying party shall, without the consent of the indemnified

party, consent to the entry of any judgment or enter into any settlement which cannot be settled in all respects by the payment of money (and such money is so paid by the indemnifying party pursuant to the terms of such settlement) or which settlement does not include as an unconditional term thereof the giving by the claimant or plaintiff to such indemnified party of a release from all liability in respect to such claim or litigation.

- 4.1.4 The indemnification provided for under this Agreement shall remain in full force and effect regardless of any investigation made by or on behalf of the indemnified party or any officer, director or controlling Person of such indemnified party and shall survive the transfer of securities. The Company and each Holder of Registrable Securities participating in an offering also agrees to make such provisions as are reasonably requested by any indemnified party for contribution to such party in the event the Company's or such Holder's indemnification is unavailable for any reason.
- 4.1.5 If the indemnification provided under Section 4.1 hereof from the indemnifying party is unavailable or insufficient to hold harmless an indemnified party in respect of any losses, claims, damages, liabilities and expenses referred to herein, then the indemnifying party, in lieu of indemnifying the indemnified party, shall contribute to the amount paid or payable by the indemnified party as a result of such losses, claims, damages, liabilities and expenses in such proportion as is appropriate to reflect the relative fault of the indemnifying party and the indemnified party, as well as any other relevant equitable considerations. The relative fault of the indemnifying party and indemnified party shall be determined by reference to, among other things, whether any action in question, including any Misstatement or alleged Misstatement, was made by, or relates to information supplied by, such indemnifying party or indemnified party, and the indemnifying party's and indemnified party's relative intent, knowledge, access to information and opportunity to correct or prevent such action and the benefits received by the such indemnifying party or indemnified party; provided, however, that the liability of any Holder under this subsection 4.1.5 shall be limited to the amount of the net proceeds received by such Holder from the sale of Registrable Securities in such offering giving rise to such liability. The amount paid or payable by a party as a result of the losses or other liabilities referred to above shall be deemed to include, subject to the limitations set forth in subsections 4.1.1, 4.1.2 and 4.1.3 above, any legal or other fees, charges or expenses reasonably incurred by such party in connection with any investigation or proceeding. The parties hereto agree that it would not be just and equitable if contribution pursuant to this subsection 4.1.5 were determined by Pro Rata allocation or by any other method of allocation, which does not take account of the equitable considerations referred to in this subsection 4.1.5. No Person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution pursuant to this subsection 4.1.5 from any Person who was not guilty of such fraudulent misrepresentation.

# ARTICLE V MISCELLANEOUS

5.1 <u>Notices</u>. Any notice or communication under this Agreement must be in writing and given by (i) deposit in the United States mail, addressed to the party to be notified, postage prepaid and registered or certified with return receipt requested, (ii) delivery in person or by courier service providing evidence of delivery, or (iii) transmission by hand delivery, electronic mail or facsimile. Each notice or communication that is mailed, delivered, or transmitted in the manner

described above shall be deemed sufficiently given, served, sent, and received, in the case of mailed notices, on the third (3rd) Business Day following the date on which it is mailed and, in the case of notices delivered by courier service, hand delivery, electronic mail (<u>provided</u> no "bounce back" or notice of non-delivery is received) or facsimile, at such time as it is delivered to the addressee (except in the case of electronic mail, with the delivery receipt or the affidavit of messenger) or at such time as delivery is refused by the addressee upon presentation. Any notice or communication under this Agreement must be addressed, if to the Company, to: 1819 West Avenue, Bay 2, Miami Beach, FL 33139, and, if to any Holder, at such Holder's address or contact information as set forth in the Company's books and records. Any party may change its address for notice at any time and from time to time by written notice to the other parties hereto, and such change of address shall become effective thirty (30) days after delivery of such notice as provided in this <u>Section 5.1</u>.

#### 5.2 Assignment; No Third Party Beneficiaries.

- 5.2.1 This Agreement and the rights, duties and obligations of the Company and the Holders hereunder may not be assigned or delegated by the Company or the Holders, as the case may be, in whole or in part.
- 5.2.2 Prior to the expiration of the Founder Shares Lock-up Period or the Private Placement Lock-up Period, as the case may be, no Existing Holder who is subject to either or both the Founder Shares Lock-up Period or the Private Placement Lock-up Period may assign or delegate such Existing Holder's rights, duties or obligations under this Agreement, in whole or in part, except in connection with a transfer of Registrable Securities by such Holder to a Permitted Transferee, to an Affiliate or as otherwise permitted pursuant to the terms of the Founder Shares Lock-up Period, the Private Placement Lock-up Period or other lock-up period, as applicable.
- 5.2.3 This Agreement and the provisions hereof shall be binding upon and shall inure to the benefit of each of the parties and its successors and the permitted assigns of the applicable Holders, which shall include Permitted Transferees.
- 5.2.4 This Agreement shall not confer any rights or benefits on any Persons that are not parties hereto, other than as expressly set forth in this Agreement and Section 5.2 hereof.
- 5.2.5 No assignment (including to a Permitted Transferee) by any party hereto of such party's rights, duties and obligations hereunder shall be binding upon or obligate the Company unless and until the Company shall have received (a) written notice of such assignment as provided in Section 5.1 hereof and (b) the written agreement of the assignee, in a form reasonably satisfactory to the Company, to be bound by the terms and provisions of this Agreement (which may be accomplished by an addendum or certificate of joinder to this Agreement). Any transfer or assignment made other than as provided in this Section 5.2 shall be null and void.
- 5.3 <u>Counterparts</u>. This Agreement may be executed in multiple counterparts (including facsimile or PDF counterparts), each of which shall be deemed an original, and all of which together shall constitute the same instrument, but only one of which need be produced.

- 5.4 Governing Law; Venue. NOTWITHSTANDING THE PLACE WHERE THIS AGREEMENT MAY BE EXECUTED BY ANY OF THE PARTIES HERETO, THE PARTIES EXPRESSLY AGREE THAT (I) THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED UNDER THE LAWS OF THE STATE OF DELAWARE AS APPLIED TO AGREEMENTS AMONG DELAWARE RESIDENTS ENTERED INTO AND TO BE PERFORMED ENTIRELY WITHIN DELAWARE, WITHOUT REGARD TO THE CONFLICT OF LAW PROVISIONS OF SUCH JURISDICTION THAT WOULD REQUIRE THE APPLICATION OF THE LAWS OF ANOTHER JURISDICTION AND (II) THE VENUE FOR ANY ACTION TAKEN WITH RESPECT TO THIS AGREEMENT SHALL BE ANY STATE OR FEDERAL COURT IN THE STATE OF DELAWARE.
- 5.5 Amendments and Modifications. Upon the written consent of the Company and the Holders of at least a majority-in-interest of the Registrable Securities at the time in question, compliance with any of the provisions, covenants and conditions set forth in this Agreement may be waived, or any of such provisions, covenants or conditions may be amended or modified; provided, however, that, notwithstanding the foregoing, any amendment hereto or waiver hereof that adversely affects either the Existing Holders as a group or the New Holders as a group, respectively, in a manner that is materially adversely different from the Existing Holders or New Holders, as applicable, shall require the consent of at least a majority-in-interest of the Registrable Securities held by such Existing Holders or New Holders, as applicable, at the time in question; provided, further, that, notwithstanding the foregoing, any amendment hereto or waiver hereof that adversely affects one Holder, solely in its capacity as a holder of the shares of capital stock of the Company, in a manner that is materially different from the other Holders (in such capacity) shall require the consent of the Holder so affected. No course of dealing between any Holder or the Company and any other party hereto or any failure or delay on the part of a Holder or the Company in exercising any rights or remedies under this Agreement shall operate as a waiver of any rights or remedies of any Holder or the Company. No single or partial exercise of any rights or remedies under this Agreement by a party shall operate as a waiver or preclude the exercise of any other rights or remedies hereunder or thereunder by such party.
- 5.6 Other Registration Rights. Other than pursuant to the terms of the Subscription Agreements in connection with the PIPE Investment, the Company represents and warrants that no Person, other than a Holder of Registrable Securities, has any right to require the Company to register any securities of the Company for sale or to include such securities of the Company in any Registration filed by the Company for the sale of securities own account or for the account of any other Person. The Company represents and warrants that this Agreement supersedes any other registration rights agreement or agreement with similar terms and conditions among the parties thereto and in the event of a conflict between any such agreement or agreements and this Agreement, the terms of this Agreement shall prevail. To the extent the Company grants any Person(s) the right to request the Company or any of its subsidiaries to register any equity securities of the Company or any of its subsidiaries or any securities convertible or exchangeable into or exercisable for such securities, the Company shall grant piggyback registration rights to the New Holders in connection therewith.
- 5.7 Term. This Agreement shall terminate upon the earlier of (a) the tenth anniversary of the date of this Agreement, (b) the date as of which all of the Registrable Securities have been sold pursuant to a Registration Statement (but in no event prior to the applicable period referred to in Section 4(a)(3) of the Securities Act and Rule 174 thereunder (or any successor rule promulgated thereafter by the Commission)) or (c) with respect to a particular Holder, the date as

of which all Registrable Securities held by such Holder have been sold (x) pursuant to a Registration Statement (but in no event prior to the applicable period referred to in Section 4(a)(3) of the Securities Act and Rule 174 thereunder (or any successor rule promulgated thereafter by the Commission)) or (y) under Rule 144 or another exemption from registration under the Securities Act; provided that, for purposes of this Section 5.7, securities constituting Registrable Securities shall be determined without regard and without giving effect to clause (D) contained in the definition of Registrable Securities. The provisions of Section 3.5 and Article IV shall survive any termination.

5.8 Rules of Construction. Any provision of this Agreement that refers to the words "includes" or "including" shall be deemed to be followed by the words "without limitation." References to numbered or letter articles, sections and subsections refer to articles, sections and subsections, respectively, of this Agreement unless expressly stated otherwise. All references to this Agreement include, whether or not expressly referenced, the exhibits and schedules attached hereto. References to a Section, paragraph, Exhibit or Schedule, such reference shall be to a Section or paragraph of, or Exhibit or Schedule to, this Agreement unless otherwise indicated. The words "hereof," "herein" and "hereunder" and words of similar import when used in this Agreement shall refer to this Agreement as a whole and not to any particular provision of this Agreement. The word "or" when used in this Agreement is not exclusive. The definitions contained in this Agreement are applicable to the singular as well as the plural forms of such terms and to the masculine as well as to the feminine and neuter genders of such term. Any agreement, instrument, law or statute defined or referred to herein or in any agreement or instrument that is referred to herein means such agreement, instrument, or statute as from time to time amended, modified or supplemented, including (in the case of agreements or instruments) by waiver or consent and (in the case of statutes) by succession of comparable successor statutes and references to all attachments thereto and instruments incorporated therein unless otherwise indicated. References to a Person are also to its permitted successors and assigns. In the event that any claim is made by any Person relating to any conflict, omission or ambiguity in this Agreement, no presumption or burden of proof or persuasion shall be implied by virtue of the fact that this Agreement was prepared by or at the request of a particular Person or its counsel.

[Signature Pages Follow]

**IN WITNESS WHEREOF**, the undersigned have caused this Agreement to be executed as of the date first written above.

#### COMPANY:

THE BEAUTY HEALTH COMPANY, a Delaware corporation

By: /s/ Brenton L. Saunders

Name: Brenton L. Saunders Title: Executive Chairman

#### **EXISTING HOLDER:**

BLS INVESTOR GROUP LLC, a Delaware limited liability company

By: /s/ Brenton L. Saunders

Name: Brenton L. Saunders Title: Managing Member

#### **NEW HOLDERS:**

LCP EDGE HOLDCO, LLC, a Delaware limited liability company

By: /s/ Kamlesh Shah

Name: Kamlesh Shah Title: Vice President, Assistant Treasurer and Assistant Secretary

DW HEALTHCARE PARTNERS IV (B), L.P., a Delaware limited partnership

By: DW Healthcare Management IV, LP

Its: General Partner

By: DW Healthcare Management IV GP, LLC

Its: General Partner

By: /s/ Doug Schillinger

Name: Doug Schillinger Title: Authorized Person

[Signature Page to Registration Rights Agreement]

#### INVESTOR RIGHTS AGREEMENT

This Investor Rights Agreement (this "<u>Agreement</u>") is entered into as of May 4, 2021 (the "<u>Effective Date</u>"), by and between The Beauty Health Company, a Delaware corporation (the "<u>Company</u>"), and LCP Edge Holdco, LLC, a Delaware limited liability company ("<u>LCP</u>"). Capitalized terms used but not otherwise defined in this Agreement have the respective meanings given to them in the Merger Agreement (as defined below).

**WHEREAS**, pursuant to that certain Agreement and Plan of Merger, dated as of December 8, 2020, by and among the Company, Hydrate Merger Sub, Inc., a Delaware corporation, LCP Edge Intermediate, Inc., a Delaware corporation, and LCP, in its capacity as the stockholders' representative thereunder (the "Merger Agreement"), LCP is receiving shares of the Company's Class A common stock, par value \$0.0001 per share (the "Common Stock"); and

**WHEREAS**, in connection with the transactions contemplated by the Merger Agreement, the Company and LCP desire to set forth certain understandings between the Company and LCP, including with respect to certain governance matters.

**NOW, THEREFORE**, in consideration of the mutual covenants contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties to this Agreement hereby agree as follows:

#### ARTICLE I

#### **BOARD OF DIRECTORS**

- Section 1.1 <u>LCP Representation</u>. LCP shall have the right, but not the obligation, to nominate to the Board (such nominees, the "<u>LCP Directors</u>") (subject to their election by the stockholders of the Company) that number of individuals that, if elected, will result in LCP having the number of directors serving on the Board that is shown below:
  - (a) From and after the Effective Date:
- (i) Three (3) directors for so long as LCP holds a number of shares of Common Stock representing at least forty percent (40%) of the shares of Common Stock then outstanding;
- (ii) Two (2) directors for so long as LCP holds a number of shares of Common Stock representing less than forty percent (40%) but at least fifteen percent (15%) of the shares of Common Stock then outstanding; and
- (iii) One (1) director for so long as LCP holds a number of shares of Common Stock representing less than fifteen percent (15%) but at least ten percent (10%) of the shares of Common Stock then outstanding.
- (b) If, at any time, LCP ceases to hold a number of shares of Common Stock representing at least ten percent (10%) of the shares of Common Stock then outstanding, LCP shall no longer have any rights under <u>Section 1.1(a)</u>.

- (i) At any time after LCP ceases to hold the relevant percentage of the shares of Common Stock then outstanding referred to in Section 1.1(a) above, if there is more than one LCP Director, LCP may by written notice to the Board, within five Business Days of ceasing to hold such number of shares of Common Stock, designate which directors shall be the LCP Directors and the other directors shall resign from the Board upon the written request of the Board (other than the LCP Directors); provided, that, if no such notice is received by the Board within 10 business days after the same has been requested in writing of LCP, the Board (other than the LCP Directors) shall determine which of the LCP Directors shall continue in office and which shall resign from the Board. At any time after LCP ceases to hold at least ten percent (10%) of the shares of Common Stock then outstanding, the LCP Director shall resign from the Board upon the written request of the Board.
- (c) So long as LCP holds ten percent (10%) or more of the then outstanding shares of Common Stock, the Company shall take all necessary and desirable actions to cause each of the Compensation Committee of the Board and the Nominating and Corporate Governance Committee of the Board to include in its membership at least one of the LCP Directors, except to the extent such membership would violate applicable securities laws or stock exchange or stock market rules.
- (d) From and after the Effective Date, unless otherwise agreed to in writing by LCP, the Company shall, as promptly as practicable, take all actions necessary (including, without limitation, calling special meetings of the Board and the stockholders of the Company and recommending, supporting and soliciting proxies) to ensure that: (i) the applicable LCP Directors are (a) designated as Class III and/or Class II directors pursuant to Section 5.2(b) of the Company's Second Amended and Restated Certificate of Incorporation and (b) included in the Board's slate of nominees to the stockholders of the Company and recommended by the Board at any meeting of stockholders called for the purpose of electing directors, in each case, to the extent necessary such that the number of LCP Directors that LCP is eligible to designate shall be designated; and (ii) each applicable LCP Director up for election in accordance with the foregoing is included in the proxy statement prepared by management of the Company in connection with the Company's solicitation of proxies or consents in favor of the foregoing for every meeting of the stockholders of the Company called with respect to the election of directors, and at every adjournment or postponement thereof, and on every action or approval by written resolution of the stockholders of the Company or the Board with respect to the election of directors; provided, that any individuals nominated under this Section 1.1 shall (x) be reasonably acceptable to the Nominating and Corporate Governance Committee of the Board (the "Nominating Committee") (or the Board, other than the LCP Directors, if there is no such Nominating Committee), and (y) in the good faith judgment of the Nominating Committee (or the Board, other than the LCP Directors, if there is no such Nominating Committee) satisfy the requirements set forth in the Company's Organizational Documents and corporate governance guidelines (as in effect from time to time), in each case as are applicable to all non-employee directors gener
- (e) If any LCP Director ceases to serve for any reason (including because of the death, disability, disqualification, resignation, or removal of such LCP Director), LCP shall, subject to LCP then being entitled to designate such individual for nomination as a director pursuant to <u>Section 1.1(a)</u>, be entitled to designate such person's successor in accordance with this Agreement and the Board shall promptly fill the vacancy with such successor LCP Director.

- Section 1.2 <u>Letter of Resignation</u>. Notwithstanding anything to the contrary in this Agreement, the election or appointment of any LCP Director to the Board shall be subject to the prior execution by such LCP Director of an irrevocable resignation letter in the form attached hereto as <u>Exhibit A</u>.
- Section 1.3 Reimbursement of Expenses; Indemnification. For so long as any LCP Director serves as a director on the Board, the Company shall (i) provide such LCP Director with the same expense reimbursement, benefits and other arrangements provided to the other members of the Board, (ii) indemnify such LCP Director on the same basis as all other members of the Board and (iii) not amend, alter or repeal any right to indemnification or exculpation covering or benefiting any LCP Director as and to the extent consistent with applicable law, including but not limited to any rights contained in the Company's Organizational Documents (except to the extent such amendment or alteration permits the Company to provide broader indemnification or exculpation rights on a retroactive basis than permitted prior thereto).
- Section 1.4 <u>Director Obligations</u>. Any LCP Director shall, upon the election or appointment of such LCP Director to the Board, execute such agreements as are required to be executed by all non-employee directors generally and shall otherwise abide by the provisions of all codes and policies of the Company that are applicable to all non-employee directors generally, including, as applicable, the Company's insider trading policy, policies requiring the pre-clearance of all securities trading activity, the Company's code of conduct and business ethics and the Company's stock ownership policy.

### ARTICLE II MISCELLANEOUS

- Section 2.1 <u>Termination</u>. This Agreement shall terminate automatically and become void and of no further force or effect, without any notice or other action by any Person, at such time as LCP ceases to hold a number of shares of Common Stock representing at least ten percent (10%) of the shares of Common Stock then outstanding.
- Section 2.2 <u>Amendment and Waiver</u>. This Agreement may be amended by the parties at any time by execution of an instrument in writing signed on behalf of each of the parties.
- Section 2.3 <u>Severability</u>. If any provision of this Agreement shall be invalid, illegal or unenforceable, the validity, legality and enforceability of the remaining provisions shall not in any way be affected or impaired thereby.
- Section 2.4 <u>Entire Agreement</u>. Except as otherwise expressly set forth herein, this document embodies the complete agreement and understanding among the parties hereto with respect to the subject matter hereof and supersedes and preempts any prior understandings, agreements or representations by or among the parties, written or oral, which may have related to the subject matter hereof in any way.

- Section 2.5 <u>Successors and Assigns</u>. The provisions of this Agreement shall inure to the benefit of, and be binding upon, the successors, assigns, heirs, executors, and administrators of the parties hereto.
- Section 2.6 <u>Counterparts</u>. This Agreement may be executed in separate counterparts each of which shall be an original and all of which taken together shall constitute one and the same agreement.
- Section 2.7 <u>Remedies</u>. The parties hereto agree and acknowledge that money damages may not be an adequate remedy for any breach of the provisions of this Agreement and that the Company and LCP shall have the right to injunctive relief or specific performance, in addition to all of its rights and remedies at law or in equity, to enforce the provisions of this Agreement.
- Section 2.8 <u>No Third Party Beneficiaries</u>. Nothing contained in this Agreement shall be construed to confer upon any Person who is not a signatory hereto any rights or benefits, as a third party beneficiary or otherwise.
- Section 2.9 Notices. All notices or other communications, including service of process, required or permitted hereunder shall be in writing and shall be deemed given or delivered and received on the earliest of (a) the day when delivered, if delivered personally, (b) one (1) Business Day after deposit with a nationally recognized courier or overnight service such as Federal Express (or upon any earlier receipt confirmed in writing by such service), (c) five (5) Business Days after mailing via U.S. certified mail, return receipt requested, or (d) the date sent, with no mail undeliverable or other rejection notice, if sent by email, in each case addressed as follows:

if to the Company, to:

1819 West Avenue, Bay 2 Miami Beach, FL 33139

Attention: Brenton L. Saunders and Manisha Narasimhan

Email: Brent.Saunders@vesperhealth.com

Manisha.Narasimhan@vesperhealth.com

with a copy (which shall not constitute notice) to:

Wachtell, Lipton, Rosen & Katz 51 West 52nd Street New York, NY 10019

Attention: Andrew R. Brownstein, Igor Kirman and DongJu Song Email: ARBrownstein@wlrk.com, IKirman@wlrk.com and

DSong@wlrk.com

if to LCP, to:

LCP Edge Holdco, LLC c/o Linden Capital Partners LLC 150 North Riverside Plaza, Suite 5100 Chicago, IL 60606

Attention: Brian Miller

Kam Shah

Email:

bmiller@lindenllc.com

kshah@lindenllc.com

with a copy (which shall not constitute notice) to:

Kirkland & Ellis LLP 300 North LaSalle Street Chicago, IL 60654

Attention: Robert A. Wilson, P.C.

Maggie D. Flores

robert.wilson@kirkland.com Email:

maggie.flores@kirkland.com

and to such other address or addressee as any such party has specified by prior written notice to the other party in accordance with this Section 2.9.

Section 2.10 Governing Law; Jurisdiction. This Agreement shall be governed by and construed in accordance with the internal laws of the State of Delaware (without reference to its choice of law rules that would cause the application of the rules of another jurisdiction). Each party hereto hereby irrevocably and unconditionally (a) agrees that all claims or causes of action based upon, arising out of, or related to this Agreement or the transactions contemplated hereby, shall only be brought in the Court of Chancery of the State of Delaware (or, if the Court of Chancery of the State of Delaware lacks subject matter jurisdiction, then in the applicable Delaware state court), or if under applicable law exclusive jurisdiction of such claim or cause of action is vested in the federal courts, then the United States District Court for the District of Delaware, (b) expressly submits to the personal jurisdiction and venue of such courts for the purposes thereof, and (c) waives and agrees not to raise (by way of motion, as a defense or otherwise) any and all jurisdictional, venue and convenience objections or defenses that such party may have in such action or proceeding.

Section 2.11 Waiver of Jury Trial. EACH OF THE PARTIES HERETO HEREBY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY WITH RESPECT TO ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF, UNDER OR IN CONNECTION WITH THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY. EACH OF THE PARTIES HERETO (A) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HERETO HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THAT FOREGOING WAIVER AND (B) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HERETO HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION <u>2.11</u>.

- Section 2.12 <u>Descriptive Headings</u>. The descriptive headings of this Agreement are inserted for convenience only and do not constitute a part of this Agreement.
- Section 2.13 Additional Securities Subject to Agreement. All shares of Common Stock that LCP hereafter acquires by means of a stock split, stock dividend, distribution, exercise of options or warrants or otherwise (other than pursuant to a public offering) whether by merger, consolidation or otherwise (including shares of a surviving corporation into which the shares of Common Stock are exchanged in such transaction) will be subject to the provisions of this Agreement to the same extent as if held on the date hereof.

#### Section 2.14 Rules of Construction.

- (a) Whenever any provision of this Agreement calls for any calculation based on a number of shares of Common Stock issued and outstanding or held by LCP, the number of shares of Common Stock deemed to be issued and outstanding or held LCP, unless specifically stated otherwise, as applicable, shall be the total number of shares of Common Stock then issued and outstanding or owned by LCP and its Permitted Transferees (as such term is defined in the Registration Rights Agreement, dated as of the date hereof, among the Company, LCP and the other parties signatory thereto).
- (b) Any provision of this Agreement that refers to the words "include," "includes" or "including" shall be deemed to be followed by the words "without limitation." References to numbered or letter articles, sections and subsections refer to articles, sections and subsections, respectively, of this Agreement unless expressly stated otherwise. The words "hereof," "herein" and "hereunder" and words of similar import when used in this Agreement shall refer to this Agreement as a whole and not to any particular provision of this Agreement. The word "or" when used in this Agreement is not exclusive. The definitions contained in this Agreement are applicable to the singular as well as the plural forms of such terms. Any agreement, instrument, law or statute defined or referred to herein or in any agreement or instrument that is referred to herein means such agreement, instrument, or statute as from time to time amended, modified or supplemented, including (in the case of agreements or instruments) by waiver or consent and (in the case of statutes) by succession of comparable successor statutes and references to all attachments thereto and instruments incorporated therein unless otherwise indicated. References to a Person are also to its permitted successors and assigns. In the event that any claim is made by any Person relating to any conflict, omission or ambiguity in this Agreement, no presumption or burden of proof or persuasion shall be implied by virtue of the fact that this Agreement was prepared by or at the request of a particular Person or its counsel.

[The remainder of this page is intentionally left blank.]

IN WITNESS WHEREOF, the parties hereto have executed this Investor Rights Agreement on the day and year first above written.

### THE BEAUTY HEALTH COMPANY

By: /s/ Brenton L. Saunders

Name: Brenton L. Saunders Title: Executive Chairman

### LCP EDGE HOLDCO, LLC

By: /s/ Kamlesh Shah

Name: Kamlesh Shah Title: Vice President, Assistant Treasurer and Assistant Secretary

[Signature Page to Investor Rights Agreement]

#### Exhibit A

#### Form of Irrevocable Resignation

#### Ladies and Gentlemen:

Brian Miller

This irrevocable resignation is delivered pursuant to Section 1.2 of the Investor Rights Agreement, dated as of May 4, 2021 (the "<u>Agreement</u>"), by and between The Beauty Health Company (the "<u>Company</u>") and LCP (as defined in the Agreement). If (i) following such time that the Agreement is terminated in accordance with its terms, the Board of Directors of the Company (the "<u>Board</u>") requests in writing that I resign as a director of the Company, or (ii) the Board requests in writing my resignation pursuant to and in accordance with Section 1.1(b)(i) of the Agreement, I hereby tender the immediate resignation of my position as a director of the Company and from any and all committees of the Board on which I serve.

This resignation may not be withdrawn by me at any time.				
Sincerely,				
/s/ Brian Miller				

#### Exhibit A

#### Form of Irrevocable Resignation

#### Ladies and Gentlemen:

Doug Schillinger

This irrevocable resignation is delivered pursuant to Section 1.2 of the Investor Rights Agreement, dated as of May 4, 2021 (the "<u>Agreement</u>"), by and between The Beauty Health Company (the "<u>Company</u>") and LCP (as defined in the Agreement). If (i) following such time that the Agreement is terminated in accordance with its terms, the Board of Directors of the Company (the "<u>Board</u>") requests in writing that I resign as a director of the Company, or (ii) the Board requests in writing my resignation pursuant to and in accordance with Section 1.1(b)(i) of the Agreement, I hereby tender the immediate resignation of my position as a director of the Company and from any and all committees of the Board on which I serve.

This resignation may not be withdrawn by me at any time.	
	Sincerely,
	/s/ Daug Schillinger

#### EMPLOYMENT AGREEMENT

THIS EMPLOYMENT AGREEMENT ("Agreement") is dated as of May 4, 2021 and effective as of the Effective Date (as defined below) between The Beauty Health Company ("Parent"), Edge Systems LLC d/b/a The HydraFacial Company (the "Company"), and Clint Carnell ("Executive").

WHEREAS, on December 8, 2020, LCP Edge Intermediate, Inc., a Delaware corporation and indirect parent of the Company, Parent and certain other parties named therein entered into an Agreement and Plan of Merger (the "*Merger Agreement*"), pursuant to which, among other things the Company will become an indirect wholly owned subsidiary of Parent (collectively, the "*Transaction*");

WHEREAS, the Company and Executive are party to that certain employment agreement, dated as of December 1, 2016 (the "Prior Agreement");

WHEREAS, the Company desires to continue employ Executive pursuant to the terms of this Agreement, and Executive desires to enter into this Agreement and to accept such employment with the Company, in each case, subject to the terms and provisions of this Agreement; and

WHEREAS, the parties intend that this Agreement shall supersede the Prior Agreement in its entirety.

NOW THEREFORE, in consideration of the mutual covenants and agreements set forth herein, the parties hereto agree as follows:

#### I. EMPLOYMENT

The Company hereby agrees to continue to employ Executive and Executive agrees to continue employment with the Company upon the terms and conditions herein set forth.

A. Employment. Executive shall serve as President and Chief Executive Officer of Parent during the Term (as defined below). Executive agrees to perform such duties as may be assigned to Executive from time to time by the Board of Directors of Parent (the "Parent Board"). Executive agrees to devote substantially all of Executive's business time and attention and best efforts to the affairs of Parent and the Company during the Term. In connection with Executive's employment by and service with the Company, Executive shall be appointed to the Parent Board effective as of the Effective Date, and thereafter shall be nominated for reelection to the Parent Board at each annual meeting of the shareholders of Parent during the Term. Executive shall have such responsibilities, power and authority as are customarily associated with the positions of president and chief executive officer in public companies, subject to the delegation of authority guidelines established by the Parent Board from time to time, and shall report solely and directly to the Parent Board.

- B. <u>Term</u>. The term of employment of Executive hereunder (the "*Term*") will be for the period commencing on the date on which the Transaction closes (the "*Effective Date*") and ending on the earliest of:
  - 1. The date of termination of Executive's employment in accordance with Article IV of this Agreement;
  - 2. The date of Executive's voluntary retirement in accordance with the Company's plans and policies; or
  - 3. The date of Executive's death.

C. <u>Principal Work Location</u>. During the Term, Executive shall perform the services required by this Agreement at the Company's offices in Long Beach, California (the "*Principal Location*"), except for travel to other locations as may be reasonably necessary to fulfill his duties and responsibilities hereunder.

### II. COMPENSATION

A. <u>Base Salary</u>. During the Term, the Company shall pay to Executive a base salary (the "*Base Salary*") at the rate of \$675,000.00 per year during 2021, payable in substantially equal semi-monthly installments pursuant to the Company's standard payroll practices. The Base Salary will be reviewed no less frequently than annually and may be adjusted upward (but not downward) by the Parent Board (or a committee thereof) in its sole discretion.

B. <u>Annual Incentive</u>: For each fiscal year of the Parent ending during the Term, Executive shall be eligible to earn a cash performance bonus (an "*Annual Bonus*") targeted at 100% of Base Salary (the "*Target Bonus*"). The actual amount of any Annual Bonus will be based upon achievement of specified levels of performance goals set by the Parent Board in consultation with Executive. Any Annual Bonus that becomes payable shall be paid at such times as annual bonuses are generally paid to senior executives of Parent, typically on or before, but in no event later than, March 15th of the calendar year following the year in which they are earned and, except as provided in Section 4, subject to Executive's continued employment through the applicable payment date. Executive's Target Bonus opportunity shall be subject to annual review by the Parent Board, and adjustments may be made at any time based upon the Parent Board's review of market trends, internal considerations and performance. The Target Bonus opportunity shall not be reduced at any time (including after any such increase).

### C. Long-Term Incentives.

1. As soon as practicable following the Effective Date, Parent will grant to Executive an option to purchase 3,100,000 shares of Parent common stock (the "*Option*") pursuant to Parent's 2021 Incentive Award Plan, as amended and/or restated from time to time (or any successor thereto) (the "*Incentive Plan*"). The Option shall have an exercise price per share equal to the Fair Market Value (as defined in the Plan) of a share of Parent common stock on the grant date. Twenty-five percent (25%) of the shares of Parent common stock subject to the Option will vest on each of the first four (4) anniversaries of the Effective Date, subject to Executive's continued employment with the Company through the applicable vesting date. The Option will be subject to the terms and conditions of the Plan and an award agreement substantially in the form attached hereto as Exhibit A (the "*Option Agreement*"), to be entered into between the Parent and Executive.

- 2. As soon as practicable following the Effective Date and the filing of the Form S-8 relating to the Incentive Plan, Parent will grant to Executive an award of performance share units covering 100,000 shares of Parent common stock ("*PSUs*") under the Incentive Plan. The PSUs will be subject to the terms and conditions (including vesting conditions set forth in the Incentive Plan and an award agreement substantially in the form attached hereto as Exhibit B, to be entered into between Parent and Executive.
- 3. For each fiscal year of Parent ending during the Term (commencing with 2022), Executive shall be eligible for one or more grants of long-term incentive awards ("*Awards*") having a grant-date value determined by the Compensation Committee of the Parent Board (the "*Compensation Committee*"), based upon the Compensation Committee's review of market trends, internal considerations and performance. The type of any such Award and the relevant terms and conditions (including vesting conditions) shall be determined by the Compensation Committee. Each Award will be subject to the terms and conditions of the Incentive Plan and an award agreement prescribed by Parent, to be entered into between Parent and Executive.
- D. <u>Reimbursement of Expenses</u>. During the Term, Executive shall be entitled to receive prompt reimbursement of all reasonable business expenses incurred by Executive in performing services hereunder, including all reasonable expenses of travel, and reasonable living expenses while away from home on business at the request of, or in the service of, the Company, *provided* that such expenses are incurred and accounted for in accordance with the policies and procedures established by the Company.
- E. <u>Legal Fees</u>. The Company has retained Fried, Frank, Harris, Shriver & Jacobson LLP ("*Fried Frank*") on behalf of Executive and other members of management. The Company will pay directly, upon delivery of customary invoices, for the reasonable fees and expenses of Fried Frank in connection with the preparation, negotiation and execution of this Agreement and the Exhibits hereto and all other advice provided by Fried Frank to or on behalf of Executive and other members of management in connection with the Transaction, up to a maximum aggregate amount of \$50,000. For the avoidance of doubt, such maximum amount shall apply to all fee and expense reimbursements for all work completed by Fried Frank in connection with its engagement by the Company since the fall of 2020, and any fees and expenses of Fried Frank in excess of such maximum amount shall not be reimbursed by the Company.
- F. Benefits. During the Term, Executive shall be eligible to participate in and be covered by all health, insurance, pension, disability insurance and other employee plans and benefits maintained by the Company for the benefit of its employees from time to time (collectively referred to herein as the "Company Benefit Plans"), on the same terms as are generally applicable to other senior executives of the Company, subject to meeting applicable eligibility requirements. Nothing contained in this Section 2.6 shall create or be deemed to create any obligation on the part of the Company to adopt or maintain any health, welfare, retirement or other benefit plan or program at any time or to create any limitation on the Company's ability to modify or terminate any such plan or program.

G. <u>Vacation and Holidays</u>. During the Term, Executive shall participate in the Company's Permissive Paid Time Off program, the terms of which can be modified by the Company at any time at the discretion of the Company. Executive shall also be entitled to such holidays as are established by the Company for all employees.

# III. NON-COMPETITION, CONFIDENTIALITY AND NONDISCLOSURE

- A. <u>Confidentiality Agreement.</u> Concurrently with the execution and delivery of this Agreement, and as part of the consideration for the promises and undertakings by Parent and the Company in this Agreement, Executive shall execute and deliver the Employee Proprietary Information and Inventions Assignment Agreement attached as Exhibit C hereto and incorporated herein by reference (the "*Confidentiality Agreement*").
- B. Other Activities. Subject to the terms of the employment hereunder, during the Term, Executive shall devote substantially all of Executive's business time and best efforts to the performance of Executive's duties and responsibilities for Parent and the Company and shall not serve on the board of directors of any for profit or not-for-profit entity without the prior consent of the Parent Board; provided, however, that Executive may serve (1) as non-executive chairman of the board of directors of OrangeTwist and may perform such duties and responsibilities as are customary for such role and (2) as a non-executive member of the board of directors of Swift Health Systems, Inc (d/b/a Inbrace). Executive agrees to resign, effective as of, or prior to, the Effective Date, from his role as (a) the Executive Chairman of OrangeTwist and (b) a member of the board of directors of SENTÉ. Executive will not, without the prior written approval of the Parent Board, engage in any other business activity or investment opportunity which is or may be competitive with the business of Parent or its affiliates, or which, individually or in the aggregate, would materially interfere or conflict with the performance of Executive's duties, services, and responsibilities hereunder, or which is in violation of applicable employee policies established from time to time by the Parent or the Company (it being agreed that, as of the date hereof, Executive's service as the non-executive chairman of the board of directors of OrangeTwist, and his investment in OrangeTwist, is not competitive with Parent or the Company and does not violate any applicable Parent or Company policy).
- C. No Violation of Other Agreements. Executive represents that, to the best of Executive's knowledge, the entrance into this Agreement and the performance of all the terms of this Agreement and as an officer of Parent and employee of the Company does not and will not breach any contract to which Executive is bound or any legal obligation of Executive:
- 1. Not to compete or to interfere with the business of a former employer (which term for purposes of this Section 3.3 shall also include persons, firms, corporations and other entities for which Executive has acted as an independent contractor or consultant); or
  - 2. Not to solicit employees, customers or vendors of any former employer.

#### IV. TERMINATION

A. <u>Definitions</u>. For purposes of this Article IV, the following definitions shall apply to the terms set forth below:

- 1. <u>Cause</u>. "Cause" shall be defined as follows:
  - a. Executive has (A) engaged in an act of theft, embezzlement or fraud or, breach of confidentiality or fiduciary duty relating to the Company; (B) breached any law, rule, regulation or policy or procedure of Parent or the Company (including but not limited to policies and procedures pertaining to harassment and discrimination); or (C) been convicted of, or plea of guilty or *nolo contendere* to, any felony;
  - b. Executive has materially breached any of the provisions of this Agreement, the Confidentiality Agreement or any other material agreement with Parent, the Company or any of their affiliates;
  - c. Actions by Executive involving willful malfeasance or gross negligence in the performance of Executive's duties which have or are reasonably likely to result in a material liability to the Company;
  - d. Executive's willful failure or refusal to perform Executive's duties as required by this Agreement if such failure to perform Executive's duties is not cured to the reasonable satisfaction of the Parent Board within ten (10) days following written notice to Executive;
  - e. Executive's willful violation of any Company policy that that is demonstrably and materially injurious to the business, financial condition or reputation of the Company or its Affiliates.

For purposes of the foregoing definition of Cause, no act or failure to act by a Participant shall be deemed willful or intentional if performed in good faith and with the reasonable belief that the action or inaction was in the best interests of the Company and its affiliates.

 <u>Change in Control</u>. "Change in Control" shall have the meaning set forth in the Incentive Plan, as amended and/or restated from time to time.

- 3. <u>Disability</u>. "Disability" shall mean that Executive has become entitled to receive benefits under the Company's applicable long-term disability plan or, if no such plan covers Executive, "Disability" means a physical or mental incapacity as a result of which Executive becomes unable to continue the proper performance of Executive's duties hereunder in substantially a full-time capacity (reasonable absences because of sickness for up to three (3) consecutive months excepted; provided, however, that any new period of incapacity or absence shall be deemed consecutive with a prior period of incapacity or absence if the new capacity or absence is determined by the Parent Board, in good faith, to be related to the prior incapacity or absence). A determination of Disability shall be subject to the certification of a qualified medical doctor agreed to by the Parent Board and Executive or, in the event of Executive's incapacity to designate a doctor, Executive's legal representative. In the absence of agreement between the Parent Board and Executive, each party shall nominate a qualified medical doctor and the two doctors so nominated shall select a third doctor, who shall make the determination as to Disability.
- Good Reason. "Good Reason" means, without Executive's written consent, the occurrence of any one or more of the following:
  - a. A material reduction in Executive's Base Salary or Target Bonus (excluding any reductions in Executive's Base Salary, and any corresponding reductions in Executive's Target Bonus, in connection with temporary across-the-board salary reductions imposed on substantially all of the Company's senior executives that do not exceed, in the aggregate, ten percent (10%) of Executive's Base Salary during any twelve (12)-month period).
  - b. The relocation of Executive's principal place of employment to a location that is greater than twenty-five (25) miles from the Principal Location if that relocation increases Executive's commute by twenty-five (25) miles or more.
  - c. A material reduction in Executive's title, duties or responsibilities as contemplated by this Agreement (other than during a period of physical or mental incapacity); *provided*, *however*, that if Executive is not re-elected as a member of the Parent Board by the shareholders of Parent after having been nominated by the Board, then this clause (iii) shall not apply.

Notwithstanding the foregoing, Executive will not be deemed to have resigned for Good Reason unless (1) Executive provides Parent with written notice setting forth in reasonable detail the facts and circumstances claimed by Executive to constitute Good Reason within sixty (60) days after the date of the occurrence of any event that Executive knows or should reasonably have known to constitute Good Reason, (2) Parent or the Company, as applicable, fails to cure such acts or omissions within thirty (30) days following Parent's receipt of such notice, and (3) the effective date of Executive's termination for Good Reason occurs no later than sixty (60) days after the expiration of such cure period.

- B. <u>Termination by Company</u>. The Company may terminate Executive's employment hereunder immediately, with or without Cause, or due to Executive's Disability. This Agreement will automatically terminate upon Executive's death during the Term.
- C. <u>Termination by Executive</u>. Executive may terminate this Agreement without Good Reason upon sixty (60) days' written notice to the Parent Board, and Executive may terminate this Agreement for Good Reason in accordance with Section 4.1(d) above.

#### D. Benefits Received Upon Termination.

- 1. If Executive's employment terminates during the Term for any reason, then the Company shall pay or provide to Executive: (i) Executive's earned but unpaid Base Salary through the Date of Termination (as defined below), (ii) to the extent required by applicable law, any vacation earned but not taken through the Date of Termination, and (iii) any vested amounts due to Executive under any plan, program or policy of the Company (collectively, the "*Accrued Obligations*"). The Accrued Obligations described in clauses (i) (ii) of the preceding sentence shall be paid within thirty (30) days after the Date of Termination (or such earlier date as may be required by applicable law) and the Accrued Obligations described in clause (iii) of the preceding sentence shall be paid in accordance with the terms of the governing plan or program. The Company and Parent shall thereafter have no further obligations to Executive under this Agreement.
- 2. If Executive's employment is terminated by the Company without Cause (excluding termination by reason of death or Disability), or Executive terminates his employment for Good Reason, in either case, prior to the consummation of a Change in Control or more than twelve (12) months after the consummation of a Change in Control, then upon Executive's "separation from service" from the Company (within the meaning of Section 409A of the Internal Revenue Code of 1986, as amended (the "Code")) (a "Separation from Service" and, the date of any such Separation from Service, the "Date of Termination"), subject to Section 4.5 below, the Company shall:

- a. pay to Executive an amount equal to Executive's earned but unpaid Annual Bonus for the fiscal year ending immediately prior to the year in which the Date of Termination occurs, said Annual Bonus to be paid as and when annual bonuses are payable for such year generally;
- b. pay to Executive as severance pay an amount equal to eighteen (18) months of Executive's Base Salary in effect as of the Date of Termination with such payments to be made in accordance with the Company's usual payroll periods during the eighteen (18) month period commencing on the Date of Termination; *provided*, that no such payments shall be made prior to the date on which the Release (as defined below) becomes effective and irrevocable and, if the aggregate period during which Executive is entitled to consider and/or revoke the Release spans two (2) calendar years, no payments under this Section 4.4(b)(ii) shall be made prior to the beginning of the second (2<sup>nd</sup>) such calendar year (and any payments otherwise payable prior thereto (if any) shall instead be paid commencing on the first regularly scheduled Company payroll date occurring in the latter such calendar year);
- c. pay to Executive a pro-rated Target Bonus amount for the year in which the Date of Termination occurs, determined by multiplying Executive's Target Bonus for the year in which the Date of Termination occurs by a fraction, the numerator of which equals the number of days Executive was employed by the Company during the calendar year in which the Date of Termination occurs and the denominator of which equals 365 or 366 (as applicable) (a "*Pro-Rata Bonus*"), payable on the first regularly scheduled payroll date following the date on which the Release becomes effective and irrevocable, *provided*, that if the aggregate period during which Executive is entitled to consider and/or revoke the Release spans two (2) calendar years, such payment shall be made on the first regularly scheduled payroll date in the second (2nd) such calendar year or, if later, the first regularly scheduled payroll date following the date the Release becomes effective and irrevocable; and
- d. subject to Executive's valid election to continue healthcare coverage under Section 4980B of the Code, during the period commencing on the Date of Termination and ending on the date that is eighteen (18) months thereafter, or, if earlier, the date on which Executive becomes covered by a group health insurance program provided by a subsequent

employer (in either case, the "COBRA Period"), the Company shall reimburse Executive for Executive's and Executive's eligible dependents with coverage under its group health plans at the same levels and the same cost to Executive as would have applied if Executive's employment had not been terminated based on Executive's elections in effect on the Date of Termination, provided, however, that (A) if any plan pursuant to which such benefits are provided is not, or ceases prior to the expiration of the period of continuation coverage to be, exempt from the application of Section 409A under Treasury Regulation Section 1.409A-1(a)(5), or (B) the Company is otherwise unable to continue to cover Executive under its group health plans without incurring penalties (including without limitation, pursuant to Section 2716 of the Public Health Service Act or the Patient Protection and Affordable Care Act), then, in either case, an amount equal to each remaining Company subsidy shall thereafter be paid to Executive in substantially equal monthly installments over the continuation coverage period (or the remaining portion thereof).

- 3. If within twelve (12) months following the consummation of a Change in Control, Executive's employment is terminated either by the Company without Cause (excluding by reason of death or Disability) or by Executive for Good Reason, then, in either case, upon Executive's Separation from Service, subject to Section 4.5 below, the Company shall:
  - a. pay to Executive an amount equal to Executive's earned but unpaid Annual Bonus for the year ending immediately prior to the year in which the Date of Termination occurs, said Annual Bonus to be paid as and when annual bonuses are payable for such year generally;
  - b. pay to Executive as severance pay an amount equal to the sum of (A) eighteen (18) months of Executive's Base Salary in effect as of the Date of Termination and (B) one and one-half (1.5) Executive's Target Bonus for the year in which the Date of Termination occurs, such payments to be made in accordance with the Company's usual payroll periods during the eighteen (18) month period commencing on the Date of Termination; *provided*, that no such payments shall be made prior to the date on which the Release becomes effective and irrevocable and, if the aggregate period during which Executive is entitled to consider and/or revoke the Release spans two (2) calendar years, no payments under this Section 4.4(c)(ii) shall be made prior to the beginning of the second (2nd) such calendar year (and any payments otherwise payable prior thereto (if any) shall instead be paid commencing on the first regularly scheduled Company payroll date occurring in the latter such calendar year);

- c. pay to Executive a Pro-Rata Bonus, payable in a lump sum on the first regularly scheduled payroll date following the date on which the Release becomes effective and irrevocable, *provided*, that if the aggregate period during which Executive is entitled to consider and/or revoke the Release spans two (2) calendar years, such payment shall be made on the first regularly scheduled payroll date in the second (2<sup>nd</sup>) such calendar year or, if later, the first regularly scheduled payroll date following the date the Release becomes effective and irrevocable; and
- d. subject to Executive's valid election to continue healthcare coverage under Section 4980B of the Code, during the COBRA Period, the Company shall reimburse Executive for Executive's and Executive's eligible dependents with coverage under its group health plans at the same levels and the same cost to Executive as would have applied if Executive's employment had not been terminated based on Executive's elections in effect on the Date of Termination, provided, however, that (A) if any plan pursuant to which such benefits are provided is not, or ceases prior to the expiration of the period of continuation coverage to be, exempt from the application of Section 409A under Treasury Regulation Section 1.409A-1(a)(5), or (B) the Company is otherwise unable to continue to cover Executive under its group health plans without incurring penalties (including without limitation, pursuant to Section 2716 of the Public Health Service Act or the Patient Protection and Affordable Care Act), then, in either case, an amount equal to each remaining Company subsidy shall thereafter be paid to Executive in substantially equal monthly installments over the continuation coverage period (or the remaining portion thereof).
- e. Executive agrees that, if (i) a Change in Control occurs within twelve (12) months following the consummation of the Transaction (or pursuant to a binding agreement entered into within such twelve (12) month period) and (ii) Executive's Termination occurs within thirty (30) days after such Change in Control, then, notwithstanding anything to the contrary in any other plan or agreement, including without limitation the Incentive Plan, the Option Agreement, and the award agreement governing the PSUs, such Termination will not result in any accelerated vesting of the Option or the PSUs.

- 4. Termination Because of Employee Death or Disability. In the event of Executive's Disability, Executive acknowledges that his employment may be terminated by the Company; provided that, during the period of the Disability prior to such termination of employment, Executive shall continue to receive all compensation and benefits as if Executive were actively employed less any sums received directly by Executive, if any, under any applicable disability income insurance policy maintained by the Company. In the event that the Company terminates Executive's employment due to his Disability, Executive shall have the right to continue to receive any payments made under any applicable disability insurance policy maintained by the Company in accordance with, and subject to the terms and conditions of, such policy. In addition, Executive (or Executive's estate) shall receive a Pro-Rata Bonus within thirty (30) days following the Date of Termination due to death or Disability and any earned but unpaid Annual Bonus for the year ending immediately prior to the year in which the Date of Termination occurs, said Annual Bonus to be paid as and when annual bonuses are payable for such year generally.
- E. <u>Release</u>. Notwithstanding the foregoing, it shall be a condition to Executive's right to receive the amounts provided for in Section 4.4(b) or (c) hereof (as applicable) that Executive execute and deliver to the Company a release of claims substantially in the form attached hereto as Exhibit D that becomes effective and irrevocable no more than sixty (60) days after the date on which Executive's employment terminates.
- F. <u>Effect of Termination</u>. Upon any termination of Executive's employment with the Company for any reason, Executive shall be deemed to have immediately resigned as Chief Executive Officer of Parent and the Company, and in any other capacity with Parent and the Company (including as an employee, officer and/or director), as well as with all subsidiaries, if applicable, without the giving of any notice or the taking of any action.

## V. <u>SUCCESSORS AND ASSIGNS</u>

A. <u>Assignment</u>. This Agreement shall inure to the benefit of and be binding upon Parent, the Company and to any person or entity which succeeds to all or substantially all of the business of Parent and/or the Company through merger, consolidation, reorganization, or other business combination or by acquisition of all or substantially all of the assets of Parent and/or the Company. To the extent that any such successor or assign (whether direct or indirect, by purchase, merger, consolidation or otherwise) to all or substantially all of the business and/or assets of Parent and/or the Company does not automatically, under applicable law, assume all obligations under this Agreement, then Parent and/or the Company will utilize its best efforts to obtain the agreement

of the successor or assign to assume all the obligations arising from this Agreement, and to perform this Agreement in the same manner and to the same extent that Parent and/or the Company would be required to perform it if no such succession or assignment had taken place. Any failure of Parent and/or the Company to obtain such agreement prior to the effectiveness of any such succession or assignment shall be deemed to be a material breach of this Agreement unless otherwise agreed to between Executive and such successor or assign. The obligations of this Article shall apply equally to Parent and the Company, as herein before defined, and to any future successor or assign to its business which automatically by operation of law or otherwise (including pursuant to such successor's or assign's agreement to assume all obligations arising from this Agreement) becomes bound by all the terms and provisions of this Agreement (*i.e.* the obligation to use best efforts to obtain the agreement of a potential second successor is assumed by the first successor when it assumes the obligations of this Agreement). The obligations of this Article V shall also apply to any corporation (*i.e.*, subsidiary or affiliated company) where Parent or the Company owns the majority of the voting securities of the corporation and the corporation becomes the employer for Executive at any time during the term of this Agreement.

B. <u>Executive Assigns</u>. This Agreement is personal to Executive and, without the prior written consent of Parent, shall not be assignable by Executive other than by will or the laws of descent and distribution. This Agreement shall inure to the benefit of and be enforceable by Executive and Executive's legal representatives, executors, heirs, distributees, devisees and legatees.

# VI. EXCESS PARACHUTE PAYMENTS; LIMITATION ON PAYMENTS

A. <u>Best Pay Cap.</u> Notwithstanding any other provision of this Agreement, in the event that any payment or benefit received or to be received by Executive (whether pursuant to the terms of this Agreement or any other plan, arrangement or agreement) (all such payments and benefits being hereinafter referred to as the "*Total Payments*") would be subject (in whole or part), to the excise tax imposed under Section 4999 of the Code (the "*Excise Tax*"), then, after taking into account any reduction in the Total Payments provided by reason of Section 280G of the Code in any other plan, arrangement or agreement, then such remaining Total Payments shall be reduced, to the extent necessary so that no portion of the Total Payments is subject to the Excise Tax but only if (i) the net amount of such Total Payments, as so reduced (and after subtracting the net amount of federal, state and local income taxes on such reduced Total Payments and after taking into account the phase out of itemized deductions and personal exemptions attributable to such reduced Total Payments and after taking into account the phase out of itemized deductions and personal exemptions attributable to such unreduced Total Payments and after taking into account the phase out of itemized deductions and personal exemptions attributable to such unreduced Total Payments and after taking into account the phase out of itemized deductions and personal exemptions attributable to such unreduced Total Payments). The reduction of the amounts payable hereunder, if applicable, shall be made by reducing the payments and benefits in the following order: (i) cash payments that may not be valued under 24(c), (iii) cash payments that may be valued under 24(c), (iv) equity-based payments that may be valued under 24(c), (iv) equity-based payments that may be valued under 24(c) and (v) other types of benefits. With respect to each category of the foregoing, such reduction shall occur first with respect to amounts that are not "deferred compensation" within the

meaning of Section 409A of the Code and next with respect to payments that are deferred compensation within the meaning of Section 409A of the Code, in each case, beginning with payments or benefits that are to be paid the farthest in time from the determination of the Independent Advisors (as defined below). All reasonable fees and expenses of the Independent Advisors shall be borne solely by the Company.

B. <u>Certain Exclusions</u>. For purposes of determining whether and the extent to which the Total Payments will be subject to the Excise Tax, (i) no portion of the Total Payments the receipt or enjoyment of which Executive shall have waived at such time and in such manner as not to constitute a "payment" within the meaning of Section 280G(b) of the Code shall be taken into account; (ii) no portion of the Total Payments shall be taken into account which, in the written opinion of an independent, nationally recognized accounting firm (the "*Independent Advisors*") selected by Parent, does not constitute a "parachute payment" within the meaning of Section 280G(b)(2) of the Code (including by reason of Section 280G(b)(4)(A) of the Code) and, in calculating the Excise Tax, no portion of such Total Payments shall be taken into account which, in the opinion of Independent Advisors, constitutes reasonable compensation for services actually rendered, within the meaning of Section 280G(b)(4)(B) of the Code, in excess of the "base amount" (as defined in Section 280G(b)(3) of the Code) allocable to such reasonable compensation; and (iii) the value of any non-cash benefit or any deferred payment or benefit included in the Total Payments shall be determined by the Independent Advisors in accordance with the principles of Sections 280G(d)(3) and (4) of the Code.

### VII. GENERAL PROVISIONS

A. <u>Notice</u>. For purposes of this Agreement, notices and all other communications provided for in the Agreement shall be in writing and shall be deemed to have been duly given when delivered in person or mailed by United States registered mail, return receipt requested, postage prepaid, as follows:

If to Parent or the Company: The Beauty Health Company

1819 West Avenue

Bay 2

Miami Beach, Florida 33139 Attn: Executive Chairman

If to Executive:

to the last address on file for Executive with the Company or such other address as either party may have furnished to the other in writing in accordance herewith, except that notices or change of address shall be effective only upon receipt.

B. <u>Amendments; No Waivers</u>. No provision of this Agreement may be amended, modified, waived or discharged unless such amendment, waiver, modification or discharge is agreed to in a writing signed by Executive, Parent and the Company. No waiver by either party hereto at any time of any breach by the other party hereto of, or compliance with, any condition or provision of this Agreement to be performed by such other party shall be deemed a waiver of similar provisions or conditions at the same or at any prior or subsequent time.

- C. <u>Governing Law</u>. This Agreement shall be governed by and construed and interpreted in accordance with the laws of the State of California, without regard to its conflicts of law principles.
- D. <u>Severability or Partial Invalidity</u>. The invalidity or unenforceability of any provisions of this Agreement shall not affect the validity or enforceability of any other provision of this Agreement, which shall remain in full force and effect.
- E. <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, each of which shall be deemed to be an original but all of which together will constitute one and the same instrument.
- F. Entire Agreement. This Agreement, together with the Confidentiality Agreement, constitutes the entire agreement of the parties and supersedes all prior written or oral and all contemporaneous oral agreements, understandings, and negotiations between the parties with respect to the subject matter hereof, including without limitation the Prior Agreement. This Agreement, together with the Confidentiality Agreement, is intended by the parties as the final expression of their agreement with respect to such terms as are included in this Agreement and may not be contradicted by evidence of any prior or contemporaneous agreement. The parties further intend that this Agreement, together with the Confidentiality Agreement, constitutes the complete and exclusive statement of its terms and that no extrinsic evidence may be introduced in any judicial proceeding involving this Agreement. Notwithstanding anything herein to the contrary, (i) this Agreement and the obligations and commitments hereunder shall neither commence nor be of any force or effect prior to the Effective Date and unless and until the Transaction is fully consummated in conformity with the terms of the Merger Agreement and (ii) in the event that the Merger Agreement is terminated in accordance with its terms, the Transaction is not consummated for any reason, or Executive's employment with the Company terminates for any reason prior to the closing of the Transaction, this Agreement will automatically terminate and be void *ab initio*.

#### G. Arbitration.

1. All disputes, controversies, and claims between Executive and Parent and/or the Company, or any of its officers, directors, employees, or agents in their capacity as such, including any controversy or dispute, whether based on contract, common law, or federal, state or local statute or regulation, that arise under or are related to this Agreement Executive's employment with the Company or the termination thereof shall be submitted to final and binding arbitration as the sole and exclusive remedy for such controversy or dispute in accordance with the rules of JAMS pursuant to its Employment Arbitration Rules and Procedures (the current version of which are available at http://www.jamsadr.com/rules-employment-arbitration/ and a copy

of which will be provided by the Company to Executive upon Executive's request). Notwithstanding the foregoing, this Agreement shall not require arbitration pursuant to this Section 7.7 of any claims: (A) under a Company benefit plan subject to the Employee Retirement Income Security Act, as amended; or (B) as to which applicable law not preempted by the Federal Arbitration Act prohibits resolution by binding arbitration. Either party may seek provisional non-monetary remedies in a court of competent jurisdiction to the extent that such remedies are not available or not available in a timely fashion through arbitration. It is the parties' intent that issues of arbitrability of any dispute shall be decided by the arbitrator.

- 2. The arbitration shall take place before a single neutral arbitrator at the JAMS office in Los Angeles, California. Such arbitrator shall be provided through JAMS by mutual agreement of the parties to the arbitration; *provided*, that, absent such agreement, the arbitrator shall be selected in accordance with the rules of JAMS then in effect. The arbitrator shall permit reasonable discovery. The award or decision of the arbitrator shall be rendered in writing; shall be final and binding on the parties; and may be enforced by judgment or order of a court of competent jurisdiction.
- 3. The Company shall pay all arbitrator fees, filing fees and other costs unique to the arbitration procedure (in each case, to the extent required by applicable law), except that if Executive initiates a claim subject to arbitration, Executive will pay any filing fee up to the amount Executive would be required to pay if Executive initiated the claim in the Superior Court of Los Angeles for the State of California for the County of Los Angeles. All other fees and costs shall be shared evenly by the Company and Executive. In the event of arbitration relating to this Agreement, the non-prevailing party shall reimburse the prevailing party for all costs incurred by the prevailing party in connection with such arbitration (including, without limitation, reasonable legal fees in connection with such arbitration, including any litigation or appeal therefrom).
- 4. EXECUTIVE, PARENT AND THE COMPANY UNDERSTAND THAT BY AGREEING TO ARBITRATE ANY ARBITRATION CLAIM, THEY WILL NOT HAVE THE RIGHT TO HAVE ANY ARBITRATION CLAIM DECIDED BY A JURY OR A COURT, BUT SHALL INSTEAD HAVE ANY ARBITRATION CLAIM DECIDED THROUGH ARBITRATION.

- 5. EXECUTIVE, PARENT AND THE COMPANY WAIVE ANY CONSTITUTIONAL OR OTHER RIGHT TO BRING CLAIMS COVERED BY THIS AGREEMENT OTHER THAN IN THEIR INDIVIDUAL CAPACITIES. EXCEPT AS MAY BE PROHIBITED BY LAW, THIS WAIVER INCLUDES THE ABILITY TO ASSERT CLAIMS AS A PLAINTIFF OR CLASS MEMBER IN ANY PURPORTED CLASS OR REPRESENTATIVE PROCEEDING.
- 6. This Section 7.7 shall be interpreted to conform to any applicable law concerning the terms and enforcement of agreements to arbitrate service disputes. To the extent any terms or conditions of this Section 7.7 would preclude its enforcement, such terms shall be severed or interpreted in a manner to allow for the enforcement of this Section 7.7. To the extent applicable law imposes additional requirements to allow enforcement of this Section 7.7, this Agreement shall be interpreted to include such terms or conditions.

H. <u>Indemnification</u>. To the extent permitted by law, applicable statutes and the Articles of Incorporation, By-laws or resolutions of Parent and the Company in effect from time to time, during the Term, Executive shall be entitled to indemnification by Parent and the Company. In addition, during the Term, Parent and/or the Company shall provide Executive with coverage under the directors and officers liability insurance policy, if any, maintained by Parent and/or the Company for the benefit of the members of the Parent Board and officers of Parent and the Company, to the same extent as such coverage is provided to members of the Parent Board and similarly-situated executive officers of Parent and the Company.

#### I. Section 409A.

1. To the extent applicable, this Agreement shall be interpreted in accordance with Section 409A of the Code and Department of Treasury regulations and other interpretative guidance issued thereunder, including without limitation any such regulations or other such guidance that may be issued after the Effective Date (collectively, "Section 409A"). Notwithstanding any provision of this Agreement to the contrary, in the event that following the Effective Date, Parent or the Company determines that any compensation or benefits payable under this Agreement may be subject to Section 409A, Parent and the Company may adopt such amendments to this Agreement or adopt other policies or procedures (including amendments, policies and procedures with retroactive effect), or take any other actions that Parent and the Company determines are necessary or appropriate to preserve the intended tax treatment of the compensation and benefits payable hereunder, including without limitation actions intended to (i) exempt the compensation and benefits payable under this Agreement from Section 409A, and/or (ii) comply with the requirements of Section 409A, provided, however, that this Section 7.9(a) does not, and shall not be construed so as to, create any obligation on the part of Parent or the Company to adopt any such amendments, policies

- or procedures or to take any other such actions or to create any liability on the part of Parent or the Company for any failure to do so. In no event shall Parent, the Company, any of their respective affiliates or any of their respective officers, directors or advisors be liable for any taxes, penalties or interest imposed under or by operation of Section 409A. Any right to a series of installment payments pursuant to this Agreement is to be treated as a right to a series of separate payments.
- 2. Notwithstanding anything to the contrary in this Agreement, no compensation or benefits (including, without limitation, any compensation or benefits provided pursuant to Section 4.4(b) or (c) above) shall be paid to Executive during the six (6)-month period following Executive's Separation from Service if the Company determines that paying such amounts at the time or times indicated in this Agreement would be a prohibited distribution under Section 409A(a)(2)(B)(i) of the Code. If the payment of any such amounts is delayed as a result of the previous sentence, then on the first business day following the end of such six (6)-month period (or such earlier date upon which such amount can be paid under Section 409A without resulting in a prohibited distribution, including as a result of Executive's death), the Company shall pay Executive a lump-sum amount equal to the cumulative amount that would have otherwise been payable to Executive during such period (without interest).

J. <u>Withholding</u>. Any payments hereunder will be subject to any required withholding of federal, state and local taxes pursuant to applicable law or regulation, and the Company and its affiliates shall be entitled to withhold any and all such taxes from amounts payable hereunder.

[Signature Page Follows]

IN WITNESS WHEREOF, the parties have executed this Agreement as of the Effective Date.

EXECUTIVE

/s/ Clinton E. Carnell
By Clint Carnell

[Signature Page to CEO Employment Agreement]

# PARENT

/s/ Brenton L. Saunders

Brent Saunders, Executive Chairman of the Parent Board

[Signature Page to CEO Employment Agreement]

COMPANY

/s/ Liyuan Woo Liyuan Woo, CFO

[Signature Page to CEO Employment Agreement]

## **EXHIBIT A**

## **OPTION AWARD AGREEMENT**

## THE BEAUTY HEALTH COMPANY

### 2021 INCENTIVE AWARD PLAN

# STOCK OPTION GRANT NOTICE CEO AND CFO FORM

The Beauty Health Company, a Delaware corporation (the "Company") has granted to the participant listed below ("Participant") the stock option (the "Option") described in this Stock Option Grant Notice (the "Grant Notice"), subject to the terms and conditions of The Beauty Health Company 2021 Incentive Award Plan (as amended from time to time, the "Plan") and the Stock Option Agreement attached hereto as Exhibit A (the "Agreement"), both of which are incorporated into this Grant Notice by reference. Capitalized terms not specifically defined in this Grant Notice or the Agreement have the meanings given to them in the Plan.

[\_\_\_]

Participant:

•	
Grant Date:	[], 2021
Exercise Price per Share:	
Shares Subject to the Option:	
Final Expiration Date:	
Vesting Commencement Date:	[], 2021
Vesting Schedule:	Twenty-five percent (25%) of the Shares subject to the Option will vest on each of the first four anniversaries of the Vesting Commencement Date, subject to Participant's continued status as a Service Provider through the applicable vesting date. Notwithstanding the foregoing, (x) if Participant incurs a Termination of Service due to Participant's death or Disability, subject to and conditioned upon Participant's (or Participant's estate's) timely execution and non-revocation of a release of claims in a form prescribed by the Company (a "Release") that becomes effective and irrevocable no later than sixty (60) days following such Termination of Service (the date such Release becomes effective and irrevocable, the "Release Effective Date"), the Option will vest in full (to the extent then-unvested) upon the Release Effective Date (and shall remain outstanding and eligible to vest through the

Release Effective Date and shall automatically be forfeited if the Release does not become effective and irrevocable on or prior to the sixtieth (60th) day following such termination), and (y) if a Change in Control is consummated more than twelve (12) months after [Closing Date], 2021 (the "Closing Date") and Participant's status as a Service Provider is terminated by the Company or its Subsidiaries without Cause (as defined in the Agreement) or due to Participant's resignation for Good Reason (as defined in the Employment Agreement between Participant and the Company, dated [\_\_\_]), in either case, within twelve (12) months following the consummation of such Change in Control, subject to and conditioned upon Participant's timely execution and non-revocation of a Release that becomes effective and irrevocable no later than sixty (60) days following such Termination of Service, the Option will vest in full (to the extent then-unvested) upon the Release Effective Date (and shall remain outstanding and eligible to vest through the Release Effective Date and shall automatically be forfeited if the Release does not become effective and irrevocable on or prior to the sixtieth (60th) day following such termination).

If (i) a Change in Control occurs within twelve (12) months following the Closing Date (or pursuant to a binding agreement entered into within such twelve (12) month period) and (ii) Participant's status as a Service Provider is terminated by the Company or its Subsidiaries without Cause (as defined in the Agreement) or due to Participant's resignation for Good Reason within twelve (12) months after such Change in Control, then, notwithstanding anything to the contrary in any other plan or agreement, such termination will not result in any accelerated vesting of the Option.

[Incentive Stock Option]/[Non-Qualified Stock Option]

**Type of Option** 

By accepting (whether in writing, electronically or otherwise) the Option, Participant agrees to be bound by the terms of this Grant Notice, the Plan and the Agreement. Participant has reviewed the Plan, this Grant Notice and the Agreement in their entirety, has had an opportunity to obtain the advice of counsel prior to executing this Grant Notice and fully understands all provisions of the Plan, this Grant Notice and the Agreement. Participant hereby agrees to accept as binding, conclusive and final all decisions or interpretations of the Administrator upon any questions arising under the Plan, this Grant Notice or the Agreement.

THE BEAUTY HEALTH COMPANY	PARTICIPANT		
By:			
Name:	[Participant Name]		
Title:			
	•		

#### STOCK OPTION AGREEMENT

Capitalized terms not specifically defined in this Agreement have the meanings specified in the Grant Notice or, if not defined in the Grant Notice, in the Plan.

# ARTICLE I. GENERAL

- 1.1 Grant of Option. The Company has granted to Participant the Option effective as of the grant date set forth in the Grant Notice (the "*Grant Date*").
- 1.2 Incorporation of Terms of Plan. The Option is subject to the terms and conditions set forth in this Agreement and the Plan, which is incorporated herein by reference. In the event of any inconsistency between the Plan and this Agreement, the terms of the Plan will control, unless it is expressly specified in in this Agreement or the Grant Notice that the specific provision of the Plan will not apply. For clarity, the foregoing sentence shall not limit the applicability of any additive language contained in this Agreement which provides supplemental or additional terms not inconsistent with the Plan.

# ARTICLE II. PERIOD OF EXERCISABILITY

- 2.1 Commencement of Exercisability. The Option will vest and become exercisable according to the vesting schedule in the Grant Notice (the "Vesting Schedule") except that any fraction of a Share as to which the Option would be vested or exercisable will be accumulated and will vest and become exercisable only when a whole vested Share has accumulated; provided, however, that, notwithstanding the foregoing or anything to the contrary in the Grant Notice or this Agreement, in no event may the Option be exercised (in whole or in part) prior to the date on which the Company files a Form S-8 Registration Statement covering the Shares subject to the Option. Except as otherwise set forth in the Grant Notice, the Plan or this Agreement, and unless the Administrator otherwise determines, the Option will immediately expire and be forfeited as to any portion of the Option that is not vested and exercisable as of Participant's Termination of Service for any reason (after taking into consideration any accelerated vesting and exercisability which may occur in connection with such Termination of Service, if any).
- 2.2 Duration of Exercisability. The Vesting Schedule is cumulative. Any portion of the Option which vests and becomes exercisable will remain vested and exercisable until the Option expires. The Option will be forfeited immediately upon its expiration.
- 2.3 Expiration of Option. Except as may be extended in accordance with Section 5.3 of the Plan, the Option may not be exercised to any extent by anyone after, and will expire on, the first of the following to occur:
  - (a) The final expiration date in the Grant Notice;
- (b) Except as the Administrator may otherwise approve, the expiration of three (3) months from the date of Participant's Termination of Service, unless Participant's Termination of Service is for Cause (as defined below), due to Participant's voluntary resignation or by reason of Participant's death or Disability;

- (c) Except as the Administrator may otherwise approve, the expiration of one year from the date of Participant's Termination of Service by reason of Participant's death or Disability;
- (d) Except as the Administrator may otherwise approve, the expiration of one (1) month from the date of Participant's Termination of Service due to Participant's voluntary resignation for any reason; and
  - (e) Except as the Administrator may otherwise approve, Participant's Termination of Service for Cause.
- 2.4 <u>Cause Definition</u>. As used in this Agreement, "*Cause*" means (i) if Participant is a party to a written employment or consulting agreement with the Company or a Subsidiary in which the term "cause" is defined (a "*Relevant Agreement*"), "Cause" as defined in the Relevant Agreement, and (ii) if no Relevant Agreement exists, the occurrence of any one or more of the following events:
- (a) Participant has engaged in an act of dishonesty, theft, embezzlement or fraud, a breach of confidentiality, an unauthorized disclosure or use of inside information, customer lists, trade secrets or other confidential information; has breached a fiduciary duty, law, rule, regulation or policy or procedure of the Company (including but not limited to policies and procedures pertaining to harassment and discrimination); Participant's commission of, or plea of guilty or *nolo contendere* to, any crime or offense (other than minor traffic violations or similar offenses), or any act by Participant constituting a felony;
- (b) Participant's gross or repeated neglect of, or repeated or willful failure to perform, his or her duties to the Company, or Participant's history of substandard performance, if such substandard performance is not cured to the satisfaction of the Board, the Chief Executive Officer of the Company (the "CEO") (unless Participant is the CEO), and/or Participant's manager within ten (10) days following notice to Participant;
- (c) Participant has breached any of the provisions of any employment, confidentiality, restrictive covenant or other agreement between Participant and the Company or an affiliate thereof;
  - (d) actions by Participant involving malfeasance or gross negligence in the performance of, Participant's duties;
- (e) Participant's failure or refusal to perform Participant's duties to the Company on an exclusive and full-time basis if such failure to perform Participant's duties is not cured to the satisfaction of the Board, the CEO (unless Participant is the CEO), and/or Participant's manager (as applicable) within ten (10) days following notice to the Participant;

- (f) Participant's insubordination or failure to follow the Board's (or the CEO's (unless Participant is the CEO) or such Participant's manager's) instructions;
- (g) Participant's violation of any rule, regulation, or policy of the Company or the Board (or Participant's manager) applicable to similarly-situated employees of the Company generally, including, without limitation, rules, regulations, or policies addressing confidentiality, non-solicitation or non-competition, if such violation is not cured to the satisfaction of the Board, the CEO (unless Participant is the CEO), and/or Participant's manager (as applicable) within ten (10) days following notice to the Participant;
- (h) Participant's use of alcohol or illicit drugs in a manner that has or would reasonably be expected to have a detrimental effect on Participant's performance, Participant's duties to Company, or the reputation of the Company or its affiliates; or
- (i) Participant's performance of acts which are or could reasonably be expected to become materially detrimental to the image, reputation, finances or business of the Company or any of its affiliates, including but limited to the Participant's commission of unlawful harassment or discrimination.

# ARTICLE III. EXERCISE OF OPTION

- 3.1 Person Eligible to Exercise. During Participant's lifetime, only Participant may exercise the Option. After Participant's death, any exercisable portion of the Option may, prior to the time the Option expires, be exercised by Participant's Designated Beneficiary as provided in the Plan.
- 3.2 Partial Exercise. Any exercisable portion of the Option or the entire Option, if then wholly exercisable, may be exercised, in whole or in part, according to the procedures in the Plan at any time prior to the time the Option or portion thereof expires, except that the Option may only be exercised for whole Shares.

## 3.3 Tax Withholding; Exercise Price.

- (a) Unless the Administrator otherwise determines, the Company shall withhold, or cause to be withheld, Shares otherwise vesting or issuable under this Option in satisfaction of any exercise price and/or applicable withholding tax obligations, in accordance with the Plan. With respect to tax withholding obligations, the number of Shares which may be so withheld or surrendered shall be limited to the number of Shares which have a fair market value on the date of withholding no greater than the aggregate amount of such liabilities based on the maximum individual statutory withholding rates in Participant's applicable jurisdictions for federal, state, local and foreign income tax and payroll tax purposes that are applicable to such taxable income, in accordance with Section 9.5 of the Plan.
- (b) Participant acknowledges that Participant is ultimately liable and responsible for all taxes owed in connection with the Option, regardless of any action the Company or any Subsidiary takes with respect to any tax withholding obligations that arise in connection with the Option. Neither the Company nor any Subsidiary makes any representation or undertaking regarding the treatment of any tax withholding in connection with the awarding, vesting or exercise of the Option or the subsequent sale of Shares. The Company and the Subsidiaries do not commit and are under no obligation to structure the Option to reduce or eliminate Participant's tax liability.

3.4 Representation. Participant represents to the Company that Participant has reviewed with Participant's own tax advisors the tax consequences of this Award and the transactions contemplated by the Grant Notice and this Agreement. Participant is relying solely on such advisors and not on any statements or representations of the Company or any of its agents.

# ARTICLE IV. OTHER PROVISIONS

- 4.1 Adjustments. Participant acknowledges that the Option is subject to adjustment, modification and termination in certain events as provided in this Agreement and the Plan.
- 4.2 Clawback. The Option and the Shares issuable hereunder shall be subject to any clawback or recoupment policy in effect on the Grant Date or as may be adopted or maintained by the Company following the Grant Date, including the Dodd-Frank Wall Street Reform and Consumer Protection Act and any rules or regulations promulgated thereunder.
- 4.3 Notices. Any notice to be given under the terms of this Agreement to the Company must be in writing and addressed to the Company in care of the Company's General Counsel at the Company's principal office or the General Counsel's then-current email address or facsimile number. Any notice to be given under the terms of this Agreement to Participant must be in writing and addressed to Participant (or, if Participant is then deceased, to the Designated Beneficiary) at Participant's last known mailing address, email address or facsimile number in the Company's personnel files. By a notice given pursuant to this Section, either party may designate a different address for notices to be given to that party. Any notice will be deemed duly given when actually received, when sent by email, when sent by certified mail (return receipt requested) and deposited with postage prepaid in a post office or branch post office regularly maintained by the United States Postal Service, when delivered by a nationally recognized express shipping company or upon receipt of a facsimile transmission confirmation.
  - 4.4 Titles. Titles are provided herein for convenience only and are not to serve as a basis for interpretation or construction of this Agreement.
- 4.5 Conformity to Securities Laws. Participant acknowledges that the Plan, the Grant Notice and this Agreement are intended to conform to the extent necessary with all Applicable Laws and, to the extent Applicable Laws permit, will be deemed amended as necessary to conform to Applicable Laws.
- 4.6 Successors and Assigns. The Company may assign any of its rights under this Agreement to single or multiple assignees, and this Agreement will inure to the benefit of the successors and assigns of the Company. Subject to the restrictions on transfer set forth in this Agreement or the Plan, this Agreement will be binding upon and inure to the benefit of the heirs, legatees, legal representatives, successors and assigns of the parties hereto.
- 4.7 Limitations Applicable to Section 16 Persons. Notwithstanding any other provision of the Plan or this Agreement, if Participant is subject to Section 16 of the Exchange Act, the Plan, the Grant Notice, this Agreement and the Option will be subject to any additional limitations set forth in any applicable exemptive rule under Section 16 of the Exchange Act (including any amendment to Rule 16b-3) that are requirements for the application of such exemptive rule. To the extent Applicable Laws permit, this Agreement will be deemed amended as necessary to conform to such applicable exemptive rule.

- 4.8 Entire Agreement. The Plan, the Grant Notice and this Agreement (including any exhibit hereto) constitute the entire agreement of the parties and supersede in their entirety all prior undertakings and agreements of the Company and Participant with respect to the subject matter hereof.
- 4.9 Severability. If any portion of the Grant Notice or this Agreement or any action taken under the Grant Notice or this Agreement, in any case is held illegal or invalid for any reason, the illegality or invalidity will not affect the remaining parts of the Grant Notice and/or this Agreement (as applicable), and the Grant Notice and/or this Agreement (as applicable) will be construed and enforced as if the illegal or invalid provisions had been excluded, and the illegal or invalid action will be null and void.
- 4.10 Limitation on Participant's Rights. Participation in the Plan confers no rights or interests other than as herein provided. This Agreement creates only a contractual obligation on the part of the Company as to amounts payable and may not be construed as creating a trust. Neither the Plan nor any underlying program, in and of itself, has any assets. Participant will have only the rights of a general unsecured creditor of the Company with respect to amounts credited and benefits payable, if any, with respect to the Option, and rights no greater than the right to receive the Shares as a general unsecured creditor with respect to the Option, as and when exercised pursuant to the terms hereof
- 4.11 Not a Contract of Employment or Service. Nothing in the Plan, the Grant Notice or this Agreement confers upon Participant any right to continue in the employ or service of the Company or any Subsidiary or interferes with or restricts in any way the rights of the Company and its Subsidiaries, which rights are hereby expressly reserved, to discharge or terminate the services of Participant at any time for any reason whatsoever, with or without Cause, except to the extent expressly provided otherwise in a written agreement between the Company or a Subsidiary and Participant.
- 4.12 Counterparts. The Grant Notice may be executed in one or more counterparts, including by way of any electronic signature, subject to Applicable Law, each of which will be deemed an original and all of which together will constitute one instrument.
  - 4.13 Incentive Stock Options. If the Option is designated as an Incentive Stock Option:
- (a) Participant acknowledges that to the extent the aggregate fair market value of shares (determined as of the time the option with respect to the shares is granted) with respect to which stock options intended to qualify as "incentive stock options" under Section 422 of the Code, including the Option, are exercisable for the first time by Participant during any calendar year exceeds \$100,000 or if for any other reason such stock options do not qualify or cease to qualify for treatment as "incentive stock options" under Section 422 of the Code, such stock options (including the Option) will be treated as non-qualified stock options. Participant further acknowledges that the rule set forth in the preceding sentence will be applied by taking the Option and other stock options into account in the order in which they were granted, as determined under Section 422(d) of the Code. Participant also acknowledges that if the Option is exercised more than three months after Participant's Termination of Service, other than by reason of death or disability, the Option will be taxed as a Non-Qualified Stock Option.

(b) Participant will give prompt written notice to the Company of any disposition or other transfer of any Shares acquired under this Agreement if such disposition or other transfer is made (i) within two years from the Grant Date or (ii) within one year after the transfer of such Shares to Participant. Such notice will specify the date of such disposition or other transfer and the amount realized, in cash, other property, assumption of indebtedness or other consideration, by Participant in such disposition or other transfer.

4.14 Governing Law. The Grant Notice and this Agreement will be governed by and interpreted in accordance with the laws of the State of Delaware, disregarding any state's choice-of-law principles requiring the application of a jurisdiction's laws other than the State of Delaware.

\*\*\*\*

## **EXHIBIT B**

### **PSU AWARD AGREEMENT**

## The BeautyHealth Company

### 2021 INCENTIVE AWARD PLAN

### PERFORMANCE-BASED RESTRICTED STOCK UNIT GRANT NOTICE

The BeautyHealth Company, a Delaware corporation (the "Company"), has granted to the participant listed below ("Participant") the Restricted Stock Units (the "RSUs") described in this Performance-Based Restricted Stock Unit Grant Notice (this "Grant Notice"), subject to the terms and conditions of [ ] 2021 Incentive Award Plan (as amended from time to time, the "Plan") and the Restricted Stock Unit Agreement attached hereto as Exhibit A (the "Agreement"), both of which are incorporated into this Grant Notice by reference. Capitalized terms not specifically defined in this Grant Notice or the Agreement have the meanings given to them in the Plan.

orant rotate or the regreement have the meanings green to them in the realist					
Participant:	[]				
Grant Date:	[], 2021				
Number of RSUs at Maximum:	[]1				
<b>Vesting Commencement Date:</b>	[], 2021				
Vesting Schedule:					
(a) <u>General</u> . Subject to clause (b) below, and further subject to and conditioned upon Participant's continued service as a Service Provider through the last day of the Performance Period, a number of RSUs shall vest on the last day of the Performance Period equal to (i) the total number of RSUs granted hereby multiplied by (ii) the applicable vesting percentage (" <i>Vesting Percentage</i> ") set forth below, which shall be determined based on greater of (x) the Company's Average Stock Price during the Year 3 Measurement Period and (y) the Company's Average Stock Price during the Year 4 Measurement Period (each forgoing capitalized term as defined below), as follows:					
Deles Threshold	Average Stock Price During the Applicable Measurement Period:	Vesting Percentage (% of Maximum):			
Below Threshold	Less than \$25.00	0%			

\$25.00

\$30.00

\$37.50 or greater

66.67%

80%

100%

**Note to Draft**: Insert number of RSUs at maximum.

Threshold/Target

Stretch

Maximum

In the event that the Company's Average Stock Price falls between the Threshold and Target values or Target and Maximum values specified in the table above, the Vesting Percentage shall be interpolated on a linear basis (for clarity, if Average Stock Price falls below the Threshold value, the Vesting Percentage shall equal 0%).

Notwithstanding the foregoing, in the event that a Change in Control is consummated during the Performance Period and Participant remains in continued service as a Service Provider until at least immediately prior to such Change in Control:

- (i) In the event that (A) the Shares do not continue to be publicly traded following the consummation of such Change in Control and (B) no Assumption of the RSUs (as defined in Section 8.3 of the Plan) occurs in connection with such Change in Control, then, immediately prior to the Change in Control, a number of RSUs will vest based solely on the per-Share consideration paid or payable (as applicable) in connection with such Change in Control (as determined by the Administrator) or, if the Change in Control is consummated after the third anniversary of the Vesting Commencement Date, the Company's Average Stock Price during the Year 3 Measurement Period (if greater); and
- (ii) In the event that (A) the Shares do not continue to be publicly traded following the consummation of such Change in Control and (B) an Assumption of the RSUs (as defined in Section 8.3 of the Plan) occurs in connection with such Change in Control, then, effective immediately prior to the closing of the Change in Control, the RSUs will be deemed to convert into a number of unvested RSUs determined based solely on the per-Share consideration paid or payable (as applicable) in connection with such Change in Control (as determined by the Administrator) or, if the Change in Control is consummated after the third anniversary of the Vesting Commencement Date, the Company's Average Stock Price during the Year 3 Measurement Period (if greater). Such unvested RSUs (as so assumed and adjusted in connection with the Change in Control) will be eligible to vest in full on the last day of the Performance Period in accordance with this clause (a) (based solely on the Participant's continued status as a Service Provider through such date) or upon Participant's Termination of Service as provided in clause (b) below.

## (b) <u>Termination of Service; Change in Control</u>. Notwithstanding clause (a) above:

(i) If Participant incurs a Termination of Service prior to the last day of the Performance Period, then the RSUs shall vest under clause (a) above or be forfeited (as applicable) in accordance with the following table. Any vesting of the RSUs pursuant to the following table shall (A) be subject to Participant (or Participant's estate, as applicable) timely executing and not revoking a release of claims in a form prescribed by the Company (a "*Release*") that becomes effective and irrevocable no later than sixty (60) days following such Termination of Service (the date such Release becomes effective and irrevocable, the "*Release Effective Date*"), and (B) be effective as of the Release Effective Date:

Reason for Termination of Service	If the Termination of Service Occurs Before the Third Anniversary of the Vesting Commencement Date, then:	If the Termination of Service Occurs On or After the Third Anniversary of the Vesting Commencement Date but Before the Fourth Anniversary of the Vesting Commencement Date, then:
Death or Disability (as defined below)	A number of RSUs will vest based on the	A number of RSUs will vest based on the
	Company's Average Stock Price over the	greater of (i) the Company's Average Stock
	Termination Measurement Period (as defined	Price over the Termination Measurement
	below).	Period and (ii) the Company's Average Stock Price over the Year 3 Measurement Period.
Without Cause [or for Good Reason (as defined below)] Prior to the Consummation of a Change in Control	All RSUs will be forfeited upon such Termination of Service without payment.	A number of RSUs will vest based on the Company's Average Stock Price over the Year 3 Measurement Period.
Without Cause [or for Good Reason] Within 24 Months After the Consummation of a Change in Control	A number of RSUs will vest based on the Company's Average Stock Price over the Termination Measurement Period.	A number of RSUs will vest based on the greater of (i) the Company's Average Stock Price over the Termination Measurement Period and (ii) the Company's Average Stock Price over the Year 3 Measurement Period.
Any Other Reason (Including for Cause or without Good Reason)	All RSUs will be forfeited upon such Termination of Service without payment.	All RSUs will be forfeited upon such Termination of Service without payment.

(ii) With respect to sub-clause (i) above, (A) the RSUs shall remain outstanding and eligible to vest following Participant's Termination of Service through the Release Effective Date and shall automatically be forfeited on the sixtieth (60<sup>th</sup>) day following such termination if the Release does not become effective and irrevocable on or prior to such date, and (B) any RSUs that do not become vested on the Release Effective Date pursuant to the applicable sub-clause shall be immediately forfeited on such date.

- (c) <u>Termination; Forfeiture</u>. Unless earlier terminated as set forth in this Grant Notice or the Agreement, any RSUs that have not become vested on or prior to the last day of the Performance Period will thereupon be automatically forfeited by Participant without payment of any consideration therefor. Except as set forth in clause (b) above, if Participant experiences a Termination of Service for any reason prior to the last day of the Performance Period, all then-unvested RSUs will thereupon be automatically forfeited by Participant without payment of any consideration therefor.
  - (d) <u>Definitions</u>. For purposes hereof, the following terms shall have the respective meanings set forth below:
  - (i) "Average Stock Price" shall mean, with respect to any Measurement Period, the average Fair Market Value of a Share over such Measurement Period.
  - (ii) "Cause" shall have the meaning set forth in [the employment agreement between Participant and [the Company], dated [\_\_\_]] / [the Company's Executive Severance Plan].<sup>2</sup>
  - (iii) "Good Reason" shall have the meaning set forth in [the employment agreement between Participant and [the Company], dated [\_\_\_\_] / [the Company's Executive Severance Plan].3
  - (iv) "Year 3 Measurement Period" means the ninety (90)-day period ending on the third (3rd) anniversary of the Vesting Commencement Date.
  - (v) "Year 4 Measurement Period" means the ninety (90)-day period ending on the fourth (4th) anniversary of the Vesting Commencement Date.
  - (vi) "*Measurement Period*" means each of the Termination Measurement Period, the Year 3 Measurement Period and the Year 4 Measurement Period.
  - (vii) "*Performance Period*" means the period commencing on the Vesting Commencement Date and ending on the fourth (4th) anniversary of the Vesting Commencement Date.
  - (viii) "Termination Measurement Period" means the ninety (90)-day period ending on (and including) the date of Participant's Termination of Service.

Note to Draft: To be updated based on whether Participant is party to an employment agreement or a participant in the Executive Severance Plan (if neither, the "Cause" definition from the Executive Severance Plan will be added here).

Note to Draft: To be updated based on whether Participant is party to an employment agreement or a participant in the Executive Severance Plan (if neither, "Good Reason" will be removed).

By accepting (whether in writing, electronically or otherwise) the RSUs, Participant agrees to be bound by the terms of this Grant Notice, the Plan and the Agreement. Participant has reviewed the Plan, this Grant Notice and the Agreement in their entirety, has had an opportunity to obtain the advice of counsel prior to executing this Grant Notice and fully understands all provisions of the Plan, this Grant Notice and the Agreement. Participant hereby agrees to accept as binding, conclusive and final all decisions or interpretations of the Administrator upon any questions arising under the Plan, this Grant Notice or the Agreement.

[0]	PARTICIPANT
By:	
Name:	[Participant Name]
Title:	

#### RESTRICTED STOCK UNIT AGREEMENT

Capitalized terms not specifically defined in this Agreement have the meanings specified in the Grant Notice or, if not defined in the Grant Notice, in the Plan.

### ARTICLE I. GENERAL

- 1.1 Award of RSUs(a). The Company has granted the RSUs to Participant effective as of the Grant Date set forth in the Grant Notice (the "*Grant Date*"). Each RSU represents the right to receive one Share as set forth in this Agreement. Participant will have no right to the distribution of any Shares until the time (if ever) the RSUs have vested.
- 1.2 Incorporation of Terms of Plan. The RSUs are subject to the terms and conditions set forth in this Agreement and the Plan, which is incorporated herein by reference. In the event of any inconsistency between the Plan and this Agreement, the terms of the Plan will control, unless it is expressly specified in in this Agreement or the Grant Notice that the specific provision of the Plan will not apply. For clarity, the foregoing sentence shall not limit the applicability of any additive language contained in this Agreement which provides supplemental or additional terms not inconsistent with the Plan.
- 1.3 Unsecured Promise. The RSUs will at all times prior to settlement represent an unsecured Company obligation payable only from the Company's general assets.

# ARTICLE II. VESTING; FORFEITURE AND SETTLEMENT

2.1 Vesting; Forfeiture. The RSUs will vest according to the vesting schedule in the Grant Notice except that any fraction of an RSU that would otherwise be vested will be accumulated and will vest only when a whole RSU has accumulated. Except as otherwise set forth in the Grant Notice, the Plan or this Agreement, and unless the Administrator otherwise determines, in the event of Participant's Termination of Service for any reason, all unvested RSUs will immediately and automatically be cancelled and forfeited (after taking into consideration any accelerated vesting which may occur in connection with such Termination of Service, if any).

### 2.2 Settlement.

- (a) RSUs that vest will be paid in Shares as soon as administratively practicable after the vesting of the applicable RSU, but in no event later than sixty (60) days following the date on which the applicable RSU vests (or, in the case of any accelerated vesting that occurs on the Release Effective Date pursuant to the Grant Notice, no later than sixty (60) days following the date on which the applicable Termination of Service occurs).
- (b) Notwithstanding the foregoing, the Company may delay any payment under this Agreement that the Company reasonably determines would violate Applicable Law or an applicable provision of the Plan until the earliest date the Company reasonably determines the making of the payment will not cause such a violation (in accordance with Treasury Regulation Section 1.409A-2(b)(7)(ii)); *provided* the Company reasonably believes the delay will not result in the imposition of excise taxes under Section 409A.

# ARTICLE III. TAXATION AND TAX WITHHOLDING

3.1 Representation. Participant represents to the Company that Participant has reviewed with Participant's own tax advisors the tax consequences of this Award and the transactions contemplated by the Grant Notice and this Agreement. Participant is relying solely on such advisors and not on any statements or representations of the Company or any of its agents.

#### 3.2 Tax Withholding.

- (a) Unless the Administrator otherwise determines, the Company shall withhold, or cause to be withheld, Shares otherwise vesting or issuable under this Award (including the RSUs) in satisfaction of any applicable withholding tax obligations, in accordance with the Plan. The number of Shares which may be so withheld or surrendered shall be limited to the number of Shares which have a Fair Market Value on the date of withholding no greater than the aggregate amount of such liabilities based on the maximum individual statutory withholding rates in Participant's applicable jurisdictions for federal, state, local and foreign income tax and payroll tax purposes that are applicable to such taxable income, in accordance with Section 9.5 of the Plan.
- (b) Participant acknowledges that Participant is ultimately liable and responsible for all taxes owed in connection with the RSUs, regardless of any action the Company or any Subsidiary takes with respect to any tax withholding obligations that arise in connection with the RSUs. Neither the Company nor any Subsidiary makes any representation or undertaking regarding the treatment of any tax withholding in connection with the awarding, vesting or payment of the RSUs or the subsequent sale of Shares. The Company and its Subsidiaries do not commit and are under no obligation to structure the RSUs to reduce or eliminate Participant's tax liability.

# ARTICLE IV. OTHER PROVISIONS

- 4.1 Adjustments. Participant acknowledges that the RSUs and the Shares subject to the RSUs are subject to adjustment, modification and termination in certain events as provided in this Agreement and the Plan.
- 4.2 Clawback. The RSUs and the Shares issuable hereunder shall be subject to clawback or recoupment in accordance with this Section 4.2. In the event that the Administrator, in its good faith discretion, determines that Participant has committed an act that constitutes Cause and such act has resulted in or would reasonably be expected to result in material harm to the Company and/or its affiliates, the Board may seek recoupment of up to the full amount of the RSUs and Shares issued upon settlement thereof and/or any proceeds received upon the sale of any such Shares. Nothing in this Section 4.2 shall limit the application of any clawback or recoupment policy in effect on the Grant Date or as may be adopted or maintained by the Company following the Grant Date, including the Dodd-Frank Wall Street Reform and Consumer Protection Act and any rules or regulations promulgated thereunder.

- 4.3 Notices. Any notice to be given under the terms of this Agreement to the Company must be in writing and addressed to the Company in care of the Company's General Counsel at the Company's principal office or the General Counsel's then-current email address or facsimile number. Any notice to be given under the terms of this Agreement to Participant must be in writing and addressed to Participant (or, if Participant is then deceased, to the Designated Beneficiary) at Participant's last known mailing address, email address or facsimile number in the Company's personnel files. By a notice given pursuant to this Section, either party may designate a different address for notices to be given to that party. Any notice will be deemed duly given when actually received, when sent by email, when sent by certified mail (return receipt requested) and deposited with postage prepaid in a post office or branch post office regularly maintained by the United States Postal Service, when delivered by a nationally recognized express shipping company or upon receipt of a facsimile transmission confirmation.
  - 4.4 Titles. Titles are provided herein for convenience only and are not to serve as a basis for interpretation or construction of this Agreement.
- 4.5 Conformity to Securities Laws. Participant acknowledges that the Plan, the Grant Notice and this Agreement are intended to conform to the extent necessary with all Applicable Laws and, to the extent Applicable Laws permit, will be deemed amended as necessary to conform to Applicable Laws.
- 4.6 Successors and Assigns. The Company may assign any of its rights under this Agreement to single or multiple assignees, and this Agreement will inure to the benefit of the successors and assigns of the Company. Subject to the restrictions on transfer set forth in this Agreement or the Plan, this Agreement will be binding upon and inure to the benefit of the heirs, legatees, legal representatives, successors and assigns of the parties hereto.
- 4.7 Limitations Applicable to Section 16 Persons. Notwithstanding any other provision of the Plan or this Agreement, if Participant is subject to Section 16 of the Exchange Act, the Plan, the Grant Notice, this Agreement and the RSUs will be subject to any additional limitations set forth in any applicable exemptive rule under Section 16 of the Exchange Act (including any amendment to Rule 16b-3) that are requirements for the application of such exemptive rule. To the extent Applicable Laws permit, this Agreement will be deemed amended as necessary to conform to such applicable exemptive rule.
- 4.8 Entire Agreement. The Plan, the Grant Notice and this Agreement (including any exhibit hereto) constitute the entire agreement of the parties and supersede in their entirety all prior undertakings and agreements of the Company and Participant with respect to the subject matter hereof.
- 4.9 Severability. If any portion of the Grant Notice or this Agreement or any action taken under the Grant Notice or this Agreement, in any case is held illegal or invalid for any reason, the illegality or invalidity will not affect the remaining parts of the Grant Notice and/or this Agreement (as applicable), and the Grant Notice and/or this Agreement (as applicable) will be construed and enforced as if the illegal or invalid provisions had been excluded, and the illegal or invalid action will be null and void.
- 4.10 Limitation on Participant's Rights. Participation in the Plan confers no rights or interests other than as herein provided. This Agreement creates only a contractual obligation on the part of the Company as to amounts payable and may not be construed as creating a trust. Neither the Plan nor any underlying program, in and of itself, has any assets. Participant will have only the rights of a general unsecured creditor of the Company with respect to amounts credited and benefits payable, if any, with respect to the RSUs, and rights no greater than the right to receive cash or the Shares as a general unsecured creditor with respect to the RSUs, as and when settled pursuant to the terms of this Agreement.

- 4.11 Not a Contract of Employment or Service. Nothing in the Plan, the Grant Notice or this Agreement confers upon Participant any right to continue in the employ or service of the Company or any Subsidiary or interferes with or restricts in any way the rights of the Company and its Subsidiaries, which rights are hereby expressly reserved, to discharge or terminate the services of Participant at any time for any reason whatsoever, with or without cause, except to the extent expressly provided otherwise in a written agreement between the Company or a Subsidiary and Participant.
- 4.12 Counterparts. The Grant Notice may be executed in one or more counterparts, including by way of any electronic signature, subject to Applicable Law, each of which will be deemed an original and all of which together will constitute one instrument.
- 4.13 Governing Law. The Grant Notice and this Agreement will be governed by and interpreted in accordance with the laws of the State of Delaware, disregarding any state's choice-of-law principles requiring the application of a jurisdiction's laws other than the State of Delaware.

\* \* \* \* \*

#### **EXHIBIT C**

## **CONFIDENTIALITY AGREEMENT**

#### THE HYDRAFACIAL COMPANY

### EMPLOYEE PROPRIETARY INFORMATION AND INVENTIONS ASSIGNMENT AGREEMENT

In consideration and as a condition of my employment by Edge Systems LLC d/b/a The HydraFacial Company (together with its parents and subsidiaries and any of their respective successors or assigns, the "Company"), and my receipt of the compensation paid to me by the Company pursuant to the employment agreement entered into between me and the Company (the "Employment Agreement") concurrently with the execution of this Employee Proprietary Information and Inventions Assignment Agreement (the "Agreement"), and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, effective as of the Effective Date (as defined in the Employment Agreement), I, the undersigned, agree as follows:

1. <u>Proprietary Information</u>. During the term of my employment with the Company, I may receive and otherwise be exposed, directly or indirectly, to confidential and proprietary information of the Company whether in graphic, written, electronic or oral form, including without limitation, information relating to the Company's business, strategies, designs, products, services and technologies and any derivatives, improvements and enhancements relating to any of the foregoing, or to the Company's suppliers, customers or business partners (collectively "Proprietary Information"). Proprietary Information may be identified at the time of disclosure as confidential or proprietary or information which by its context would reasonably be deemed to be confidential or proprietary. "Proprietary Information" may also include without limitation (i)(a) unpublished patent disclosures and patent applications and other filings, know-how, trade secrets, works of authorship and other intellectual property, as well as any information regarding ideas, Work Product (as defined below), technology, and processes, including without limitation assays, sketches, schematics, techniques, drawings, designs, descriptions, specifications and technical documentation, (b) specifications, protocols, models, designs, equipment, engineering, algorithms, software programs, software source documents, formulae, (c) information concerning or resulting from any research and development or other project, including without limitation, experimental work, and product development plans, regulatory compliance information, and research, development and regulatory strategies, and (d) business and financial information, including without limitation purchasing, procurement, manufacturing, customer lists, information relating to investors, employees, business and contractual relationships, business forecasts, sales and merchandising, business and marketing plans, product plans, and business strategies, including without limitation, information the Company provides regarding third parties, such as, but not limited to, suppliers, customers, employees, investors, or vendors; and (ii) any other information, to the extent such information contains, reflects or is based upon any of the foregoing Proprietary Information. The Proprietary Information may also include information of a third party that is disclosed to me by the Company or such third party at the Company's direction. Any information disclosed by any of the Company's affiliated companies or by any company, person or other entity participating with the Company in any consortium, partnership, joint venture or similar business combination, which would otherwise constitute Proprietary Information if disclosed by the Company, shall be deemed to constitute Proprietary Information under this Agreement, and the rights of the Company under this Agreement may be enforced by any such affiliate or participating entity (as well as by the Company) with respect to any violation relating to the Proprietary Information disclosed by such affiliate or entity, as if that affiliate or entity were also a party to this Agreement.

- 2. Obligations of Non-Use and Nondisclosure. I acknowledge the confidential and secret character of the Proprietary Information, and agree that the Proprietary Information is the sole, exclusive and valuable property of the Company. Accordingly, I agree not to use the Proprietary Information except in the performance of my authorized duties as an employee of the Company, and not to disclose all or any part of the Proprietary Information in any form to any third party, either during or after the term of my employment with the Company, without the prior written consent of the Company on a case-by-case basis, and to cooperate with the Company and use my best efforts not to prevent the unauthorized us or disclosure or reproductions of any Proprietary Information. In addition, I agree not to copy or remove any tangible materials containing Proprietary Information from the premises of the Company, except in the proper performance of my duties as an employee of the Company or with the prior written consent of the Company on a case-by-case basis. Upon termination of my employment with the Company, I agree to cease using and to return to the Company all whole and partial copies and derivatives of the Proprietary Information, whether in my possession or under my direct or indirect control, provided that I am entitled to retain my personal copies of (a) my compensation and benefits records, and (b) this Agreement. I understand that my obligations of nondisclosure with respect to Proprietary Information shall not apply to information that I can establish by competent proof (i) was actually in the public domain at the time of disclosure or enters the public domain following disclosure other than as a result of a breach of this Agreement, (ii) is already in my possession without breach of any obligations of confidentiality at the time of disclosure by the Company as shown by my files and records immediately prior to the time of disclosure, or (iii) is obtained by me from a third party not under confidentiality obligations and without a breach of any obligations of confidentiality. If I become compelled by law, regulation (including without limitation the rules of any applicable securities exchange), court order, or other governmental authority to disclose any Proprietary Information, I shall, to the extent possible and permissible under applicable law, first give notice to the Company. I agree to cooperate reasonably with the Company (at the Company's request) in any proceeding to obtain a protective order or other remedy. If such protective order or other remedy is not obtained, I shall only disclose that portion of such Proprietary Information required to be disclosed, in the opinion of my legal counsel. I shall request that confidential treatment be accorded such Proprietary Information, where available. Compulsory disclosures made pursuant to this section shall not relieve me of my obligations of confidentiality and non-use with respect to non-compulsory disclosures. If I learn of any possible unauthorized use or disclosure of Proprietary Information, I shall cooperate fully with the Company to enforce its rights in such information. Notwithstanding the foregoing or anything herein to the contrary, nothing contained herein shall prohibit me from (x) filing a charge with, reporting possible violations of federal law or regulation to, participating in any investigation by, or cooperating with any governmental agency or entity or making other disclosures that are protected under the whistleblower provisions of applicable law or regulation; (y) communicating directly with, cooperating with, or providing information (including trade secrets) in confidence to, any federal, state or local government regulator (including, but not limited to, the U.S. Securities and Exchange Commission, the U.S. Commodity Futures Trading Commission, or the U.S. Department of Justice) for the purpose of reporting or investigating a suspected violation of law, or from providing such information to my attorney or in a sealed complaint or other document filed in a lawsuit or other governmental proceeding; and/or (z) making disclosures that are protected by the National Labor Relations Act or similar applicable law.
- 3. <u>Defend Trade Secrets Act Notice of Immunity Rights</u>. I acknowledge that the Company has provided me with the following notice of immunity rights in compliance with the requirements of the Defend Trade Secrets Act: (a) I shall not be held criminally or civilly liable under any Federal or State trade

secret law for the disclosure of Proprietary Information that is made in confidence to a Federal, State, or local government official or to an attorney solely for the purpose of reporting or investigating a suspected violation of law, (b) I shall not be held criminally or civilly liable under any Federal or State trade secret law for the disclosure of Proprietary Information that is made in a complaint or other document filed in a lawsuit or other proceeding, if such filing is made under seal and (c) if I file a lawsuit for retaliation by the Company for reporting a suspected violation of law, I may disclose the Proprietary Information to my attorney and use the Proprietary Information in the court proceeding, if I file any document containing the Proprietary Information under seal, and do not disclose the Proprietary Information, except pursuant to court order.

4. <u>Property of the Company</u>. I acknowledge and agree that all notes, memoranda, reports, drawings, blueprints, manuals, materials, data, emails and other papers and records of every kind, or other tangible or intangible materials which shall come into my possession in the course of my employment with the Company, relating to any Proprietary Information, shall be the sole and exclusive property of the Company and I hereby assign any rights or interests I may obtain in any of the foregoing. I agree to surrender this property to the Company immediately upon termination of my employment with the Company, or at any time upon request by the Company. I further agree that any property situated on the Company's data systems or on the Company's premises and owned by the Company, including without limitation electronic storage media, filing cabinets or other work areas, is subject to inspection by Company personnel at any time with or without notice. I further agree that in the event of termination of my employment with the Company I will execute a Termination Certificate substantially in the form attached hereto as <u>Exhibit A</u>.

#### 5. Inventions.

- (a) <u>Disclosure and Assignment of Inventions</u>. For purposes of this Agreement, an "<u>Invention</u>" shall mean any idea, invention or work of authorship, including, without limitation, any documentation, formula, design, device, code, method, software, technique, process, discovery, concept, improvement, enhancement, development, machine or contribution, in each case whether or not patentable or copyrightable. I will disclose all Inventions promptly in writing to an officer of the Company or to attorneys of the Company in accordance with the Company's policies and procedures, I will, and hereby do, assign to the Company, without requirement of further writing, without royalty or any other further consideration, my entire right, title and interest throughout the world in and to all Inventions created, conceived, made, developed, and/or reduced to practice by me in the course of my employment by the Company and all intellectual property rights therein. I hereby waive, and agree to waive, any moral rights I may have in any copyrightable work I create or have created on behalf of the Company. I also hereby agree, that for a period of one year after my employment with the Company, I shall disclose to the Company any Inventions that I create, conceive, make, develop, reduce to practice or work on that relate to the work I performed for the Company. The Company agrees that it will use commercially reasonable measures to keep Inventions disclosed to it pursuant to this Section 5.1 that do not constitute Inventions are assigned or assignable to the Company pursuant to this Section 5.1 or otherwise.
- (b) <u>Certain Exemptions</u>. The obligations to assign Inventions set forth in Section 5.1 apply with respect to all Inventions (a) whether or not such Inventions are conceived, made, developed or worked on by me during my regular hours of employment with the Company; (b) whether or not the Invention

was made at the suggestion of the Company; (c) whether or not the Invention was reduced to drawings, written description, documentation, models or other tangible form; and (d) whether or not the Invention is related to the general line of business engaged in by the Company, but do not apply to Inventions that (x) I develop entirely on my own time or after the date of this Agreement without using the Company's equipment, supplies, facilities or Proprietary Information; (y) do not relate to the Company's business, or actual or demonstrably anticipated research or development of the Company at the time of conception or reduction to practice of the Invention; and (z) do not result from and are not related to any work performed by me for the Company. I hereby acknowledge and agree that the Company has notified me that, if I reside in the state of California, assignments provided for in Section 5.1 do not apply to any Invention which qualifies fully for exemption from assignment under the provisions of Section 2870 of the California Labor Code ("Section 2870"), a copy of which is attached as Exhibit B. If applicable, at the time of disclosure of an Invention that I believe qualifies under Section 2870, I shall provide to the Company, in writing, evidence to substantiate the belief that such Invention qualifies under Section 2870. I further understand that, to the extent this Agreement shall be construed in accordance with the laws of any state which precludes a requirement in an employee agreement to assign certain classes of inventions made by an employee, Section 5.1 shall be interpreted not to apply to any Invention which a court rules and/or the Company agrees falls within such classes.

- (c) Records. I will make and maintain adequate and current written records of all Inventions covered by Section 5.1. These records may be in the form of notes, sketches, drawings, flow charts, electronic data or recordings, notebooks and any other format. These records shall be and remain the property of the Company at all times and shall be made available to the Company at all times.
- (d) Patents and Other Rights. I agree to assist the Company in obtaining, maintaining and enforcing patents, invention assignments and copyright assignments, and other proprietary rights in connection with any Invention covered by Section 5.3, and will otherwise assist the Company as reasonably required by the Company to perfect in the Company the rights, title and other interests in my work product granted to the Company under this Agreement (both in the United States and foreign countries). I further agree that my obligations under this Section 5.4 shall continue beyond the termination of my employment with the Company, but if I am requested by the Company to render such assistance after the termination of such employment, I shall be entitled to a fair and reasonable rate of compensation for such assistance, and to reimbursement of any expenses incurred at the request of the Company relating to such assistance. If the Company is unable for any reason, after reasonable effort, to secure my signature on any document needed in connection with the actions specified above, I hereby irrevocably designate and appoint the Company and its duly authorized officers and agents as my agent and attorney in fact, which appointment is coupled with an interest, to act for and in my behalf to execute, verify and file any such documents and to do all other lawfully permitted acts to further the purposes of this Section 5.4 with the same legal force and effect as if executed by me.
- (e) <u>Prior Contracts Inventions; Information Belonging to Third Parties.</u> I represent and warrant that, except as set forth on <u>Exhibit C</u>, I am not required, and I have not been required during the course of work for the Company or its predecessors, to assign Inventions under any other contracts that are now or were previously in existence between me and any other person or entity. I further represent that (i) I am not obligated under any consulting, employment or other agreement that would affect the Company's rights or my duties under this Agreement, and I shall not enter into any such agreement or obligation during the period of my employment by the Company, (ii) there

is no action, investigation, or proceeding pending or threatened, or any basis therefor known to me involving my prior employment or any consultancy or the use of any information or techniques alleged to be proprietary to any former employer, and (iii) the performance of my duties as an employee of the Company do not and will not breach, or constitute a default under any agreement to which I am bound, including any agreement limiting the use or disclosure of proprietary information acquired in confidence prior to engagement by the Company or if applicable, any agreement to refrain from competing, directly or indirectly, with the business of such previous employer or any other party or to refrain from soliciting employees, customers or suppliers of such previous employer or other party. I will not, in connection with my employment by the Company, use or disclose to the Company any confidential, trade secret or other proprietary information of any previous employer or other person to which I am not lawfully entitled. As a matter of record, I attach as Exhibit C a brief description of all Inventions made or conceived by me prior to my employment with the Company which I desire to be excluded from this Agreement ("Background Technology."). If full disclosure of any Background Technology would breach or constitute a default under any agreement to which I am bound, including any agreement limiting the use or disclosure of proprietary information acquired in confidence prior to engagement by the Company, I understand that I am to describe such Background Technology in Exhibit C at the most specific level possible without violating any such prior agreement. Without limiting my obligations or representations under this Section 5.5, if I use any Background Technology in the course of my employment or incorporate any Background Technology in any product, service or other offering of the Company, I hereby grant the Company a non-exclusive, royalty-free, perpetual and irrevocable, worldwide right to use and sublicense the use of Background Technology for the purpose of developing, marketing, selling and supporting Company technology, products and services, either directly or through multiple tiers of distribution, but not for the purpose of marketing Background Technology separately from Company products or services.

- (f) Works Made for Hire. I acknowledge that all original works of authorship which are made by me (solely or jointly with others) within the scope of my employment with the Company and which are eligible for copyright protection are "works made for hire" as that term is defined in the United States Copyright Act (17 U.S.C., Section 101).
- 6. <u>Restrictive Covenants</u>. I agree to fully comply with the covenants set forth in this Section 6 (the "<u>Restrictive Covenants</u>"). I further acknowledge and agree that the Restrictive Covenants are reasonable and necessary to protect the Company's legitimate business interests, including its Proprietary Information and goodwill.
- (a) Non-Competition. During the term of my employment by the Company, I will not without the prior written approval of the Board of Directors of the Company, (a) engage in any other professional employment or consulting, or (b) directly or indirectly participate in or assist any business (whether as owner, partner, officer, director, employee, consultant, investor, lender or otherwise, except as the holder of not more than 1% of the outstanding stock of a publicly-held company), in each case, which is a current or potential supplier, customer or competitor of the Company, including but not limited to any business or enterprise that develops, manufactures, markets, licenses, sells or provides any product or service that competes with any product or service developed, manufactured, marketed, licensed, sold or provided, or planned to be developed, manufactured, marketed, licensed, sold or provided, by the Company while I was employed by the Company.
- (b) Non-Solicitation.

- (i) During the term of my employment with the Company and for a period of one (1) year thereafter I will not (i) solicit for employment or engagement, any employee, consultant or independent contractor of the Company, or (ii) solicit, encourage, induce or attempt to induce or assist others to solicit, encourage, induce or attempt to induce any employees, consultants or independent contractors of the Company who were employed or engaged by the Company at any time during the term of my employment with the Company to terminate their employment or engagement with the Company.
- (ii) During the term of my employment with the Company, I will not solicit, divert or take away, or attempt to divert or take away, the business of any customer or client of the Company (served by the Company during the twelve (12)-month period prior to the termination of my employment with the Company) on my own behalf or on behalf of any person or entity other than the Company.
- (c) No Defamatory Communications. During the term of my employment with the Company and thereafter, I agree that I will not make any public or private statement which would reasonably be expected to defame or disparage the Company or any of its employees, officers, managers or directors. Notwithstanding the foregoing, this Section 6.3 shall not preclude me from making any statement to the extent required by law or legal process.
- (d) <u>Extension</u>. Without limiting the Company's ability to seek other remedies available in law or equity, if I violate the provisions of Section 6.2(a), I shall continue to be bound by the restrictions set forth in such section until a period of one year has expired without any violation of such provisions.
- (e) <u>Interpretation</u>. If any restriction set forth in Section 6.2 is found by any court of competent jurisdiction to be unenforceable because it extends for too long a period of time or over too great a range of activities or in too broad a geographic area, it shall be interpreted to extend only over the maximum period of time, range of activities or geographic area as to which it may be enforceable.
- 7. <u>Notification to Other Parties</u>. In the event of termination of my employment with the Company, I hereby consent to notification by the Company to my new employer or other party for whom I work about my rights and obligations under this Agreement.
- 8. <u>Employment at Will</u>. I understand and agree that my employment with the Company is at will. Accordingly, my employment can be terminated for any lawful reason or for no reason, without cause or notice, at my option or the Company's option, subject to and except as otherwise expressly set forth in the Employment Agreement. The Restrictive Covenants will remain in effect for the periods specified in this Agreement, unless such Restrictive Covenants are modified by a further written agreement signed by both an authorized officer of the Company and me which expressly alters such Restrictive Covenants.

## 9. Miscellaneous.

(a) The parties' rights and obligations under this Agreement will bind and inure to the benefit of their respective successors, heirs, executors, and administrators and permitted assigns. I will not assign this Agreement or my obligations hereunder without the prior written consent of the Company, which consent may be withheld in the Company's sole discretion, and any such purported assignment without consent shall be null and void from the beginning. I agree that the Company may freely assign this Agreement to any affiliate or successor in interest, including any person or entity that, whether by way of merger, sale, acquisition, corporate re-organization or otherwise, directly or indirectly acquires all or substantially all of the business or assets of the Company.

- (b) This Agreement, together with the Employment Agreement, constitutes the parties' final, exclusive and complete understanding and agreement with respect to the subject matter hereof, and supersedes all prior and contemporaneous understandings and agreements, whether oral or written, relating to its subject matter.
- (c) Any subsequent change or changes in my duties, obligations, rights or compensation will not affect the validity or scope of this Agreement. This Agreement may not be waived, modified or amended unless mutually agreed upon in writing by both parties. No delay or omission by the Company in exercising any right under this Agreement will operate as a waiver of that or any other right. A waiver or consent given by the Company on any one occasion is effective only in that instance and will not be construed as a bar to or waiver of any right on any other occasion.
- (d) The provisions of this Agreement are severable, and the invalidity or unenforceability of any provision(s) of this Agreement shall not impact the validity or enforceability of any other provision(s) of this Agreement, which shall remain in full force and effect.
- (e) I acknowledge that the Company will suffer substantial damages not readily ascertainable or compensable in terms of money in the event of the breach of any of my obligations under this Agreement. I therefore agree that the Company shall be entitled (without limitation of any other rights or remedies otherwise available to the Company) to obtain an injunction from any court of competent jurisdiction prohibiting the continuance or recurrence of any breach of this Agreement. I also agree that if the Company prevails in any action or proceeding to enforce my obligations under this Agreement, I will pay all of the Company's expenses relating to any such action or proceeding including, without limitation, all reasonable attorney's fees, if so authorized by applicable state and/or federal law.
- (f) The rights and obligations of the parties under this Agreement shall be governed in all respects by the laws of the State of California exclusively, without reference to any conflict laws rule that would result in the application of the laws of any other jurisdiction. I agree that upon the Company's request, all disputes arising hereunder shall be adjudicated in the state and federal courts having jurisdiction over disputes arising in Los Angeles County, California, and I hereby agree to consent to the personal jurisdiction of such courts. The Company and I each hereby irrevocably waive any right to a trial by jury in any action, suit or other legal proceeding arising under or relating to any provision of this Agreement.
- (g) Any notices required or permitted hereunder shall be given to the appropriate party at the address specified on the signature page to this Agreement or at such other address as the party shall specify in writing. Such notice shall be deemed given upon personal delivery, or sent by certified or registered mail, postage prepaid, three (3) days after the date of mailing.
- (h) Except as otherwise provided herein, the provisions of this Agreement shall survive the termination of my employment with the Company for any reason.
- (i) This Agreement may be executed in any number of counterparts, each of which shall be deemed to be an original and all of which together shall be deemed to be one and the same instrument. A facsimile, PDF (or any electronic signature complying with the U.S. federal ESIGN Act of 2000, e.g., www.docusign.com) or any other type of copy of an executed version of this Agreement signed by a party is binding upon the signing party to the same extent as the original of the signed agreement.

I ACKNOWLEDGE THAT I HAVE HAD THE OPPORTUNITY TO SEEK THE ADVICE OF INDEPENDENT LEGAL COUNSEL PRIOR TO SIGNING THIS AGREEMENT, AND THAT I HAVE EITHER CONSULTED WITH OR ON MY OWN VOLITION CHOSEN NOT TO CONSULT WITH SUCH COUNSEL. I FURTHER ACKNOWLEDGE THAT I HAVE READ THIS AGREEMENT CAREFULLY AND I UNDERSTAND AND ACCEPT THE OBLIGATIONS WHICH IT IMPOSES UPON ME WITHOUT RESERVATION. NO PROMISES OR REPRESENTATIONS HAVE BEEN MADE TO ME TO INDUCE ME TO SIGN THIS AGREEMENT. I SIGN THIS AGREEMENT VOLUNTARILY AND FREELY, IN DUPLICATE, WITH THE UNDERSTANDING THAT THE COMPANY WILL RETAIN ONE COUNTERPART AND THE OTHER COUNTERPART WILL BE RETAINED BY ME.

(Signature Page Follows)

IN WITNESS WHEREOF, I have executed this document as of	, 20	
		Employee:
		Address:
AGREED AND ACKNOWLEDGED:		
		EDGE SYSTEMS LLC d/b/a THE HYDRAFACIAL COMPANY
		Ву:
		Name:
		Title:
		Address: The HydraFacial Company 2165 E Spring Street Long Beach, CA 90806
1	3	

#### Exhibit A

### **Termination Certificate**

I, the undersigned, hereby certify that I do not have in my possession, nor have I failed to return, any documents or materials relating to the business of The HydraFacial Company or its affiliates (together, the "Company"), or copies thereof, including, without limitation, any item of Proprietary Information listed in Section 4 of the Company's Employee Proprietary Information and Inventions Assignment Agreement (the "Agreement") to which I am a party, but not including copies of my own compensation and benefits records (in each case, to the extent expressly permitted by the Agreement).

(Printed or Typed Name of Employee)

### Exhibit B

## California Labor Code

California Labor Code § 2870. Application of provision providing that employee shall assign or offer to assign rights in invention to employer.

- (a) Any provision in an employment agreement which provides that an employee shall assign, or offer to assign, any of his or her rights in an invention to his or her employer shall not apply to an invention that the employee developed entirely on his or her own time without using the employer's equipment, supplies, facilities, or trade secret information except for those inventions that either:
  - (1) Relate at the time of conception or reduction to practice of the invention to the employer's business, or actual or demonstrably anticipated research or development of the employer; or
  - (2) Result from any work performed by the employee for the employer.
- (b) To the extent a provision in an employment agreement purports to require an employee to assign an invention otherwise excluded from being required to be assigned under subdivision (a), the provision is against the public policy of this state and is unenforceable.

# Exhibit C

# Background Technology

List here prior contracts to assign Inventions that are now in existence between any other person or entity and you.

List here previous Inventions which you desire to have specifically excluded from the operation of this Agreement. Continue on reverse side if necessary.

### **EXHIBIT D**

## **RELEASE**

### GENERAL RELEASE OF ALL CLAIMS

This General Release of all Claims (this "*Release*") is entered into on [•], 20[•] by and between Vesper Healthcare Acquisition Corp. ("*Parent*"), Edge Systems LLC d/b/a The HydraFacial Company (the "*Company*"), and [•] ("*Executive*"). In consideration of the payments and benefits set forth in the Employment Agreement (the "*Employment Agreement*") between Executive, Parent and the Company, effective May 4, 2021 (the "*Effective Date*"), to which Executive first became legally entitled following the Effective Date, Executive agrees as follows:

#### General Release and Waiver of Claims.

- (a) For valuable consideration, the receipt and adequacy of which are hereby acknowledged, Executive and each of Executive's respective heirs, executors, administrators, representatives, agents, successors, assigns and representatives (the "*Releasor*") hereby irrevocably and unconditionally releases and forever discharges each of Parent, the Company, and their respective partners, subsidiaries, associates, affiliates, successors, heirs, assigns, agents, directors, officers, employees, representatives, lawyers, insurers, and all persons acting by, through, under or in concert with them (collectively, the "*Releasees*"), of and from any and all manner of action or actions, cause or causes of action, judgments, obligations, in law or in equity, suits, debts, liens, contracts, agreements, promises, liability, claims, demands, damages, losses, costs, attorneys' fees or expenses, of any nature whatsoever, known or unknown, fixed or contingent, which Releasor now has, ever had or may hereafter have against any Releasee, by reason of any act, omission, practice, conduct, event, cause, or other matter whatsoever from the beginning of time up to and including the date that Executive executes this Release, to the fullest extent permitted by law
- (b) Without limiting the generality of the foregoing, Releasor releases and discharges Releasees from any and all claims in any way arising out of, based upon, or related to Executive's employment with Parent and/or the Company, the termination of employment of Executive by Parent and/or the Company and/or the events surrounding the circumstances relating to that termination, including, but not limited to:
  (i) any and all claims arising under tort, contract and quasi-contract law, including, but not limited to, claims of breach of contract (express or implied), tortious interference with contract or prospective business advantage, breach of the covenant of good faith and fair dealing, promissory estoppel, detrimental reliance, invasion of privacy, wrongful or retaliatory discharge, fraud, defamation, slander, libel, negligent or intentional infliction of emotional distress or compensatory or punitive damages; (ii) any and all claims for monetary or equitable relief, including, but not limited to, attorneys' fees, back pay, front pay, reinstatement, experts' fees, medical fees or expenses, costs, and disbursements; and (iii) and any claim under Title VII of the Civil Rights Act of 1964, the Age Discrimination In

Employment Act ("ADEA"), the Americans With Disabilities Act, the Family and Medical Leave Act, the Equal Pay Act, the False Claims Act, the Employee Retirement Income Security Act, the Federal Worker Retraining and Notification Act, the Fair Labor Standards Act, the Civil Rights Act of 1991, Section 1981 of U.S.C. Title 42, the Sarbanes-Oxley Act of 2002, the California Fair Employment and Housing Act, the California Unfair Competition Law, the California Equal Pay Law, the Moore-Brown-Roberti Family Rights Act of 1991, the California Labor Code, the California Worker Adjustment and Retraining Notification Act, California Wage and Hour laws, the California False Claims Act, the California Constitution and the California Corporate Criminal Liability Act, and any other federal, state or local law or ordinance prohibiting employment discrimination, harassment or retaliation. This Release does not release claims arising after the date Executive executes this Release, nor claims that cannot be released as a matter of law, including, but not limited to, Executive's right to file a charge with or participate in a charge by the Equal Employment Opportunity Commission ("EEOC"), the National Labor Relations Board ("NLRB"), or any other local, state, or federal administrative body or government agency that is authorized to enforce or administer laws related to employment, against Parent and/or the Company (with the understanding that any such filing or participation does not give Executive the right to recover any monetary damages against Parent and/or the Company; Executive's release of claims herein bars Executive from recovering such monetary relief from Parent and/or the Company before the EEOC, NLRB, or other administrative body). Notwithstanding the foregoing, this Release does not apply to (i) any lawsuit brought to challenge the validity of this Release under the ADEA, (ii) payments or benefits under Article IV of the Employment Agreement, which payments and benefits (among other good and valuable consideration) are provided in exchange for this Release, (iii) any claims for indemnification arising under any applicable indemnification obligation of Parent and/or the Company, (iv) accrued or vested benefits under any applicable Parent and/or Company employee benefit plan (within the meaning of Section 3(3) of the Employment Retirement Income Security Act) and (v) any rights as a shareholder of the Parent or pursuant to any option, restricted stock unit or other equity compensation award agreement or plan.

(c) Executive acknowledges that Executive has been advised by legal counsel and is familiar with the provisions of California Civil Code Section 1542, which provides as follows:

"A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS THAT THE CREDITOR OR RELEASING PARTY DOES NOT KNOW OR SUSPECT TO EXIST IN HIS OR HER FAVOR AT THE TIME OF EXECUTING THE RELEASE AND THAT, IF KNOWN BY HIM OR HER, WOULD HAVE MATERIALLY AFFECTED HIS OR HER SETTLEMENT WITH THE DEBTOR OR RELEASED PARTY."

Executive, being aware of, understanding and acknowledging the significance and consequence of specifically waiving California Civil Code Section 1542, hereby expressly waives and relinquishes all rights and benefits Executive may have thereunder, as well as any other applicable statutes or common law principles of similar effect, in order to effect a full and complete general release as described above. Thus, notwithstanding the provisions of California Civil Code Section 1542, and to implement a full and complete release, Executive expressly acknowledges this Release is intended to include in its effect, without limitation, all claims he does not know or suspect to exist in his favor at the time of signing this Release, and that this Release contemplates the extinguishment of any such claims.

- 2. <u>Consideration and Revocation Period</u>. By signing this Release, Executive represents and warrants that:
  - (a) Under the Federal Age Discrimination in Employment Act of 1967, as amended, and the applicable rules and regulations promulgated thereunder, Parent and the Company advise Executive that he should consult with independent counsel before executing this Release; and Executive acknowledges that he has been so advised. Executive further acknowledges that he has had at least [twenty-one (21)][forty-five (45)] days to consider this Release before signing it and Executive further acknowledges that if he signs this Release prior to the expiration of the [twenty-one (21)][forty-five (45)] day period, Executive waives the remainder of that period.
  - (b) Executive acknowledges that he has carefully read this Release in its entirety; that he has had an adequate opportunity to consider it; that he understands all its terms; and that he knowingly and voluntarily assents to all the terms and conditions contained herein, including, without limitation, the waiver and release contained herein.
  - (c) Executive further acknowledges that he has seven (7) calendar days following the date he signs this Release to revoke it and this Release shall not become effective until the eighth day following the date on which Executive signs this Release. Executive understands that if he wishes to revoke this Release, Executive must deliver written notice of revocation [(which may be by email)], stating Executive's intent to revoke this Release on or before 5:00 p.m. (PST) of the seventh (7th) day after the date on which Executive signs this Release to [TITLE], at [ADDRESS]. Executive acknowledges that if Executive revokes this Release, Executive will not receive any payments or benefits pursuant to Article IV of the Employment Agreement.
- 3. No Assignment. Executive represents and warrants that there has been no assignment or other transfer of any interest in any claim released hereunder which Executive may have against each Releasee and Executive agrees to indemnify and hold each Releasee harmless from any liability, claims, demands, damages, costs, expenses and attorneys' fees incurred by any Releasee as the result of any such assignment or transfer or any rights or claims under any such assignment or transfer. It is the intention of the parties that this indemnity does not require payment as a condition precedent to recovery by any Releasee against Executive under this indemnity

- 4. <u>Proceedings</u>. Executive agrees that if Executive hereafter commences any suit arising out of, based upon, or relating to any of the claims released hereunder or in any manner asserts against any Releasee any of the claims released hereunder, then Executive agrees to pay to such Releasee, in addition to any other damages caused to such Releasee thereby, all attorneys' fees incurred by such Releasee in defending or otherwise responding to said suit or claim. Notwithstanding the foregoing, the foregoing sentence shall not apply to the extent such attorneys' fees are attributable to Executive's good faith challenge to or a request for declaratory relief with respect to the validity of the waiver herein under the ADEA.
- 5. <u>Nonadmission</u>. Executive further understands and agrees that neither the payment of any sum of money nor the execution of this Release shall constitute or be construed as an admission of any liability whatsoever by any of the Releasees, all of whom have consistently taken the position that they have no liability whatsoever to Executive.
- 6. <u>Confidential Information</u>. Executive acknowledges and agrees that Executive is bound by that certain Confidentiality Agreement (as defined in the Employment Agreement). Executive hereby reaffirms the covenants, terms and conditions set forth in the Confidentiality Agreement, and acknowledges and agrees that the Confidentiality Agreement remains in full force and effect in accordance with its terms
- 7. <u>Severability.</u> In the event any provision or part of this Release is found to be invalid or unenforceable, only that particular provision or part so found, and not the entire Release, will be inoperative
- 8. <u>Governing Law</u>. This Release shall be governed by and construed in accordance with the laws of the State of California, without regard to conflicts of laws principles thereof.

EXECUTIVE ACKNOWLEDGES THAT HE HAS READ THIS RELEASE AGREEMENT AND THAT HE FULLY KNOWS, UNDERSTANDS AND APPRECIATES ITS CONTENTS, AND THAT HE HEREBY EXECUTES THE SAME AND MAKES THIS RELEASE AGREEMENT AND THE RELEASE AND AGREEMENTS PROVIDED FOR HEREIN VOLUNTARILY AND OF HIS OWN FREE WILL.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, the undersigned has executed this General Release of all Claims this day of	20	
		[Name]

### **EMPLOYMENT AGREEMENT**

THIS EMPLOYMENT AGREEMENT ("Agreement") is dated as of May 4, 2021 and effective as of the Effective Date (as defined below) between The Beauty Health Company ("Parent"), Edge Systems LLC d/b/a The HydraFacial Company (the "Company"), and Liyuan Woo ("Executive").

WHEREAS, on December 8, 2020, LCP Edge Intermediate, Inc., a Delaware corporation and indirect parent of the Company, Parent and certain other parties named therein entered into an Agreement and Plan of Merger (the "*Merger Agreement*"), pursuant to which, among other things the Company will become an indirect wholly owned subsidiary of Parent (collectively, the "*Transaction*");

WHEREAS, the Company and Executive are party to that certain employment agreement, dated as of September 21, 2020 (the "*Prior Agreement*");

WHEREAS, the Company desires to continue employ Executive pursuant to the terms of this Agreement, and Executive desires to enter into this Agreement and to accept such employment with the Company, in each case, subject to the terms and provisions of this Agreement; and

WHEREAS, the parties intend that this Agreement shall supersede the Prior Agreement in its entirety.

NOW THEREFORE, in consideration of the mutual covenants and agreements set forth herein, the parties hereto agree as follows:

## I. EMPLOYMENT

The Company hereby agrees to continue to employ Executive and Executive agrees to continue employment with the Company upon the terms and conditions herein set forth.

A. <u>Employment</u>. Executive shall serve as Chief Financial Officer of Parent during the Term (as defined below). Executive agrees to perform such duties as may be assigned to Executive from time to time by the Board of Directors of Parent (the "*Parent Board*"). Executive agrees to devote substantially all of Executive's business time and attention and best efforts to the affairs of Parent and the Company during the Term. Executive shall have such responsibilities, power and authority as are customarily associated with the positions of chief financial officer in public companies, subject to the delegation of authority guidelines established by the Parent Board from time to time, and shall report solely and directly to the Parent Board.

- B. <u>Term</u>. The term of employment of Executive hereunder (the "*Term*") will be for the period commencing on the date on which the Transaction closes (the "*Effective Date*") and ending on the earliest of:
  - 1. The date of termination of Executive's employment in accordance with Article IV of this Agreement;
  - 2. The date of Executive's voluntary retirement in accordance with the Company's plans and policies; or
  - 3. The date of Executive's death.
- C. <u>Principal Work Location</u>. During the Term, Executive shall perform the services required by this Agreement at the Company's offices in Long Beach, California (the "*Principal Location*"), except for travel to other locations as may be reasonably necessary to fulfill his duties and responsibilities hereunder.

# II. COMPENSATION

- A. <u>Base Salary</u>. During the Term, the Company shall pay to Executive a base salary (the "*Base Salary*") at the rate of \$415,000.00 per year during 2021, payable in substantially equal semi-monthly installments pursuant to the Company's standard payroll practices. The Base Salary will be reviewed no less frequently than annually and may be adjusted upward (but not downward) by the Parent Board (or a committee thereof) in its sole discretion.
- B. <u>Annual Incentive</u>: For each fiscal year of the Parent ending during the Term, Executive shall be eligible to earn a cash performance bonus (an "*Annual Bonus*") targeted at 60% of Base Salary (the "*Target Bonus*"). The actual amount of any Annual Bonus will be based upon achievement of specified levels of performance goals set by the Parent Board in consultation with Executive. Any Annual Bonus that becomes payable shall be paid at such times as annual bonuses are generally paid to senior executives of Parent, typically on or before, but in no event later than, March 15<sup>th</sup> of the calendar year following the year in which they are earned and, except as provided in Section 4, subject to Executive's continued employment through the applicable payment date. Executive's Target Bonus opportunity shall be subject to annual review by the Parent Board, and adjustments may be made at any time based upon the Parent Board's review of market trends, internal considerations and performance. The Target Bonus opportunity shall not be reduced at any time (including after any such increase).

# C. Long-Term Incentives.

1. As soon as practicable following the Effective Date, Parent will grant to Executive an option to purchase seven hundred forty-four thousand (744,000) shares of Parent common stock (the "*Option*") pursuant to Parent's 2021 Incentive Award Plan, as amended and/or restated from time (or any successor thereto) (the "*Incentive Plan*"). The Option shall have an exercise price per share equal to the Fair Market Value (as defined in the Plan) of a share of Parent common stock on the grant date. Twenty-five percent (25%) of the shares of Parent common stock subject to the Option will vest on each of the first four (4) anniversaries of the Effective Date, subject to Executive's continued employment with the Company through the applicable vesting date. The Option will be subject to the terms and conditions of the Plan and an award agreement substantially in the form attached hereto as Exhibit A (the "*Option Agreement*"), to be entered into between the Parent and Executive.

- 2. As soon as practicable following the Effective Date and the filing of the Form S-8 relating to the Incentive Plan, Parent will grant to Executive an award of performance share units covering 50,000 shares of Parent common stock ("*PSUs*") under the Incentive Plan. The PSUs will be subject to the terms and conditions (including vesting conditions set forth in the Incentive Plan and an award agreement substantially in the form attached hereto as Exhibit B, to be entered into between Parent and Executive.
- 3. For each fiscal year of Parent ending during the Term (commencing with 2022), Executive shall be eligible for one or more grants of long-term incentive awards ("*Awards*") having a grant-date value determined by the Compensation Committee of the Parent Board (the "*Compensation Committee*"), based upon the Compensation Committee's review of market trends, internal considerations and performance. The type of any such Award and the relevant terms and conditions (including vesting conditions) shall be determined by the Compensation Committee. Each Award will be subject to the terms and conditions of the Incentive Plan and an award agreement prescribed by Parent, to be entered into between Parent and Executive.
- D. <u>Reimbursement of Expenses</u>. During the Term, Executive shall be entitled to receive prompt reimbursement of all reasonable business expenses incurred by Executive in performing services hereunder, including all reasonable expenses of travel, and reasonable living expenses while away from home on business at the request of, or in the service of, the Company, *provided* that such expenses are incurred and accounted for in accordance with the policies and procedures established by the Company.
- E. <u>Benefits</u>. During the Term, Executive shall be eligible to participate in and be covered by all health, insurance, pension, disability insurance and other employee plans and benefits maintained by the Company for the benefit of its employees from time to time (collectively referred to herein as the "*Company Benefit Plans*"), on the same terms as are generally applicable to other senior executives of the Company, subject to meeting applicable eligibility requirements. Nothing contained in this Section 2.6 shall create or be deemed to create any obligation on the part of the Company to adopt or maintain any health, welfare, retirement or other benefit plan or program at any time or to create any limitation on the Company's ability to modify or terminate any such plan or program.
- F. <u>Vacation and Holidays</u>. During the Term, Executive shall participate in the Company's Permissive Paid Time Off program, the terms of which can be modified by the Company at any time at the discretion of the Company. Executive shall also be entitled to such holidays as are established by the Company for all employees.

# III. NON-COMPETITION, CONFIDENTIALITY AND NONDISCLOSURE

A. <u>Confidentiality Agreement.</u> Concurrently with the execution and delivery of this Agreement, and as part of the consideration for the promises and undertakings by Parent and the Company in this Agreement, Executive shall execute and deliver the Employee Proprietary Information and Inventions Assignment Agreement attached as Exhibit C hereto and incorporated herein by reference (the "*Confidentiality Agreement*").

- B. Other Activities. Subject to the terms of the employment hereunder, during the Term, Executive shall devote substantially all of Executive's business time and best efforts to the performance of Executive's duties and responsibilities for Parent and the Company and shall not serve on the board of directors of any for profit or not-for-profit entity without the prior consent of the Parent Board. Executive will not, without the prior written approval of the Parent Board, engage in any other business activity or investment opportunity which is or may be competitive with the business of Parent or its affiliates, or which, individually or in the aggregate, would materially interfere or conflict with the performance of Executive's duties, services, and responsibilities hereunder, or which is in violation of applicable employee policies established from time to time by the Parent or the Company.
- C. <u>No Violation of Other Agreements</u>. Executive represents that, to the best of Executive's knowledge, the entrance into this Agreement and the performance of all the terms of this Agreement and as an officer of Parent and employee of the Company does not and will not breach any contract to which Executive is bound or any legal obligation of Executive:
- 1. Not to compete or to interfere with the business of a former employer (which term for purposes of this Section 3.3 shall also include persons, firms, corporations and other entities for which Executive has acted as an independent contractor or consultant); or
  - 2. Not to solicit employees, customers or vendors of any former employer.

# IV. TERMINATION

- A. <u>Definitions</u>. For purposes of this Article IV, the following definitions shall apply to the terms set forth below:
- 1. <u>Cause</u>. "Cause" shall be defined as follows:
  - a. Executive has (A) engaged in an act of theft, embezzlement or fraud or, breach of confidentiality or fiduciary duty relating to the Company; (B) breached any law, rule, regulation or policy or procedure of Parent or the Company (including but not limited to policies and procedures pertaining to harassment and discrimination); or (C) been convicted of, or plea of guilty or nolo contendere to, any felony;
  - b. Executive has materially breached any of the provisions of this Agreement, the Confidentiality Agreement or any other material agreement with Parent, the Company or any of their affiliates;
  - c. Actions by Executive involving willful malfeasance or gross negligence in the performance of Executive's duties which have or are reasonably likely to result in a material liability to the Company;

- d. Executive's willful failure or refusal to perform Executive's duties as required by this Agreement if such failure to perform Executive's duties is not cured to the reasonable satisfaction of the Parent Board within ten (10) days following written notice to Executive;
- e. Executive's willful violation of any Company policy that that is demonstrably and materially injurious to the business, financial condition or reputation of the Company or its Affiliates.

For purposes of the foregoing definition of Cause, no act or failure to act by a Participant shall be deemed willful or intentional if performed in good faith and with the reasonable belief that the action or inaction was in the best interests of the Company and its affiliates.

- 2. <u>Change in Control</u>. "Change in Control" shall have the meaning set forth in the Incentive Plan, as amended and/or restated from time to time.
- 3. <u>Disability</u>. "*Disability*" shall mean that Executive has become entitled to receive benefits under the Company's applicable long-term disability plan or, if no such plan covers Executive, "Disability" means a physical or mental incapacity as a result of which Executive becomes unable to continue the proper performance of Executive's duties hereunder in substantially a full-time capacity (reasonable absences because of sickness for up to three (3) consecutive months excepted; *provided*, *however*, that any new period of incapacity or absence is determined by the Parent Board, in good faith, to be related to the prior incapacity or absence). A determination of Disability shall be subject to the certification of a qualified medical doctor agreed to by the Parent Board and Executive or, in the event of Executive's incapacity to designate a doctor, Executive's legal representative. In the absence of agreement between the Parent Board and Executive, each party shall nominate a qualified medical doctor and the two doctors so nominated shall select a third doctor, who shall make the determination as to Disability.
- 4. <u>Good Reason</u>. "Good Reason" means, without Executive's written consent, the occurrence of any one or more of the following:
  - a. A material reduction in Executive's Base Salary or Target Bonus (excluding any reductions in Executive's Base Salary, and any corresponding reductions in Executive's Target Bonus, in connection with temporary across-the-board salary reductions imposed on substantially all of the Company's senior executives that do not exceed, in the aggregate, ten percent (10%) of Executive's Base Salary during any twelve (12)-month period).

- b. The relocation of Executive's principal place of employment to a location that is greater than twenty-five (25) miles from the Principal Location if that relocation increases Executive's commute by twenty-five (25) miles or more.
- c. A material reduction in Executive's title, duties or responsibilities as contemplated by this Agreement (other than during a period of physical or mental incapacity); *provided*, *however*, that if Executive is not re-elected as a member of the Parent Board by the shareholders of Parent after having been nominated by the Board, then this clause (iii) shall not apply.

Notwithstanding the foregoing, Executive will not be deemed to have resigned for Good Reason unless (1) Executive provides Parent with written notice setting forth in reasonable detail the facts and circumstances claimed by Executive to constitute Good Reason within sixty (60) days after the date of the occurrence of any event that Executive knows or should reasonably have known to constitute Good Reason, (2) Parent or the Company, as applicable, fails to cure such acts or omissions within thirty (30) days following Parent's receipt of such notice, and (3) the effective date of Executive's termination for Good Reason occurs no later than sixty (60) days after the expiration of such cure period.

- B. <u>Termination by Company</u>. The Company may terminate Executive's employment hereunder immediately, with or without Cause, or due to Executive's Disability. This Agreement will automatically terminate upon Executive's death during the Term.
- C. <u>Termination by Executive</u>. Executive may terminate this Agreement without Good Reason upon sixty (60) days' written notice to the Parent Board, and Executive may terminate this Agreement for Good Reason in accordance with Section 4.1(d) above.

## D. Benefits Received Upon Termination.

1. If Executive's employment terminates during the Term for any reason, then the Company shall pay or provide to Executive: (i) Executive's earned but unpaid Base Salary through the Date of Termination (as defined below), (ii) to the extent required by applicable law, any vacation earned but not taken through the Date of Termination, and (iii) any vested amounts due to Executive under any plan, program or policy of the Company (collectively, the "Accrued Obligations"). The Accrued Obligations described in

- clauses (i) (ii) of the preceding sentence shall be paid within thirty (30) days after the Date of Termination (or such earlier date as may be required by applicable law) and the Accrued Obligations described in clause (iii) of the preceding sentence shall be paid in accordance with the terms of the governing plan or program. The Company and Parent shall thereafter have no further obligations to Executive under this Agreement.
- 2. If Executive's employment is terminated by the Company without Cause (excluding termination by reason of death or Disability), or Executive terminates his employment for Good Reason, in either case, prior to the consummation of a Change in Control or more than twelve (12) months after the consummation of a Change in Control, then upon Executive's "separation from service" from the Company (within the meaning of Section 409A of the Internal Revenue Code of 1986, as amended (the "Code")) (a "Separation from Service" and, the date of any such Separation from Service, the "Date of Termination"), subject to Section 4.5 below, the Company shall:
  - a. pay to Executive an amount equal to Executive's earned but unpaid Annual Bonus for the fiscal year ending immediately prior to the year in which the Date of Termination occurs, said Annual Bonus to be paid as and when annual bonuses are payable for such year generally;
  - b. pay to Executive as severance pay an amount equal to eighteen (18) months of Executive's Base Salary in effect as of the Date of Termination with such payments to be made in accordance with the Company's usual payroll periods during the eighteen (18) month period commencing on the Date of Termination; *provided*, that no such payments shall be made prior to the date on which the Release (as defined below) becomes effective and irrevocable and, if the aggregate period during which Executive is entitled to consider and/or revoke the Release spans two (2) calendar years, no payments under this Section 4.4(b)(ii) shall be made prior to the beginning of the second (2nd) such calendar year (and any payments otherwise payable prior thereto (if any) shall instead be paid commencing on the first regularly scheduled Company payroll date occurring in the latter such calendar year);
  - c. pay to Executive a pro-rated Target Bonus amount for the year in which the Date of Termination occurs, determined by multiplying Executive's Target Bonus for the year in which the Date of Termination occurs by a fraction, the numerator of which equals the number of days Executive was employed

by the Company during the calendar year in which the Date of Termination occurs and the denominator of which equals 365 or 366 (as applicable) (a "*Pro-Rata Bonus*"), payable on the first regularly scheduled payroll date following the date on which the Release becomes effective and irrevocable, *provided*, that if the aggregate period during which Executive is entitled to consider and/or revoke the Release spans two (2) calendar years, such payment shall be made on the first regularly scheduled payroll date in the second (2nd) such calendar year or, if later, the first regularly scheduled payroll date following the date the Release becomes effective and irrevocable; and

d. subject to Executive's valid election to continue healthcare coverage under Section 4980B of the Code, during the period commencing on the Date of Termination and ending on the date that is eighteen (18) months thereafter, or, if earlier, the date on which Executive becomes covered by a group health insurance program provided by a subsequent employer (in either case, the "COBRA Period"), the Company shall reimburse Executive for Executive's and Executive's eligible dependents with coverage under its group health plans at the same levels and the same cost to Executive as would have applied if Executive's employment had not been terminated based on Executive's elections in effect on the Date of Termination, provided, however, that (A) if any plan pursuant to which such benefits are provided is not, or ceases prior to the expiration of the period of continuation coverage to be, exempt from the application of Section 409A under Treasury Regulation Section 1.409A-1(a)(5), or (B) the Company is otherwise unable to continue to cover Executive under its group health plans without incurring penalties (including without limitation, pursuant to Section 2716 of the Public Health Service Act or the Patient Protection and Affordable Care Act), then, in either case, an amount equal to each remaining Company subsidy shall thereafter be paid to Executive in substantially equal monthly installments over the continuation coverage period (or the remaining portion thereof).

- 3. If within twelve (12) months following the consummation of a Change in Control, Executive's employment is terminated either by the Company without Cause (excluding by reason of death or Disability) or by Executive for Good Reason, then, in either case, upon Executive's Separation from Service, subject to Section 4.5 below, the Company shall:
  - a. pay to Executive an amount equal to Executive's earned but unpaid Annual Bonus for the year ending immediately prior to the year in which the Date of Termination occurs, said Annual Bonus to be paid as and when annual bonuses are payable for such year generally;
  - b. pay to Executive as severance pay an amount equal to the sum of (A) eighteen (18) months of Executive's Base Salary in effect as of the Date of Termination and (B) one and one-half (1.5) Executive's Target Bonus for the year in which the Date of Termination occurs, such payments to be made in accordance with the Company's usual payroll periods during the eighteen (18) month period commencing on the Date of Termination; *provided*, that no such payments shall be made prior to the date on which the Release becomes effective and irrevocable and, if the aggregate period during which Executive is entitled to consider and/or revoke the Release spans two (2) calendar years, no payments under this Section 4.4(c)(ii) shall be made prior to the beginning of the second (2nd) such calendar year (and any payments otherwise payable prior thereto (if any) shall instead be paid commencing on the first regularly scheduled Company payroll date occurring in the latter such calendar year):
  - c. pay to Executive a Pro-Rata Bonus, payable in a lump sum on the first regularly scheduled payroll date following the date on which the Release becomes effective and irrevocable, *provided*, that if the aggregate period during which Executive is entitled to consider and/or revoke the Release spans two (2) calendar years, such payment shall be made on the first regularly scheduled payroll date in the second (2<sup>nd</sup>) such calendar year or, if later, the first regularly scheduled payroll date following the date the Release becomes effective and irrevocable; and
  - d. subject to Executive's valid election to continue healthcare coverage under Section 4980B of the Code, during the COBRA Period, the Company shall reimburse Executive for Executive's and Executive's eligible dependents with coverage under its group health plans at the same levels and the same cost to Executive as would have applied if Executive's employment had not been terminated based on Executive's elections in effect on the Date of Termination, *provided, however*, that (A) if any plan pursuant to which such benefits are provided is not, or ceases prior to the expiration of the period of continuation coverage to be, exempt from the application of Section 409A under Treasury

Regulation Section 1.409A-1(a)(5), or (B) the Company is otherwise unable to continue to cover Executive under its group health plans without incurring penalties (including without limitation, pursuant to Section 2716 of the Public Health Service Act or the Patient Protection and Affordable Care Act), then, in either case, an amount equal to each remaining Company subsidy shall thereafter be paid to Executive in substantially equal monthly installments over the continuation coverage period (or the remaining portion thereof).

- e. Executive agrees that, if (i) a Change in Control occurs within twelve (12) months following the consummation of the Transaction (or pursuant to a binding agreement entered into within such twelve (12) month period) and (ii) Executive's Termination occurs within thirty (30) days after such Change in Control, then, notwithstanding anything to the contrary in any other plan or agreement, including without limitation the Incentive Plan, the Option Agreement, and the award agreement governing the PSUs, such Termination will not result in any accelerated vesting of the Option or the PSUs.
- 4. <u>Termination Because of Employee Death or Disability</u>. In the event of Executive's Disability, Executive acknowledges that his employment may be terminated by the Company; *provided* that, during the period of the Disability prior to such termination of employment, Executive shall continue to receive all compensation and benefits as if Executive were actively employed less any sums received directly by Executive, if any, under any applicable disability income insurance policy maintained by the Company. In the event that the Company terminates Executive's employment due to his Disability, Executive shall have the right to continue to receive any payments made under any applicable disability insurance policy maintained by the Company in accordance with, and subject to the terms and conditions of, such policy. In addition, Executive (or Executive's estate) shall receive a Pro-Rata Bonus within thirty (30) days following the Date of Termination due to death or Disability and any earned but unpaid Annual Bonus for the year ending immediately prior to the year in which the Date of Termination occurs, said Annual Bonus to be paid as and when annual bonuses are payable for such year generally.
- E. <u>Release</u>. Notwithstanding the foregoing, it shall be a condition to Executive's right to receive the amounts provided for in Section 4.4(b) or (c) hereof (as applicable) that Executive execute and deliver to the Company a release of claims substantially in the form attached hereto as Exhibit D that becomes effective and irrevocable no more than sixty (60) days after the date on which Executive's employment terminates.

F. <u>Effect of Termination</u>. Upon any termination of Executive's employment with the Company for any reason, Executive shall be deemed to have immediately resigned as Chief Financial Officer of Parent and the Company, and in any other capacity with Parent and the Company (including as an employee, officer and/or director), as well as with all subsidiaries, if applicable, without the giving of any notice or the taking of any action.

# V. SUCCESSORS AND ASSIGNS

A. <u>Assignment</u>. This Agreement shall inure to the benefit of and be binding upon Parent, the Company and to any person or entity which succeeds to all or substantially all of the business of Parent and/or the Company through merger, consolidation, reorganization, or other business combination or by acquisition of all or substantially all of the assets of Parent and/or the Company. To the extent that any such successor or assign (whether direct or indirect, by purchase, merger, consolidation or otherwise) to all or substantially all of the business and/or assets of Parent and/or the Company does not automatically, under applicable law, assume all obligations under this Agreement, then Parent and/or the Company will utilize its best efforts to obtain the agreement of the successor or assign to assume all the obligations arising from this Agreement, and to perform this Agreement in the same manner and to the same extent that Parent and/or the Company would be required to perform it if no such succession or assignment had taken place. Any failure of Parent and/or the Company to obtain such agreement prior to the effectiveness of any such succession or assignment shall be deemed to be a material breach of this Agreement unless otherwise agreed to between Executive and such successor or assign. The obligations of this Article shall apply equally to Parent and the Company, as herein before defined, and to any future successor or assign to its business which automatically by operation of law or otherwise (including pursuant to such successor's or assign's agreement to assume all obligations arising from this Agreement) becomes bound by all the terms and provisions of this Agreement (*i.e.*, the obligations of this Agreement or the Company owns the majority of the voting securities of the corporation and the corporation becomes the employer for Executive at any time during the term of this Agreement.

B. <u>Executive Assigns</u>. This Agreement is personal to Executive and, without the prior written consent of Parent, shall not be assignable by Executive other than by will or the laws of descent and distribution. This Agreement shall inure to the benefit of and be enforceable by Executive and Executive's legal representatives, executors, heirs, distributees, devisees and legatees.

# VI. EXCESS PARACHUTE PAYMENTS; LIMITATION ON PAYMENTS

A. Best Pay Cap. Notwithstanding any other provision of this Agreement, in the event that any payment or benefit received or to be received by Executive (whether pursuant to the terms of this Agreement or any other plan, arrangement or agreement) (all such payments and benefits being hereinafter referred to as the "Total Payments") would be subject (in whole or part), to the excise tax imposed under Section 4999 of the Code (the "Excise Tax"), then, after taking into account any reduction in the Total Payments provided by reason of Section 280G of the Code in any other plan, arrangement or agreement, then such remaining Total Payments shall be reduced, to the extent necessary so that no portion of the Total Payments is subject to the Excise Tax but only if (i) the net amount of such Total Payments, as so reduced (and after subtracting the net amount of federal, state and local income taxes on such reduced Total Payments and after taking into account the phase out of itemized deductions and personal exemptions attributable to such reduced Total Payments) is greater than or equal to (ii) the net amount of such Total Payments without such reduction (but after subtracting the net amount of federal, state and local income taxes on such Total Payments and the amount of Excise Tax to which Executive would be subject in respect of such unreduced Total Payments and after taking into account the phase out of itemized deductions and personal exemptions attributable to such unreduced Total Payments). The reduction of the amounts payable hereunder, if applicable, shall be made by reducing the payments and benefits in the following order: (i) cash payments that may not be valued under Treas. Reg. § 1.280G-1, Q&A-24(c) ("24(c)"), (ii) equity-based payments that may not be valued under 24(c), (iii) cash payments that may be valued under 24(c), (iv) equity-based payments that may be valued under 24(c) and (v) other types of benefits. With respect to each category of the foregoing, such reduction shall occur first with respect to amounts that are not "deferred compensation" within the meaning of Section 409A of the Code and next with respect to payments that are deferred compensation within the meaning of Section 409A of the Code, in each case, beginning with payments or benefits that are to be paid the farthest in time from the determination of the Independent Advisors (as defined below). All reasonable fees and expenses of the Independent Advisors shall be borne solely by the Company.

B. <u>Certain Exclusions</u>. For purposes of determining whether and the extent to which the Total Payments will be subject to the Excise Tax, (i) no portion of the Total Payments the receipt or enjoyment of which Executive shall have waived at such time and in such manner as not to constitute a "payment" within the meaning of Section 280G(b) of the Code shall be taken into account; (ii) no portion of the Total Payments shall be taken into account which, in the written opinion of an independent, nationally recognized accounting firm (the "*Independent Advisors*") selected by Parent, does not constitute a "parachute payment" within the meaning of Section 280G(b)(2) of the Code (including by reason of Section 280G(b)(4)(A) of the Code) and, in calculating the Excise Tax, no portion of such Total Payments shall be taken into account which, in the opinion of Independent Advisors, constitutes reasonable compensation for services actually rendered, within the meaning of Section 280G(b)(4)(B) of the Code, in excess of the "base amount" (as defined in Section 280G(b)(3) of the Code) allocable to such reasonable compensation; and (iii) the value of any non-cash benefit or any deferred payment or benefit included in the Total Payments shall be determined by the Independent Advisors in accordance with the principles of Sections 280G(d)(3) and (4) of the Code.

## VII. GENERAL PROVISIONS

A. <u>Notice</u>. For purposes of this Agreement, notices and all other communications provided for in the Agreement shall be in writing and shall be deemed to have been duly given when delivered in person or mailed by United States registered mail, return receipt requested, postage prepaid, as follows:

If to Parent or the Company:

The Beauty Health Company 1819 West Avenue Bay 2 Miami Beach, Florida 33139 Attn: Executive Chairman

If to Executive:

to the last address on file for Executive with the Company or such other address as either party may have furnished to the other in writing in accordance herewith, except that notices or change of address shall be effective only upon receipt.

- B. <u>Amendments; No Waivers</u>. No provision of this Agreement may be amended, modified, waived or discharged unless such amendment, waiver, modification or discharge is agreed to in a writing signed by Executive, Parent and the Company. No waiver by either party hereto at any time of any breach by the other party hereto of, or compliance with, any condition or provision of this Agreement to be performed by such other party shall be deemed a waiver of similar or dissimilar provisions or conditions at the same or at any prior or subsequent time.
- C. <u>Governing Law</u>. This Agreement shall be governed by and construed and interpreted in accordance with the laws of the State of California, without regard to its conflicts of law principles.
- D. <u>Severability or Partial Invalidity</u>. The invalidity or unenforceability of any provisions of this Agreement shall not affect the validity or enforceability of any other provision of this Agreement, which shall remain in full force and effect.
- E. <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, each of which shall be deemed to be an original but all of which together will constitute one and the same instrument.
- F. Entire Agreement. This Agreement, together with the Confidentiality Agreement, constitutes the entire agreement of the parties and supersedes all prior written or oral and all contemporaneous oral agreements, understandings, and negotiations between the parties with respect to the subject matter hereof, including without limitation the Prior Agreement. This Agreement, together with the Confidentiality Agreement, is intended by the parties as the final expression of their agreement with respect to such terms as are included in this Agreement and may not be contradicted by evidence of any prior or contemporaneous agreement. The parties further intend that this Agreement, together with the Confidentiality Agreement, constitutes the complete and exclusive statement of its terms and that no extrinsic evidence may be introduced in any judicial proceeding involving this Agreement. Notwithstanding anything herein to the contrary, (i) this Agreement and the obligations and commitments hereunder shall neither commence nor be of any force or effect prior to the Effective Date and unless and until the Transaction is fully consummated in conformity with the terms of the Merger Agreement and (ii) in the event that the Merger Agreement is terminated in accordance with its terms, the Transaction is not consummated for any reason, or Executive's employment with the Company terminates for any reason prior to the closing of the Transaction, this Agreement will automatically terminate and be void *ab initio*.

### G. Arbitration.

- 1. All disputes, controversies, and claims between Executive and Parent and/or the Company, or any of its officers, directors, employees, or agents in their capacity as such, including any controversy or dispute, whether based on contract, common law, or federal, state or local statute or regulation, that arise under or are related to this Agreement Executive's employment with the Company or the termination thereof shall be submitted to final and binding arbitration as the sole and exclusive remedy for such controversy or dispute in accordance with the rules of JAMS pursuant to its Employment Arbitration Rules and Procedures (the current version of which are available at http://www.jamsadr.com/rules-employment-arbitration/ and a copy of which will be provided by the Company to Executive upon Executive's request). Notwithstanding the foregoing, this Agreement shall not require arbitration pursuant to this Section 7.7 of any claims: (A) under a Company benefit plan subject to the Employee Retirement Income Security Act, as amended; or (B) as to which applicable law not preempted by the Federal Arbitration Act prohibits resolution by binding arbitration. Either party may seek provisional non-monetary remedies in a court of competent jurisdiction to the extent that such remedies are not available or not available in a timely fashion through arbitration. It is the parties' intent that issues of arbitrability of any dispute shall be decided by the arbitrator.
- 2. The arbitration shall take place before a single neutral arbitrator at the JAMS office in Los Angeles, California. Such arbitrator shall be provided through JAMS by mutual agreement of the parties to the arbitration; *provided*, that, absent such agreement, the arbitrator shall be selected in accordance with the rules of JAMS then in effect. The arbitrator shall permit reasonable discovery. The award or decision of the arbitrator shall be rendered in writing; shall be final and binding on the parties; and may be enforced by judgment or order of a court of competent jurisdiction.
- 3. The Company shall pay all arbitrator fees, filing fees and other costs unique to the arbitration procedure (in each case, to the extent required by applicable law), except that if Executive initiates a claim subject to arbitration, Executive will pay any filing fee up to the amount Executive would be required to pay if Executive initiated

the claim in the Superior Court of Los Angeles for the State of California for the County of Los Angeles. All other fees and costs shall be shared evenly by the Company and Executive. In the event of arbitration relating to this Agreement, the non-prevailing party shall reimburse the prevailing party for all costs incurred by the prevailing party in connection with such arbitration (including, without limitation, reasonable legal fees in connection with such arbitration, including any litigation or appeal therefrom).

- 4. EXECUTIVE, PARENT AND THE COMPANY UNDERSTAND THAT BY AGREEING TO ARBITRATE ANY ARBITRATION CLAIM, THEY WILL NOT HAVE THE RIGHT TO HAVE ANY ARBITRATION CLAIM DECIDED BY A JURY OR A COURT, BUT SHALL INSTEAD HAVE ANY ARBITRATION CLAIM DECIDED THROUGH ARBITRATION.
- 5. EXECUTIVE, PARENT AND THE COMPANY WAIVE ANY CONSTITUTIONAL OR OTHER RIGHT TO BRING CLAIMS COVERED BY THIS AGREEMENT OTHER THAN IN THEIR INDIVIDUAL CAPACITIES. EXCEPT AS MAY BE PROHIBITED BY LAW, THIS WAIVER INCLUDES THE ABILITY TO ASSERT CLAIMS AS A PLAINTIFF OR CLASS MEMBER IN ANY PURPORTED CLASS OR REPRESENTATIVE PROCEEDING.
- 6. This Section 7.7 shall be interpreted to conform to any applicable law concerning the terms and enforcement of agreements to arbitrate service disputes. To the extent any terms or conditions of this Section 7.7 would preclude its enforcement, such terms shall be severed or interpreted in a manner to allow for the enforcement of this Section 7.7. To the extent applicable law imposes additional requirements to allow enforcement of this Section 7.7, this Agreement shall be interpreted to include such terms or conditions.

H. <u>Indemnification</u>. To the extent permitted by law, applicable statutes and the Articles of Incorporation, By-laws or resolutions of Parent and the Company in effect from time to time, during the Term, Executive shall be entitled to indemnification by Parent and the Company. In addition, during the Term, Parent and/or the Company shall provide Executive with coverage under the directors and officers liability insurance policy, if any, maintained by Parent and/or the Company for the benefit of the members of the Parent Board and officers of Parent and the Company, to the same extent as such coverage is provided to members of the Parent Board and similarly-situated executive officers of Parent and the Company.

### I. Section 409A.

- 1. To the extent applicable, this Agreement shall be interpreted in accordance with Section 409A of the Code and Department of Treasury regulations and other interpretative guidance issued thereunder, including without limitation any such regulations or other such guidance that may be issued after the Effective Date (collectively, "Section 409A"). Notwithstanding any provision of this Agreement to the contrary, in the event that following the Effective Date, Parent or the Company determines that any compensation or benefits payable under this Agreement may be subject to Section 409A, Parent and the Company may adopt such amendments to this Agreement or adopt other policies or procedures (including amendments, policies and procedures with retroactive effect), or take any other actions that Parent and the Company determines are necessary or appropriate to preserve the intended tax treatment of the compensation and benefits payable hereunder, including without limitation actions intended to (i) exempt the compensation and benefits payable under this Agreement from Section 409A, and/or (ii) comply with the requirements of Section 409A, provided, however, that this Section 7.9(a) does not, and shall not be construed so as to, create any obligation on the part of Parent or the Company to adopt any such amendments, policies or procedures or to take any other such actions or to create any liability on the part of Parent or the Company for any failure to do so. In no event shall Parent, the Company, any of their respective affiliates or any of their respective officers, directors or advisors be liable for any taxes, penalties or interest imposed under or by operation of Section 409A. Any right to a series of installment payments pursuant to this Agreement is to be treated as a right to a series of separate payments.
- 2. Notwithstanding anything to the contrary in this Agreement, no compensation or benefits (including, without limitation, any compensation or benefits provided pursuant to Section 4.4(b) or (c) above) shall be paid to Executive during the six (6)-month period following Executive's Separation from Service if the Company determines that paying such amounts at the time or times indicated in this Agreement would be a prohibited distribution under Section 409A(a)(2)(B)(i) of the Code. If the payment of any such amounts is delayed as a result of the previous sentence, then on the first business day following the end of such six (6)-month period (or such earlier date upon which such amount can be paid under Section 409A without resulting in a prohibited distribution, including as a result of Executive's death), the Company shall pay Executive a lump-sum amount equal to the cumulative amount that would have otherwise been payable to Executive during such period (without interest).
- J. <u>Withholding</u>. Any payments hereunder will be subject to any required withholding of federal, state and local taxes pursuant to applicable law or regulation, and the Company and its affiliates shall be entitled to withhold any and all such taxes from amounts payable hereunder.

[Signature Page Follows]

IN WITNESS WHEREOF, the parties have executed this Agreement as of the Effective Date.

EXECUTIVE

/s/ Liyuan Woo

By Liyuan Woo

[Signature Page to CFO Employment Agreement]

# PARENT

# /s/ Brenton L. Saunders

Brent Saunders, Executive Chairman of the Parent Board

[Signature Page to CFO Employment Agreement]

**COMPANY** 

/s/ Clinton E. Carnell

Clint Carnell, CEO

[Signature Page to CFO Employment Agreement]

## **EXHIBIT A**

## **OPTION AWARD AGREEMENT**

THE BEAUTY HEALTH COMPANY	
2021 INCENTIVE AWARD PLAN	

## STOCK OPTION GRANT NOTICE CEO and CFO FORM

The Beauty Health Company, a Delaware corporation (the "Company") has granted to the participant listed below ("Participant") the stock option (the "Option") described in this Stock Option Grant Notice (the "Grant Notice"), subject to the terms and conditions of The Beauty Health Company 2021 Incentive Award Plan (as amended from time to time, the "Plan") and the Stock Option Agreement attached hereto as Exhibit A (the "Agreement"), both of which are incorporated into this Grant Notice by reference. Capitalized terms not specifically defined in this Grant Notice or the Agreement have the meanings given to them in the Plan.

Participant:	[]
Grant Date:	[], 2021
Exercise Price per Share:	[]
Shares Subject to the Option:	
Final Expiration Date:	[]
Vesting Commencement Date:	[], 2021

**Vesting Schedule:** 

Twenty-five percent (25%) of the Shares subject to the Option will vest on each of the first four anniversaries of the Vesting Commencement Date, subject to Participant's continued status as a Service Provider through the applicable vesting date. Notwithstanding the foregoing, (x) if Participant incurs a Termination of Service due to Participant's death or Disability, subject to and conditioned upon Participant's (or Participant's estate's) timely execution and non-revocation of a release of claims in a form prescribed by the Company (a "Release") that becomes effective and irrevocable no later than sixty (60) days following such Termination of Service (the date such Release becomes effective and irrevocable, the "Release Effective Date"), the Option will vest in full (to the extent then-unvested) upon the Release Effective Date (and shall remain outstanding and eligible to vest through the Release Effective Date and shall automatically be forfeited if the Release does not become effective and irrevocable on or prior to the sixtieth (60th) day following such termination), and (y) if a Change in Control is consummated more than twelve (12) months after [Closing

Date], 2021 (the "Closing Date") and Participant's status as a Service Provider is terminated by the Company or its Subsidiaries without Cause (as defined in the Agreement) or due to Participant's resignation for Good Reason (as defined in the Employment Agreement between Participant and the Company, dated [\_\_\_]), in either case, within twelve (12) months following the consummation of such Change in Control, subject to and conditioned upon Participant's timely execution and non-revocation of a Release that becomes effective and irrevocable no later than sixty (60) days following such Termination of Service, the Option will vest in full (to the extent then-unvested) upon the Release Effective Date (and shall remain outstanding and eligible to vest through the Release Effective Date and shall automatically be forfeited if the Release does not become effective and irrevocable on or prior to the sixtieth (60th) day following such termination).

If (i) a Change in Control occurs within twelve (12) months following the Closing Date (or pursuant to a binding agreement entered into within such twelve (12) month period) and (ii) Participant's status as a Service Provider is terminated by the Company or its Subsidiaries without Cause (as defined in the Agreement) or due to Participant's resignation for Good Reason within twelve (12) months after such Change in Control, then, notwithstanding anything to the contrary in any other plan or agreement, such termination will not result in any accelerated vesting of the Option.

[Incentive Stock Option]/[Non-Qualified Stock Option]

## **Type of Option**

By accepting (whether in writing, electronically or otherwise) the Option, Participant agrees to be bound by the terms of this Grant Notice, the Plan and the Agreement. Participant has reviewed the Plan, this Grant Notice and the Agreement in their entirety, has had an opportunity to obtain the advice of counsel prior to executing this Grant Notice and fully understands all provisions of the Plan, this Grant Notice and the Agreement. Participant hereby agrees to accept as binding, conclusive and final all decisions or interpretations of the Administrator upon any questions arising under the Plan, this Grant Notice or the Agreement.

THE BEAUTY HEALTH COMPANY	PARTICIPANT
By:	
Name:	[Participant Name]
Title:	

#### STOCK OPTION AGREEMENT

Capitalized terms not specifically defined in this Agreement have the meanings specified in the Grant Notice or, if not defined in the Grant Notice, in the Plan.

#### • GENERAL

- <u>Grant of Option</u>. The Company has granted to Participant the Option effective as of the grant date set forth in the Grant Notice (the "*Grant Date*").
- <u>Incorporation of Terms of Plan</u>. The Option is subject to the terms and conditions set forth in this Agreement and the Plan, which is incorporated herein by reference. In the event of any inconsistency between the Plan and this Agreement, the terms of the Plan will control, unless it is expressly specified in in this Agreement or the Grant Notice that the specific provision of the Plan will not apply. For clarity, the foregoing sentence shall not limit the applicability of any additive language contained in this Agreement which provides supplemental or additional terms not inconsistent with the Plan.

# PERIOD OF EXERCISABILITY

- Commencement of Exercisability. The Option will vest and become exercisable according to the vesting schedule in the Grant Notice (the "Vesting Schedule") except that any fraction of a Share as to which the Option would be vested or exercisable will be accumulated and will vest and become exercisable only when a whole vested Share has accumulated; provided, however, that, notwithstanding the foregoing or anything to the contrary in the Grant Notice or this Agreement, in no event may the Option be exercised (in whole or in part) prior to the date on which the Company files a Form S-8 Registration Statement covering the Shares subject to the Option. Except as otherwise set forth in the Grant Notice, the Plan or this Agreement, and unless the Administrator otherwise determines, the Option will immediately expire and be forfeited as to any portion of the Option that is not vested and exercisable as of Participant's Termination of Service for any reason (after taking into consideration any accelerated vesting and exercisability which may occur in connection with such Termination of Service, if any).
- Duration of Exercisability. The Vesting Schedule is cumulative. Any portion of the Option which vests and becomes exercisable will remain vested and exercisable until the Option expires. The Option will be forfeited immediately upon its expiration.
- <u>Expiration of Option</u>. Except as may be extended in accordance with Section 5.3 of the Plan, the Option may not be exercised to any extent by anyone after, and will expire on, the first of the following to occur:
  - The final expiration date in the Grant Notice;
- Except as the Administrator may otherwise approve, the expiration of three (3) months from the date of Participant's Termination of Service, unless Participant's Termination of Service is for Cause (as defined below), due to Participant's voluntary resignation or by reason of Participant's death or Disability;
- Except as the Administrator may otherwise approve, the expiration of one year from the date of Participant's Termination of Service by reason of Participant's death or Disability;

- Except as the Administrator may otherwise approve, the expiration of one (1) month from the date of Participant's Termination of Service due to Participant's voluntary resignation for any reason; and
  - Except as the Administrator may otherwise approve, Participant's Termination of Service for Cause.
  - 2.4 <u>Cause Definition</u>. As used in this Agreement, "*Cause*" means (i) if Participant is a party to a written employment or consulting agreement with the Company or a Subsidiary in which the term "cause" is defined (a "*Relevant Agreement*"), "Cause" as defined in the Relevant Agreement, and (ii) if no Relevant Agreement exists, the occurrence of any one or more of the following events:
- (a) Participant has engaged in an act of dishonesty, theft, embezzlement or fraud, a breach of confidentiality, an unauthorized disclosure or use of inside information, customer lists, trade secrets or other confidential information; has breached a fiduciary duty, law, rule, regulation or policy or procedure of the Company (including but not limited to policies and procedures pertaining to harassment and discrimination); Participant's commission of, or plea of guilty or *nolo contendere* to, any crime or offense (other than minor traffic violations or similar offenses), or any act by Participant constituting a felony;
- (b) Participant's gross or repeated neglect of, or repeated or willful failure to perform, his or her duties to the Company, or Participant's history of substandard performance, if such substandard performance is not cured to the satisfaction of the Board, the Chief Executive Officer of the Company (the "*CEO*") (unless Participant is the CEO), and/or Participant's manager within ten (10) days following notice to Participant;
- (c) Participant has breached any of the provisions of any employment, confidentiality, restrictive covenant or other agreement between Participant and the Company or an affiliate thereof;
  - (d) actions by Participant involving malfeasance or gross negligence in the performance of, Participant's duties;
- (e) Participant's failure or refusal to perform Participant's duties to the Company on an exclusive and full-time basis if such failure to perform Participant's duties is not cured to the satisfaction of the Board, the CEO (unless Participant is the CEO), and/or Participant's manager (as applicable) within ten (10) days following notice to the Participant;
- (f) Participant's insubordination or failure to follow the Board's (or the CEO's (unless Participant is the CEO) or such Participant's manager's) instructions;
- (g) Participant's violation of any rule, regulation, or policy of the Company or the Board (or Participant's manager) applicable to similarly-situated employees of the Company generally, including, without limitation, rules, regulations, or policies addressing confidentiality, non-solicitation or non-competition, if such violation is not cured to the satisfaction of the Board, the CEO (unless Participant is the CEO), and/or Participant's manager (as applicable) within ten (10) days following notice to the Participant;
- (h) Participant's use of alcohol or illicit drugs in a manner that has or would reasonably be expected to have a detrimental effect on Participant's performance, Participant's duties to Company, or

the reputation of the Company or its affiliates; or

(i) Participant's performance of acts which are or could reasonably be expected to become materially detrimental to the image, reputation, finances or business of the Company or any of its affiliates, including but limited to the Participant's commission of unlawful harassment or discrimination.

# EXERCISE OF OPTION

- <u>Person Eligible to Exercise</u>. During Participant's lifetime, only Participant may exercise the Option. After Participant's death, any exercisable portion of the Option may, prior to the time the Option expires, be exercised by Participant's Designated Beneficiary as provided in the Plan.
- <u>Partial Exercise</u>. Any exercisable portion of the Option or the entire Option, if then wholly exercisable, may be exercised, in whole or in part, according to the procedures in the Plan at any time prior to the time the Option or portion thereof expires, except that the Option may only be exercised for whole Shares.
  - Tax Withholding; Exercise Price.
- Unless the Administrator otherwise determines, the Company shall withhold, or cause to be withheld, Shares otherwise vesting or issuable under this Option in satisfaction of any exercise price and/or applicable withholding tax obligations, in accordance with the Plan. With respect to tax withholding obligations, the number of Shares which may be so withheld or surrendered shall be limited to the number of Shares which have a fair market value on the date of withholding no greater than the aggregate amount of such liabilities based on the maximum individual statutory withholding rates in Participant's applicable jurisdictions for federal, state, local and foreign income tax and payroll tax purposes that are applicable to such taxable income, in accordance with Section 9.5 of the Plan.
- Participant acknowledges that Participant is ultimately liable and responsible for all taxes owed in connection with the Option, regardless of any action the Company or any Subsidiary takes with respect to any tax withholding obligations that arise in connection with the Option. Neither the Company nor any Subsidiary makes any representation or undertaking regarding the treatment of any tax withholding in connection with the awarding, vesting or exercise of the Option or the subsequent sale of Shares. The Company and the Subsidiaries do not commit and are under no obligation to structure the Option to reduce or eliminate Participant's tax liability.
- <u>Representation</u>. Participant represents to the Company that Participant has reviewed with Participant's own tax advisors the tax consequences of this Award and the transactions contemplated by the Grant Notice and this Agreement. Participant is relying solely on such advisors and not on any statements or representations of the Company or any of its agents.

# OTHER PROVISIONS

- <u>Adjustments</u>. Participant acknowledges that the Option is subject to adjustment, modification and termination in certain events as provided in this Agreement and the Plan.
- <u>Clawback</u>. The Option and the Shares issuable hereunder shall be subject to any clawback or recoupment policy in effect on the Grant Date or as may be adopted or maintained by the Company following the Grant Date, including the Dodd-Frank Wall Street Reform and Consumer Protection Act and any rules or regulations promulgated thereunder.

- Notices. Any notice to be given under the terms of this Agreement to the Company must be in writing and addressed to the Company in care of the Company's General Counsel at the Company's principal office or the General Counsel's then-current email address or facsimile number. Any notice to be given under the terms of this Agreement to Participant must be in writing and addressed to Participant (or, if Participant is then deceased, to the Designated Beneficiary) at Participant's last known mailing address, email address or facsimile number in the Company's personnel files. By a notice given pursuant to this Section, either party may designate a different address for notices to be given to that party. Any notice will be deemed duly given when actually received, when sent by email, when sent by certified mail (return receipt requested) and deposited with postage prepaid in a post office or branch post office regularly maintained by the United States Postal Service, when delivered by a nationally recognized express shipping company or upon receipt of a facsimile transmission confirmation.
  - <u>Titles</u>. Titles are provided herein for convenience only and are not to serve as a basis for interpretation or construction of this Agreement.
- <u>Conformity to Securities Laws</u>. Participant acknowledges that the Plan, the Grant Notice and this Agreement are intended to conform to the extent necessary with all Applicable Laws and, to the extent Applicable Laws permit, will be deemed amended as necessary to conform to Applicable Laws.
- <u>Successors and Assigns</u>. The Company may assign any of its rights under this Agreement to single or multiple assignees, and this Agreement will inure to the benefit of the successors and assigns of the Company. Subject to the restrictions on transfer set forth in this Agreement or the Plan, this Agreement will be binding upon and inure to the benefit of the heirs, legatees, legal representatives, successors and assigns of the parties hereto.
- <u>Limitations Applicable to Section 16 Persons</u>. Notwithstanding any other provision of the Plan or this Agreement, if Participant is subject to Section 16 of the Exchange Act, the Plan, the Grant Notice, this Agreement and the Option will be subject to any additional limitations set forth in any applicable exemptive rule under Section 16 of the Exchange Act (including any amendment to Rule 16b-3) that are requirements for the application of such exemptive rule. To the extent Applicable Laws permit, this Agreement will be deemed amended as necessary to conform to such applicable exemptive rule.
- Entire Agreement. The Plan, the Grant Notice and this Agreement (including any exhibit hereto) constitute the entire agreement of the parties and supersede in their entirety all prior undertakings and agreements of the Company and Participant with respect to the subject matter hereof.
- Severability. If any portion of the Grant Notice or this Agreement or any action taken under the Grant Notice or this Agreement, in any case is held illegal or invalid for any reason, the illegality or invalidity will not affect the remaining parts of the Grant Notice and/or this Agreement (as applicable), and the Grant Notice and/or this Agreement (as applicable) will be construed and enforced as if the illegal or invalid provisions had been excluded, and the illegal or invalid action will be null and void.

- <u>Limitation on Participant's Rights</u>. Participation in the Plan confers no rights or interests other than as herein provided. This Agreement creates only a contractual obligation on the part of the Company as to amounts payable and may not be construed as creating a trust. Neither the Plan nor any underlying program, in and of itself, has any assets. Participant will have only the rights of a general unsecured creditor of the Company with respect to amounts credited and benefits payable, if any, with respect to the Option, and rights no greater than the right to receive the Shares as a general unsecured creditor with respect to the Option, as and when exercised pursuant to the terms hereof
- Not a Contract of Employment or Service. Nothing in the Plan, the Grant Notice or this Agreement confers upon Participant any right to continue in the employ or service of the Company or any Subsidiary or interferes with or restricts in any way the rights of the Company and its Subsidiaries, which rights are hereby expressly reserved, to discharge or terminate the services of Participant at any time for any reason whatsoever, with or without Cause, except to the extent expressly provided otherwise in a written agreement between the Company or a Subsidiary and Participant.
- <u>Counterparts</u>. The Grant Notice may be executed in one or more counterparts, including by way of any electronic signature, subject to Applicable Law, each of which will be deemed an original and all of which together will constitute one instrument.
  - Incentive Stock Options. If the Option is designated as an Incentive Stock Option:
- Participant acknowledges that to the extent the aggregate fair market value of shares (determined as of the time the option with respect to the shares is granted) with respect to which stock options intended to qualify as "incentive stock options" under Section 422 of the Code, including the Option, are exercisable for the first time by Participant during any calendar year exceeds \$100,000 or if for any other reason such stock options do not qualify or cease to qualify for treatment as "incentive stock options" under Section 422 of the Code, such stock options (including the Option) will be treated as non-qualified stock options. Participant further acknowledges that the rule set forth in the preceding sentence will be applied by taking the Option and other stock options into account in the order in which they were granted, as determined under Section 422(d) of the Code. Participant also acknowledges that if the Option is exercised more than three months after Participant's Termination of Service, other than by reason of death or disability, the Option will be taxed as a Non-Qualified Stock Option.
- Participant will give prompt written notice to the Company of any disposition or other transfer of any Shares acquired under this Agreement if such disposition or other transfer is made (i) within two years from the Grant Date or (ii) within one year after the transfer of such Shares to Participant. Such notice will specify the date of such disposition or other transfer and the amount realized, in cash, other property, assumption of indebtedness or other consideration, by Participant in such disposition or other transfer.
- <u>Governing Law</u>. The Grant Notice and this Agreement will be governed by and interpreted in accordance with the laws of the State of Delaware, disregarding any state's choice-of-law principles requiring the application of a jurisdiction's laws other than the State of Delaware.

\*\*\*\*

## **EXHIBIT B**

## **PSU AWARD AGREEMENT**

# The BeautyHealth Company 2021 INCENTIVE AWARD PLAN

## PERFORMANCE-BASED RESTRICTED STOCK UNIT GRANT NOTICE

The BeautyHealth Company, a Delaware corporation (the "*Company*"), has granted to the participant listed below ("*Participant*") the Restricted Stock Units (the "*RSUs*") described in this Performance-Based Restricted Stock Unit Grant Notice (this "*Grant Notice*"), subject to the terms and conditions of [ ] 2021 Incentive Award Plan (as amended from time to time, the "*Plan*") and the Restricted Stock Unit Agreement attached hereto as **Exhibit A** (the "*Agreement*"), both of which are incorporated into this Grant Notice by reference. Capitalized terms not specifically defined in this Grant Notice or the Agreement have the meanings given to them in the Plan.

Number of RSUs at Maximum:	
Vesting Commencement Date:	[], 2021
Vesting Schedule:	
(a) <u>General</u> . Subject to clause (b) belo	ow, and further subject to and conditioned upon Participant's continued service as a Service
Provider through the last day of the Perforn	nance Period, a number of RSUs shall vest on the last day of the Performance Period equal to
(i) the total number of RSUs granted hereby	y multiplied by (ii) the applicable vesting percentage ("Vesting Percentage") set forth below,

\_\_\_], 2021

	Average Stock Price During the Applicable Measurement Period:	Vesting Percentage (% of Maximum):
Below Threshold	Less than \$25.00	0%
Threshold/Target	\$25.00	66.67%
Stretch	\$30.00	80%
Maximum	\$37.50 or greater	100%

which shall be determined based on greater of (x) the Company's Average Stock Price during the Year 3 Measurement Period and (y) the Company's Average Stock Price during the Year 4 Measurement Period (each forgoing capitalized term as defined below), as follows:

Participant:

**Grant Date:** 

Note to Draft: Insert number of RSUs at maximum.

In the event that the Company's Average Stock Price falls between the Threshold and Target values or Target and Maximum values specified in the table above, the Vesting Percentage shall be interpolated on a linear basis (for clarity, if Average Stock Price falls below the Threshold value, the Vesting Percentage shall equal 0%).

Notwithstanding the foregoing, in the event that a Change in Control is consummated during the Performance Period and Participant remains in continued service as a Service Provider until at least immediately prior to such Change in Control:

- (i) In the event that (A) the Shares do not continue to be publicly traded following the consummation of such Change in Control and (B) no Assumption of the RSUs (as defined in Section 8.3 of the Plan) occurs in connection with such Change in Control, then, immediately prior to the Change in Control, a number of RSUs will vest based solely on the per-Share consideration paid or payable (as applicable) in connection with such Change in Control (as determined by the Administrator) or, if the Change in Control is consummated after the third anniversary of the Vesting Commencement Date, the Company's Average Stock Price during the Year 3 Measurement Period (if greater); and
- (ii) In the event that (A) the Shares do not continue to be publicly traded following the consummation of such Change in Control and (B) an Assumption of the RSUs (as defined in Section 8.3 of the Plan) occurs in connection with such Change in Control, then, effective immediately prior to the closing of the Change in Control, the RSUs will be deemed to convert into a number of unvested RSUs determined based solely on the per-Share consideration paid or payable (as applicable) in connection with such Change in Control (as determined by the Administrator) or, if the Change in Control is consummated after the third anniversary of the Vesting Commencement Date, the Company's Average Stock Price during the Year 3 Measurement Period (if greater). Such unvested RSUs (as so assumed and adjusted in connection with the Change in Control) will be eligible to vest in full on the last day of the Performance Period in accordance with this clause (a) (based solely on the Participant's continued status as a Service Provider through such date) or upon Participant's Termination of Service as provided in clause (b) below.
  - (b) Termination of Service; Change in Control. Notwithstanding clause (a) above:
- (i) If Participant incurs a Termination of Service prior to the last day of the Performance Period, then the RSUs shall vest under clause (a) above or be forfeited (as applicable) in accordance with the following table. Any vesting of the RSUs pursuant to the following table shall (A) be subject to Participant (or Participant's estate, as applicable) timely executing and not revoking a release of claims in a form prescribed by the Company (a "*Release*") that becomes effective and irrevocable no later than sixty (60) days following such Termination of Service (the date such Release becomes effective and irrevocable, the "*Release Effective Date*"), and (B) be effective as of the Release Effective Date:

Reason for Termination of Service	If the Termination of Service Occurs Before the Third Anniversary of the Vesting Commencement Date, then:	Occurs On or After the Third Anniversary of the Vesting Commencement Date but Before the Fourth Anniversary of the Vesting Commencement Date, then:
Death or Disability (as defined below)	A number of RSUs will vest based on the Company's Average Stock Price over the Termination Measurement Period (as defined below).	A number of RSUs will vest based on the greater of (i) the Company's Average Stock Price over the Termination Measurement Period and (ii) the Company's Average Stock Price over the Year 3 Measurement Period.
Without Cause [or for Good Reason (as defined below)] Prior to the Consummation of a Change in Control	All RSUs will be forfeited upon such Termination of Service without payment.	A number of RSUs will vest based on the Company's Average Stock Price over the Year 3 Measurement Period.
Without Cause [or for Good Reason] Within 24 Months After the Consummation of a Change in Control	A number of RSUs will vest based on the Company's Average Stock Price over the Termination Measurement Period.	A number of RSUs will vest based on the greater of (i) the Company's Average Stock Price over the Termination Measurement Period and (ii) the Company's Average Stock Price over the Year 3 Measurement Period.
Any Other Reason (Including for Cause or without Good Reason)	All RSUs will be forfeited upon such Termination of Service without payment.	All RSUs will be forfeited upon such Termination of Service without payment.

If the Termination of Service

(ii) With respect to sub-clause (i) above, (A) the RSUs shall remain outstanding and eligible to vest following Participant's Termination of Service through the Release Effective Date and shall automatically be forfeited on the sixtieth (60th) day following such termination if the Release does not become effective and irrevocable on or prior to such date, and (B) any RSUs that do not become vested on the Release Effective Date pursuant to the applicable sub-clause shall be immediately forfeited on such date.

<sup>(</sup>c) <u>Termination</u>; <u>Forfeiture</u>. Unless earlier terminated as set forth in this Grant Notice or the Agreement, any RSUs that have not become vested on or prior to the last day of the Performance Period will thereupon be automatically forfeited by Participant without payment of any consideration therefor. Except as set forth in clause (b) above, if Participant experiences a Termination of Service for any reason prior to the last day of the Performance Period, all then-unvested RSUs will thereupon be automatically forfeited by Participant without payment of any consideration therefor.

- (d) Definitions. For purposes hereof, the following terms shall have the respective meanings set forth below:
- (i) "Average Stock Price" shall mean, with respect to any Measurement Period, the average Fair Market Value of a Share over such Measurement Period.
- (ii) "*Cause*" shall have the meaning set forth in [the employment agreement between Participant and [the Company], dated [\_\_\_]] / [the Company's Executive Severance Plan].<sup>2</sup>
- (iii) "*Good Reason*" shall have the meaning set forth in [the employment agreement between Participant and [the Company], dated [\_\_\_]] / [the Company's Executive Severance Plan].<sup>3</sup>
- (iv) "*Year 3 Measurement Period*" means the ninety (90)-day period ending on the third (3<sup>rd</sup>) anniversary of the Vesting Commencement Date.
- (v) "Year 4 Measurement Period" means the ninety (90)-day period ending on the fourth (4th) anniversary of the Vesting Commencement Date.
- (vi) "*Measurement Period*" means each of the Termination Measurement Period, the Year 3 Measurement Period and the Year 4 Measurement Period.
- (vii) "*Performance Period*" means the period commencing on the Vesting Commencement Date and ending on the fourth (4th) anniversary of the Vesting Commencement Date.
- (viii) "*Termination Measurement Period*" means the ninety (90)-day period ending on (and including) the date of Participant's Termination of Service.

By accepting (whether in writing, electronically or otherwise) the RSUs, Participant agrees to be bound by the terms of this Grant Notice, the Plan and the Agreement. Participant has reviewed the Plan, this Grant Notice and the Agreement in their entirety, has had an opportunity to obtain the advice of counsel prior to executing this Grant Notice and fully understands all provisions of the Plan, this Grant Notice and the Agreement. Participant hereby agrees to accept as binding, conclusive and final all decisions or interpretations of the Administrator upon any questions arising under the Plan, this Grant Notice or the Agreement.

Note to Draft: To be updated based on whether Participant is party to an employment agreement or a participant in the Executive Severance Plan (if neither, the "Cause" definition from the Executive Severance Plan will be added here).

Note to Draft: To be updated based on whether Participant is party to an employment agreement or a participant in the Executive Severance Plan (if neither, "Good Reason" will be removed).

[•]	PARTICIPANT
By: Name: Title:	[Participant Name]

#### RESTRICTED STOCK UNIT AGREEMENT

Capitalized terms not specifically defined in this Agreement have the meanings specified in the Grant Notice or, if not defined in the Grant Notice, in the Plan.

## ARTICLE I. GENERAL

- 1.1 <u>Award of RSUs</u>. The Company has granted the RSUs to Participant effective as of the Grant Date set forth in the Grant Notice (the "*Grant Date*"). Each RSU represents the right to receive one Share as set forth in this Agreement. Participant will have no right to the distribution of any Shares until the time (if ever) the RSUs have vested.
- 1.2 <u>Incorporation of Terms of Plan</u>. The RSUs are subject to the terms and conditions set forth in this Agreement and the Plan, which is incorporated herein by reference. In the event of any inconsistency between the Plan and this Agreement, the terms of the Plan will control, unless it is expressly specified in in this Agreement or the Grant Notice that the specific provision of the Plan will not apply. For clarity, the foregoing sentence shall not limit the applicability of any additive language contained in this Agreement which provides supplemental or additional terms not inconsistent with the Plan.
- 1.3 <u>Unsecured Promise</u>. The RSUs will at all times prior to settlement represent an unsecured Company obligation payable only from the Company's general assets.

# ARTICLE II. VESTING; FORFEITURE AND SETTLEMENT

2.1 <u>Vesting; Forfeiture</u>. The RSUs will vest according to the vesting schedule in the Grant Notice except that any fraction of an RSU that would otherwise be vested will be accumulated and will vest only when a whole RSU has accumulated. Except as otherwise set forth in the Grant Notice, the Plan or this Agreement, and unless the Administrator otherwise determines, in the event of Participant's Termination of Service for any reason, all unvested RSUs will immediately and automatically be cancelled and forfeited (after taking into consideration any accelerated vesting which may occur in connection with such Termination of Service, if any).

#### 2.2 Settlement.

- (a) RSUs that vest will be paid in Shares as soon as administratively practicable after the vesting of the applicable RSU, but in no event later than sixty (60) days following the date on which the applicable RSU vests (or, in the case of any accelerated vesting that occurs on the Release Effective Date pursuant to the Grant Notice, no later than sixty (60) days following the date on which the applicable Termination of Service occurs).
- (b) Notwithstanding the foregoing, the Company may delay any payment under this Agreement that the Company reasonably determines would violate Applicable Law or an applicable provision of the Plan until the earliest date the Company reasonably determines the making of the payment will not cause such a violation (in accordance with Treasury Regulation Section 1.409A-2(b)(7)(ii)); *provided* the Company reasonably believes the delay will not result in the imposition of excise taxes under Section 409A.

# ARTICLE III. TAXATION AND TAX WITHHOLDING

3.1 <u>Representation</u>. Participant represents to the Company that Participant has reviewed with Participant's own tax advisors the tax consequences of this Award and the transactions contemplated by the Grant Notice and this Agreement. Participant is relying solely on such advisors and not on any statements or representations of the Company or any of its agents.

#### 3.2 Tax Withholding.

- (a) Unless the Administrator otherwise determines, the Company shall withhold, or cause to be withheld, Shares otherwise vesting or issuable under this Award (including the RSUs) in satisfaction of any applicable withholding tax obligations, in accordance with the Plan. The number of Shares which may be so withheld or surrendered shall be limited to the number of Shares which have a Fair Market Value on the date of withholding no greater than the aggregate amount of such liabilities based on the maximum individual statutory withholding rates in Participant's applicable jurisdictions for federal, state, local and foreign income tax and payroll tax purposes that are applicable to such taxable income, in accordance with Section 9.5 of the Plan.
- (b) Participant acknowledges that Participant is ultimately liable and responsible for all taxes owed in connection with the RSUs, regardless of any action the Company or any Subsidiary takes with respect to any tax withholding obligations that arise in connection with the RSUs. Neither the Company nor any Subsidiary makes any representation or undertaking regarding the treatment of any tax withholding in connection with the awarding, vesting or payment of the RSUs or the subsequent sale of Shares. The Company and its Subsidiaries do not commit and are under no obligation to structure the RSUs to reduce or eliminate Participant's tax liability.

# ARTICLE IV. OTHER PROVISIONS

- 4.1 <u>Adjustments</u>. Participant acknowledges that the RSUs and the Shares subject to the RSUs are subject to adjustment, modification and termination in certain events as provided in this Agreement and the Plan.
- 4.2 <u>Clawback</u>. The RSUs and the Shares issuable hereunder shall be subject to clawback or recoupment in accordance with this Section 4.2. In the event that the Administrator, in its good faith discretion, determines that Participant has committed an act that constitutes Cause and such act has resulted in or would reasonably be expected to result in material harm to the Company and/or its affiliates, the Board may seek <u>recoupment of up to the full amount of the RSUs and Shares issued upon settlement thereof and/or any proceeds received upon the sale of any such Shares. Nothing in this Section 4.2 shall limit the application of any clawback or recoupment policy in effect on the Grant Date or as may be adopted or maintained by the Company following the Grant Date, including the Dodd-Frank Wall Street Reform and Consumer Protection Act and any rules or regulations promulgated thereunder.</u>
- 4.3 <u>Notices</u>. Any notice to be given under the terms of this Agreement to the Company must be in writing and addressed to the Company in care of the Company's General Counsel at the Company's principal office or the General Counsel's then-current email address or facsimile number. Any notice to be given under the terms of this Agreement to Participant must be in writing and addressed to Participant (or, if Participant is then deceased, to the Designated Beneficiary) at Participant's last known mailing address, email address or facsimile number in the Company's personnel files. By a notice given pursuant to this Section, either party may designate a different address for notices to be given to that party. Any notice will

be deemed duly given when actually received, when sent by email, when sent by certified mail (return receipt requested) and deposited with postage prepaid in a post office or branch post office regularly maintained by the United States Postal Service, when delivered by a nationally recognized express shipping company or upon receipt of a facsimile transmission confirmation.

- 4.4 <u>Titles</u>. Titles are provided herein for convenience only and are not to serve as a basis for interpretation or construction of this Agreement.
- 4.5 <u>Conformity to Securities Laws</u>. Participant acknowledges that the Plan, the Grant Notice and this Agreement are intended to conform to the extent necessary with all Applicable Laws and, to the extent Applicable Laws permit, will be deemed amended as necessary to conform to Applicable Laws.
- 4.6 <u>Successors and Assigns</u>. The Company may assign any of its rights under this Agreement to single or multiple assignees, and this Agreement will inure to the benefit of the successors and assigns of the Company. Subject to the restrictions on transfer set forth in this Agreement or the Plan, this Agreement will be binding upon and inure to the benefit of the heirs, legatees, legal representatives, successors and assigns of the parties hereto.
- 4.7 <u>Limitations Applicable to Section 16 Persons</u>. Notwithstanding any other provision of the Plan or this Agreement, if Participant is subject to Section 16 of the Exchange Act, the Plan, the Grant Notice, this Agreement and the RSUs will be subject to any additional limitations set forth in any applicable exemptive rule under Section 16 of the Exchange Act (including any amendment to Rule 16b-3) that are requirements for the application of such exemptive rule. To the extent Applicable Laws permit, this Agreement will be deemed amended as necessary to conform to such applicable exemptive rule.
- 4.8 Entire Agreement. The Plan, the Grant Notice and this Agreement (including any exhibit hereto) constitute the entire agreement of the parties and supersede in their entirety all prior undertakings and agreements of the Company and Participant with respect to the subject matter hereof.
- 4.9 Severability. If any portion of the Grant Notice or this Agreement or any action taken under the Grant Notice or this Agreement, in any case is held illegal or invalid for any reason, the illegality or invalidity will not affect the remaining parts of the Grant Notice and/or this Agreement (as applicable), and the Grant Notice and/or this Agreement (as applicable) will be construed and enforced as if the illegal or invalid provisions had been excluded, and the illegal or invalid action will be null and void.
- 4.10 <u>Limitation on Participant's Rights</u>. Participation in the Plan confers no rights or interests other than as herein provided. This Agreement creates only a contractual obligation on the part of the Company as to amounts payable and may not be construed as creating a trust. Neither the Plan nor any underlying program, in and of itself, has any assets. Participant will have only the rights of a general unsecured creditor of the Company with respect to amounts credited and benefits payable, if any, with respect to the RSUs, and rights no greater than the right to receive cash or the Shares as a general unsecured creditor with respect to the RSUs, as and when settled pursuant to the terms of this Agreement.
- 4.11 <u>Not a Contract of Employment or Service</u>. Nothing in the Plan, the Grant Notice or this Agreement confers upon Participant any right to continue in the employ or service of the Company or any Subsidiary or interferes with or restricts in any way the rights of the Company and its Subsidiaries, which rights are hereby expressly reserved, to discharge or terminate the services of Participant at any time for any reason whatsoever, with or without cause, except to the extent expressly provided otherwise in a written agreement between the Company or a Subsidiary and Participant.

- 4.12 <u>Counterparts</u>. The Grant Notice may be executed in one or more counterparts, including by way of any electronic signature, subject to Applicable Law, each of which will be deemed an original and all of which together will constitute one instrument.
- 4.13 <u>Governing Law</u>. The Grant Notice and this Agreement will be governed by and interpreted in accordance with the laws of the State of Delaware, disregarding any state's choice-of-law principles requiring the application of a jurisdiction's laws other than the State of Delaware.

\*\*\*\*

#### **EXHIBIT C**

## **CONFIDENTIALITY AGREEMENT**

#### THE HYDRAFACIAL COMPANY

#### EMPLOYEE PROPRIETARY INFORMATION AND INVENTIONS ASSIGNMENT AGREEMENT

In consideration and as a condition of my employment by Edge Systems LLC d/b/a The HydraFacial Company (together with its parents and subsidiaries and any of their respective successors or assigns, the "Company"), and my receipt of the compensation paid to me by the Company pursuant to the employment agreement entered into between me and the Company (the "Employment Agreement") concurrently with the execution of this Employee Proprietary Information and Inventions Assignment Agreement (the "Agreement"), and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, effective as of the Effective Date (as defined in the Employment Agreement), I, the undersigned, agree as follows:

1. Proprietary Information. During the term of my employment with the Company, I may receive and otherwise be exposed, directly or indirectly, to confidential and proprietary information of the Company whether in graphic, written, electronic or oral form, including without limitation, information relating to the Company's business, strategies, designs, products, services and technologies and any derivatives, improvements and enhancements relating to any of the foregoing, or to the Company's suppliers, customers or business partners (collectively "Proprietary Information"). Proprietary Information may be identified at the time of disclosure as confidential or proprietary or information which by its context would reasonably be deemed to be confidential or proprietary. "Proprietary Information" may also include without limitation (i)(a) unpublished patent disclosures and patent applications and other filings, know-how, trade secrets, works of authorship and other intellectual property, as well as any information regarding ideas, Work Product (as defined below), technology, and processes, including without limitation assays, sketches, schematics, techniques, drawings, designs, descriptions, specifications and technical documentation, (b) specifications, protocols, models, designs, equipment, engineering, algorithms, software programs, software source documents, formulae, (c) information concerning or resulting from any research and development or other project, including without limitation, experimental work, and product development plans, regulatory compliance information, and research, development and regulatory strategies, and (d) business and financial information, including without limitation purchasing, procurement, manufacturing, customer lists, information relating to investors, employees, business and contractual relationships, business forecasts, sales and merchandising, business and marketing plans, product plans, and business strategies, including without limitation, information the Company provides regarding third parties, such as, but not limited to, suppliers, customers, employees, investors, or vendors; and (ii) any other information, to the extent such information contains, reflects or is based upon any of the foregoing Proprietary Information. The Proprietary Information may also include information of a third party that is disclosed to me by the Company or such third party at the Company's direction. Any information disclosed by any of the Company's affiliated companies or by any company, person or other entity participating with the Company in any consortium, partnership, joint venture or similar business combination, which would otherwise constitute Proprietary Information if disclosed by the Company, shall be deemed to constitute Proprietary Information under this Agreement, and the rights of the Company under this Agreement may be enforced by any such affiliate or participating entity (as well as by the Company) with respect to any violation relating to the Proprietary Information disclosed by such affiliate or entity, as if that affiliate or entity were also a party to this Agreement.

2. Obligations of Non-Use and Nondisclosure. I acknowledge the confidential and secret character of the Proprietary Information, and agree that the Proprietary Information is the sole, exclusive and valuable property of the Company. Accordingly, I agree not to use the Proprietary Information except in the performance of my authorized duties as an employee of the Company, and not to disclose all or any part of the Proprietary Information in any form to any third party, either during or after the term of my employment with the Company, without the prior written consent of the Company on a case-by-case basis, and to cooperate with the Company and use my best efforts not to prevent the unauthorized us or disclosure or reproductions of any Proprietary Information. In addition, I agree not to copy or remove any tangible materials containing Proprietary Information from the premises of the Company, except in the proper performance of my duties as an employee of the Company or with the prior written consent of the Company on a case-by-case basis. Upon termination of my employment with the Company, I agree to cease using and to return to the Company all whole and partial copies and derivatives of the Proprietary Information, whether in my possession or under my direct or indirect control, provided that I am entitled to retain my personal copies of (a) my compensation and benefits records, and (b) this Agreement. I understand that my obligations of nondisclosure with respect to Proprietary Information shall not apply to information that I can establish by competent proof (i) was actually in the public domain at the time of disclosure or enters the public domain following disclosure other than as a result of a breach of this Agreement, (ii) is already in my possession without breach of any obligations of confidentiality at the time of disclosure by the Company as shown by my files and records immediately prior to the time of disclosure, or (iii) is obtained by me from a third party not under confidentiality obligations and without a breach of any obligations of confidentiality. If I become compelled by law, regulation (including without limitation the rules of any applicable securities exchange), court order, or other governmental authority to disclose any Proprietary Information, I shall, to the extent possible and permissible under applicable law, first give notice to the Company. I agree to cooperate reasonably with the Company (at the Company's request) in any proceeding to obtain a protective order or other remedy. If such protective order or other remedy is not obtained, I shall only disclose that portion of such Proprietary Information required to be disclosed, in the opinion of my legal counsel. I shall request that confidential treatment be accorded such Proprietary Information, where available. Compulsory disclosures made pursuant to this section shall not relieve me of my obligations of confidentiality and non-use with respect to non-compulsory disclosures. If I learn of any possible unauthorized use or disclosure of Proprietary Information, I shall cooperate fully with the Company to enforce its rights in such information. Notwithstanding the foregoing or anything herein to the contrary, nothing contained herein shall prohibit me from (x) filing a charge with, reporting possible violations of federal law or regulation to, participating in any investigation by, or cooperating with any governmental agency or entity or making other disclosures that are protected under the whistleblower provisions of applicable law or regulation; (y) communicating directly with, cooperating with, or providing information (including trade secrets) in confidence to, any federal, state or local government regulator (including, but not limited to, the U.S. Securities and Exchange Commission, the U.S. Commodity Futures Trading Commission, or the U.S. Department of Justice) for the purpose of reporting or investigating a suspected violation of law, or from providing such information to my attorney or in a sealed complaint or other document filed in a lawsuit or other governmental proceeding; and/or (z) making disclosures that are protected by the National Labor Relations Act or similar applicable law.

- 3. <u>Defend Trade Secrets Act Notice of Immunity Rights</u>. I acknowledge that the Company has provided me with the following notice of immunity rights in compliance with the requirements of the Defend Trade Secrets Act: (a) I shall not be held criminally or civilly liable under any Federal or State trade secret law for the disclosure of Proprietary Information that is made in confidence to a Federal, State, or local government official or to an attorney solely for the purpose of reporting or investigating a suspected violation of law, (b) I shall not be held criminally or civilly liable under any Federal or State trade secret law for the disclosure of Proprietary Information that is made in a complaint or other document filed in a lawsuit or other proceeding, if such filing is made under seal and (c) if I file a lawsuit for retaliation by the Company for reporting a suspected violation of law, I may disclose the Proprietary Information to my attorney and use the Proprietary Information in the court proceeding, if I file any document containing the Proprietary Information under seal, and do not disclose the Proprietary Information, except pursuant to court order.
- 4. <u>Property of the Company</u>. I acknowledge and agree that all notes, memoranda, reports, drawings, blueprints, manuals, materials, data, emails and other papers and records of every kind, or other tangible or intangible materials which shall come into my possession in the course of my employment with the Company, relating to any Proprietary Information, shall be the sole and exclusive property of the Company and I hereby assign any rights or interests I may obtain in any of the foregoing. I agree to surrender this property to the Company immediately upon termination of my employment with the Company, or at any time upon request by the Company. I further agree that any property situated on the Company's data systems or on the Company's premises and owned by the Company, including without limitation electronic storage media, filing cabinets or other work areas, is subject to inspection by Company personnel at any time with or without notice. I further agree that in the event of termination of my employment with the Company I will execute a Termination Certificate substantially in the form attached hereto as <u>Exhibit A</u>.

## 5. Inventions.

(a) <u>Disclosure and Assignment of Inventions</u>. For purposes of this Agreement, an "<u>Invention</u>" shall mean any idea, invention or work of authorship, including, without limitation, any documentation, formula, design, device, code, method, software, technique, process, discovery, concept, improvement, enhancement, development, machine or contribution, in each case whether or not patentable or copyrightable. I will disclose all Inventions promptly in writing to an officer of the Company or to attorneys of the Company in accordance with the Company's policies and procedures, I will, and hereby do, assign to the Company, without requirement of further writing, without royalty or any other further consideration, my entire right, title and interest throughout the world in and to all Inventions created, conceived, made, developed, and/or reduced to practice by me in the course of my employment by the Company and all intellectual property rights therein. I hereby waive, and agree to waive, any moral rights I may have in any copyrightable work I create or have created on behalf of the Company. I also hereby agree, that for a period of one year after my employment with the Company, I shall disclose to the Company any Inventions that I create, conceive, make, develop, reduce to practice or work on that relate to the work I performed for the Company. The Company agrees that it will use commercially reasonable measures to keep Inventions disclosed to it pursuant to this Section 5.1 that do not constitute Inventions are assigned or assignable to the Company pursuant to this Section 5.1 or otherwise.

- (b) Certain Exemptions. The obligations to assign Inventions set forth in Section 5.1 apply with respect to all Inventions (a) whether or not such Inventions are conceived, made, developed or worked on by me during my regular hours of employment with the Company; (b) whether or not the Invention was made at the suggestion of the Company; (c) whether or not the Invention was reduced to drawings, written description, documentation, models or other tangible form; and (d) whether or not the Invention is related to the general line of business engaged in by the Company, but do not apply to Inventions that (x) I develop entirely on my own time or after the date of this Agreement without using the Company's equipment, supplies, facilities or Proprietary Information; (y) do not relate to the Company's business, or actual or demonstrably anticipated research or development of the Company at the time of conception or reduction to practice of the Invention; and (z) do not result from and are not related to any work performed by me for the Company. I hereby acknowledge and agree that the Company has notified me that, if I reside in the state of California, assignments provided for in Section 5.1 do not apply to any Invention which qualifies fully for exemption from assignment under the provisions of Section 2870 of the California Labor Code ("Section 2870"), a copy of which is attached as Exhibit B. If applicable, at the time of disclosure of an Invention that I believe qualifies under Section 2870. I further understand that, to the extent this Agreement shall be construed in accordance with the laws of any state which precludes a requirement in an employee agreement to assign certain classes of inventions made by an employee, Section 5.1 shall be interpreted not to apply to any Invention which a court rules and/or the Company agrees falls within such classes.
- (c) Records. I will make and maintain adequate and current written records of all Inventions covered by Section 5.1. These records may be in the form of notes, sketches, drawings, flow charts, electronic data or recordings, notebooks and any other format. These records shall be and remain the property of the Company at all times and shall be made available to the Company at all times.
- (d) Patents and Other Rights. I agree to assist the Company in obtaining, maintaining and enforcing patents, invention assignments and copyright assignments, and other proprietary rights in connection with any Invention covered by Section 5.3, and will otherwise assist the Company as reasonably required by the Company to perfect in the Company the rights, title and other interests in my work product granted to the Company under this Agreement (both in the United States and foreign countries). I further agree that my obligations under this Section 5.4 shall continue beyond the termination of my employment with the Company, but if I am requested by the Company to render such assistance after the termination of such employment, I shall be entitled to a fair and reasonable rate of compensation for such assistance, and to reimbursement of any expenses incurred at the request of the Company relating to such assistance. If the Company is unable for any reason, after reasonable effort, to secure my signature on any document needed in connection with the actions specified above, I hereby irrevocably designate and appoint the Company and its duly authorized officers and agents as my agent and attorney in fact, which appointment is coupled with an interest, to act for and in my behalf to execute, verify and file any such documents and to do all other lawfully permitted acts to further the purposes of this Section 5.4 with the same legal force and effect as if executed by me.
- (e) Prior Contracts Inventions; Information Belonging to Third Parties. I represent and warrant that, except as set forth on Exhibit C, I am not required, and I have not been required during the course of work for the Company or its predecessors, to assign Inventions under any other contracts that are now or were previously in existence between me and any other person or entity. I further represent that (i) I am not obligated under any consulting, employment or other agreement that would affect the Company's rights or my duties under this Agreement, and I shall not enter into

any such agreement or obligation during the period of my employment by the Company, (ii) there is no action, investigation, or proceeding pending or threatened, or any basis therefor known to me involving my prior employment or any consultancy or the use of any information or techniques alleged to be proprietary to any former employer, and (iii) the performance of my duties as an employee of the Company do not and will not breach, or constitute a default under any agreement to which I am bound, including any agreement limiting the use or disclosure of proprietary information acquired in confidence prior to engagement by the Company or if applicable, any agreement to refrain from competing, directly or indirectly, with the business of such previous employer or any other party or to refrain from soliciting employees, customers or suppliers of such previous employer or other party. I will not, in connection with my employment by the Company, use or disclose to the Company any confidential, trade secret or other proprietary information of any previous employer or other person to which I am not lawfully entitled. As a matter of record, I attach as Exhibit C a brief description of all Inventions made or conceived by me prior to my employment with the Company which I desire to be excluded from this Agreement ("Background Technology"). If full disclosure of any Background Technology would breach or constitute a default under any agreement to which I am bound, including any agreement limiting the use or disclosure of proprietary information acquired in confidence prior to engagement by the Company, I understand that I am to describe such Background Technology in Exhibit C at the most specific level possible without violating any such prior agreement. Without limiting my obligations or representations under this Section 5.5, if I use any Background Technology in the course of my employment or incorporate any Background Technology in any product, service or other offering of the Company, I hereby grant the Company a non-exclusive, royalty-free, perpetual and irrevocable, worldwide right to use and sublicense the use of Background Technology for the purpose of developing, marketing, selling and supporting Company technology, products and services, either directly or through multiple tiers of distribution, but not for the purpose of marketing Background Technology separately from Company products or services.

- (f) Works Made for Hire. I acknowledge that all original works of authorship which are made by me (solely or jointly with others) within the scope of my employment with the Company and which are eligible for copyright protection are "works made for hire" as that term is defined in the United States Copyright Act (17 U.S.C., Section 101).
- 6. <u>Restrictive Covenants</u>. I agree to fully comply with the covenants set forth in this Section 6 (the "<u>Restrictive Covenants</u>"). I further acknowledge and agree that the Restrictive Covenants are reasonable and necessary to protect the Company's legitimate business interests, including its Proprietary Information and goodwill.
- (a) Non-Competition. During the term of my employment by the Company, I will not without the prior written approval of the Board of Directors of the Company, (a) engage in any other professional employment or consulting, or (b) directly or indirectly participate in or assist any business (whether as owner, partner, officer, director, employee, consultant, investor, lender or otherwise, except as the holder of not more than 1% of the outstanding stock of a publicly-held company), in each case, which is a current or potential supplier, customer or competitor of the Company, including but not limited to any business or enterprise that develops, manufactures, markets, licenses, sells or provides any product or service that competes with any product or service developed, manufactured, marketed, licensed, sold or provided, or planned to be developed, manufactured, marketed, licensed, sold or provided, by the Company while I was employed by the Company.

#### (b) Non-Solicitation.

- (i) During the term of my employment with the Company and for a period of one (1) year thereafter I will not (i) solicit for employment or engagement, any employee, consultant or independent contractor of the Company, or (ii) solicit, encourage, induce or attempt to induce or assist others to solicit, encourage, induce or attempt to induce any employees, consultants or independent contractors of the Company who were employed or engaged by the Company at any time during the term of my employment with the Company to terminate their employment or engagement with the Company.
- (ii) During the term of my employment with the Company, I will not solicit, divert or take away, or attempt to divert or take away, the business of any customer or client of the Company (served by the Company during the twelve (12)-month period prior to the termination of my employment with the Company) on my own behalf or on behalf of any person or entity other than the Company.
- (c) <u>No Defamatory Communications</u>. During the term of my employment with the Company and thereafter, I agree that I will not make any public or private statement which would reasonably be expected to defame or disparage the Company or any of its employees, officers, managers or directors. Notwithstanding the foregoing, this Section 6.3 shall not preclude me from making any statement to the extent required by law or legal process.
- (d) <u>Extension</u>. Without limiting the Company's ability to seek other remedies available in law or equity, if I violate the provisions of Section 6.2(a), I shall continue to be bound by the restrictions set forth in such section until a period of one year has expired without any violation of such provisions.
- (e) <u>Interpretation</u>. If any restriction set forth in Section 6.2 is found by any court of competent jurisdiction to be unenforceable because it extends for too long a period of time or over too great a range of activities or in too broad a geographic area, it shall be interpreted to extend only over the maximum period of time, range of activities or geographic area as to which it may be enforceable.
- 7. <u>Notification to Other Parties</u>. In the event of termination of my employment with the Company, I hereby consent to notification by the Company to my new employer or other party for whom I work about my rights and obligations under this Agreement.
- 8. Employment at Will. I understand and agree that my employment with the Company is at will. Accordingly, my employment can be terminated for any lawful reason or for no reason, without cause or notice, at my option or the Company's option, subject to and except as otherwise expressly set forth in the Employment Agreement. The Restrictive Covenants will remain in effect for the periods specified in this Agreement, unless such Restrictive Covenants are modified by a further written agreement signed by both an authorized officer of the Company and me which expressly alters such Restrictive Covenants.

## 9. Miscellaneous.

(a) The parties' rights and obligations under this Agreement will bind and inure to the benefit of their respective successors, heirs, executors, and administrators and permitted assigns. I will not assign this Agreement or my obligations hereunder without the prior written consent of the Company, which consent may be withheld in the Company's sole discretion, and any such purported

assignment without consent shall be null and void from the beginning. I agree that the Company may freely assign this Agreement to any affiliate or successor in interest, including any person or entity that, whether by way of merger, sale, acquisition, corporate re-organization or otherwise, directly or indirectly acquires all or substantially all of the business or assets of the Company.

- (b) This Agreement, together with the Employment Agreement, constitutes the parties' final, exclusive and complete understanding and agreement with respect to the subject matter hereof, and supersedes all prior and contemporaneous understandings and agreements, whether oral or written, relating to its subject matter.
- (c) Any subsequent change or changes in my duties, obligations, rights or compensation will not affect the validity or scope of this Agreement. This Agreement may not be waived, modified or amended unless mutually agreed upon in writing by both parties. No delay or omission by the Company in exercising any right under this Agreement will operate as a waiver of that or any other right. A waiver or consent given by the Company on any one occasion is effective only in that instance and will not be construed as a bar to or waiver of any right on any other occasion.
- (d) The provisions of this Agreement are severable, and the invalidity or unenforceability of any provision(s) of this Agreement shall not impact the validity or enforceability of any other provision(s) of this Agreement, which shall remain in full force and effect.
- (e) I acknowledge that the Company will suffer substantial damages not readily ascertainable or compensable in terms of money in the event of the breach of any of my obligations under this Agreement. I therefore agree that the Company shall be entitled (without limitation of any other rights or remedies otherwise available to the Company) to obtain an injunction from any court of competent jurisdiction prohibiting the continuance or recurrence of any breach of this Agreement. I also agree that if the Company prevails in any action or proceeding to enforce my obligations under this Agreement, I will pay all of the Company's expenses relating to any such action or proceeding including, without limitation, all reasonable attorney's fees, if so authorized by applicable state and/or federal law.
- (f) The rights and obligations of the parties under this Agreement shall be governed in all respects by the laws of the State of California exclusively, without reference to any conflict laws rule that would result in the application of the laws of any other jurisdiction. I agree that upon the Company's request, all disputes arising hereunder shall be adjudicated in the state and federal courts having jurisdiction over disputes arising in Los Angeles County, California, and I hereby agree to consent to the personal jurisdiction of such courts. The Company and I each hereby irrevocably waive any right to a trial by jury in any action, suit or other legal proceeding arising under or relating to any provision of this Agreement.
- (g) Any notices required or permitted hereunder shall be given to the appropriate party at the address specified on the signature page to this Agreement or at such other address as the party shall specify in writing. Such notice shall be deemed given upon personal delivery, or sent by certified or registered mail, postage prepaid, three (3) days after the date of mailing.
- (h) Except as otherwise provided herein, the provisions of this Agreement shall survive the termination of my employment with the Company for any reason.

(i) This Agreement may be executed in any number of counterparts, each of which shall be deemed to be an original and all of which together shall be deemed to be one and the same instrument. A facsimile, PDF (or any electronic signature complying with the U.S. federal ESIGN Act of 2000, e.g., www.docusign.com) or any other type of copy of an executed version of this Agreement signed by a party is binding upon the signing party to the same extent as the original of the signed agreement.

I ACKNOWLEDGE THAT I HAVE HAD THE OPPORTUNITY TO SEEK THE ADVICE OF INDEPENDENT LEGAL COUNSEL PRIOR TO SIGNING THIS AGREEMENT, AND THAT I HAVE EITHER CONSULTED WITH OR ON MY OWN VOLITION CHOSEN NOT TO CONSULT WITH SUCH COUNSEL. I FURTHER ACKNOWLEDGE THAT I HAVE READ THIS AGREEMENT CAREFULLY AND I UNDERSTAND AND ACCEPT THE OBLIGATIONS WHICH IT IMPOSES UPON ME WITHOUT RESERVATION. NO PROMISES OR REPRESENTATIONS HAVE BEEN MADE TO ME TO INDUCE ME TO SIGN THIS AGREEMENT. I SIGN THIS AGREEMENT VOLUNTARILY AND FREELY, IN DUPLICATE, WITH THE UNDERSTANDING THAT THE COMPANY WILL RETAIN ONE COUNTERPART AND THE OTHER COUNTERPART WILL BE RETAINED BY ME.

(Signature Page Follows)

N WITNESS WHEREOF, I have executed this document as of	. 20 .	
	Employee:	
	Address:	
	_	
GREED AND ACKNOWLEDGED:		
	EDGE SYSTEM	MS LLC d/b/a
	THE HYDRAI	FACIAL COMPANY
	By:	
	Name:	
	Title:	
		The HydraFacial Company 2165 E Spring Street

#### Exhibit A

## **Termination Certificate**

I, the undersigned, hereby certify that I do not have in my possession, nor have I failed to return, any documents or materials relating to the business of The HydraFacial Company or its affiliates (together, the "Company"), or copies thereof, including, without limitation, any item of Proprietary Information listed in Section 4 of the Company's Employee Proprietary Information and Inventions Assignment Agreement (the "Agreement") to which I am a party, but not including copies of my own compensation and benefits records (in each case, to the extent expressly permitted by the Agreement).

I further certify that I have complied with all of the terms of the Agreement signed by me. I further agree that in compliance with the Agreement, I will preserve as confidential any information relating to the Company or any of it business partners, clients, consultants or licensees which has been disclosed to me in confidence during the course of my employment by the Company unless authorized in writing to disclose such information (i) by an executive officer of the Company, in the event that I am not an executive officer of the Company, or (ii) by the Board of Directors of the Company, in the event that I am an executive officer of the Company. I understand that nothing herein is intended to or shall prevent me from communicating directly with, cooperating with, or providing information to, any federal, state or local government regulator, including, but not limited to, the U.S. Securities and Exchange Commission, the U.S. Commodity Futures Trading Commission, or the U.S. Department of Justice.

Date:	
	(Employee's Signature)
	(Printed or Typed Name of Employee)

#### Exhibit B

## California Labor Code

California Labor Code § 2870. Application of provision providing that employee shall assign or offer to assign rights in invention to employer.

- (a) Any provision in an employment agreement which provides that an employee shall assign, or offer to assign, any of his or her rights in an invention to his or her employer shall not apply to an invention that the employee developed entirely on his or her own time without using the employer's equipment, supplies, facilities, or trade secret information except for those inventions that either:
  - (1) Relate at the time of conception or reduction to practice of the invention to the employer's business, or actual or demonstrably anticipated research or development of the employer; or
  - (2) Result from any work performed by the employee for the employer.
- (b) To the extent a provision in an employment agreement purports to require an employee to assign an invention otherwise excluded from being required to be assigned under subdivision (a), the provision is against the public policy of this state and is unenforceable.

# Exhibit C

# Background Technology

List here prior contracts to assign Inventions that are now in existence between any other person or entity and you.

List here previous Inventions which you desire to have specifically excluded from the operation of this Agreement. Continue on reverse side if necessary.

#### **EXHIBIT D**

#### **RELEASE**

#### GENERAL RELEASE OF ALL CLAIMS

This General Release of all Claims (this "*Release*") is entered into on [•], 20[•] by and between Vesper Healthcare Acquisition Corp. ("*Parent*"), Edge Systems LLC d/b/a The HydraFacial Company (the "*Company*"), and [•] ("*Executive*"). In consideration of the payments and benefits set forth in the Employment Agreement (the "*Employment Agreement*") between Executive, Parent and the Company, effective May 4, 2021 (the "*Effective Date*"), to which Executive first became legally entitled following the Effective Date, Executive agrees as follows:

#### 1. General Release and Waiver of Claims.

- (a) For valuable consideration, the receipt and adequacy of which are hereby acknowledged, Executive and each of Executive's respective heirs, executors, administrators, representatives, agents, successors, assigns and representatives (the "*Releasor*") hereby irrevocably and unconditionally releases and forever discharges each of Parent, the Company, and their respective partners, subsidiaries, associates, affiliates, successors, heirs, assigns, agents, directors, officers, employees, representatives, lawyers, insurers, and all persons acting by, through, under or in concert with them (collectively, the "*Releasees*"), of and from any and all manner of action or actions, cause or causes of action, judgments, obligations, in law or in equity, suits, debts, liens, contracts, agreements, promises, liability, claims, demands, damages, losses, costs, attorneys' fees or expenses, of any nature whatsoever, known or unknown, fixed or contingent, which Releasor now has, ever had or may hereafter have against any Releasee, by reason of any act, omission, practice, conduct, event, cause, or other matter whatsoever from the beginning of time up to and including the date that Executive executes this Release, to the fullest extent permitted by law.
- (b) Without limiting the generality of the foregoing, Releasor releases and discharges Releasees from any and all claims in any way arising out of, based upon, or related to Executive's employment with Parent and/or the Company, the termination of employment of Executive by Parent and/or the Company and/or the events surrounding the circumstances relating to that termination, including, but not limited to:
  (i) any and all claims arising under tort, contract and quasi-contract law, including, but not limited to, claims of breach of contract (express or implied), tortious interference with contract or prospective business advantage, breach of the covenant of good faith and fair dealing, promissory estoppel, detrimental reliance, invasion of privacy, wrongful or retaliatory discharge, fraud, defamation, slander, libel, negligent or intentional infliction of emotional distress or compensatory or punitive damages; (ii) any and all claims for monetary or equitable relief, including, but not limited to, attorneys' fees, back pay, front pay, reinstatement, experts' fees, medical fees or expenses, costs, and disbursements; and (iii) and any claim under Title VII of the Civil Rights Act of 1964, the Age Discrimination In

Employment Act ("ADEA"), the Americans With Disabilities Act, the Family and Medical Leave Act, the Equal Pay Act, the False Claims Act, the Employee Retirement Income Security Act, the Federal Worker Retraining and Notification Act, the Fair Labor Standards Act, the Civil Rights Act of 1991, Section 1981 of U.S.C. Title 42, the Sarbanes-Oxley Act of 2002, the California Fair Employment and Housing Act, the California Unfair Competition Law, the California Equal Pay Law, the Moore-Brown-Roberti Family Rights Act of 1991, the California Labor Code, the California Worker Adjustment and Retraining Notification Act, California Wage and Hour laws, the California False Claims Act, the California Constitution and the California Corporate Criminal Liability Act, and any other federal, state or local law or ordinance prohibiting employment discrimination, harassment or retaliation. This Release does not release claims arising after the date Executive executes this Release, nor claims that cannot be released as a matter of law, including, but not limited to, Executive's right to file a charge with or participate in a charge by the Equal Employment Opportunity Commission ("EEOC"), the National Labor Relations Board ("NLRB"), or any other local, state, or federal administrative body or government agency that is authorized to enforce or administer laws related to employment, against Parent and/or the Company (with the understanding that any such filing or participation does not give Executive the right to recover any monetary damages against Parent and/or the Company; Executive's release of claims herein bars Executive from recovering such monetary relief from Parent and/or the Company before the EEOC, NLRB, or other administrative body). Notwithstanding the foregoing, this Release does not apply to (i) any lawsuit brought to challenge the validity of this Release under the ADEA, (ii) payments or benefits under Article IV of the Employment Agreement, which payments and benefits (among other good and valuable consideration) are provided in exchange for this Release, (iii) any claims for indemnification arising under any applicable indemnification obligation of Parent and/or the Company, (iv) accrued or vested benefits under any applicable Parent and/or Company employee benefit plan (within the meaning of Section 3(3) of the Employment Retirement Income Security Act) and (v) any rights as a shareholder of the Parent or pursuant to any option, restricted stock unit or other equity compensation award agreement or plan.

- (c) Executive acknowledges that Executive has been advised by legal counsel and is familiar with the provisions of California Civil Code Section 1542, which provides as follows:
  - "A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS THAT THE CREDITOR OR RELEASING PARTY DOES NOT KNOW OR SUSPECT TO EXIST IN HIS OR HER FAVOR AT THE TIME OF EXECUTING THE RELEASE AND THAT, IF KNOWN BY HIM OR HER, WOULD HAVE MATERIALLY AFFECTED HIS OR HER SETTLEMENT WITH THE DEBTOR OR RELEASED PARTY."

Executive, being aware of, understanding and acknowledging the significance and consequence of specifically waiving California Civil Code Section 1542, hereby expressly waives and relinquishes all rights and benefits Executive may have thereunder, as well as any other applicable statutes or common law principles of similar effect, in order to effect a full and complete general release as described above. Thus, notwithstanding the provisions of California Civil Code Section 1542, and to implement a full and complete release, Executive expressly acknowledges this Release is intended to include in its effect, without limitation, all claims he does not know or suspect to exist in his favor at the time of signing this Release, and that this Release contemplates the extinguishment of any such claims.

- 2. <u>Consideration and Revocation Period</u>. By signing this Release, Executive represents and warrants that:
  - (a) Under the Federal Age Discrimination in Employment Act of 1967, as amended, and the applicable rules and regulations promulgated thereunder, Parent and the Company advise Executive that he should consult with independent counsel before executing this Release; and Executive acknowledges that he has been so advised. Executive further acknowledges that he has had at least [twenty-one (21)][forty-five (45)] days to consider this Release before signing it and Executive further acknowledges that if he signs this Release prior to the expiration of the [twenty-one (21)][forty-five (45)] day period, Executive waives the remainder of that period.
  - (b) Executive acknowledges that he has carefully read this Release in its entirety; that he has had an adequate opportunity to consider it; that he understands all its terms; and that he knowingly and voluntarily assents to all the terms and conditions contained herein, including, without limitation, the waiver and release contained herein.
  - (c) Executive further acknowledges that he has seven (7) calendar days following the date he signs this Release to revoke it and this Release shall not become effective until the eighth day following the date on which Executive signs this Release. Executive understands that if he wishes to revoke this Release, Executive must deliver written notice of revocation [(which may be by email)], stating Executive's intent to revoke this Release on or before 5:00 p.m. (PST) of the seventh (7th) day after the date on which Executive signs this Release to [TITLE], at [ADDRESS]. Executive acknowledges that if Executive revokes this Release, Executive will not receive any payments or benefits pursuant to Article IV of the Employment Agreement.
- 3. No Assignment. Executive represents and warrants that there has been no assignment or other transfer of any interest in any claim released hereunder which Executive may have against each Releasee and Executive agrees to indemnify and hold each Releasee harmless from any liability, claims, demands, damages, costs, expenses and attorneys' fees incurred by any Releasee as the result of any such assignment or transfer or any rights or claims under any such assignment or transfer. It is the intention of the parties that this indemnity does not require payment as a condition precedent to recovery by any Releasee against Executive under this indemnity

- 4. <u>Proceedings</u>. Executive agrees that if Executive hereafter commences any suit arising out of, based upon, or relating to any of the claims released hereunder or in any manner asserts against any Releasee any of the claims released hereunder, then Executive agrees to pay to such Releasee, in addition to any other damages caused to such Releasee thereby, all attorneys' fees incurred by such Releasee in defending or otherwise responding to said suit or claim. Notwithstanding the foregoing, the foregoing sentence shall not apply to the extent such attorneys' fees are attributable to Executive's good faith challenge to or a request for declaratory relief with respect to the validity of the waiver herein under the ADEA.
- 5. <u>Nonadmission</u>. Executive further understands and agrees that neither the payment of any sum of money nor the execution of this Release shall constitute or be construed as an admission of any liability whatsoever by any of the Releasees, all of whom have consistently taken the position that they have no liability whatsoever to Executive.
- 6. <u>Confidential Information</u>. Executive acknowledges and agrees that Executive is bound by that certain Confidentiality Agreement (as defined in the Employment Agreement). Executive hereby reaffirms the covenants, terms and conditions set forth in the Confidentiality Agreement, and acknowledges and agrees that the Confidentiality Agreement remains in full force and effect in accordance with its terms
- 7. <u>Severability.</u> In the event any provision or part of this Release is found to be invalid or unenforceable, only that particular provision or part so found, and not the entire Release, will be inoperative
- 8. <u>Governing Law</u>. This Release shall be governed by and construed in accordance with the laws of the State of California, without regard to conflicts of laws principles thereof.

EXECUTIVE ACKNOWLEDGES THAT HE HAS READ THIS RELEASE AGREEMENT AND THAT HE FULLY KNOWS, UNDERSTANDS AND APPRECIATES ITS CONTENTS, AND THAT HE HEREBY EXECUTES THE SAME AND MAKES THIS RELEASE AGREEMENT AND THE RELEASE AND AGREEMENTS PROVIDED FOR HEREIN VOLUNTARILY AND OF HIS OWN FREE WILL.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

N WITNESS WHEREOF, the undersigned has executed this General Release of all Claims this	day of	20	
			[Name

April 29, 2021

Daniel Watson dwatson@hydrafacial.com

Dear Dan.

As you know, on December 8, 2020, The Beauty Health Company (formerly known as Vesper Healthcare Acquisition Corp.) (as may be referred to herein together with its subsidiaries, the "*Company*"), LCP Intermediate, Inc., the indirect parent of Edge Systems LLC d/b/a The HydraFacial Company ("*Hydrafacial*"), and certain other parties entered into an Agreement and Plan of Merger whereby (among other things) Hydrafacial will become an indirect wholly owned subsidiary of the Company (the "*Transaction*").

In connection with the Transaction, the Company desires to employ you under the following terms, subject to the consummation of the Transaction, effective as of the date on which the Transaction is consummated (the "Effective Date"):

- 1. <u>Title</u>; <u>Duties</u>; <u>Location</u>. You will serve as Executive Vice President of Americas Sales of the Company and will perform such duties as are usual and customary for your position or as otherwise may be prescribed by the Company. You will continue to work out of Long Beach, CA, except for travel as may be necessary to fulfill your duties to the Company.
- 2. <u>Base Compensation</u>. Your base compensation will be \$371,196.54 per year (your "*Base Salary*")), payable in accordance with the Company's standard payroll practices (but no less often than monthly).
- 3. <u>Annual Bonus</u>. For each fiscal year of the Company ending during your employment with the Company (commencing with 2021), you will be eligible to participate in, and earn a cash performance bonus (an "*Annual Bonus*") under, the Company's Management Incentive Plan or any successor plan (the "*Bonus Plan*"). Your target Annual Bonus will be 60% of your Base Salary. The actual amount of any Annual Bonus will be determined based on the satisfaction of Company, individual and/or other performance metrics established by the Board of Directors of the Company (the "*Board*") or the Compensation Committee thereof (the "*Compensation Committee*"). The Bonus Plan may be amended or changed by the Company from time to time. Any Annual Bonus that becomes payable to you will be paid at such time(s) as annual bonuses are generally paid to senior leaders of the Company with respect to the applicable fiscal year, but in no event later than March 15<sup>th</sup> of the calendar year following the year to which the Annual Bonus relates, subject to your continued employment with the Company through the applicable payment date.
- 4. Option. As soon as practicable following the Effective Date, subject to and contingent on the approval of the Company's 2021 Incentive Award Plan (the "Incentive Plan") by the Board and the stockholders of the Company, you will be eligible to receive a grant of options to purchase 310,000 shares of the Company's common stock (the "Option") pursuant to the Incentive Plan. The grant of the Option is further subject to and contingent upon the approval by the Board or the Compensation Committee of the Option. If granted, the Option shall have an exercise price per share equal to the Fair Market Value (as defined in the Incentive Plan) of a share of the Company common stock on the grant date. Twenty-five percent (25%) of the Shares subject to the Option will vest on each of the first four (4) anniversaries of the

Effective Date, subject to your continued employment with the Company through the applicable vesting date. The Option will be subject to the terms and conditions of the Incentive Plan and an award agreement, substantially in the form attached hereto as Exhibit A (the "Option Agreement"), to be entered into between the Company and you.

- 5. <u>Equity Awards</u>. Without limiting Section 4 above, for each fiscal year of your employment with the Company, beginning in 2022, you will be eligible for one or more grants of long-term incentive awards ("*Awards*"). The amount (if any), type and terms and conditions (including vesting conditions) of any such Award shall be determined by the Compensation Committee in its sole discretion. Each Award will be subject to the terms and conditions of the Incentive Plan, subject to and contingent on the approval of the Incentive Plan by the Board and the stockholders of the Company, or any successor plan, and an award agreement entered into between the Company and you.
- 6. <u>Benefits</u>. During your employment with the Company, you will continue to be eligible to participate in the Company's employee benefit plans and programs generally applicable to similarly-situated employees of the Company, as in effect from time to time, subject to the terms and conditions of the applicable plans and programs. These plans and programs currently include the Company's permissive paid time off plan, health and welfare plans and a 401(k) retirement savings plan. The Company reserves the right to change, discontinue or amend its benefits and benefit plans at any time and from time to time.
- 7. <u>Confidentiality Agreement</u>. As a condition of your continued employment with the Company, you are required to execute and return to the Company, and to comply with, the Employee Proprietary Information and Inventions Assignment Agreement attached hereto as <u>Exhibit B</u> (the "Confidentiality Agreement").
- 8. <u>Severance Plan</u>. It is anticipated that the Company will adopt the Executive Severance Plan (the "Severance Plan"), in substantially the form attached hereto as <u>Exhibit C</u>, effective on or shortly following the Effective Date. As a senior leader of the organization, if the Severance Plan is adopted, you will be designated as a "Tier One" participant in the Severance Plan. As a participant in the Severance Plan, you would be eligible to receive certain severance payments and benefits upon your Qualifying Termination or CIC Termination (each as defined in the Severance Plan) in accordance with the terms of the Severance Plan as in effect from time to time, subject to the requirements, terms and conditions set forth therein.
- 9. <u>Policies</u>. As an employee of the Company, you will continue to be subject to, and you agree to comply with, the Company's policies and procedures as they may be amended from time to time.
- 10. <u>At-Will Employment</u>. Your employment with the Company is "at-will" and is not for a specified period of time. Either you or the Company may terminate your employment at any time and for any reason whatsoever (or for no reason). It also means the Company may prospectively change your title, reporting line, duties, or other terms of your employment at any time in its discretion. This at-will employment relationship may not be changed, modified, rescinded or superseded, except pursuant to a written agreement executed by you and the Company.

- 11. Withholding. The Company may withhold from any amounts payable under this letter such federal, state, local or foreign taxes as are required to be withheld pursuant to any applicable law or regulation.
- 12. <u>Effectiveness</u>. The effectiveness of this offer letter is subject to and conditioned upon the closing of the Transaction. If the Transaction does not close for any reason, this offer letter will be null and void and neither party will have any rights or obligations hereunder.
- 13. <u>Representations</u>. You hereby represent and warrant to the Company that: (i) you have consulted with or have had the opportunity to consult with independent counsel of your own choice concerning this offer letter and have been advised to do so by the Company; (ii) you have read and understand this letter, are fully aware of its legal effect, and have entered into it freely based on your own judgment and (iii) your entry into this offer letter and performance of services for the Company will not violate any noncompetition, restrictive covenant or other contract between you and any prior employer.
- 14. <u>Miscellaneous</u>. This offer letter, together with the Option Agreement and the Confidentiality Agreement, constitutes the final, complete and exclusive agreement between you and the Company with respect to the subject matter hereof and supersedes all prior understandings and agreements concerning such subject matter, whether written or oral, including, without limitation, any prior service agreement between you and the Company, its affiliates or any predecessor employer and any prior versions of this offer letter. No provisions of this offer letter may be amended, modified, or waived unless agreed to in writing and signed by you and by a duly authorized officer of the Company. No waiver by either party of any breach by the other party of any condition or provision of this Agreement shall be deemed a waiver of similar or dissimilar provisions or conditions at the same or at any prior or subsequent time. This offer letter shall be governed by and construed in accordance with the laws of the State of California, without regard to principles of conflicts of law thereof.

I look forward to you continued contributions and look forward to receiving your acceptance by May 3, 2021. Please let me know if you have any questions about this offer.

Sincerely,

/s/ Clinton E. Carnell

Clint Carnell President & Chief Executive Officer The Beauty Health Company

/s/ Clinton E. Carnell Clint Carnell Chief Executive Officer

Edge Systems LLC d/b/a The HydraFacial Company

Accepted and agreed to on April 29, 2021

/s/ Daniel Watson

Signature

# **Daniel Watson**

# Attachments:

Exhibit A: Form of Option Agreement Exhibit B: Employee Proprietary Information and Inventions Assignment Agreement Exhibit C: Proposed Executive Severance Plan

## **EXHIBIT A**

## FORM OF OPTION AGREEMENT

THE DEALTHY HE ALTHY COMPANY
THE BEAUTY HEALTH COMPANY
2021 INCENTIVE AWARD PLAN
STOCK OPTION OF ANT NOTICE
STOCK OPTION GRANT NOTICE

The Beauty Health Company, a Delaware corporation (the "Company") has granted to the participant listed below ("Participant") the stock option (the "Option") described in this Stock Option Grant Notice (the "Grant Notice"), subject to the terms and conditions of The Beauty Health Company 2021 Incentive Award Plan (as amended from time to time, the "Plan") and the Stock Option Agreement attached hereto as Exhibit A (the "Agreement"), both of which are incorporated into this Grant Notice by reference. Capitalized terms not specifically defined in this Grant Notice or the Agreement have the meanings given to them in the Plan.

Participant:	[]
Grant Date:	[], 2021
Exercise Price per Share:	[]
Shares Subject to the Option:	
Final Expiration Date:	[]
<b>Vesting Commencement Date:</b>	[], 2021
Vesting Schedule:	Twenty-five p

Twenty-five percent (25%) of the Shares subject to the Option will vest on each of the first four anniversaries of the Vesting Commencement Date, subject to Participant's continued status as a Service Provider through the applicable vesting date. Notwithstanding the foregoing, (x) if Participant incurs a Termination of Service due to Participant's death or Disability, subject to and conditioned upon Participant's (or Participant's estate's) timely execution and non-revocation of a release of claims in a form prescribed by the Company (a "*Release*") that becomes effective and irrevocable no later than sixty (60) days following such Termination of Service (the date such Release becomes effective and irrevocable, the "*Release Effective Date*"), the Option will vest in full (to the extent then-unvested) upon the Release Effective Date (and shall remain outstanding and eligible to vest through the Release Effective Date and shall automatically be forfeited if the Release does not become effective and irrevocable on or prior to the sixtieth (60th) day following such termination), and (y) if a Change in Control is consummated more than twelve (12) months after [Closing Date], 2021 (the "*Closing Date*") (and such Change in Control does not occur pursuant to a binding agreement entered into

within the twelve (12) months after Closing Date which, for clarity, is addressed solely in the following paragraph) and Participant's status as a Service Provider is terminated by the Company or its Subsidiaries without Cause (as defined in the Agreement) or due to Participant's resignation for Good Reason (as defined in the Company's Executive Severance Plan), in either case, within twelve (12) months following the consummation of such Change in Control, subject to and conditioned upon Participant's timely execution and non-revocation of a Release that becomes effective and irrevocable no later than sixty (60) days following such Termination of Service, the Option will vest in full (to the extent then-unvested) upon the Release Effective Date (and shall remain outstanding and eligible to vest through the Release Effective Date and shall automatically be forfeited if the Release does not become effective and irrevocable on or prior to the sixtieth (60th) day following such termination).

If (i) a Change in Control occurs within twelve (12) months following Closing Date (or pursuant to a binding agreement entered into within such twelve (12) month period) and (ii) Participant's status as a Service Provider is terminated by the Company or its Subsidiaries without Cause or due to Participant's resignation for Good Reason, in either case, within twelve (12) months after the consummation of such Change in Control, then, subject to and conditioned upon Participant's timely execution and non-revocation of a Release that becomes effective and irrevocable no later than sixty (60) days following such Termination of Service, then a number of Shares subject to the Option equal to (A) the product of (x) 1/48th of the total number of Shares subject to the Option by (y) the number of full and partial months elapsed between the Grant Date and date of such Termination of Service less (B) the total number of Shares subject to the Option that have vested prior to the date of Termination of Service will vest upon the Release Effective Date (and shall remain outstanding and eligible to vest through the Release Effective Date and shall automatically be forfeited if the Release does not become effective and irrevocable on or prior to the sixtieth (60th) day following such termination).

### **Type of Option**

[Incentive Stock Option]/[Non-Qualified Stock Option]

By accepting (whether in writing, electronically or otherwise) the Option, Participant agrees to be bound by the terms of this Grant Notice, the Plan and the Agreement. Participant has reviewed the Plan, this Grant Notice and the Agreement in their entirety, has had an opportunity to obtain the advice of counsel prior to executing this Grant Notice and fully understands all provisions of the Plan, this Grant Notice and the Agreement. Participant hereby agrees to accept as binding, conclusive and final all decisions or interpretations of the Administrator upon any questions arising under the Plan, this Grant Notice or the Agreement.

THE BEAUTY HEALTH COMPANY	PARTICIPANT
Ву:	
Name:	[Participant Name]

#### STOCK OPTION AGREEMENT

Capitalized terms not specifically defined in this Agreement have the meanings specified in the Grant Notice or, if not defined in the Grant Notice, in the Plan.

# GENERAL

- <u>Grant of Option</u>. The Company has granted to Participant the Option effective as of the grant date set forth in the Grant Notice (the "*Grant Date*").
- <u>Incorporation of Terms of Plan</u>. The Option is subject to the terms and conditions set forth in this Agreement and the Plan, which is incorporated herein by reference. In the event of any inconsistency between the Plan and this Agreement, the terms of the Plan will control, unless it is expressly specified in in this Agreement or the Grant Notice that the specific provision of the Plan will not apply. For clarity, the foregoing sentence shall not limit the applicability of any additive language contained in this Agreement which provides supplemental or additional terms not inconsistent with the Plan.

## PERIOD OF EXERCISABILITY

- <u>Commencement of Exercisability</u>. The Option will vest and become exercisable according to the vesting schedule in the Grant Notice (the "Vesting Schedule") except that any fraction of a Share as to which the Option would be vested or exercisable will be accumulated and will vest and become exercisable only when a whole vested Share has accumulated; provided, however, that, notwithstanding the foregoing or anything to the contrary in the Grant Notice or this Agreement, in no event may the Option be exercised (in whole or in part) prior to the date on which the Company files a Form S-8 Registration Statement covering the Shares subject to the Option. Except as otherwise set forth in the Grant Notice, the Plan or this Agreement, and unless the Administrator otherwise determines, the Option will immediately expire and be forfeited as to any portion of the Option that is not vested and exercisable as of Participant's Termination of Service for any reason (after taking into consideration any accelerated vesting and exercisability which may occur in connection with such Termination of Service, if any).
- <u>Duration of Exercisability</u>. The Vesting Schedule is cumulative. Any portion of the Option which vests and becomes exercisable will remain vested and exercisable until the Option expires. The Option will be forfeited immediately upon its expiration.
- Expiration of Option. Except as may be extended in accordance with Section 5.3 of the Plan, the Option may not be exercised to any extent by anyone after, and will expire on, the first of the following to occur:
  - The final expiration date in the Grant Notice;
- Except as the Administrator may otherwise approve, the expiration of three (3) months from the date of Participant's Termination of Service, unless Participant's Termination of Service is for Cause (as defined below), due to Participant's voluntary resignation or by reason of Participant's death or Disability;

- Except as the Administrator may otherwise approve, the expiration of one year from the date of Participant's Termination of Service by reason of Participant's death or Disability;
- Except as the Administrator may otherwise approve, the expiration of one (1) month from the date of Participant's Termination of Service due to Participant's voluntary resignation for any reason; and
  - Except as the Administrator may otherwise approve, Participant's Termination of Service for Cause.
- 2.4 <u>Cause Definition</u>. As used in this Agreement, "*Cause*" means (i) if Participant is a party to a written employment or consulting agreement with the Company or a Subsidiary in which the term "cause" is defined (a "*Relevant Agreement*"), "Cause" as defined in the Relevant Agreement, and (ii) if no Relevant Agreement exists, the occurrence of any one or more of the following events:
- (a) Participant has engaged in an act of dishonesty, theft, embezzlement or fraud, a breach of confidentiality, an unauthorized disclosure or use of inside information, customer lists, trade secrets or other confidential information; has breached a fiduciary duty, law, rule, regulation or policy or procedure of the Company (including but not limited to policies and procedures pertaining to harassment and discrimination); Participant's commission of, or plea of guilty or *nolo contendere* to, any crime or offense (other than minor traffic violations or similar offenses), or any act by Participant constituting a felony;
- (b) Participant's gross or repeated neglect of, or repeated or willful failure to perform, his or her duties to the Company, or Participant's history of substandard performance, if such substandard performance is not cured to the satisfaction of the Board, the Chief Executive Officer of the Company (the "CEO") (unless Participant is the CEO), and/or Participant's manager within ten (10) days following notice to Participant;
- (c) Participant has breached any of the provisions of any employment, confidentiality, restrictive covenant or other agreement between Participant and the Company or an affiliate thereof;
  - (d) actions by Participant involving malfeasance or gross negligence in the performance of, Participant's duties;
- (e) Participant's failure or refusal to perform Participant's duties to the Company on an exclusive and full-time basis if such failure to perform Participant's duties is not cured to the satisfaction of the Board, the CEO (unless Participant is the CEO), and/or Participant's manager (as applicable) within ten (10) days following notice to the Participant;
- (f) Participant's insubordination or failure to follow the Board's (or the CEO's (unless Participant is the CEO) or such Participant's manager's) instructions;

- (g) Participant's violation of any rule, regulation, or policy of the Company or the Board (or Participant's manager) applicable to similarly-situated employees of the Company generally, including, without limitation, rules, regulations, or policies addressing confidentiality, non-solicitation or non-competition, if such violation is not cured to the satisfaction of the Board, the CEO (unless Participant is the CEO), and/or Participant's manager (as applicable) within ten (10) days following notice to the Participant;
- (h) Participant's use of alcohol or illicit drugs in a manner that has or would reasonably be expected to have a detrimental effect on Participant's performance, Participant's duties to Company, or the reputation of the Company or its affiliates; or
- (i) Participant's performance of acts which are or could reasonably be expected to become materially detrimental to the image, reputation, finances or business of the Company or any of its affiliates, including but limited to the Participant's commission of unlawful harassment or discrimination.

# EXERCISE OF OPTION

- <u>Person Eligible to Exercise</u>. During Participant's lifetime, only Participant may exercise the Option. After Participant's death, any exercisable portion of the Option may, prior to the time the Option expires, be exercised by Participant's Designated Beneficiary as provided in the Plan.
- <u>Partial Exercise</u>. Any exercisable portion of the Option or the entire Option, if then wholly exercisable, may be exercised, in whole or in part, according to the procedures in the Plan at any time prior to the time the Option or portion thereof expires, except that the Option may only be exercised for whole Shares.
  - Tax Withholding; Exercise Price.
- Unless the Administrator otherwise determines, the Company shall withhold, or cause to be withheld, Shares otherwise vesting or issuable under this Option in satisfaction of any exercise price and/or applicable withholding tax obligations, in accordance with the Plan. With respect to tax withholding obligations, the number of Shares which may be so withheld or surrendered shall be limited to the number of Shares which have a fair market value on the date of withholding no greater than the aggregate amount of such liabilities based on the maximum individual statutory withholding rates in Participant's applicable jurisdictions for federal, state, local and foreign income tax and payroll tax purposes that are applicable to such taxable income, in accordance with Section 9.5 of the Plan.
- Participant acknowledges that Participant is ultimately liable and responsible for all taxes owed in connection with the Option, regardless of any action the Company or any Subsidiary takes with respect to any tax withholding obligations that arise in connection with the Option. Neither the Company nor any Subsidiary makes any representation or undertaking regarding the treatment of any tax withholding in connection with the awarding, vesting or exercise of the Option or the subsequent sale of Shares. The Company and the Subsidiaries do not commit and are under no obligation to structure the Option to reduce or eliminate Participant's tax liability.
  - <u>Representation</u>. Participant represents to the Company that Participant has reviewed with Participant's own tax advisors the tax consequences of this Award and the transactions contemplated by the Grant Notice and this Agreement. Participant is relying solely on such advisors and not on any statements or representations of the Company or any of its agents.

## OTHER PROVISIONS

- <u>Adjustments</u>. Participant acknowledges that the Option is subject to adjustment, modification and termination in certain events as provided in this Agreement and the Plan.
- <u>Clawback</u>. The Option and the Shares issuable hereunder shall be subject to any clawback or recoupment policy in effect on the Grant Date or as may be adopted or maintained by the Company following the Grant Date, including the Dodd-Frank Wall Street Reform and Consumer Protection Act and any rules or regulations promulgated thereunder.
- <u>Notices</u>. Any notice to be given under the terms of this Agreement to the Company must be in writing and addressed to the Company in care of the Company's General Counsel at the Company's principal office or the General Counsel's then-current email address or facsimile number. Any notice to be given under the terms of this Agreement to Participant must be in writing and addressed to Participant (or, if Participant is then deceased, to the Designated Beneficiary) at Participant's last known mailing address, email address or facsimile number in the Company's personnel files. By a notice given pursuant to this Section, either party may designate a different address for notices to be given to that party. Any notice will be deemed duly given when actually received, when sent by email, when sent by certified mail (return receipt requested) and deposited with postage prepaid in a post office or branch post office regularly maintained by the United States Postal Service, when delivered by a nationally recognized express shipping company or upon receipt of a facsimile transmission confirmation.
  - <u>Titles</u>. Titles are provided herein for convenience only and are not to serve as a basis for interpretation or construction of this Agreement.
- <u>Conformity to Securities Laws</u>. Participant acknowledges that the Plan, the Grant Notice and this Agreement are intended to conform to the extent necessary with all Applicable Laws and, to the extent Applicable Laws permit, will be deemed amended as necessary to conform to Applicable Laws.
- <u>Successors and Assigns</u>. The Company may assign any of its rights under this Agreement to single or multiple assignees, and this Agreement will inure to the benefit of the successors and assigns of the Company. Subject to the restrictions on transfer set forth in this Agreement or the Plan, this Agreement will be binding upon and inure to the benefit of the heirs, legatees, legal representatives, successors and assigns of the parties hereto.
- <u>Limitations Applicable to Section 16 Persons</u>. Notwithstanding any other provision of the Plan or this Agreement, if Participant is subject to Section 16 of the Exchange Act, the Plan, the Grant Notice, this Agreement and the Option will be subject to any additional limitations set forth in any applicable exemptive rule under Section 16 of the Exchange Act (including any amendment to Rule 16b-3) that are requirements for the application of such exemptive rule. To the extent Applicable Laws permit, this Agreement will be deemed amended as necessary to conform to such applicable exemptive rule.

- Entire Agreement. The Plan, the Grant Notice and this Agreement (including any exhibit hereto) constitute the entire agreement of the parties and supersede in their entirety all prior undertakings and agreements of the Company and Participant with respect to the subject matter hereof.
- <u>Severability</u>. If any portion of the Grant Notice or this Agreement or any action taken under the Grant Notice or this Agreement, in any case is held illegal or invalid for any reason, the illegality or invalidity will not affect the remaining parts of the Grant Notice and/or this Agreement (as applicable), and the Grant Notice and/or this Agreement (as applicable) will be construed and enforced as if the illegal or invalid provisions had been excluded, and the illegal or invalid action will be null and void.
- <u>Limitation on Participant's Rights</u>. Participation in the Plan confers no rights or interests other than as herein provided. This Agreement creates only a contractual obligation on the part of the Company as to amounts payable and may not be construed as creating a trust. Neither the Plan nor any underlying program, in and of itself, has any assets. Participant will have only the rights of a general unsecured creditor of the Company with respect to amounts credited and benefits payable, if any, with respect to the Option, and rights no greater than the right to receive the Shares as a general unsecured creditor with respect to the Option, as and when exercised pursuant to the terms hereof
- <u>Not a Contract of Employment or Service</u>. Nothing in the Plan, the Grant Notice or this Agreement confers upon Participant any right to continue in the employ or service of the Company or any Subsidiary or interferes with or restricts in any way the rights of the Company and its Subsidiaries, which rights are hereby expressly reserved, to discharge or terminate the services of Participant at any time for any reason whatsoever, with or without Cause, except to the extent expressly provided otherwise in a written agreement between the Company or a Subsidiary and Participant.
- <u>Counterparts</u>. The Grant Notice may be executed in one or more counterparts, including by way of any electronic signature, subject to Applicable Law, each of which will be deemed an original and all of which together will constitute one instrument.
  - <u>Incentive Stock Options</u>. If the Option is designated as an Incentive Stock Option:
- Participant acknowledges that to the extent the aggregate fair market value of shares (determined as of the time the option with respect to the shares is granted) with respect to which stock options intended to qualify as "incentive stock options" under Section 422 of the Code, including the Option, are exercisable for the first time by Participant during any calendar year exceeds \$100,000 or if for any other reason such stock options do not qualify or cease to qualify for treatment as "incentive stock options" under Section 422 of the Code, such stock options (including the Option) will be treated as non-qualified stock options. Participant further acknowledges that the rule set forth in the preceding sentence will be applied by taking the Option and other stock options into account in the order in which they were granted, as determined under Section 422(d) of the Code. Participant also acknowledges that if the Option is exercised more than three months after Participant's Termination of Service, other than by reason of death or disability, the Option will be taxed as a Non-Qualified Stock Option.
- Participant will give prompt written notice to the Company of any disposition or other transfer of any Shares acquired under this Agreement if such disposition or other transfer is made (i) within two years from the Grant Date or (ii) within one year after the transfer of such Shares

to Participant. Such notice will specify the date of such disposition or other transfer and the amount realized, in cash, other property, assumption of indebtedness or other consideration, by Participant in such disposition or other transfer.

• <u>Governing Law</u>. The Grant Notice and this Agreement will be governed by and interpreted in accordance with the laws of the State of Delaware, disregarding any state's choice-of-law principles requiring the application of a jurisdiction's laws other than the State of Delaware.

\*\*\*\*

#### **EXHIBIT B**

# EMPLOYEE PROPRIETARY INFORMATION AND INVENTIONS ASSIGNMENT AGREEMENT

#### THE HYDRAFACIAL COMPANY

# EMPLOYEE PROPRIETARY INFORMATION AND INVENTIONS ASSIGNMENT AGREEMENT

In consideration and as a condition of my employment by Edge Systems LLC d/b/a The HydraFacial Company (together with its parents and subsidiaries and any of their respective successors or assigns, the "Company"), and my receipt of the compensation paid to me by the Company pursuant to the employment agreement entered into between me and the Company (the "Employment Agreement") concurrently with the execution of this Employee Proprietary Information and Inventions Assignment Agreement (the "Agreement"), and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, effective as of the Effective Date (as defined in the Employment Agreement), I, the undersigned, agree as follows:

1. Proprietary Information. During the term of my employment with the Company, I may receive and otherwise be exposed, directly or indirectly, to confidential and proprietary information of the Company whether in graphic, written, electronic or oral form, including without limitation, information relating to the Company's business, strategies, designs, products, services and technologies and any derivatives, improvements and enhancements relating to any of the foregoing, or to the Company's suppliers, customers or business partners (collectively "Proprietary Information"). Proprietary Information may be identified at the time of disclosure as confidential or proprietary or information which by its context would reasonably be deemed to be confidential or proprietary. "Proprietary Information" may also include without limitation (i)(a) unpublished patent disclosures and patent applications and other filings, know-how, trade secrets, works of authorship and other intellectual property, as well as any information regarding ideas, Work Product (as defined below), technology, and processes, including without limitation assays, sketches, schematics, techniques, drawings, designs, descriptions, specifications and technical documentation, (b) specifications, protocols, models, designs, equipment, engineering, algorithms, software programs, software source documents, formulae, (c) information concerning or resulting from any research and development or other project, including without limitation, experimental work, and product development plans, regulatory compliance information, and research, development and regulatory strategies, and (d) business and financial information, including without limitation purchasing, procurement, manufacturing, customer lists, information relating to investors, employees, business and contractual relationships, business forecasts, sales and merchandising, business and marketing plans, product plans, and business strategies, including without limitation, information the Company provides regarding third parties, such as, but not limited to, suppliers, customers, employees, investors, or vendors; and (ii) any other information, to the extent such information contains, reflects or is based upon any of the foregoing Proprietary Information. The Proprietary Information may also include information of

a third party that is disclosed to me by the Company or such third party at the Company's direction. Any information disclosed by any of the Company's affiliated companies or by any company, person or other entity participating with the Company in any consortium, partnership, joint venture or similar business combination, which would otherwise constitute Proprietary Information if disclosed by the Company, shall be deemed to constitute Proprietary Information under this Agreement, and the rights of the Company under this Agreement may be enforced by any such affiliate or participating entity (as well as by the Company) with respect to any violation relating to the Proprietary Information disclosed by such affiliate or entity, as if that affiliate or entity were also a party to this Agreement.

2. Obligations of Non-Use and Nondisclosure. I acknowledge the confidential and secret character of the Proprietary Information, and agree that the Proprietary Information is the sole, exclusive and valuable property of the Company. Accordingly, I agree not to use the Proprietary Information except in the performance of my authorized duties as an employee of the Company, and not to disclose all or any part of the Proprietary Information in any form to any third party, either during or after the term of my employment with the Company, without the prior written consent of the Company on a case-by-case basis, and to cooperate with the Company and use my best efforts not to prevent the unauthorized us or disclosure or reproductions of any Proprietary Information. In addition, I agree not to copy or remove any tangible materials containing Proprietary Information from the premises of the Company, except in the proper performance of my duties as an employee of the Company or with the prior written consent of the Company on a case-by-case basis. Upon termination of my employment with the Company, I agree to cease using and to return to the Company all whole and partial copies and derivatives of the Proprietary Information, whether in my possession or under my direct or indirect control, provided that I am entitled to retain my personal copies of (a) my compensation and benefits records, and (b) this Agreement. I understand that my obligations of nondisclosure with respect to Proprietary Information shall not apply to information that I can establish by competent proof (i) was actually in the public domain at the time of disclosure or enters the public domain following disclosure other than as a result of a breach of this Agreement, (ii) is already in my possession without breach of any obligations of confidentiality at the time of disclosure by the Company as shown by my files and records immediately prior to the time of disclosure, or (iii) is obtained by me from a third party not under confidentiality obligations and without a breach of any obligations of confidentiality. If I become compelled by law, regulation (including without limitation the rules of any applicable securities exchange), court order, or other governmental authority to disclose any Proprietary Information, I shall, to the extent possible and permissible under applicable law, first give notice to the Company. I agree to cooperate reasonably with the Company (at the Company's request) in any proceeding to obtain a protective order or other remedy. If such protective order or other remedy is not obtained, I shall only disclose that portion of such Proprietary Information required to be disclosed, in the opinion of my legal counsel. I shall request that confidential treatment be accorded such Proprietary Information, where available. Compulsory disclosures made pursuant to this section shall not relieve me of my obligations of confidentiality and non-use with respect to non-compulsory disclosures. If I learn of any possible unauthorized use or disclosure of Proprietary Information, I shall cooperate fully with the Company to

enforce its rights in such information. Notwithstanding the foregoing or anything herein to the contrary, nothing contained herein shall prohibit me from (x) filing a charge with, reporting possible violations of federal law or regulation to, participating in any investigation by, or cooperating with any governmental agency or entity or making other disclosures that are protected under the whistleblower provisions of applicable law or regulation; (y) communicating directly with, cooperating with, or providing information (including trade secrets) in confidence to, any federal, state or local government regulator (including, but not limited to, the U.S. Securities and Exchange Commission, the U.S. Commodity Futures Trading Commission, or the U.S. Department of Justice) for the purpose of reporting or investigating a suspected violation of law, or from providing such information to my attorney or in a sealed complaint or other document filed in a lawsuit or other governmental proceeding; and/or (z) making disclosures that are protected by the National Labor Relations Act or similar applicable law.

- 3. <u>Defend Trade Secrets Act Notice of Immunity Rights</u>. I acknowledge that the Company has provided me with the following notice of immunity rights in compliance with the requirements of the Defend Trade Secrets Act: (a) I shall not be held criminally or civilly liable under any Federal or State trade secret law for the disclosure of Proprietary Information that is made in confidence to a Federal, State, or local government official or to an attorney solely for the purpose of reporting or investigating a suspected violation of law, (b) I shall not be held criminally or civilly liable under any Federal or State trade secret law for the disclosure of Proprietary Information that is made in a complaint or other document filed in a lawsuit or other proceeding, if such filing is made under seal and (c) if I file a lawsuit for retaliation by the Company for reporting a suspected violation of law, I may disclose the Proprietary Information to my attorney and use the Proprietary Information in the court proceeding, if I file any document containing the Proprietary Information under seal, and do not disclose the Proprietary Information, except pursuant to court order.
- 4. <u>Property of the Company</u>. I acknowledge and agree that all notes, memoranda, reports, drawings, blueprints, manuals, materials, data, emails and other papers and records of every kind, or other tangible or intangible materials which shall come into my possession in the course of my employment with the Company, relating to any Proprietary Information, shall be the sole and exclusive property of the Company and I hereby assign any rights or interests I may obtain in any of the foregoing. I agree to surrender this property to the Company immediately upon termination of my employment with the Company, or at any time upon request by the Company. I further agree that any property situated on the Company's data systems or on the Company's premises and owned by the Company, including without limitation electronic storage media, filing cabinets or other work areas, is subject to inspection by Company personnel at any time with or without notice. I further agree that in the event of termination of my employment with the Company I will execute a Termination Certificate substantially in the form attached hereto as <u>Exhibit A</u>.

## 5. Inventions.

- Disclosure and Assignment of Inventions. For purposes of this Agreement, an "Invention" shall mean any idea, invention or work of authorship, including, without limitation, any documentation, formula, design, device, code, method, software, technique, process, discovery, concept, improvement, enhancement, development, machine or contribution, in each case whether or not patentable or copyrightable. I will disclose all Inventions promptly in writing to an officer of the Company or to attorneys of the Company in accordance with the Company's policies and procedures, I will, and hereby do, assign to the Company, without requirement of further writing, without royalty or any other further consideration, my entire right, title and interest throughout the world in and to all Inventions created, conceived, made, developed, and/or reduced to practice by me in the course of my employment by the Company and all intellectual property rights therein. I hereby waive, and agree to waive, any moral rights I may have in any copyrightable work I create or have created on behalf of the Company. I also hereby agree, that for a period of one year after my employment with the Company, I shall disclose to the Company any Inventions that I create, conceive, make, develop, reduce to practice or work on that relate to the work I performed for the Company. The Company agrees that it will use commercially reasonable measures to keep Inventions disclosed to it pursuant to this Section 5.1 that do not constitute Inventions to be owned by the Company in confidence and shall not use any Inventions for its own advantage, unless in either case those Inventions are assigned or assignable to the Company pursuant to this Section 5.1 or otherwise.
- Certain Exemptions. The obligations to assign Inventions set forth in Section 5.1 apply with respect to all Inventions (a) whether or not such 5.2 Inventions are conceived, made, developed or worked on by me during my regular hours of employment with the Company; (b) whether or not the Invention was made at the suggestion of the Company; (c) whether or not the Invention was reduced to drawings, written description, documentation, models or other tangible form; and (d) whether or not the Invention is related to the general line of business engaged in by the Company, but do not apply to Inventions that (x) I develop entirely on my own time or after the date of this Agreement without using the Company's equipment, supplies, facilities or Proprietary Information; (y) do not relate to the Company's business, or actual or demonstrably anticipated research or development of the Company at the time of conception or reduction to practice of the Invention; and (z) do not result from and are not related to any work performed by me for the Company. I hereby acknowledge and agree that the Company has notified me that, if I reside in the state of California, assignments provided for in Section 5.1 do not apply to any Invention which qualifies fully for exemption from assignment under the provisions of Section 2870 of the California Labor Code ("Section 2870"), a copy of which is attached as Exhibit B. If applicable, at the time of disclosure of an Invention that I believe qualifies under Section 2870, I shall provide to the Company, in writing, evidence to substantiate the belief that such Invention qualifies under Section 2870. I further understand that, to the extent this Agreement shall be construed in accordance with the laws of any state which precludes a requirement in an employee agreement to assign certain classes of inventions made by an employee, Section 5.1 shall be interpreted not to apply to any Invention which a court rules and/or the Company agrees falls within such classes.
- 5.3 Records. I will make and maintain adequate and current written records of all Inventions covered by Section 5.1. These records may be in the form of notes, sketches, drawings, flow charts, electronic data or recordings, notebooks and any other format. These records shall be and remain the property of the Company at all times and shall be made available to the Company at all times.
- 5.4 <u>Patents and Other Rights</u>. I agree to assist the Company in obtaining, maintaining and enforcing patents, invention assignments and copyright assignments, and other proprietary rights in connection with any Invention covered by Section 5.3, and will otherwise assist the Company as reasonably required by the Company to perfect in the Company the rights, title and other interests

in my work product granted to the Company under this Agreement (both in the United States and foreign countries). I further agree that my obligations under this Section 5.4 shall continue beyond the termination of my employment with the Company, but if I am requested by the Company to render such assistance after the termination of such employment, I shall be entitled to a fair and reasonable rate of compensation for such assistance, and to reimbursement of any expenses incurred at the request of the Company relating to such assistance. If the Company is unable for any reason, after reasonable effort, to secure my signature on any document needed in connection with the actions specified above, I hereby irrevocably designate and appoint the Company and its duly authorized officers and agents as my agent and attorney in fact, which appointment is coupled with an interest, to act for and in my behalf to execute, verify and file any such documents and to do all other lawfully permitted acts to further the purposes of this Section 5.4 with the same legal force and effect as if executed by me.

5.5 Prior Contracts Inventions; Information Belonging to Third Parties. I represent and warrant that, except as set forth on Exhibit C, I am not required, and I have not been required during the course of work for the Company or its predecessors, to assign Inventions under any other contracts that are now or were previously in existence between me and any other person or entity. I further represent that (i) I am not obligated under any consulting, employment or other agreement that would affect the Company's rights or my duties under this Agreement, and I shall not enter into any such agreement or obligation during the period of my employment by the Company, (ii) there is no action, investigation, or proceeding pending or threatened, or any basis therefor known to me involving my prior employment or any consultancy or the use of any information or techniques alleged to be proprietary to any former employer, and (iii) the performance of my duties as an employee of the Company do not and will not breach, or constitute a default under any agreement to which I am bound, including any agreement limiting the use or disclosure of proprietary information acquired in confidence prior to engagement by the Company or if applicable, any agreement to refrain from competing, directly or indirectly, with the business of such previous employer or any other party or to refrain from soliciting employees, customers or suppliers of such previous employer or other party. I will not, in connection with my employment by the Company, use or disclose to the Company any confidential, trade secret or other proprietary information of any previous employer or other person to which I am not lawfully entitled. As a matter of record, I attach as Exhibit C a brief description of all Inventions made or conceived by me prior to my employment with the Company which I desire to be excluded from this Agreement ("Background Technology."). If full disclosure of any Background Technology would breach or constitute a default under any agreement to which I am bound, including any agreement limiting the use or disclosure of proprietary information acquired in confidence prior to engagement by the Company, I understand that I am to describe such Background Technology in Exhibit C at the most specific level possible without violating any such prior agreement. Without limiting my obligations or representations under this Section 5.5, if I use any Background Technology in the course of my employment or incorporate any Background Technology in any product, service or other offering of the Company, I hereby grant the Company a non-exclusive, royalty-free, perpetual and irrevocable, worldwide right to use and sublicense the use of Background Technology for the purpose of developing, marketing, selling and supporting Company technology, products and services, either directly or through multiple tiers of distribution, but not for the purpose of marketing Background Technology separately from Company products or services.

- 5.6 Works Made for Hire. I acknowledge that all original works of authorship which are made by me (solely or jointly with others) within the scope of my employment with the Company and which are eligible for copyright protection are "works made for hire" as that term is defined in the United States Copyright Act (17 U.S.C., Section 101).
  - 6. <u>Restrictive Covenants</u>. I agree to fully comply with the covenants set forth in this Section 6 (the "<u>Restrictive Covenants</u>"). I further acknowledge and agree that the Restrictive Covenants are reasonable and necessary to protect the Company's legitimate business interests, including its Proprietary Information and goodwill.
- 6.1 Non-Competition. During the term of my employment by the Company, I will not without the prior written approval of the Board of Directors of the Company, (a) engage in any other professional employment or consulting, or (b) directly or indirectly participate in or assist any business (whether as owner, partner, officer, director, employee, consultant, investor, lender or otherwise, except as the holder of not more than 1% of the outstanding stock of a publicly-held company), in each case, which is a current or potential supplier, customer or competitor of the Company, including but not limited to any business or enterprise that develops, manufactures, markets, licenses, sells or provides any product or service that competes with any product or service developed, manufactured, marketed, licensed, sold or provided, or planned to be developed, manufactured, marketed, licensed, sold or provided by the Company.

## 6.2 Non-Solicitation.

- (a) During the term of my employment with the Company and for a period of one (1) year thereafter I will not (i) solicit for employment or engagement, any employee, consultant or independent contractor of the Company, or (ii) solicit, encourage, induce or attempt to induce or assist others to solicit, encourage, induce or attempt to induce any employees, consultants or independent contractors of the Company who were employed or engaged by the Company at any time during the term of my employment with the Company to terminate their employment or engagement with the Company.
- (b) During the term of my employment with the Company, I will not solicit, divert or take away, or attempt to divert or take away, the business of any customer or client of the Company (served by the Company during the twelve (12)-month period prior to the termination of my employment with the Company) on my own behalf or on behalf of any person or entity other than the Company.
- 6.3 <u>No Defamatory Communications</u>. During the term of my employment with the Company and thereafter, I agree that I will not make any public or private statement which would reasonably be expected to defame or disparage the Company or any of its employees, officers, managers or directors. Notwithstanding the foregoing, this Section 6.3 shall not preclude me from making any statement to the extent required by law or legal process.
- 6.4 <u>Extension</u>. Without limiting the Company's ability to seek other remedies available in law or equity, if I violate the provisions of Section 6.2(a), I shall continue to be bound by the restrictions set forth in such section until a period of one year has expired without any violation of such provisions.
- 6.5 <u>Interpretation</u>. If any restriction set forth in Section 6.2 is found by any court of competent jurisdiction to be unenforceable because it extends for too long a period of time or over too great a range of activities or in too broad a geographic area, it shall be interpreted to extend only over the maximum period of time, range of activities or geographic area as to which it may be enforceable.

- 7. <u>Notification to Other Parties</u>. In the event of termination of my employment with the Company, I hereby consent to notification by the Company to my new employer or other party for whom I work about my rights and obligations under this Agreement.
- 8. Employment at Will. I understand and agree that my employment with the Company is at will. Accordingly, my employment can be terminated for any lawful reason or for no reason, without cause or notice, at my option or the Company's option, subject to and except as otherwise expressly set forth in the Employment Agreement. The Restrictive Covenants will remain in effect for the periods specified in this Agreement, unless such Restrictive Covenants are modified by a further written agreement signed by both an authorized officer of the Company and me which expressly alters such Restrictive Covenants.

## 9. Miscellaneous.

- 9.1 The parties' rights and obligations under this Agreement will bind and inure to the benefit of their respective successors, heirs, executors, and administrators and permitted assigns. I will not assign this Agreement or my obligations hereunder without the prior written consent of the Company, which consent may be withheld in the Company's sole discretion, and any such purported assignment without consent shall be null and void from the beginning. I agree that the Company may freely assign this Agreement to any affiliate or successor in interest, including any person or entity that, whether by way of merger, sale, acquisition, corporate re-organization or otherwise, directly or indirectly acquires all or substantially all of the business or assets of the Company.
- 9.2 This Agreement, together with the Employment Agreement, constitutes the parties' final, exclusive and complete understanding and agreement with respect to the subject matter hereof, and supersedes all prior and contemporaneous understandings and agreements, whether oral or written, relating to its subject matter.
- 9.3 Any subsequent change or changes in my duties, obligations, rights or compensation will not affect the validity or scope of this Agreement. This Agreement may not be waived, modified or amended unless mutually agreed upon in writing by both parties. No delay or omission by the Company in exercising any right under this Agreement will operate as a waiver of that or any other right. A waiver or consent given by the Company on any one occasion is effective only in that instance and will not be construed as a bar to or waiver of any right on any other occasion.
- 9.4 The provisions of this Agreement are severable, and the invalidity or unenforceability of any provision(s) of this Agreement shall not impact the validity or enforceability of any other provision(s) of this Agreement, which shall remain in full force and effect.
- 9.5 I acknowledge that the Company will suffer substantial damages not readily ascertainable or compensable in terms of money in the event of the breach of any of my obligations under this Agreement. I therefore agree that the Company shall be entitled (without limitation of any other rights or remedies otherwise available to the Company) to obtain an injunction from any court of competent jurisdiction prohibiting the continuance or recurrence of any breach of this Agreement. I also agree that if the Company prevails in any action or proceeding to enforce my obligations under this Agreement, I will pay all of the Company's expenses relating to any such action or proceeding including, without limitation, all reasonable attorney's fees, if so authorized by applicable state and/or federal law.

- 9.6 The rights and obligations of the parties under this Agreement shall be governed in all respects by the laws of the State of California exclusively, without reference to any conflict laws rule that would result in the application of the laws of any other jurisdiction. I agree that upon the Company's request, all disputes arising hereunder shall be adjudicated in the state and federal courts having jurisdiction over disputes arising in Los Angeles County, California, and I hereby agree to consent to the personal jurisdiction of such courts. The Company and I each hereby irrevocably waive any right to a trial by jury in any action, suit or other legal proceeding arising under or relating to any provision of this Agreement.
- 9.7 Any notices required or permitted hereunder shall be given to the appropriate party at the address specified on the signature page to this Agreement or at such other address as the party shall specify in writing. Such notice shall be deemed given upon personal delivery, or sent by certified or registered mail, postage prepaid, three (3) days after the date of mailing.
- 9.8 Except as otherwise provided herein, the provisions of this Agreement shall survive the termination of my employment with the Company for any reason.
- 9.9 This Agreement may be executed in any number of counterparts, each of which shall be deemed to be an original and all of which together shall be deemed to be one and the same instrument. A facsimile, PDF (or any electronic signature complying with the U.S. federal ESIGN Act of 2000, e.g., www.docusign.com) or any other type of copy of an executed version of this Agreement signed by a party is binding upon the signing party to the same extent as the original of the signed agreement.

I ACKNOWLEDGE THAT I HAVE HAD THE OPPORTUNITY TO SEEK THE ADVICE OF INDEPENDENT LEGAL COUNSEL PRIOR TO SIGNING THIS AGREEMENT, AND THAT I HAVE EITHER CONSULTED WITH OR ON MY OWN VOLITION CHOSEN NOT TO CONSULT WITH SUCH COUNSEL. I FURTHER ACKNOWLEDGE THAT I HAVE READ THIS AGREEMENT CAREFULLY AND I UNDERSTAND AND ACCEPT THE OBLIGATIONS WHICH IT IMPOSES UPON ME WITHOUT RESERVATION. NO PROMISES OR REPRESENTATIONS HAVE BEEN MADE TO ME TO INDUCE ME TO SIGN THIS AGREEMENT. I SIGN THIS AGREEMENT VOLUNTARILY AND FREELY, IN DUPLICATE, WITH THE UNDERSTANDING THAT THE COMPANY WILL RETAIN ONE COUNTERPART AND THE OTHER COUNTERPART WILL BE RETAINED BY ME.

(Signature Page Follows)

IN WITNESS WHEREOF, I have executed this document as of	, 20
	Employee:
	Address:
AGREED AND ACKNOWLEDGED:	
	EDGE SYSTEMS LLC d/b/a THE HYDRAFACIAL COMPANY
	Ву:
	Name:
	Title:
	Address: The HydraFacial Company 2165 E Spring Street Long Beach, CA 90806

## Exhibit A

## **Termination Certificate**

I, the undersigned, hereby certify that I do not have in my possession, nor have I failed to return, any documents or materials relating to the business of The HydraFacial Company or its affiliates (together, the "Company"), or copies thereof, including, without limitation, any item of Proprietary Information listed in Section 4 of the Company's Employee Proprietary Information and Inventions Assignment Agreement (the "Agreement") to which I am a party, but not including copies of my own compensation and benefits records (in each case, to the extent expressly permitted by the Agreement).

I further certify that I have complied with all of the terms of the Agreement signed by me. I further agree that in compliance with the Agreement, I will preserve as confidential any information relating to the Company or any of it business partners, clients, consultants or licensees which has been disclosed to me in confidence during the course of my employment by the Company unless authorized in writing to disclose such information (i) by an executive officer of the Company, in the event that I am not an executive officer of the Company, or (ii) by the Board of Directors of the Company, in the event that I am an executive officer of the Company. I understand that nothing herein is intended to or shall prevent me from communicating directly with, cooperating with, or providing information to, any federal, state or local government regulator, including, but not limited to, the U.S. Securities and Exchange Commission, the U.S. Commodity Futures Trading Commission, or the U.S. Department of Justice.

Date:	
	(Employee's Signature)
	(Printed or Typed Name of Employee)

## Exhibit B

## California Labor Code

California Labor Code § 2870. Application of provision providing that employee shall assign or offer to assign rights in invention to employer.

- (a) Any provision in an employment agreement which provides that an employee shall assign, or offer to assign, any of his or her rights in an invention to his or her employer shall not apply to an invention that the employee developed entirely on his or her own time without using the employer's equipment, supplies, facilities, or trade secret information except for those inventions that either:
  - (1) Relate at the time of conception or reduction to practice of the invention to the employer's business, or actual or demonstrably anticipated research or development of the employer; or
  - (2) Result from any work performed by the employee for the employer.
- (b) To the extent a provision in an employment agreement purports to require an employee to assign an invention otherwise excluded from being required to be assigned under subdivision (a), the provision is against the public policy of this state and is unenforceable.

# Exhibit C

# Background Technology

List here prior contracts to assign Inventions that are now in existence between any other person or entity and you.

List here previous Inventions which you desire to have specifically excluded from the operation of this Agreement. Continue on reverse side if necessary.

## **EXHIBIT C**

## PROPOSED EXECUTIVE SEVERANCE PLAN

# THE BEAUTY HEALTH COMPANY EXECUTIVE SEVERANCE PLAN

The Beauty Health Company (the "<u>Company</u>"), has adopted this The Beauty Health Company Executive Severance Plan, including the attached Exhibits (the "<u>Plan</u>"), for the benefit of Participants (as defined below) on the terms and conditions hereinafter stated. The Plan, as set forth herein, is intended to provide severance protections to a select group of management or highly compensated employees (within the meaning of ERISA (as defined below)) in connection with qualifying terminations of employment.

- 1. **Defined Terms**. Capitalized terms used but not otherwise defined herein shall have their respective meanings set forth below:
- 1.1 "Base Salary." means, with respect to any Participant, the Participant's annual base salary rate in effect immediately prior to a Qualifying Termination.
  - 1.2 "Board" means the Board of Directors of the Company.
  - 1.3 "Cause" means, with respect to any Participant, the occurrence of any one or more of the following events:
- (a) the Participant has engaged in an act of dishonesty, theft, embezzlement or fraud, a breach of confidentiality, an unauthorized disclosure or use of inside information, customer lists, trade secrets or other confidential information, or has breached a fiduciary duty, law, rule, regulation or policy or procedure of the Company (including but not limited to policies and procedures pertaining to harassment and discrimination);
- (b) the Participant's commission of, or plea of guilty or *nolo contendere* to, any crime or offense (other than minor traffic violations or similar offenses), or any act by the Participant constituting a felony;
- (c) the Participant's gross or repeated neglect of, or repeated or willful failure to perform, his or her duties to the Company, or the Participant's history of substandard performance, if such substandard performance is not cured to the satisfaction of the Board, the Chief Executive Officer of the Company (the "CEO"), and/or the Participant's manager within ten (10) days following notice to the Participant;
- (d) the Participant has breached any of the provisions of any employment, confidentiality, restrictive covenant or other agreement between the Participant and the Company or an affiliate thereof;
  - (e) actions by the Participant involving malfeasance or gross negligence in the performance of, the Participant's duties;

- (f) the Participant's failure or refusal to perform the Participant's duties to the Company on an exclusive and full-time basis if such failure to perform the Participant's duties is not cured to the satisfaction of the Board, the CEO, and/or the Participant's manager (as applicable) within ten (10) days following notice to the Participant;
- (g) the Participant's willful violation of any material rule, regulation, or policy of the Company or the Board (or such Participant's manager) applicable to similarly-situated employees of the Company generally, including, without limitation, rules, regulations, or policies addressing confidentiality, non-solicitation or non-competition, if such violation is not cured to the satisfaction of the Board, the CEO, and/or the Participant's manager (as applicable) within ten (10) days following notice to the Participant;
- (h) the Participant's use of alcohol or illicit drugs in a manner that has or would reasonably be expected to have a detrimental effect on the Participant's performance, the Participant's duties to Company, or the reputation of the Company or its affiliates; or
- (i) the Participant's performance of acts which are or could reasonably be expected to become materially detrimental to the image, reputation, finances or business of the Company or any of its affiliates, including but limited to the Participant's commission of unlawful harassment or discrimination.

For purposes of the foregoing definition of Cause, no act or failure to act by a Participant shall be deemed willful or intentional if performed in good faith and with the reasonable belief that the action or inaction was in the best interests of the Company and its affiliates.

- 1.4 "Change in Control" shall have the meaning set forth in the Company's 2021 Incentive Award Plan, as may be amended from time to time, or any successor equity incentive plan established by the Company.
- 1.5 "<u>CIC Protection Period</u>" means the period beginning on (and including) the date on which a Change in Control is consummated and ending on (and including) the nine (9)-month anniversary thereof.
  - 1.6 "CIC Termination" means a Qualifying Termination which occurs during the CIC Protection Period.
  - 1.7 "COBRA" means the Consolidated Omnibus Budget Reconciliation Act of 1985.
  - 1.8 "Code" means the Internal Revenue Code of 1986, as amended from time to time, or any successor thereto.
- 1.9 "Committee" means the Compensation Committee of the Board, or such other committee as may be appointed by the Board to administer the Plan.
  - 1.10 "Date of Termination" means the effective date of the termination of the Participant's employment.
- 1.11 "Employee" means an individual who is an employee (within the meaning of Code Section 3401(c)) of the Company or any of its subsidiaries.
  - 1.12 "ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and the regulations promulgated thereunder.
- 1.13 "Good Reason" means, with respect to any Participant and without the Participant's written consent, the occurrence of any one or more of the following events:

- (a) a material reduction in the Participant's Base Salary or Target Bonus except in connection with across-the-board salary reductions (and corresponding target bonus reductions) imposed on substantially all of the Company's senior executives not to exceed, in the aggregate, ten percent (10%) during any twelve (12)-month period;
- (b) the relocation of the Participant's principal place of employment to a location that is greater than fifty (50) miles from the Participant's principal place of employment as of the date on which he or she becomes a Participant in the Plan if that relocation increases the Participant's commute by fifty (50) miles or more; or
- (c) a material reduction in the Participant's title, duties or responsibilities to the Company (other than during a period of physical or mental incapacity, and specifically <u>not to include</u> any change in Participant's title, duties or responsibilities as a result of any Change in Control or any other transaction that does not qualify as a Change in Control).

Notwithstanding the foregoing, the Participant will not be deemed to have resigned for Good Reason unless (1) the Participant provides the Company with written notice setting forth in reasonable detail the facts and circumstances claimed by the Participant to constitute Good Reason within thirty (30) days after the date of the occurrence of any event that the Participant knows or reasonably should have known to constitute Good Reason, (2) the Company fails to cure such acts or omissions within thirty (30) days following its receipt of such notice, and (3) the effective date of the Participant's termination for Good Reason occurs no later than sixty (60) days after the expiration of the Company's cure period.

- 1.14 "Participant" means each Employee with a title of Vice President or higher who is selected by the Administrator (or designee thereof in accordance with Section 3 hereof) to participate in the Plan and is provided with (and, if applicable, countersigns) a Participation Notice in accordance with Section 13.2 hereof, other than any Employee who, at the time of his or her termination of employment, is covered by an employment or other individual agreement with the Company or a subsidiary thereof that provides for cash severance or termination benefits. For the avoidance of doubt, retention bonus payments, change in control bonus payments and other similar payments shall not constitute "cash severance" for purposes of this definition.
- 1.15 "Prior-Year Bonus" means, with respect to any Participant, the Participant's earned but unpaid annual cash performance bonus, if any, for the Company's fiscal year ending immediately prior to the fiscal year in which the Date of Termination occurs. For clarity, if the Date of Termination occurs on or after the date on which the Company pays annual cash performance bonuses for the Company's fiscal year ending immediately prior to the fiscal year in which the Date of Termination occurs, then there shall be no Prior-Year Bonus.
- 1.16 "<u>Pro-Rata Target Bonus</u>" means, with respect to any Participant, a pro-rated portion of the Participant's Target Bonus amount for the year in which the Date of Termination occurs, determined by multiplying the Participant's Target Bonus for the year in which the Date of Termination occurs by a fraction, the numerator of which equals the number of days during which the Participant was employed by the Company or its subsidiaries during the calendar year in which the Date of Termination occurs and the denominator of which equals 365 or 366 (as applicable).

- 1.17 "Qualifying Termination" means a termination of the Participant's employment with the Company or a subsidiary thereof, as applicable, (a) by the Company or such subsidiary, as applicable, without Cause or (b) by the Participant for Good Reason. Notwithstanding anything contained herein, in no event shall a Participant be deemed to have experienced a Qualifying Termination if, (a) if such Participant is offered and/or accepts a comparable employment position with the Company or any subsidiary, or (b) if in connection with a Change in Control or any other corporate transaction or sale of assets involving the Company or any subsidiary, such Participant is offered and accepts a comparable employment position with the successor or purchaser entity (or an affiliate thereof), as applicable. A Qualifying Termination shall not include a termination due to the Participant's death or disability.
- 1.18 "Severance Benefits" means, collectively, the Prior-Year Bonus, the Cash Salary Severance, the Pro-Rata Target Bonus, and the COBRA Benefits to which a Participant may become entitled pursuant to the Plan.
- 1.19 "Severance Classification" means, with respect to any Participant, his or her designation as a "Tier 1," "Tier 2" or "Tier 3" participant in the Plan.
- 1.20 "<u>Severance Period</u>" means, with respect to any Participant, the number of months following the Participant's Date of Termination during which the Participant is entitled to the Cash Salary Severance and COBRA Benefits, as determined in accordance with <u>Exhibit A</u> or Exhibit B, as applicable, attached hereto (based on the Participant's Severance Classification).
- 1.21 "<u>Target Bonus</u>" means, with respect to any Participant, the Participant's target annual cash performance bonus, if any, for the year in which the Date of Termination occurs. For the avoidance of doubt, with respect to any Participant who is not eligible to receive an annual cash performance bonus as part of his or her annual compensation, such Participant's Target Bonus shall be equal to zero.
- 2. **Effectiveness of the Plan; Notification**. The Plan shall become effective on the date on which it is adopted by the Board. The Administrator shall, pursuant to a Participation Notice, notify each Participant that such Participant has been selected to participate in the Plan and of such Participant's Severance Classification.
- 3. **Administration**. Subject to Section 13.4 hereof, the Plan shall be interpreted, administered and operated by the Committee (the "<u>Administrator</u>"), which shall have complete authority, subject to the express provisions of the Plan, to interpret the Plan, to prescribe, amend and rescind rules and regulations relating to the Plan, and to make all other determinations necessary or advisable for the administration of the Plan. The Administrator may delegate any of its duties hereunder to a subcommittee, or to such person or persons from time to time as it may designate (other than to any Participant in the Plan). All decisions, interpretations and other actions of the Administrator (including with respect to whether a Qualifying Termination has occurred) shall be final, conclusive and binding on all parties who have an interest in the Plan.

## 4. Severance Benefits.

- 4.1 <u>Eligibility</u>. Each Employee who qualifies as a Participant and who experiences a Qualifying Termination is eligible to receive Severance Benefits under the Plan.
- 4.2 <u>Qualifying Termination Payment</u>. In the event that a Participant experiences a Qualifying Termination (other than a CIC Termination), then, subject to Sections 4.5 and 4.6 hereof and further subject to the Participant's execution of a Release that becomes effective and irrevocable in accordance with Section 4.4 hereof, and subject to any additional requirements specified in the Plan, the Company shall pay or provide to the Participant the following Severance Benefits:

- (a) <u>Prior-Year Bonus</u>. The Company shall pay to the Participant an amount equal to the Participant's Prior-Year Bonus, payable within seventy (70) days following the Date of Termination; *provided*, that if the aggregate period during which the Participant is entitled to consider and/or revoke the Release spans two calendar years, the Prior-Year Bonus shall be paid in the second (2<sup>nd</sup>) such calendar year;
- (b) <u>Cash Salary Severance</u>. The Company shall pay to the Participant an amount equal to the amount determined in accordance with <u>Exhibit A</u> attached hereto (based on the Participant's Severance Classification) (the "<u>Cash Salary Severance</u>"). Subject to Section 6.2 hereof, the Cash Salary Severance (as set forth on <u>Exhibit A</u>) shall be paid to the Participant in accordance with the Company's usual payroll periods during the Severance Period (as set forth on <u>Exhibit A</u> and based on the Participant's Severance Classification) commencing on the Date of Termination; *provided*, that no such payments shall be made prior to the date on which the Release becomes effective and irrevocable and, if the aggregate period during which the Participant is entitled to consider and/or revoke the Release spans two calendar years, no payments under this Section 4.2(b) shall be made prior to the beginning of the second (2nd) such calendar year (and any such payments otherwise payable prior thereto (if any) shall instead be paid on the first (1st) regularly scheduled Company payroll date in the second (2nd) such calendar year).
- (c) <u>Pro-Rata Target Bonus</u>. The Company shall pay to the Participant an amount equal to his or her Pro-Rated Target Bonus, payable within seventy (70) days following the Date of Termination; *provided*, that if the aggregate period during which the Participant is entitled to consider and/or revoke the Release spans two calendar years, the Pro-Rata Target Bonus will be paid in the second (2<sup>nd</sup>) such calendar year).
- (d) <u>COBRA</u>. Subject to the Participant's valid election to continue health care coverage under Section 4980B of the Code, to the extent that the Participant is eligible to do so, then the Company shall reimburse the Participant for the Participant and the Participant's eligible dependents with coverage under its group health plans at the same levels and at the same cost to Participant as would have applied if the Participant's employment had not been terminated based on Participant's elections in effect on the Date of Termination until the earlier of the end of the month during which the Participant's Severance Period (as determined in accordance with <u>Exhibit A</u> attached hereto and based on the Participant's Severance Classification)) ends or the date the Participant becomes covered by a group health insurance program provided by a subsequent employer (the "<u>COBRA Benefits</u>"). Notwithstanding the foregoing, (i) if any plan pursuant to which the COBRA Benefits are provided is not, or ceases prior to the expiration of the period of continuation coverage to be, exempt from the application of Code Section 409A under Treasury Regulation Section 1.409A-1(a)(5), or (ii) the Company is otherwise unable to continue to cover the Participant under its group health plans without incurring penalties (including without limitation, pursuant to Section 2716 of the Public Health Service Act or the Patient Protection and Affordable Care Act), then, in either case, an amount equal to each remaining Company reimbursement shall thereafter be paid to the Participant in substantially equal monthly installments over the Severance Period (or the remaining portion thereof).

- 4.3 <u>CIC Termination Payment</u>. In the event that a Participant experiences a CIC Termination, then, subject to Sections 4.5 and 4.6 hereof and further subject to the Participant's execution of a Release that becomes effective and irrevocable in accordance with Section 4.4 hereof, and subject to any additional requirements specified in the Plan, then the Company shall pay or provide to the Participant, as applicable, the Severance Benefits set forth in Sections 4.2(a)-(d) hereof; *provided*, *however*, that the amount of the Cash Salary Severance and the Severance Period shall be determined in accordance with <u>Exhibit B</u> attached hereto (instead of in accordance with <u>Exhibit A</u>) based on the Participant's Severance Classification.
- 4.4 <u>Release</u>. Notwithstanding anything herein to the contrary, no Participant shall be eligible or entitled to receive or retain any Severance Benefits under the Plan unless he or she executes a general release of claims in a form prescribed by the Company (the "<u>Release</u>") that becomes effective and irrevocable no more than sixty (60) days after the Date of Termination.
- 4.5 <u>Impact of Subsequent Employment</u>. Notwithstanding anything herein to the contrary, unless otherwise determined by the Administrator, in the event that, after his or her Date of Termination and prior to the end of his or her Severance Period, a Participant commences paid employment with any other entity, the payments and benefits described in Section 4.2 or 4.3 above, as applicable, shall cease immediately upon the Participant's commencement of such employment. Each Participant hereby agrees to give the Company prompt written notice of such subsequent paid employment no later than the date on which such employment commences.
- 4.6 Non-U.S. Employees. Notwithstanding anything in the Plan to the contrary, any Participant that resides outside of the United States (each, a "Non-U.S. Participant") and is entitled to receive severance, notice or similar termination payments and/or benefits under the laws of his or her country of residence upon his or her termination of employment with the Company and its subsidiaries (collectively, "Statutory Severance") and that becomes eligible to receive Severance Benefits under the Plan shall be entitled to receive either (i) the payments and benefits described in Section 4.2 or 4.3 above, as applicable, or (ii) such Non-US Participant's Statutory Severance, whichever is greater.
- 5. **Limitations**. Notwithstanding any provision of the Plan to the contrary, if a Participant's status as an Employee is terminated for any reason other than due to a Qualifying Termination, the Participant shall not be entitled to receive any Severance Benefits under the Plan, and the Company shall not have any obligation to such Participant under the Plan.

## 6. Section 409A.

- 6.1 General. To the extent applicable, the Plan shall be interpreted and applied consistent and in accordance with Code Section 409A and Department of Treasury regulations and other interpretive guidance issued thereunder. Notwithstanding any provision of the Plan to the contrary, to the extent that the Administrator determines that any payments or benefits under the Plan may not be either compliant with or exempt from Code Section 409A and related Department of Treasury guidance, the Administrator may in its sole discretion adopt such amendments to the Plan or take such other actions that the Administrator determines are necessary or appropriate to (a) exempt the compensation and benefits payable under the Plan from Code Section 409A and/or preserve the intended tax treatment of such compensation and benefits, or (b) comply with the requirements of Code Section 409A and related Department of Treasury guidance; *provided*, *however*, that this Section 6.1 shall not create any obligation on the part of the Administrator to adopt any such amendment or take any other action, nor shall the Company have any liability for failing to do so.
- 6.2 <u>Potential Six-Month Delay.</u> Notwithstanding anything to the contrary in the Plan, no amounts shall be paid to any Participant under the Plan during the six (6)-month period following such Participant's "separation from service" (within the meaning of Code Section 409A(a)(2)(A)(i) and Treasury Regulation Section 1.409A-1(h)) to the extent that the Administrator determines that paying such amounts at the time or times indicated in the Plan would result in a prohibited distribution under Code Section 409A(a)(2)(B)(i). If the payment of any such amounts is delayed as a result of the previous sentence, then

on the first business day following the end of such six (6)-month period (or such earlier date upon which such amount can be paid under Code Section 409A without resulting in a prohibited distribution, including as a result of the Participant's death), the Participant shall receive payment of a lump-sum amount equal to the cumulative amount that would have otherwise been payable to the Participant during such six (6)-month period without interest thereon.

- 6.3 <u>Separation from Service</u>. A termination of employment shall not be deemed to have occurred for purposes of any provision of the Plan providing for the payment of any amounts or benefits that constitute "nonqualified deferred compensation" under Code Section 409A upon or following a termination of employment unless such termination is also a "separation from service" within the meaning of Code Section 409A and, for purposes of any such provision of the Plan, references to a "termination," "termination of employment" or like terms shall mean "separation from service".
- 6.4 Reimbursements. To the extent that any payments or reimbursements provided to a Participant under the Plan are deemed to constitute compensation to the Participant to which Treasury Regulation Section 1.409A-3(i)(1)(iv) would apply, such amounts shall be paid or reimbursed reasonably promptly, but not later than December 31st of the year following the year in which the expense was incurred. The amount of any such payments eligible for reimbursement in one (1) year shall not affect the payments or expenses that are eligible for payment or reimbursement in any other taxable year, and the Participant's right to such payments or reimbursement of any such expenses shall not be subject to liquidation or exchange for any other benefit.
- 6.5 <u>Installments</u>. For purposes of applying the provisions of Code Section 409A to the Plan, each separately identified amount to which a Participant is entitled under the Plan shall be treated as a separate payment. In addition, to the extent permissible under Code Section 409A, the right to receive any installment payments under the Plan shall be treated as a right to receive a series of separate payments and, accordingly, each such installment payment shall at all times be considered a separate and distinct payment as permitted under Treasury Regulation Section 1.409A-2(b)(2)(iii). Whenever a payment under the Plan specifies a payment period with reference to a number of days, the actual date of payment within the specified period shall be within the sole discretion of the Company.

## 7. Limitation on Payments.

7.1 <u>Best Pay Cap.</u> Notwithstanding any other provision of the Plan, in the event that any payment or benefit received or to be received by a Participant (whether pursuant to the terms of the Plan or any other plan, arrangement or agreement) (all such payments and benefits, including the Severance Benefits, being hereinafter referred to as the "<u>Total Payments</u>") would be subject (in whole or part), to the excise tax imposed under Code Section 4999 (the "<u>Excise Tax</u>"), then, after taking into account any reduction in the Total Payments provided by reason of Code Section 280G in any other plan, arrangement or agreement, the cash Severance Benefits under the Plan shall first be reduced, and any non-cash Severance Benefits hereunder shall thereafter be reduced, to the extent necessary so that no portion of the Total Payments is subject to the Excise Tax but only if (a) the net amount of such Total Payments, as so reduced (and after subtracting the net amount of federal, state and local income taxes on such reduced Total Payments) is greater than or equal to (b) the net amount of such Total Payments without such reduction (but after subtracting the net amount of federal, state and local income taxes on such Total Payments and the amount of Excise Tax to which the Participant would be subject in respect of such unreduced Total Payments and after taking into account the phase out of itemized deductions and personal exemptions attributable to such unreduced Total Payments).

- 7.2 <u>Certain Exclusions</u>. For purposes of determining whether and the extent to which the Total Payments will be subject to the Excise Tax, (a) no portion of the Total Payments, the receipt or retention of which the Participant has waived at such time and in such manner so as not to constitute a "payment" within the meaning of Code Section 280G(b), will be taken into account; (b) no portion of the Total Payments will be taken into account which, in the written opinion of an independent, nationally recognized accounting firm (the "<u>Independent Advisors</u>") selected by the Company, does not constitute a "parachute payment" within the meaning of Code Section 280G(b)(2) (including by reason of Code Section 280G(b)(4)(A)) and, in calculating the Excise Tax, no portion of such Total Payments will be taken into account which, in the opinion of Independent Advisors, constitutes reasonable compensation for services actually rendered, within the meaning of Code Section 280G(b)(4)(B), in excess of the "base amount" (as defined in Code Section 280G(b)(3)) allocable to such reasonable compensation; and (c) the value of any non-cash benefit or any deferred payment or benefit included in the Total Payments shall be determined by the Independent Advisors in accordance with the principles of Code Sections 280G(d)(3) and (4).
- 8. **No Mitigation**. No Participant shall be required to seek other employment or attempt in any way to reduce or mitigate any Severance Benefits payable under the Plan and the amount of any such Severance Benefits shall not be reduced by any other compensation paid or provided to any Participant following such Participant's termination of service.

## 9. Successors.

- 9.1 <u>Company Successors</u>. The Plan shall inure to the benefit of and shall be binding upon the Company and its successors and assigns. Any successor (whether direct or indirect and whether by purchase, lease, merger, consolidation, liquidation or otherwise) to all or substantially all of the Company's business and/or assets shall assume and agree to perform the obligations of the Company under the Plan.
- 9.2 <u>Participant Successors</u>. The Plan shall inure to the benefit of and be enforceable by each Participant's personal or legal representatives, executors, administrators, successors, heirs, distributees, devisees, legatees or other beneficiaries. If a Participant dies while any amount remains payable to such Participant hereunder, all such amounts shall be paid in accordance with the terms of the Plan to the executors, personal representatives or administrators of such Participant's estate.
- 10. **Notices**. All communications relating to matters arising under the Plan shall be in writing and shall be deemed to have been duly given when hand delivered, faxed, emailed or mailed by reputable overnight carrier or United States certified mail, return receipt requested, addressed, if to a Participant, to the address on file with the Company or to such other address as the Participant may have furnished to the other in writing in accordance herewith and, if to the Company, to such address as may be specified from time to time by the Administrator, except that notice of change of address shall be effective only upon actual receipt.

## 11. Claims Procedure; Arbitration.

11.1 <u>Claims</u>. Generally, Participants are not required to present a formal claim in order to receive benefits under the Plan. If, however, any person (the "<u>Claimant</u>") believes that benefits are being denied improperly, that the Plan is not being operated properly, that fiduciaries of the Plan have breached their duties, or that the Claimant's legal rights are being violated with respect to the Plan, the Claimant must file a formal claim, in writing, with the Administrator. This requirement applies to all claims that any Claimant has with respect to the Plan, including claims against fiduciaries and former fiduciaries, except to the extent the Administrator determines, in its sole discretion that it does not have the power to grant all relief reasonably being sought by the Claimant. A formal claim must be filed within ninety (90) days after the date the Claimant first knew or should have known of the facts on which the claim is based, unless the Administrator consents otherwise in writing. The Administrator shall provide a Claimant, on request, with a copy of the claims procedures established under Section 11.2 hereof.

11.2 <u>Claims Procedure</u>. The Administrator has adopted procedures for considering claims (which are set forth in <u>Exhibit C</u> attached hereto), which it may amend or modify from time to time, as it sees fit. These procedures shall comply with all applicable legal requirements. These procedures may provide that final and binding arbitration shall be the ultimate means of contesting a denied claim (even if the Administrator or its delegates have failed to follow the prescribed procedures with respect to the claim). The right to receive benefits under the Plan is contingent on a Claimant using the prescribed claims and arbitration procedures to resolve any claim.

#### 12. Covenants.

- 12.1 <u>Restrictive Covenants</u>. A Participant's right to receive and/or retain the Severance Benefits payable under this Plan is conditioned upon and subject to the Participant's continued compliance with any restrictive covenants (e.g., confidentiality, non-solicitation, non-competition, non-disparagement) contained in any other written agreement between the Participant and the Company, as in effect on the date of the Participant's Qualifying Termination.
- 12.2 <u>Return of Property</u>. A Participant's right to receive and/or retain the Severance Benefits payable under the Plan is conditioned upon the Participant's return to the Company of all Company documents (and all copies thereof) and other Company property (in each case, whether physical, electronic or otherwise) in the Participant's possession or control.

## 13. Miscellaneous.

- 13.1 Entire Plan; Relation to Other Agreements. The Plan, together with any Participation Notice issued in connection with the Plan, contains the entire understanding of the parties relating to the subject matter hereof and supersedes any prior agreement, arrangement and understanding between any Participant, on the one hand, and the Company and/or any subsidiary, on the other hand, with respect to the subject matter hereof. Severance Benefits payable under the Plan are not intended to duplicate any other severance benefits payable to a Participant by the Company. By participating in the Plan and accepting the Severance Benefits hereunder, the Participant acknowledges and agrees that any prior agreement, arrangement and understanding between any Participant, on the one hand, and the Company and/or any subsidiary, on the other hand, with respect to the subject matter hereof is hereby revoked and ineffective with respect to the Participant.
- 13.2 <u>Participation Notices</u>. The Administrator shall have the authority, in its sole discretion, to select Employees to participate in the Plan and to provide written notice to any such Employee that he or she is a Participant in, and eligible to receive Severance Benefits under, the Plan (a "<u>Participation Notice</u>") at or any time prior to his or her termination of employment.
- 13.3 No Right to Continued Service. Nothing contained in the Plan shall (a) confer upon any Participant any right to continue as an employee of the Company or any subsidiary, (b) constitute any contract of employment or agreement to continue employment for any particular period, or (c) interfere in any way with the right of the Company to terminate a service relationship with any Participant, with or without Cause.

- 13.4 <u>Termination and Amendment of Plan</u>. Prior to the consummation of a Change in Control, the Plan may be amended or terminated by the Administrator at any time and from time to time, in its sole discretion. For a period of nine (9) months from and after the consummation of a Change in Control, the Plan may not be amended, modified, suspended or terminated except with the express written consent of each Participant who would be adversely affected by any such amendment, modification, suspension or termination. After the expiration of such nine (9)-month period, and subject to Section 2 hereof, the Plan may again be amended or terminated by the Administrator at any time and from time to time, in its sole discretion (provided, that no such amendment or termination shall adversely affect the rights of any Participant who has experienced a Qualifying Termination on or prior to such amendment or termination).
- 13.5 <u>Survival</u>. Section 7 (Limitation on Payments), Section 11 (Claims Procedure; Arbitration) and Section 12 (Covenants) hereof shall survive the termination or expiration of the Plan and shall continue in effect.
- 13.6 <u>Severance Benefit Obligations</u>. Notwithstanding anything contained herein, Severance Benefits paid or provided under the Plan may be paid or provided by the Company or any subsidiary employer, as applicable.
- 13.7 Withholding. The Company shall have the authority and the right to deduct and withhold an amount sufficient to satisfy federal, state, local and foreign taxes required by law to be withheld with respect to any Severance Benefits payable under the Plan.
- 13.8 <u>Benefits Not Assignable</u>. Except as otherwise provided herein or by law, no right or interest of any Participant under the Plan shall be assignable or transferable, in whole or in part, either directly or by operation of law or otherwise, including without limitation by execution, levy, garnishment, attachment, pledge or in any manner; no attempted assignment or transfer thereof shall be effective; and no right or interest of any Participant under the Plan shall be liable for, or subject to, any obligation or liability of such Participant. When a payment is due under the Plan to a Participant who is unable to care for his or her affairs, payment may be made directly to his or her legal guardian or personal representative.
- 13.9 <u>Applicable Law</u>. The Plan is intended to be an unfunded "top hat" pension plan within the meaning of U.S. Department of Labor Regulation Section 2520.104-23 and shall be interpreted, administered, and enforced as such in accordance with ERISA. To the extent that state law is applicable, the statutes and common law of the State of Delaware, excluding any that mandate the use of another jurisdiction's laws, will apply.
- 13.10 <u>Validity</u>. The invalidity or unenforceability of any provision of the Plan shall not affect the validity or enforceability of any other provision of the Plan, which shall remain in full force and effect.
- 13.11 <u>Captions</u>. The captions contained in the Plan are for convenience only and shall have no bearing on the meaning, construction or interpretation of the Plan's provisions.
  - 13.12 Expenses. The expenses of administering the Plan shall be borne by the Company or its successor, as applicable.
- 13.13 <u>Unfunded Plan</u>. The Plan shall be maintained in a manner to be considered "unfunded" for purposes of ERISA. The Company shall be required to make payments only as benefits become due and payable. No person shall have any right, other than the right of an unsecured general creditor against the Company, with respect to the benefits payable hereunder, or which may be payable hereunder, to any Participant, surviving spouse or beneficiary hereunder. If the Company, acting in its sole discretion,

establishes a reserve or other fund associated with the Plan, no person shall have any right to or interest in any specific amount or asset of such reserve or fund by reason of amounts which may be payable to such person under the Plan, nor shall such person have any right to receive any payment under the Plan except as and to the extent expressly provided in the Plan. The assets in any such reserve or fund shall be part of the general assets of the Company, subject to the control of the Company.

\* \* \* \* \*

Exhibit A

# CALCULATION OF NON-CHANGE IN CONTROL SEVERANCE AMOUNTS

Severance Classification	Cash Salary Severance	Severance Period
Tier 1	12 months of Base Salary	12 months
Tier 2	6 months of Base Salary	6 months
Tier 3	3 months of Base Salary	3 months

# CALCULATION OF CHANGE IN CONTROL SEVERANCE AMOUNTS

Severance Classification	Cash Salary Severance	Severance Period
Tier 1	1.12 months of Base Salary 2.100% Target Bonus	12 months
Tier 2	<ul><li>1.6 months of Base Salary</li><li>2.50% Target Bonus</li></ul>	6 months
Tier 3	<ul><li>1.3 months of Base Salary</li><li>2.25% Target Bonus</li></ul>	3 months

#### DETAILED CLAIMS PROCEDURES

Section 1.1. <u>Claim Procedure</u>. Claims for benefits under the Plan shall be administered in accordance with Section 503 of ERISA and the Department of Labor Regulations thereunder. The Administrator shall have the right to delegate its duties under this Exhibit and all references to the Administrator shall be a reference to any such delegate, as well. The Administrator shall make all determinations as to the rights of any Participant, beneficiary, alternate payee or other person who makes a claim for benefits under the Plan (each, a "<u>Claimant</u>"). A Claimant may authorize a representative to act on his or her behalf with respect to any claim under the Plan. A Claimant who asserts a right to any benefit under the Plan he has not received, in whole or in part, must file a written claim with the Administrator. All written claims shall be submitted to Deborah Rodriguez, Chief Talent Officer; 2165 E Spring St., Long Beach, CA 90806; DRodriguez@hydrafacial.com.

- (a) Regular Claims Procedure. The claims procedure in this subsection (a) shall apply to all claims for Plan benefits.
- (1) <u>Timing of Denial</u>. If the Administrator denies a claim in whole or in part (an "<u>adverse benefit determination</u>"), then the Administrator will provide notice of the decision to the Claimant within a reasonable period of time, not to exceed ninety (90) days after the Administrator receives the claim, unless the Administrator determines that any extension of time for processing is required. In the event that the Administrator determines that such an extension is required, written notice of the extension will be furnished to the Claimant before the end of the initial ninety (90) day review period. The extension will not exceed a period of ninety (90) days from the end of the initial ninety (90) day period, and the extension notice will indicate the special circumstances requiring such extension of time and the date by which the Administrator expects to render the benefit decision.
- (2) <u>Denial Notice</u>. The Administrator shall provide every Claimant who is denied a claim for benefits with a written or electronic notice of its decision. The notice will set forth, in a manner to be understood by the Claimant:
  - i. the specific reason or reasons for the adverse benefit determination;
  - ii. reference to the specific Plan provisions on which the determination is based;
  - iii. a description of any additional material or information necessary for the Claimant to perfect the claim and an explanation as to why such information is necessary; and
  - iv. an explanation of the Plan's appeal procedure and the time limits applicable to such procedures, including a statement of the Claimant's right to bring an action under Section 502(a) of ERISA after receiving a final adverse benefit determination upon appeal.

- (3) <u>Appeal of Denial</u>. The Claimant may appeal an initial adverse benefit determination by submitting a written appeal to the Administrator within sixty (60) days of receiving notice of the denial of the claim. The Claimant:
  - i. may submit written comments, documents, records and other information relating to the claim for benefits;
  - ii. will be provided, upon request and without charge, reasonable access to and copies of all documents, records and other information relevant to the Claimant's claim for benefits; and
  - iii. will receive a review that takes into account all comments, documents, records and other information submitted by the Claimant relating to the appeal, without regard to whether such information was submitted or considered in the initial benefit determination.
- (4) <u>Decision on Appeal</u>. The Administrator will conduct a full and fair review of the claim and the initial adverse benefit determination. The Administrator holds regularly scheduled meetings at least quarterly. The Administrator shall make a benefit determination no later than the date of the regularly scheduled meeting that immediately follows the Plan's receipt of an appeal request, unless the appeal request is filed within thirty (30) days preceding the date of such meeting. In such case, a benefit determination may be made by no later than the date of the second (2nd) regularly scheduled meeting following the Plan's receipt of the appeal request. If special circumstances require a further extension of time for processing, a benefit determination shall be rendered no later than the third (3rd) regularly scheduled meeting of the Administrator following the Plan's receipt of the appeal request. If such an extension of time for review is required, the Administrator shall provide the Claimant with written notice of the extension, describing the special circumstances and the date as of which the benefit determination will be made, prior to the commencement of the extension. The Administrator generally cannot extend the review period any further unless the Claimant voluntarily agrees to a longer extension. The Administrator shall notify the Claimant of the benefit determination as soon as possible but not later than five (5) days after it has been made.
- (5) <u>Notice of Determination on Appeal</u>. The Administrator shall provide the Claimant with written or electronic notification of its benefit determination on review. In the case of an adverse benefit determination, the notice shall set forth, in a manner intended to be understood by the Claimant:
  - i. the specific reason or reasons for the adverse benefit determination;
  - ii. reference to the specific Plan provisions on which the adverse benefit determination is based;

- iii. a statement that the Claimant is entitled to receive, upon request and without charge, reasonable access to, and copies of, all documents, records and other information relevant to the claim for benefits;
- iv. a statement describing any voluntary appeal procedures offered by the Plan and the Claimant's right to obtain the information about such procedures; and
- v. a statement of the Claimant's right to bring an action under Section 502(a) of ERISA.
- (c) Exhaustion; Judicial Proceedings. No action at law or in equity shall be brought to recover benefits under the Plan until the claim and appeal rights described in the Plan have been exercised and the Plan benefits requested in such appeal have been denied in whole or in part. If any judicial proceeding is undertaken to appeal the denial of a claim or bring any other action under ERISA other than a breach of fiduciary claim, the evidence presented may be strictly limited to the evidence timely presented to the Administrator. Any such judicial proceeding must be filed by the earlier of:
  (a) one (1) year after the Administrator's final decision regarding the claim appeal or (b) one (1) year after the Participant or other Claimant commenced payment of the Plan benefits at issue in the judicial proceeding.
- (d) <u>Administrator's Decision is Binding</u>. Benefits under the Plan shall be paid only if the Administrator decides in its sole discretion that a Claimant is entitled to them. In determining claims for benefits, the Administrator has the authority to interpret the Plan, to resolve ambiguities, to make factual determinations, and to resolve questions relating to eligibility for and amount of benefits. Subject to applicable law, any decision made in accordance with the above claims procedures is final and binding on all parties and shall be given the maximum possible deference allowed by law. A misstatement or other mistake of fact shall be corrected when it becomes known and the Administrator shall make such adjustment on account thereof as it considers equitable and practicable.

THE P	REA	IITV	HEA1	TH	CON	ЛРΑ	NV
-------	-----	------	------	----	-----	-----	----

## 2021 INCENTIVE AWARD PLAN

## STOCK OPTION GRANT NOTICE

The Beauty Health Company, a Delaware corporation (the "Company") has granted to the participant listed below ("Participant") the stock option (the "Option") described in this Stock Option Grant Notice (the "Grant Notice"), subject to the terms and conditions of The Beauty Health Company 2021 Incentive Award Plan (as amended from time to time, the "Plan") and the Stock Option Agreement attached hereto as Exhibit A (the "Agreement"), both of which are incorporated into this Grant Notice by reference. Capitalized terms not specifically defined in this Grant Notice or the Agreement have the meanings given to them in the Plan.

z ur uespuner	L	,
Grant Date:	[	], 2021
Exercise Price per Share:	[	]
Shares Subject to the Option:		
Final Expiration Date:	[	]
Vesting Commencement Date:	Г	1 2021

Participant:

**Vesting Schedule:** 

Twenty-five percent (25%) of the Shares subject to the Option will vest on each of the first four anniversaries of the Vesting Commencement Date, subject to Participant's continued status as a Service Provider through the applicable vesting date. Notwithstanding the foregoing, (x) if Participant incurs a Termination of Service due to Participant's death or Disability, subject to and conditioned upon Participant's (or Participant's estate's) timely execution and non-revocation of a release of claims in a form prescribed by the Company (a "Release") that becomes effective and irrevocable no later than sixty (60) days following such Termination of Service (the date such Release becomes effective and irrevocable, the "Release Effective Date"), the Option will vest in full (to the extent then-unvested) upon the Release Effective Date (and shall remain outstanding and eligible to vest through the Release Effective Date and shall automatically be forfeited if the Release does not become effective and irrevocable on or prior to the sixtieth (60th) day following such termination), and (y) if a Change in Control is consummated more than twelve (12) months after [Closing Date], 2021 (the "Closing Date") and Participant's status as a Service Provider is terminated by the Company or its Subsidiaries without Cause (as defined in the Agreement) or due to Participant's resignation for Good Reason (as defined in the Employment Agreement between Participant and the Company, dated [ ]), in either case, within twelve (12) months following the consummation of such Change in

Control, subject to and conditioned upon Participant's timely execution and non-revocation of a Release that becomes effective and irrevocable no later than sixty (60) days following such Termination of Service, the Option will vest in full (to the extent then-unvested) upon the Release Effective Date (and shall remain outstanding and eligible to vest through the Release Effective Date and shall automatically be forfeited if the Release does not become effective and irrevocable on or prior to the sixtieth (60th) day following such termination).

If (i) a Change in Control occurs within twelve (12) months following the Closing Date (or pursuant to a binding agreement entered into within such twelve (12) month period) and (ii) Participant's status as a Service Provider is terminated by the Company or its Subsidiaries without Cause (as defined in the Agreement) or due to Participant's resignation for Good Reason within twelve (12) months after such Change in Control, then, notwithstanding anything to the contrary in any other plan or agreement, such termination will not result in any accelerated vesting of the Option.

[Incentive Stock Option]/[Non-Qualified Stock Option]

## **Type of Option**

By accepting (whether in writing, electronically or otherwise) the Option, Participant agrees to be bound by the terms of this Grant Notice, the Plan and the Agreement. Participant has reviewed the Plan, this Grant Notice and the Agreement in their entirety, has had an opportunity to obtain the advice of counsel prior to executing this Grant Notice and fully understands all provisions of the Plan, this Grant Notice and the Agreement. Participant hereby agrees to accept as binding, conclusive and final all decisions or interpretations of the Administrator upon any questions arising under the Plan, this Grant Notice or the Agreement.

THE BEAUTY HEALTH COMPANY	PARTICIPANT
By:	
Name:	[Participant Name]
Title:	

#### STOCK OPTION AGREEMENT

Capitalized terms not specifically defined in this Agreement have the meanings specified in the Grant Notice or, if not defined in the Grant Notice, in the Plan.

## ARTICLE I. GENERAL

- 1.1 <u>Grant of Option</u>. The Company has granted to Participant the Option effective as of the grant date set forth in the Grant Notice (the "*Grant Date*").
- 1.2 Incorporation of Terms of Plan. The Option is subject to the terms and conditions set forth in this Agreement and the Plan, which is incorporated herein by reference. In the event of any inconsistency between the Plan and this Agreement, the terms of the Plan will control, unless it is expressly specified in in this Agreement or the Grant Notice that the specific provision of the Plan will not apply. For clarity, the foregoing sentence shall not limit the applicability of any additive language contained in this Agreement which provides supplemental or additional terms not inconsistent with the Plan.

# ARTICLE II. PERIOD OF EXERCISABILITY

- 2.1 <u>Commencement of Exercisability.</u> The Option will vest and become exercisable according to the vesting schedule in the Grant Notice (the "Vesting Schedule") except that any fraction of a Share as to which the Option would be vested or exercisable will be accumulated and will vest and become exercisable only when a whole vested Share has accumulated; provided, however, that, notwithstanding the foregoing or anything to the contrary in the Grant Notice or this Agreement, in no event may the Option be exercised (in whole or in part) prior to the date on which the Company files a Form S-8 Registration Statement covering the Shares subject to the Option. Except as otherwise set forth in the Grant Notice, the Plan or this Agreement, and unless the Administrator otherwise determines, the Option will immediately expire and be forfeited as to any portion of the Option that is not vested and exercisable as of Participant's Termination of Service for any reason (after taking into consideration any accelerated vesting and exercisability which may occur in connection with such Termination of Service, if any).
- 2.2 <u>Duration of Exercisability.</u> The Vesting Schedule is cumulative. Any portion of the Option which vests and becomes exercisable will remain vested and exercisable until the Option expires. The Option will be forfeited immediately upon its expiration.
- 2.3 <u>Expiration of Option</u>. Except as may be extended in accordance with Section 5.3 of the Plan, the Option may not be exercised to any extent by anyone after, and will expire on, the first of the following to occur:
  - (a) The final expiration date in the Grant Notice;
- (b) Except as the Administrator may otherwise approve, the expiration of three (3) months from the date of Participant's Termination of Service, unless Participant's Termination of Service is for Cause (as defined below), due to Participant's voluntary resignation or by reason of Participant's death or Disability;
- (c) Except as the Administrator may otherwise approve, the expiration of one year from the date of Participant's Termination of Service by reason of Participant's death or Disability;

- (d) Except as the Administrator may otherwise approve, the expiration of one (1) month from the date of Participant's Termination of Service due to Participant's voluntary resignation for any reason; and
  - (e) Except as the Administrator may otherwise approve, Participant's Termination of Service for Cause.
- 2.4 <u>Cause Definition</u>. As used in this Agreement, "*Cause*" means (i) if Participant is a party to a written employment or consulting agreement with the Company or a Subsidiary in which the term "cause" is defined (a "*Relevant Agreement*"), "Cause" as defined in the Relevant Agreement, and (ii) if no Relevant Agreement exists, the occurrence of any one or more of the following events:
- (a) Participant has engaged in an act of dishonesty, theft, embezzlement or fraud, a breach of confidentiality, an unauthorized disclosure or use of inside information, customer lists, trade secrets or other confidential information; has breached a fiduciary duty, law, rule, regulation or policy or procedure of the Company (including but not limited to policies and procedures pertaining to harassment and discrimination); Participant's commission of, or plea of guilty or *nolo contendere* to, any crime or offense (other than minor traffic violations or similar offenses), or any act by Participant constituting a felony;
- (b) Participant's gross or repeated neglect of, or repeated or willful failure to perform, his or her duties to the Company, or Participant's history of substandard performance, if such substandard performance is not cured to the satisfaction of the Board, the Chief Executive Officer of the Company (the "CEO") (unless Participant is the CEO), and/or Participant's manager within ten (10) days following notice to Participant;
- (c) Participant has breached any of the provisions of any employment, confidentiality, restrictive covenant or other agreement between Participant and the Company or an affiliate thereof;
  - (d) actions by Participant involving malfeasance or gross negligence in the performance of, Participant's duties;
- (e) Participant's failure or refusal to perform Participant's duties to the Company on an exclusive and full-time basis if such failure to perform Participant's duties is not cured to the satisfaction of the Board, the CEO (unless Participant is the CEO), and/or Participant's manager (as applicable) within ten (10) days following notice to the Participant;
- (f) Participant's insubordination or failure to follow the Board's (or the CEO's (unless Participant is the CEO) or such Participant's manager's) instructions;
- (g) Participant's violation of any rule, regulation, or policy of the Company or the Board (or Participant's manager) applicable to similarly-situated employees of the Company generally, including, without limitation, rules, regulations, or policies addressing confidentiality, non-solicitation or non-competition, if such violation is not cured to the satisfaction of the Board, the CEO (unless Participant is the CEO), and/or Participant's manager (as applicable) within ten (10) days following notice to the Participant;
- (h) Participant's use of alcohol or illicit drugs in a manner that has or would reasonably be expected to have a detrimental effect on Participant's performance, Participant's duties to Company, or the reputation of the Company or its affiliates; or

(i) Participant's performance of acts which are or could reasonably be expected to become materially detrimental to the image, reputation, finances or business of the Company or any of its affiliates, including but limited to the Participant's commission of unlawful harassment or discrimination.

# ARTICLE III. EXERCISE OF OPTION

- 3.1 <u>Person Eligible to Exercise</u>. During Participant's lifetime, only Participant may exercise the Option. After Participant's death, any exercisable portion of the Option may, prior to the time the Option expires, be exercised by Participant's Designated Beneficiary as provided in the Plan.
- 3.2 <u>Partial Exercise</u>. Any exercisable portion of the Option or the entire Option, if then wholly exercisable, may be exercised, in whole or in part, according to the procedures in the Plan at any time prior to the time the Option or portion thereof expires, except that the Option may only be exercised for whole Shares.

## 3.3 <u>Tax Withholding; Exercise Price</u>.

- (a) Unless the Administrator otherwise determines, the Company shall withhold, or cause to be withheld, Shares otherwise vesting or issuable under this Option in satisfaction of any exercise price and/or applicable withholding tax obligations, in accordance with the Plan. With respect to tax withholding obligations, the number of Shares which may be so withheld or surrendered shall be limited to the number of Shares which have a fair market value on the date of withholding no greater than the aggregate amount of such liabilities based on the maximum individual statutory withholding rates in Participant's applicable jurisdictions for federal, state, local and foreign income tax and payroll tax purposes that are applicable to such taxable income, in accordance with Section 9.5 of the Plan.
- (b) Participant acknowledges that Participant is ultimately liable and responsible for all taxes owed in connection with the Option, regardless of any action the Company or any Subsidiary takes with respect to any tax withholding obligations that arise in connection with the Option. Neither the Company nor any Subsidiary makes any representation or undertaking regarding the treatment of any tax withholding in connection with the awarding, vesting or exercise of the Option or the subsequent sale of Shares. The Company and the Subsidiaries do not commit and are under no obligation to structure the Option to reduce or eliminate Participant's tax liability.
- 3.4 <u>Representation</u>. Participant represents to the Company that Participant has reviewed with Participant's own tax advisors the tax consequences of this Award and the transactions contemplated by the Grant Notice and this Agreement. Participant is relying solely on such advisors and not on any statements or representations of the Company or any of its agents.

# ARTICLE IV. OTHER PROVISIONS

- 4.1 <u>Adjustments</u>. Participant acknowledges that the Option is subject to adjustment, modification and termination in certain events as provided in this Agreement and the Plan.
- 4.2 <u>Clawback</u>. The Option and the Shares issuable hereunder shall be subject to any clawback or recoupment policy in effect on the Grant Date or as may be adopted or maintained by the Company following the Grant Date, including the Dodd-Frank Wall Street Reform and Consumer Protection Act and any rules or regulations promulgated thereunder.

- 4.3 Notices. Any notice to be given under the terms of this Agreement to the Company must be in writing and addressed to the Company in care of the Company's General Counsel at the Company's principal office or the General Counsel's then-current email address or facsimile number. Any notice to be given under the terms of this Agreement to Participant must be in writing and addressed to Participant (or, if Participant is then deceased, to the Designated Beneficiary) at Participant's last known mailing address, email address or facsimile number in the Company's personnel files. By a notice given pursuant to this Section, either party may designate a different address for notices to be given to that party. Any notice will be deemed duly given when actually received, when sent by email, when sent by certified mail (return receipt requested) and deposited with postage prepaid in a post office or branch post office regularly maintained by the United States Postal Service, when delivered by a nationally recognized express shipping company or upon receipt of a facsimile transmission confirmation.
  - 4.4 <u>Titles</u>. Titles are provided herein for convenience only and are not to serve as a basis for interpretation or construction of this Agreement.
- 4.5 <u>Conformity to Securities Laws</u>. Participant acknowledges that the Plan, the Grant Notice and this Agreement are intended to conform to the extent necessary with all Applicable Laws and, to the extent Applicable Laws permit, will be deemed amended as necessary to conform to Applicable Laws.
- 4.6 <u>Successors and Assigns</u>. The Company may assign any of its rights under this Agreement to single or multiple assignees, and this Agreement will inure to the benefit of the successors and assigns of the Company. Subject to the restrictions on transfer set forth in this Agreement or the Plan, this Agreement will be binding upon and inure to the benefit of the heirs, legatees, legal representatives, successors and assigns of the parties hereto.
- 4.7 <u>Limitations Applicable to Section 16 Persons.</u> Notwithstanding any other provision of the Plan or this Agreement, if Participant is subject to Section 16 of the Exchange Act, the Plan, the Grant Notice, this Agreement and the Option will be subject to any additional limitations set forth in any applicable exemptive rule under Section 16 of the Exchange Act (including any amendment to Rule 16b-3) that are requirements for the application of such exemptive rule. To the extent Applicable Laws permit, this Agreement will be deemed amended as necessary to conform to such applicable exemptive rule.
- 4.8 <u>Entire Agreement</u>. The Plan, the Grant Notice and this Agreement (including any exhibit hereto) constitute the entire agreement of the parties and supersede in their entirety all prior undertakings and agreements of the Company and Participant with respect to the subject matter hereof.
- 4.9 <u>Severability</u>. If any portion of the Grant Notice or this Agreement or any action taken under the Grant Notice or this Agreement, in any case is held illegal or invalid for any reason, the illegality or invalidity will not affect the remaining parts of the Grant Notice and/or this Agreement (as applicable), and the Grant Notice and/or this Agreement (as applicable) will be construed and enforced as if the illegal or invalid provisions had been excluded, and the illegal or invalid action will be null and void.
- 4.10 <u>Limitation on Participant's Rights</u>. Participation in the Plan confers no rights or interests other than as herein provided. This Agreement creates only a contractual obligation on the part of the Company as to amounts payable and may not be construed as creating a trust. Neither the Plan nor any underlying program, in and of itself, has any assets. Participant will have only the rights of a general unsecured creditor of the Company with respect to amounts credited and benefits payable, if any, with respect to the Option, and rights no greater than the right to receive the Shares as a general unsecured creditor with respect to the Option, as and when exercised pursuant to the terms hereof

- 4.11 <u>Not a Contract of Employment or Service</u>. Nothing in the Plan, the Grant Notice or this Agreement confers upon Participant any right to continue in the employ or service of the Company or any Subsidiary or interferes with or restricts in any way the rights of the Company and its Subsidiaries, which rights are hereby expressly reserved, to discharge or terminate the services of Participant at any time for any reason whatsoever, with or without Cause, except to the extent expressly provided otherwise in a written agreement between the Company or a Subsidiary and Participant.
- 4.12 <u>Counterparts</u>. The Grant Notice may be executed in one or more counterparts, including by way of any electronic signature, subject to Applicable Law, each of which will be deemed an original and all of which together will constitute one instrument.
  - 4.13 <u>Incentive Stock Options</u>. If the Option is designated as an Incentive Stock Option:
- (a) Participant acknowledges that to the extent the aggregate fair market value of shares (determined as of the time the option with respect to the shares is granted) with respect to which stock options intended to qualify as "incentive stock options" under Section 422 of the Code, including the Option, are exercisable for the first time by Participant during any calendar year exceeds \$100,000 or if for any other reason such stock options do not qualify or cease to qualify for treatment as "incentive stock options" under Section 422 of the Code, such stock options (including the Option) will be treated as non-qualified stock options. Participant further acknowledges that the rule set forth in the preceding sentence will be applied by taking the Option and other stock options into account in the order in which they were granted, as determined under Section 422(d) of the Code. Participant also acknowledges that if the Option is exercised more than three months after Participant's Termination of Service, other than by reason of death or disability, the Option will be taxed as a Non-Qualified Stock Option.
- (b) Participant will give prompt written notice to the Company of any disposition or other transfer of any Shares acquired under this Agreement if such disposition or other transfer is made (i) within two years from the Grant Date or (ii) within one year after the transfer of such Shares to Participant. Such notice will specify the date of such disposition or other transfer and the amount realized, in cash, other property, assumption of indebtedness or other consideration, by Participant in such disposition or other transfer.
- 4.14 <u>Governing Law</u>. The Grant Notice and this Agreement will be governed by and interpreted in accordance with the laws of the State of Delaware, disregarding any state's choice-of-law principles requiring the application of a jurisdiction's laws other than the State of Delaware.

\*\*\*\*

THE BE	AUTY	HEALTH	COMP	ANV
--------	------	--------	------	-----

## 2021 INCENTIVE AWARD PLAN

## STOCK OPTION GRANT NOTICE

The Beauty Health Company, a Delaware corporation (the "*Company*") has granted to the participant listed below ("*Participant*") the stock option (the "*Option*") described in this Stock Option Grant Notice (the "*Grant Notice*"), subject to the terms and conditions of The Beauty Health Company 2021 Incentive Award Plan (as amended from time to time, the "*Plan*") and the Stock Option Agreement attached hereto as *Exhibit A* (the "*Agreement*"), both of which are incorporated into this Grant Notice by reference. Capitalized terms not specifically defined in this Grant Notice or the Agreement have the meanings given to them in the Plan.

Participant:	[	]
Grant Date:	[	], 2021
<b>Exercise Price per Share:</b>	[	]
Shares Subject to the Option:		
Final Expiration Date:	[	]
Vesting Commencement Date:	[	1. 2021

**Vesting Schedule:** 

Twenty-five percent (25%) of the Shares subject to the Option will vest on each of the first four anniversaries of the Vesting Commencement Date, subject to Participant's continued status as a Service Provider through the applicable vesting date. Notwithstanding the foregoing, (x) if Participant incurs a Termination of Service due to Participant's death or Disability, subject to and conditioned upon Participant's (or Participant's estate's) timely execution and non-revocation of a release of claims in a form prescribed by the Company (a "Release") that becomes effective and irrevocable no later than sixty (60) days following such Termination of Service (the date such Release becomes effective and irrevocable, the "Release Effective Date"), the Option will vest in full (to the extent then-unvested) upon the Release Effective Date (and shall remain outstanding and eligible to vest through the Release Effective Date and shall automatically be forfeited if the Release does not become effective and irrevocable on or prior to the sixtieth (60th) day following such termination), and (y) if a Change in Control is consummated more than twelve (12) months after [Closing Date], 2021 (the "Closing Date") (and such Change in Control does not occur pursuant to a binding agreement entered into within the twelve (12) months after Closing Date which, for clarity, is addressed solely in the following paragraph) and Participant's status as a Service Provider is terminated by the Company or its Subsidiaries without Cause (as defined in the Agreement) or due to Participant's resignation for Good Reason

(as defined in the Company's Executive Severance Plan), in either case, within twelve (12) months following the consummation of such Change in Control, subject to and conditioned upon Participant's timely execution and non-revocation of a Release that becomes effective and irrevocable no later than sixty (60) days following such Termination of Service, the Option will vest in full (to the extent then-unvested) upon the Release Effective Date (and shall remain outstanding and eligible to vest through the Release Effective Date and shall automatically be forfeited if the Release does not become effective and irrevocable on or prior to the sixtieth (60th) day following such termination).

If (i) a Change in Control occurs within twelve (12) months following Closing Date (or pursuant to a binding agreement entered into within such twelve (12) month period) and (ii) Participant's status as a Service Provider is terminated by the Company or its Subsidiaries without Cause or due to Participant's resignation for Good Reason, in either case, within twelve (12) months after the consummation of such Change in Control, then, subject to and conditioned upon Participant's timely execution and non-revocation of a Release that becomes effective and irrevocable no later than sixty (60) days following such Termination of Service, then a number of Shares subject to the Option equal to (A) the product of (x) 1/48th of the total number of Shares subject to the Option by (y) the number of full and partial months elapsed between the Grant Date and date of such Termination of Service less (B) the total number of Shares subject to the Option that have vested prior to the date of Termination of Service will vest upon the Release Effective Date (and shall remain outstanding and eligible to vest through the Release Effective Date and shall automatically be forfeited if the Release does not become effective and irrevocable on or prior to the sixtieth (60th) day following such termination).

[Incentive Stock Option]/[Non-Qualified Stock Option]

## **Type of Option**

By accepting (whether in writing, electronically or otherwise) the Option, Participant agrees to be bound by the terms of this Grant Notice, the Plan and the Agreement. Participant has reviewed the Plan, this Grant Notice and the Agreement in their entirety, has had an opportunity to obtain the advice of counsel prior to executing this Grant Notice and fully understands all provisions of the Plan, this Grant Notice and the Agreement. Participant hereby agrees to accept as binding, conclusive and final all decisions or interpretations of the Administrator upon any questions arising under the Plan, this Grant Notice or the Agreement.

THE BEAUTY HEALTH COMPANY	PARTICIPANT	
_		
By:		
Name:	[Participant Name]	
Title:	•	

#### STOCK OPTION AGREEMENT

Capitalized terms not specifically defined in this Agreement have the meanings specified in the Grant Notice or, if not defined in the Grant Notice, in the Plan.

## ARTICLE I. GENERAL

- 1.1 <u>Grant of Option</u>. The Company has granted to Participant the Option effective as of the grant date set forth in the Grant Notice (the "*Grant Date*").
- 1.2 <u>Incorporation of Terms of Plan</u>. The Option is subject to the terms and conditions set forth in this Agreement and the Plan, which is incorporated herein by reference. In the event of any inconsistency between the Plan and this Agreement, the terms of the Plan will control, unless it is expressly specified in in this Agreement or the Grant Notice that the specific provision of the Plan will not apply. For clarity, the foregoing sentence shall not limit the applicability of any additive language contained in this Agreement which provides supplemental or additional terms not inconsistent with the Plan.

# ARTICLE II. PERIOD OF EXERCISABILITY

- 2.1 <u>Commencement of Exercisability.</u> The Option will vest and become exercisable according to the vesting schedule in the Grant Notice (the "Vesting Schedule") except that any fraction of a Share as to which the Option would be vested or exercisable will be accumulated and will vest and become exercisable only when a whole vested Share has accumulated; provided, however, that, notwithstanding the foregoing or anything to the contrary in the Grant Notice or this Agreement, in no event may the Option be exercised (in whole or in part) prior to the date on which the Company files a Form S-8 Registration Statement covering the Shares subject to the Option. Except as otherwise set forth in the Grant Notice, the Plan or this Agreement, and unless the Administrator otherwise determines, the Option will immediately expire and be forfeited as to any portion of the Option that is not vested and exercisable as of Participant's Termination of Service for any reason (after taking into consideration any accelerated vesting and exercisability which may occur in connection with such Termination of Service, if any).
- 2.2 <u>Duration of Exercisability.</u> The Vesting Schedule is cumulative. Any portion of the Option which vests and becomes exercisable will remain vested and exercisable until the Option expires. The Option will be forfeited immediately upon its expiration.
- 2.3 <u>Expiration of Option</u>. Except as may be extended in accordance with Section 5.3 of the Plan, the Option may not be exercised to any extent by anyone after, and will expire on, the first of the following to occur:
  - (a) The final expiration date in the Grant Notice;
- (b) Except as the Administrator may otherwise approve, the expiration of three (3) months from the date of Participant's Termination of Service, unless Participant's Termination of Service is for Cause (as defined below), due to Participant's voluntary resignation or by reason of Participant's death or Disability;
- (c) Except as the Administrator may otherwise approve, the expiration of one year from the date of Participant's Termination of Service by reason of Participant's death or Disability;

- (d) Except as the Administrator may otherwise approve, the expiration of one (1) month from the date of Participant's Termination of Service due to Participant's voluntary resignation for any reason; and
  - (e) Except as the Administrator may otherwise approve, Participant's Termination of Service for Cause.
- 2.4 <u>Cause Definition</u>. As used in this Agreement, "*Cause*" means (i) if Participant is a party to a written employment or consulting agreement with the Company or a Subsidiary in which the term "cause" is defined (a "*Relevant Agreement*"), "Cause" as defined in the Relevant Agreement, and (ii) if no Relevant Agreement exists, the occurrence of any one or more of the following events:
- (a) Participant has engaged in an act of dishonesty, theft, embezzlement or fraud, a breach of confidentiality, an unauthorized disclosure or use of inside information, customer lists, trade secrets or other confidential information; has breached a fiduciary duty, law, rule, regulation or policy or procedure of the Company (including but not limited to policies and procedures pertaining to harassment and discrimination); Participant's commission of, or plea of guilty or *nolo contendere* to, any crime or offense (other than minor traffic violations or similar offenses), or any act by Participant constituting a felony;
- (b) Participant's gross or repeated neglect of, or repeated or willful failure to perform, his or her duties to the Company, or Participant's history of substandard performance, if such substandard performance is not cured to the satisfaction of the Board, the Chief Executive Officer of the Company (the "CEO") (unless Participant is the CEO), and/or Participant's manager within ten (10) days following notice to Participant;
- (c) Participant has breached any of the provisions of any employment, confidentiality, restrictive covenant or other agreement between Participant and the Company or an affiliate thereof;
  - (d) actions by Participant involving malfeasance or gross negligence in the performance of, Participant's duties;
- (e) Participant's failure or refusal to perform Participant's duties to the Company on an exclusive and full-time basis if such failure to perform Participant's duties is not cured to the satisfaction of the Board, the CEO (unless Participant is the CEO), and/or Participant's manager (as applicable) within ten (10) days following notice to the Participant;
- (f) Participant's insubordination or failure to follow the Board's (or the CEO's (unless Participant is the CEO) or such Participant's manager's) instructions;
- (g) Participant's violation of any rule, regulation, or policy of the Company or the Board (or Participant's manager) applicable to similarly-situated employees of the Company generally, including, without limitation, rules, regulations, or policies addressing confidentiality, non-solicitation or non-competition, if such violation is not cured to the satisfaction of the Board, the CEO (unless Participant is the CEO), and/or Participant's manager (as applicable) within ten (10) days following notice to the Participant;
- (h) Participant's use of alcohol or illicit drugs in a manner that has or would reasonably be expected to have a detrimental effect on Participant's performance, Participant's duties to Company, or the reputation of the Company or its affiliates; or

(i) Participant's performance of acts which are or could reasonably be expected to become materially detrimental to the image, reputation, finances or business of the Company or any of its affiliates, including but limited to the Participant's commission of unlawful harassment or discrimination.

# ARTICLE III. EXERCISE OF OPTION

- 3.1 <u>Person Eligible to Exercise</u>. During Participant's lifetime, only Participant may exercise the Option. After Participant's death, any exercisable portion of the Option may, prior to the time the Option expires, be exercised by Participant's Designated Beneficiary as provided in the Plan.
- 3.2 <u>Partial Exercise</u>. Any exercisable portion of the Option or the entire Option, if then wholly exercisable, may be exercised, in whole or in part, according to the procedures in the Plan at any time prior to the time the Option or portion thereof expires, except that the Option may only be exercised for whole Shares.

# 3.3 <u>Tax Withholding; Exercise Price</u>.

- (a) Unless the Administrator otherwise determines, the Company shall withhold, or cause to be withheld, Shares otherwise vesting or issuable under this Option in satisfaction of any exercise price and/or applicable withholding tax obligations, in accordance with the Plan. With respect to tax withholding obligations, the number of Shares which may be so withheld or surrendered shall be limited to the number of Shares which have a fair market value on the date of withholding no greater than the aggregate amount of such liabilities based on the maximum individual statutory withholding rates in Participant's applicable jurisdictions for federal, state, local and foreign income tax and payroll tax purposes that are applicable to such taxable income, in accordance with Section 9.5 of the Plan.
- (b) Participant acknowledges that Participant is ultimately liable and responsible for all taxes owed in connection with the Option, regardless of any action the Company or any Subsidiary takes with respect to any tax withholding obligations that arise in connection with the Option. Neither the Company nor any Subsidiary makes any representation or undertaking regarding the treatment of any tax withholding in connection with the awarding, vesting or exercise of the Option or the subsequent sale of Shares. The Company and the Subsidiaries do not commit and are under no obligation to structure the Option to reduce or eliminate Participant's tax liability.
- 3.4 <u>Representation</u>. Participant represents to the Company that Participant has reviewed with Participant's own tax advisors the tax consequences of this Award and the transactions contemplated by the Grant Notice and this Agreement. Participant is relying solely on such advisors and not on any statements or representations of the Company or any of its agents.

# ARTICLE IV. OTHER PROVISIONS

- 4.1 <u>Adjustments</u>. Participant acknowledges that the Option is subject to adjustment, modification and termination in certain events as provided in this Agreement and the Plan.
- 4.2 <u>Clawback</u>. The Option and the Shares issuable hereunder shall be subject to any clawback or recoupment policy in effect on the Grant Date or as may be adopted or maintained by the Company following the Grant Date, including the Dodd-Frank Wall Street Reform and Consumer Protection Act and any rules or regulations promulgated thereunder.

- 4.3 Notices. Any notice to be given under the terms of this Agreement to the Company must be in writing and addressed to the Company in care of the Company's General Counsel at the Company's principal office or the General Counsel's then-current email address or facsimile number. Any notice to be given under the terms of this Agreement to Participant must be in writing and addressed to Participant (or, if Participant is then deceased, to the Designated Beneficiary) at Participant's last known mailing address, email address or facsimile number in the Company's personnel files. By a notice given pursuant to this Section, either party may designate a different address for notices to be given to that party. Any notice will be deemed duly given when actually received, when sent by email, when sent by certified mail (return receipt requested) and deposited with postage prepaid in a post office or branch post office regularly maintained by the United States Postal Service, when delivered by a nationally recognized express shipping company or upon receipt of a facsimile transmission confirmation.
  - 4.4 <u>Titles</u>. Titles are provided herein for convenience only and are not to serve as a basis for interpretation or construction of this Agreement.
- 4.5 <u>Conformity to Securities Laws</u>. Participant acknowledges that the Plan, the Grant Notice and this Agreement are intended to conform to the extent necessary with all Applicable Laws and, to the extent Applicable Laws permit, will be deemed amended as necessary to conform to Applicable Laws.
- 4.6 <u>Successors and Assigns</u>. The Company may assign any of its rights under this Agreement to single or multiple assignees, and this Agreement will inure to the benefit of the successors and assigns of the Company. Subject to the restrictions on transfer set forth in this Agreement or the Plan, this Agreement will be binding upon and inure to the benefit of the heirs, legatees, legal representatives, successors and assigns of the parties hereto.
- 4.7 <u>Limitations Applicable to Section 16 Persons.</u> Notwithstanding any other provision of the Plan or this Agreement, if Participant is subject to Section 16 of the Exchange Act, the Plan, the Grant Notice, this Agreement and the Option will be subject to any additional limitations set forth in any applicable exemptive rule under Section 16 of the Exchange Act (including any amendment to Rule 16b-3) that are requirements for the application of such exemptive rule. To the extent Applicable Laws permit, this Agreement will be deemed amended as necessary to conform to such applicable exemptive rule.
- 4.8 <u>Entire Agreement</u>. The Plan, the Grant Notice and this Agreement (including any exhibit hereto) constitute the entire agreement of the parties and supersede in their entirety all prior undertakings and agreements of the Company and Participant with respect to the subject matter hereof.
- 4.9 <u>Severability</u>. If any portion of the Grant Notice or this Agreement or any action taken under the Grant Notice or this Agreement, in any case is held illegal or invalid for any reason, the illegality or invalidity will not affect the remaining parts of the Grant Notice and/or this Agreement (as applicable), and the Grant Notice and/or this Agreement (as applicable) will be construed and enforced as if the illegal or invalid provisions had been excluded, and the illegal or invalid action will be null and void.
- 4.10 <u>Limitation on Participant's Rights</u>. Participation in the Plan confers no rights or interests other than as herein provided. This Agreement creates only a contractual obligation on the part of the Company as to amounts payable and may not be construed as creating a trust. Neither the Plan nor any underlying program, in and of itself, has any assets. Participant will have only the rights of a general unsecured creditor of the Company with respect to amounts credited and benefits payable, if any, with respect to the Option, and rights no greater than the right to receive the Shares as a general unsecured creditor with respect to the Option, as and when exercised pursuant to the terms hereof

- 4.11 <u>Not a Contract of Employment or Service</u>. Nothing in the Plan, the Grant Notice or this Agreement confers upon Participant any right to continue in the employ or service of the Company or any Subsidiary or interferes with or restricts in any way the rights of the Company and its Subsidiaries, which rights are hereby expressly reserved, to discharge or terminate the services of Participant at any time for any reason whatsoever, with or without Cause, except to the extent expressly provided otherwise in a written agreement between the Company or a Subsidiary and Participant.
- 4.12 <u>Counterparts</u>. The Grant Notice may be executed in one or more counterparts, including by way of any electronic signature, subject to Applicable Law, each of which will be deemed an original and all of which together will constitute one instrument.
  - 4.13 <u>Incentive Stock Options</u>. If the Option is designated as an Incentive Stock Option:
- (a) Participant acknowledges that to the extent the aggregate fair market value of shares (determined as of the time the option with respect to the shares is granted) with respect to which stock options intended to qualify as "incentive stock options" under Section 422 of the Code, including the Option, are exercisable for the first time by Participant during any calendar year exceeds \$100,000 or if for any other reason such stock options do not qualify or cease to qualify for treatment as "incentive stock options" under Section 422 of the Code, such stock options (including the Option) will be treated as non-qualified stock options. Participant further acknowledges that the rule set forth in the preceding sentence will be applied by taking the Option and other stock options into account in the order in which they were granted, as determined under Section 422(d) of the Code. Participant also acknowledges that if the Option is exercised more than three months after Participant's Termination of Service, other than by reason of death or disability, the Option will be taxed as a Non-Qualified Stock Option.
- (b) Participant will give prompt written notice to the Company of any disposition or other transfer of any Shares acquired under this Agreement if such disposition or other transfer is made (i) within two years from the Grant Date or (ii) within one year after the transfer of such Shares to Participant. Such notice will specify the date of such disposition or other transfer and the amount realized, in cash, other property, assumption of indebtedness or other consideration, by Participant in such disposition or other transfer.
- 4.14 <u>Governing Law</u>. The Grant Notice and this Agreement will be governed by and interpreted in accordance with the laws of the State of Delaware, disregarding any state's choice-of-law principles requiring the application of a jurisdiction's laws other than the State of Delaware.

\*\*\*\*

# The BeautyHealth Company

# 2021 INCENTIVE AWARD PLAN

#### PERFORMANCE-BASED RESTRICTED STOCK UNIT GRANT NOTICE

The BeautyHealth Company, a Delaware corporation (the "Company"), has granted to the participant listed below ("Participant") the Restricted Stock Units (the "RSUs") described in this Performance-Based Restricted Stock Unit Grant Notice (this "Grant Notice"), subject to the terms and conditions of [ ● ] 2021 Incentive Award Plan (as amended from time to time, the "Plan") and the Restricted Stock Unit Agreement attached hereto as Exhibit A (the "Agreement"), both of which are incorporated into this Grant Notice by reference. Capitalized terms not specifically defined in this Grant Notice or the Agreement have the meanings given to them in the Plan.

Participant:	[	]
Grant Date:	[	], 2021
Number of RSUs at Maximum:	[	]1
Vesting Commencement Date:	[	], 2021
**		

**Vesting Schedule:** 

(a) <u>General</u>. Subject to clause (b) below, and further subject to and conditioned upon Participant's continued service as a Service Provider through the last day of the Performance Period, a number of RSUs shall vest on the last day of the Performance Period equal to (i) the total number of RSUs granted hereby multiplied by (ii) the applicable vesting percentage ("*Vesting Percentage*") set forth below, which shall be determined based on greater of (x) the Company's Average Stock Price during the Year 3 Measurement Period and (y) the Company's Average Stock Price during the Year 4 Measurement Period (each forgoing capitalized term as defined below), as follows:

	Average Stock	
	Price During the	
	Applicable	Vesting
	Measurement	Percentage (% of
	Period:	Maximum):
Below Threshold	Less than \$25.00	0%
Threshold/Target	\$25.00	66.67%
Stretch	\$30.00	80%
Maximum	\$37.50 or greater	100%

In the event that the Company's Average Stock Price falls between the Threshold and Target values or Target and Maximum values specified in the table above, the Vesting Percentage shall be interpolated on a linear basis (for clarity, if Average Stock Price falls below the Threshold value, the Vesting Percentage shall equal 0%).

Notwithstanding the foregoing, in the event that a Change in Control is consummated during the Performance Period and Participant remains in continued service as a Service Provider until at least immediately prior to such Change in Control:

[Signature Page to Restricted Stock Unit Grant Notice]

Note to Draft: Insert number of RSUs at maximum.

- (i) In the event that (A) the Shares do not continue to be publicly traded following the consummation of such Change in Control and (B) no Assumption of the RSUs (as defined in Section 8.3 of the Plan) occurs in connection with such Change in Control, then, immediately prior to the Change in Control, a number of RSUs will vest based solely on the per-Share consideration paid or payable (as applicable) in connection with such Change in Control (as determined by the Administrator) or, if the Change in Control is consummated after the third anniversary of the Vesting Commencement Date, the Company's Average Stock Price during the Year 3 Measurement Period (if greater); and
- (ii) In the event that (A) the Shares do not continue to be publicly traded following the consummation of such Change in Control and (B) an Assumption of the RSUs (as defined in Section 8.3 of the Plan) occurs in connection with such Change in Control, then, effective immediately prior to the closing of the Change in Control, the RSUs will be deemed to convert into a number of unvested RSUs determined based solely on the per-Share consideration paid or payable (as applicable) in connection with such Change in Control (as determined by the Administrator) or, if the Change in Control is consummated after the third anniversary of the Vesting Commencement Date, the Company's Average Stock Price during the Year 3 Measurement Period (if greater). Such unvested RSUs (as so assumed and adjusted in connection with the Change in Control) will be eligible to vest in full on the last day of the Performance Period in accordance with this clause (a) (based solely on the Participant's continued status as a Service Provider through such date) or upon Participant's Termination of Service as provided in clause (b) below.

# (b) <u>Termination of Service; Change in Control</u>. Notwithstanding clause (a) above:

(i) If Participant incurs a Termination of Service prior to the last day of the Performance Period, then the RSUs shall vest under clause (a) above or be forfeited (as applicable) in accordance with the following table. Any vesting of the RSUs pursuant to the following table shall (A) be subject to Participant (or Participant's estate, as applicable) timely executing and not revoking a release of claims in a form prescribed by the Company (a "*Release*") that becomes effective and irrevocable no later than sixty (60) days following such Termination of Service (the date such Release becomes effective and irrevocable, the "*Release Effective Date*"), and (B) be effective as of the Release Effective Date:

If the Termination of Service Occurs On or After

Reason for Termination of Service	If the Termination of Service Occurs Before the Third Anniversary of the Vesting Commencement Date, then:	the Third Anniversary of the Vesting Commencement Date but Before the Fourth Anniversary of the Vesting Commencement Date, then:
Death or Disability (as defined below)	A number of RSUs will vest based on the Company's Average Stock Price over the Termination Measurement Period (as defined below).	A number of RSUs will vest based on the greater of (i) the Company's Average Stock Price over the Termination Measurement Period and (ii) the Company's Average Stock Price over the Year 3 Measurement Period.
Without Cause [or for Good Reason (as defined below)] Prior to the Consummation of a Change in Control	All RSUs will be forfeited upon such Termination of Service without payment.	A number of RSUs will vest based on the Company's Average Stock Price over the Year 3 Measurement Period.
Without Cause [or for Good Reason] Within 24 Months After the Consummation of a Change in Control	A number of RSUs will vest based on the Company's Average Stock Price over the Termination Measurement Period.	A number of RSUs will vest based on the greater of (i) the Company's Average Stock Price over the Termination Measurement Period and (ii) the Company's Average Stock Price over the Year 3 Measurement Period.
Any Other Reason (Including for Cause or without Good Reason)	All RSUs will be forfeited upon such Termination of Service without payment.	All RSUs will be forfeited upon such Termination of Service without payment.

- (ii) With respect to sub-clause (i) above, (A) the RSUs shall remain outstanding and eligible to vest following Participant's Termination of Service through the Release Effective Date and shall automatically be forfeited on the sixtieth (60th) day following such termination if the Release does not become effective and irrevocable on or prior to such date, and (B) any RSUs that do not become vested on the Release Effective Date pursuant to the applicable sub-clause shall be immediately forfeited on such date.
- (c) <u>Termination; Forfeiture</u>. Unless earlier terminated as set forth in this Grant Notice or the Agreement, any RSUs that have not become vested on or prior to the last day of the Performance Period will thereupon be automatically forfeited by Participant without payment of any consideration therefor. Except as set forth in clause (b) above, if Participant experiences a Termination of Service for any reason prior to the last day of the Performance Period, all then-unvested RSUs will thereupon be automatically forfeited by Participant without payment of any consideration therefor.
  - (d) <u>Definitions</u>. For purposes hereof, the following terms shall have the respective meanings set forth below:
  - (i) "Average Stock Price" shall mean, with respect to any Measurement Period, the average Fair Market Value of a Share over such Measurement Period.
  - (ii) "Cause" shall have the meaning set forth in [the employment agreement between Participant and [the Company], dated [ ]] / [the Company's Executive Severance Plan].<sup>2</sup>
  - (iii) "*Good Reason*" shall have the meaning set forth in [the employment agreement between Participant and [the Company], dated [ ]] / [the Company's Executive Severance Plan].<sup>3</sup>
  - (iv) "Year 3 Measurement Period" means the ninety (90)-day period ending on the third (3<sup>rd</sup>) anniversary of the Vesting Commencement Date.

Note to Draft: To be updated based on whether Participant is party to an employment agreement or a participant in the Executive Severance Plan (if neither, the "Cause" definition from the Executive Severance Plan will be added here).

Note to Draft: To be updated based on whether Participant is party to an employment agreement or a participant in the Executive Severance Plan (if neither, "Good Reason" will be removed).

- (v) "Year 4 Measurement Period" means the ninety (90)-day period ending on the fourth (4th) anniversary of the Vesting Commencement Date.
- (vi) "*Measurement Period*" means each of the Termination Measurement Period, the Year 3 Measurement Period and the Year 4 Measurement Period.
- (vii) "*Performance Period*" means the period commencing on the Vesting Commencement Date and ending on the fourth (4th) anniversary of the Vesting Commencement Date.
- (viii) "Termination Measurement Period" means the ninety (90)-day period ending on (and including) the date of Participant's Termination of Service.

By accepting (whether in writing, electronically or otherwise) the RSUs, Participant agrees to be bound by the terms of this Grant Notice, the Plan and the Agreement. Participant has reviewed the Plan, this Grant Notice and the Agreement in their entirety, has had an opportunity to obtain the advice of counsel prior to executing this Grant Notice and fully understands all provisions of the Plan, this Grant Notice and the Agreement. Participant hereby agrees to accept as binding, conclusive and final all decisions or interpretations of the Administrator upon any questions arising under the Plan, this Grant Notice or the Agreement.

[•]	PARTICIPANT
By:	
Name:	[Participant Name]
Title:	<del>-</del>

#### RESTRICTED STOCK UNIT AGREEMENT

Capitalized terms not specifically defined in this Agreement have the meanings specified in the Grant Notice or, if not defined in the Grant Notice, in the Plan.

# ARTICLE I. GENERAL

- 1.1 <u>Award of RSUs</u>. The Company has granted the RSUs to Participant effective as of the Grant Date set forth in the Grant Notice (the "*Grant Date*"). Each RSU represents the right to receive one Share as set forth in this Agreement. Participant will have no right to the distribution of any Shares until the time (if ever) the RSUs have vested.
- 1.2 <u>Incorporation of Terms of Plan</u>. The RSUs are subject to the terms and conditions set forth in this Agreement and the Plan, which is incorporated herein by reference. In the event of any inconsistency between the Plan and this Agreement, the terms of the Plan will control, unless it is expressly specified in in this Agreement or the Grant Notice that the specific provision of the Plan will not apply. For clarity, the foregoing sentence shall not limit the applicability of any additive language contained in this Agreement which provides supplemental or additional terms not inconsistent with the Plan.
- 1.3 <u>Unsecured Promise</u>. The RSUs will at all times prior to settlement represent an unsecured Company obligation payable only from the Company's general assets.

# ARTICLE II. VESTING; FORFEITURE AND SETTLEMENT

2.1 <u>Vesting; Forfeiture</u>. The RSUs will vest according to the vesting schedule in the Grant Notice except that any fraction of an RSU that would otherwise be vested will be accumulated and will vest only when a whole RSU has accumulated. Except as otherwise set forth in the Grant Notice, the Plan or this Agreement, and unless the Administrator otherwise determines, in the event of Participant's Termination of Service for any reason, all unvested RSUs will immediately and automatically be cancelled and forfeited (after taking into consideration any accelerated vesting which may occur in connection with such Termination of Service, if any).

### 2.2 Settlement.

- (a) RSUs that vest will be paid in Shares as soon as administratively practicable after the vesting of the applicable RSU, but in no event later than sixty (60) days following the date on which the applicable RSU vests (or, in the case of any accelerated vesting that occurs on the Release Effective Date pursuant to the Grant Notice, no later than sixty (60) days following the date on which the applicable Termination of Service occurs).
- (b) Notwithstanding the foregoing, the Company may delay any payment under this Agreement that the Company reasonably determines would violate Applicable Law or an applicable provision of the Plan until the earliest date the Company reasonably determines the making of the payment will not cause such a violation (in accordance with Treasury Regulation Section 1.409A-2(b)(7)(ii)); *provided* the Company reasonably believes the delay will not result in the imposition of excise taxes under Section 409A.

# ARTICLE III. TAXATION AND TAX WITHHOLDING

3.1 <u>Representation</u>. Participant represents to the Company that Participant has reviewed with Participant's own tax advisors the tax consequences of this Award and the transactions contemplated by the Grant Notice and this Agreement. Participant is relying solely on such advisors and not on any statements or representations of the Company or any of its agents.

#### 3.2 <u>Tax Withholding</u>.

- (a) Unless the Administrator otherwise determines, the Company shall withhold, or cause to be withheld, Shares otherwise vesting or issuable under this Award (including the RSUs) in satisfaction of any applicable withholding tax obligations, in accordance with the Plan. The number of Shares which may be so withheld or surrendered shall be limited to the number of Shares which have a Fair Market Value on the date of withholding no greater than the aggregate amount of such liabilities based on the maximum individual statutory withholding rates in Participant's applicable jurisdictions for federal, state, local and foreign income tax and payroll tax purposes that are applicable to such taxable income, in accordance with Section 9.5 of the Plan.
- (b) Participant acknowledges that Participant is ultimately liable and responsible for all taxes owed in connection with the RSUs, regardless of any action the Company or any Subsidiary takes with respect to any tax withholding obligations that arise in connection with the RSUs. Neither the Company nor any Subsidiary makes any representation or undertaking regarding the treatment of any tax withholding in connection with the awarding, vesting or payment of the RSUs or the subsequent sale of Shares. The Company and its Subsidiaries do not commit and are under no obligation to structure the RSUs to reduce or eliminate Participant's tax liability.

# ARTICLE IV. OTHER PROVISIONS

- 4.1 <u>Adjustments</u>. Participant acknowledges that the RSUs and the Shares subject to the RSUs are subject to adjustment, modification and termination in certain events as provided in this Agreement and the Plan.
- 4.2 <u>Clawback</u>. The RSUs and the Shares issuable hereunder shall be subject to clawback or recoupment in accordance with this Section 4.2. In the event that the Administrator, in its good faith discretion, determines that Participant has committed an act that constitutes Cause and such act has resulted in or would reasonably be expected to result in material harm to the Company and/or its affiliates, the Board may seek <u>recoupment of up to the full amount of the RSUs and Shares issued upon settlement thereof and/or any proceeds received upon the sale of any such Shares. Nothing in this Section 4.2 shall limit the application of any clawback or recoupment policy in effect on the Grant Date or as may be adopted or maintained by the Company following the Grant Date, including the Dodd-Frank Wall Street Reform and Consumer Protection Act and any rules or regulations promulgated thereunder.</u>
- 4.3 <u>Notices</u>. Any notice to be given under the terms of this Agreement to the Company must be in writing and addressed to the Company in care of the Company's General Counsel at the Company's principal office or the General Counsel's then-current email address or facsimile number. Any notice to be given under the terms of this Agreement to Participant must be in writing and addressed to Participant (or, if Participant is then deceased, to the Designated Beneficiary) at Participant's last known mailing address, email address or facsimile number in the Company's personnel files. By a notice given pursuant to this Section, either party may designate a different address for notices to be given to that party. Any notice will

be deemed duly given when actually received, when sent by email, when sent by certified mail (return receipt requested) and deposited with postage prepaid in a post office or branch post office regularly maintained by the United States Postal Service, when delivered by a nationally recognized express shipping company or upon receipt of a facsimile transmission confirmation.

- 4.4 <u>Titles</u>. Titles are provided herein for convenience only and are not to serve as a basis for interpretation or construction of this Agreement.
- 4.5 <u>Conformity to Securities Laws</u>. Participant acknowledges that the Plan, the Grant Notice and this Agreement are intended to conform to the extent necessary with all Applicable Laws and, to the extent Applicable Laws permit, will be deemed amended as necessary to conform to Applicable Laws.
- 4.6 <u>Successors and Assigns</u>. The Company may assign any of its rights under this Agreement to single or multiple assignees, and this Agreement will inure to the benefit of the successors and assigns of the Company. Subject to the restrictions on transfer set forth in this Agreement or the Plan, this Agreement will be binding upon and inure to the benefit of the heirs, legalees, legal representatives, successors and assigns of the parties hereto.
- 4.7 <u>Limitations Applicable to Section 16 Persons</u>. Notwithstanding any other provision of the Plan or this Agreement, if Participant is subject to Section 16 of the Exchange Act, the Plan, the Grant Notice, this Agreement and the RSUs will be subject to any additional limitations set forth in any applicable exemptive rule under Section 16 of the Exchange Act (including any amendment to Rule 16b-3) that are requirements for the application of such exemptive rule. To the extent Applicable Laws permit, this Agreement will be deemed amended as necessary to conform to such applicable exemptive rule.
- 4.8 <u>Entire Agreement</u>. The Plan, the Grant Notice and this Agreement (including any exhibit hereto) constitute the entire agreement of the parties and supersede in their entirety all prior undertakings and agreements of the Company and Participant with respect to the subject matter hereof.
- 4.9 <u>Severability</u>. If any portion of the Grant Notice or this Agreement or any action taken under the Grant Notice or this Agreement, in any case is held illegal or invalid for any reason, the illegality or invalidity will not affect the remaining parts of the Grant Notice and/or this Agreement (as applicable), and the Grant Notice and/or this Agreement (as applicable) will be construed and enforced as if the illegal or invalid provisions had been excluded, and the illegal or invalid action will be null and void.
- 4.10 <u>Limitation on Participant's Rights</u>. Participation in the Plan confers no rights or interests other than as herein provided. This Agreement creates only a contractual obligation on the part of the Company as to amounts payable and may not be construed as creating a trust. Neither the Plan nor any underlying program, in and of itself, has any assets. Participant will have only the rights of a general unsecured creditor of the Company with respect to amounts credited and benefits payable, if any, with respect to the RSUs, and rights no greater than the right to receive cash or the Shares as a general unsecured creditor with respect to the RSUs, as and when settled pursuant to the terms of this Agreement.
- 4.11 <u>Not a Contract of Employment or Service</u>. Nothing in the Plan, the Grant Notice or this Agreement confers upon Participant any right to continue in the employ or service of the Company or any Subsidiary or interferes with or restricts in any way the rights of the Company and its Subsidiaries, which rights are hereby expressly reserved, to discharge or terminate the services of Participant at any time for any reason whatsoever, with or without cause, except to the extent expressly provided otherwise in a written agreement between the Company or a Subsidiary and Participant.

- 4.12 <u>Counterparts</u>. The Grant Notice may be executed in one or more counterparts, including by way of any electronic signature, subject to Applicable Law, each of which will be deemed an original and all of which together will constitute one instrument.
- 4.13 <u>Governing Law</u>. The Grant Notice and this Agreement will be governed by and interpreted in accordance with the laws of the State of Delaware, disregarding any state's choice-of-law principles requiring the application of a jurisdiction's laws other than the State of Delaware.

\* \* \* \* \*

# THE BEAUTY HEALTH COMPANY EXECUTIVE SEVERANCE PLAN

The Beauty Health Company (the "<u>Company</u>"), has adopted this The Beauty Health Company Executive Severance Plan, including the attached Exhibits (the "<u>Plan</u>"), for the benefit of Participants (as defined below) on the terms and conditions hereinafter stated. The Plan, as set forth herein, is intended to provide severance protections to a select group of management or highly compensated employees (within the meaning of ERISA (as defined below)) in connection with qualifying terminations of employment.

- 1. **Defined Terms**. Capitalized terms used but not otherwise defined herein shall have their respective meanings set forth below:
- 1.1 "<u>Base Salary</u>" means, with respect to any Participant, the Participant's annual base salary rate in effect immediately prior to a Qualifying Termination.
  - 1.2 "Board" means the Board of Directors of the Company.
  - 1.3 "Cause" means, with respect to any Participant, the occurrence of any one or more of the following events:
- (a) the Participant has engaged in an act of dishonesty, theft, embezzlement or fraud, a breach of confidentiality, an unauthorized disclosure or use of inside information, customer lists, trade secrets or other confidential information, or has breached a fiduciary duty, law, rule, regulation or policy or procedure of the Company (including but not limited to policies and procedures pertaining to harassment and discrimination);
- (b) the Participant's commission of, or plea of guilty or *nolo contendere* to, any crime or offense (other than minor traffic violations or similar offenses), or any act by the Participant constituting a felony;
- (c) the Participant's gross or repeated neglect of, or repeated or willful failure to perform, his or her duties to the Company, or the Participant's history of substandard performance, if such substandard performance is not cured to the satisfaction of the Board, the Chief Executive Officer of the Company (the "CEO"), and/or the Participant's manager within ten (10) days following notice to the Participant;
- (d) the Participant has breached any of the provisions of any employment, confidentiality, restrictive covenant or other agreement between the Participant and the Company or an affiliate thereof;
  - (e) actions by the Participant involving malfeasance or gross negligence in the performance of, the Participant's duties;
- (f) the Participant's failure or refusal to perform the Participant's duties to the Company on an exclusive and full-time basis if such failure to perform the Participant's duties is not cured to the satisfaction of the Board, the CEO, and/or the Participant's manager (as applicable) within ten (10) days following notice to the Participant;
- (g) the Participant's willful violation of any material rule, regulation, or policy of the Company or the Board (or such Participant's manager) applicable to similarly-situated employees of the

Company generally, including, without limitation, rules, regulations, or policies addressing confidentiality, non-solicitation or non-competition, if such violation is not cured to the satisfaction of the Board, the CEO, and/or the Participant's manager (as applicable) within ten (10) days following notice to the Participant;

- (h) the Participant's use of alcohol or illicit drugs in a manner that has or would reasonably be expected to have a detrimental effect on the Participant's performance, the Participant's duties to Company, or the reputation of the Company or its affiliates; or
- (i) the Participant's performance of acts which are or could reasonably be expected to become materially detrimental to the image, reputation, finances or business of the Company or any of its affiliates, including but limited to the Participant's commission of unlawful harassment or discrimination.

For purposes of the foregoing definition of Cause, no act or failure to act by a Participant shall be deemed willful or intentional if performed in good faith and with the reasonable belief that the action or inaction was in the best interests of the Company and its affiliates.

- 1.4 "Change in Control" shall have the meaning set forth in the Company's 2021 Incentive Award Plan, as may be amended from time to time, or any successor equity incentive plan established by the Company.
- 1.5 "<u>CIC Protection Period</u>" means the period beginning on (and including) the date on which a Change in Control is consummated and ending on (and including) the nine (9)-month anniversary thereof.
  - 1.6 "CIC Termination" means a Qualifying Termination which occurs during the CIC Protection Period.
  - 1.7 "COBRA" means the Consolidated Omnibus Budget Reconciliation Act of 1985.
  - 1.8 "Code" means the Internal Revenue Code of 1986, as amended from time to time, or any successor thereto.
- 1.9 "<u>Committee</u>" means the Compensation Committee of the Board, or such other committee as may be appointed by the Board to administer the Plan.
  - 1.10 "Date of Termination" means the effective date of the termination of the Participant's employment.
- 1.11 "Employee" means an individual who is an employee (within the meaning of Code Section 3401(c)) of the Company or any of its subsidiaries.
  - 1.12 "ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and the regulations promulgated thereunder.
- 1.13 "Good Reason" means, with respect to any Participant and without the Participant's written consent, the occurrence of any one or more of the following events:
- (a) a material reduction in the Participant's Base Salary or Target Bonus except in connection with across-the-board salary reductions (and corresponding target bonus reductions) imposed on substantially all of the Company's senior executives not to exceed, in the aggregate, ten percent (10%) during any twelve (12)-month period;

- (b) the relocation of the Participant's principal place of employment to a location that is greater than fifty (50) miles from the Participant's principal place of employment as of the date on which he or she becomes a Participant in the Plan if that relocation increases the Participant's commute by fifty (50) miles or more; or
- (c) a material reduction in the Participant's title, duties or responsibilities to the Company (other than during a period of physical or mental incapacity, and specifically <u>not to</u> include any change in Participant's title, duties or responsibilities as a result of any Change in Control or any other transaction that does not qualify as a Change in Control).

Notwithstanding the foregoing, the Participant will not be deemed to have resigned for Good Reason unless (1) the Participant provides the Company with written notice setting forth in reasonable detail the facts and circumstances claimed by the Participant to constitute Good Reason within thirty (30) days after the date of the occurrence of any event that the Participant knows or reasonably should have known to constitute Good Reason, (2) the Company fails to cure such acts or omissions within thirty (30) days following its receipt of such notice, and (3) the effective date of the Participant's termination for Good Reason occurs no later than sixty (60) days after the expiration of the Company's cure period.

- 1.14 "Participant" means each Employee with a title of Vice President or higher who is selected by the Administrator (or designee thereof in accordance with Section 3 hereof) to participate in the Plan and is provided with (and, if applicable, countersigns) a Participation Notice in accordance with Section 13.2 hereof, other than any Employee who, at the time of his or her termination of employment, is covered by an employment or other individual agreement with the Company or a subsidiary thereof that provides for cash severance or termination benefits. For the avoidance of doubt, retention bonus payments, change in control bonus payments and other similar payments shall not constitute "cash severance" for purposes of this definition.
- 1.15 "Prior-Year Bonus" means, with respect to any Participant, the Participant's earned but unpaid annual cash performance bonus, if any, for the Company's fiscal year ending immediately prior to the fiscal year in which the Date of Termination occurs. For clarity, if the Date of Termination occurs on or after the date on which the Company pays annual cash performance bonuses for the Company's fiscal year ending immediately prior to the fiscal year in which the Date of Termination occurs, then there shall be no Prior-Year Bonus.
- 1.16 "Pro-Rata Target Bonus" means, with respect to any Participant, a pro-rated portion of the Participant's Target Bonus amount for the year in which the Date of Termination occurs, determined by multiplying the Participant's Target Bonus for the year in which the Date of Termination occurs by a fraction, the numerator of which equals the number of days during which the Participant was employed by the Company or its subsidiaries during the calendar year in which the Date of Termination occurs and the denominator of which equals 365 or 366 (as applicable).
- 1.17 "Qualifying Termination" means a termination of the Participant's employment with the Company or a subsidiary thereof, as applicable, (a) by the Company or such subsidiary, as applicable, without Cause or (b) by the Participant for Good Reason. Notwithstanding anything contained herein, in no event shall a Participant be deemed to have experienced a Qualifying Termination if, (a) if such Participant is offered and/or accepts a comparable employment position with the Company or any subsidiary, or (b) if in connection with a Change in Control or any other corporate transaction or sale of assets involving the Company or any subsidiary, such Participant is offered and accepts a comparable employment position with the successor or purchaser entity (or an affiliate thereof), as applicable. A Qualifying Termination shall not include a termination due to the Participant's death or disability.

- 1.18 "Severance Benefits" means, collectively, the Prior-Year Bonus, the Cash Salary Severance, the Pro-Rata Target Bonus, and the COBRA Benefits to which a Participant may become entitled pursuant to the Plan.
- 1.19 "Severance Classification" means, with respect to any Participant, his or her designation as a "Tier 1," "Tier 2" or "Tier 3" participant in the Plan.
- 1.20 "Severance Period" means, with respect to any Participant, the number of months following the Participant's Date of Termination during which the Participant is entitled to the Cash Salary Severance and COBRA Benefits, as determined in accordance with Exhibit A or Exhibit B, as applicable, attached hereto (based on the Participant's Severance Classification).
- 1.21 "<u>Target Bonus</u>" means, with respect to any Participant, the Participant's target annual cash performance bonus, if any, for the year in which the Date of Termination occurs. For the avoidance of doubt, with respect to any Participant who is not eligible to receive an annual cash performance bonus as part of his or her annual compensation, such Participant's Target Bonus shall be equal to zero.
- 2. **Effectiveness of the Plan; Notification**. The Plan shall become effective on the date on which it is adopted by the Board. The Administrator shall, pursuant to a Participation Notice, notify each Participant that such Participant has been selected to participate in the Plan and of such Participant's Severance Classification.
- 3. **Administration**. Subject to Section 13.4 hereof, the Plan shall be interpreted, administered and operated by the Committee (the "<u>Administrator</u>"), which shall have complete authority, subject to the express provisions of the Plan, to interpret the Plan, to prescribe, amend and rescind rules and regulations relating to the Plan, and to make all other determinations necessary or advisable for the administration of the Plan. The Administrator may delegate any of its duties hereunder to a subcommittee, or to such person or persons from time to time as it may designate (other than to any Participant in the Plan). All decisions, interpretations and other actions of the Administrator (including with respect to whether a Qualifying Termination has occurred) shall be final, conclusive and binding on all parties who have an interest in the Plan.

#### 4. Severance Benefits.

- 4.1 <u>Eligibility.</u> Each Employee who qualifies as a Participant and who experiences a Qualifying Termination is eligible to receive Severance Benefits under the Plan.
- 4.2 <u>Qualifying Termination Payment</u>. In the event that a Participant experiences a Qualifying Termination (other than a CIC Termination), then, subject to Sections 4.5 and 4.6 hereof and further subject to the Participant's execution of a Release that becomes effective and irrevocable in accordance with Section 4.4 hereof, and subject to any additional requirements specified in the Plan, the Company shall pay or provide to the Participant the following Severance Benefits:
  - (a) <u>Prior-Year Bonus</u>. The Company shall pay to the Participant an amount equal to the Participant's Prior-Year Bonus, payable within seventy (70) days following the Date of Termination; *provided*, that if the aggregate period during which the Participant is entitled to consider and/or revoke the Release spans two calendar years, the Prior-Year Bonus shall be paid in the second (2<sup>nd</sup>) such calendar year;

- (b) <u>Cash Salary Severance</u>. The Company shall pay to the Participant an amount equal to the amount determined in accordance with <u>Exhibit A</u> attached hereto (based on the Participant's Severance Classification) (the "<u>Cash Salary Severance</u>"). Subject to Section 6.2 hereof, the Cash Salary Severance (as set forth on <u>Exhibit A</u>) shall be paid to the Participant in accordance with the Company's usual payroll periods during the Severance Period (as set forth on <u>Exhibit A</u> and based on the Participant's Severance Classification) commencing on the Date of Termination; *provided*, that no such payments shall be made prior to the date on which the Release becomes effective and irrevocable and, if the aggregate period during which the Participant is entitled to consider and/or revoke the Release spans two calendar years, no payments under this Section 4.2(b) shall be made prior to the beginning of the second (2nd) such calendar year (and any such payments otherwise payable prior thereto (if any) shall instead be paid on the first (1st) regularly scheduled Company payroll date in the second (2nd) such calendar year).
- (c) <u>Pro-Rata Target Bonus</u>. The Company shall pay to the Participant an amount equal to his or her Pro-Rated Target Bonus, payable within seventy (70) days following the Date of Termination; *provided*, that if the aggregate period during which the Participant is entitled to consider and/or revoke the Release spans two calendar years, the Pro-Rata Target Bonus will be paid in the second (2<sup>nd</sup>) such calendar year).
- (d) <u>COBRA</u>. Subject to the Participant's valid election to continue health care coverage under Section 4980B of the Code, to the extent that the Participant is eligible to do so, then the Company shall reimburse the Participant for the Participant and the Participant's eligible dependents with coverage under its group health plans at the same levels and at the same cost to Participant as would have applied if the Participant's employment had not been terminated based on Participant's elections in effect on the Date of Termination until the earlier of the end of the month during which the Participant's Severance Period (as determined in accordance with <u>Exhibit A</u> attached hereto and based on the Participant's Severance Classification)) ends or the date the Participant becomes covered by a group health insurance program provided by a subsequent employer (the "<u>COBRA Benefits</u>"). Notwithstanding the foregoing, (i) if any plan pursuant to which the COBRA Benefits are provided is not, or ceases prior to the expiration of the period of continuation coverage to be, exempt from the application of Code Section 409A under Treasury Regulation Section 1.409A-1(a)(5), or (ii) the Company is otherwise unable to continue to cover the Participant under its group health plans without incurring penalties (including without limitation, pursuant to Section 2716 of the Public Health Service Act or the Patient Protection and Affordable Care Act), then, in either case, an amount equal to each remaining Company reimbursement shall thereafter be paid to the Participant in substantially equal monthly installments over the Severance Period (or the remaining portion thereof).
- 4.3 <u>CIC Termination Payment</u>. In the event that a Participant experiences a CIC Termination, then, subject to Sections 4.5 and 4.6 hereof and further subject to the Participant's execution of a Release that becomes effective and irrevocable in accordance with Section 4.4 hereof, and subject to any additional requirements specified in the Plan, then the Company shall pay or provide to the Participant, as applicable, the Severance Benefits set forth in Sections 4.2(a)-(d) hereof; *provided*, *however*, that the amount of the Cash Salary Severance and the Severance Period shall be determined in accordance with <u>Exhibit B</u> attached hereto (instead of in accordance with <u>Exhibit A</u>) based on the Participant's Severance Classification.

- 4.4 <u>Release</u>. Notwithstanding anything herein to the contrary, no Participant shall be eligible or entitled to receive or retain any Severance Benefits under the Plan unless he or she executes a general release of claims in a form prescribed by the Company (the "<u>Release</u>") that becomes effective and irrevocable no more than sixty (60) days after the Date of Termination.
- 4.5 <u>Impact of Subsequent Employment</u>. Notwithstanding anything herein to the contrary, unless otherwise determined by the Administrator, in the event that, after his or her Date of Termination and prior to the end of his or her Severance Period, a Participant commences paid employment with any other entity, the payments and benefits described in Section 4.2 or 4.3 above, as applicable, shall cease immediately upon the Participant's commencement of such employment. Each Participant hereby agrees to give the Company prompt written notice of such subsequent paid employment no later than the date on which such employment commences.
- 4.6 <u>Non-U.S. Employees</u>. Notwithstanding anything in the Plan to the contrary, any Participant that resides outside of the United States (each, a "<u>Non-U.S. Participant</u>") and is entitled to receive severance, notice or similar termination payments and/or benefits under the laws of his or her country of residence upon his or her termination of employment with the Company and its subsidiaries (collectively, "<u>Statutory Severance</u>") and that becomes eligible to receive Severance Benefits under the Plan shall be entitled to receive either (i) the payments and benefits described in Section 4.2 or 4.3 above, as applicable, or (ii) such Non-US Participant's Statutory Severance, whichever is greater.
- 5. **Limitations**. Notwithstanding any provision of the Plan to the contrary, if a Participant's status as an Employee is terminated for any reason other than due to a Qualifying Termination, the Participant shall not be entitled to receive any Severance Benefits under the Plan, and the Company shall not have any obligation to such Participant under the Plan.

#### 6. Section 409A.

- 6.1 <u>General</u>. To the extent applicable, the Plan shall be interpreted and applied consistent and in accordance with Code Section 409A and Department of Treasury regulations and other interpretive guidance issued thereunder. Notwithstanding any provision of the Plan to the contrary, to the extent that the Administrator determines that any payments or benefits under the Plan may not be either compliant with or exempt from Code Section 409A and related Department of Treasury guidance, the Administrator may in its sole discretion adopt such amendments to the Plan or take such other actions that the Administrator determines are necessary or appropriate to (a) exempt the compensation and benefits payable under the Plan from Code Section 409A and/or preserve the intended tax treatment of such compensation and benefits, or (b) comply with the requirements of Code Section 409A and related Department of Treasury guidance; *provided*, *however*, that this Section 6.1 shall not create any obligation on the part of the Administrator to adopt any such amendment or take any other action, nor shall the Company have any liability for failing to do so.
- 6.2 <u>Potential Six-Month Delay.</u> Notwithstanding anything to the contrary in the Plan, no amounts shall be paid to any Participant under the Plan during the six (6)-month period following such Participant's "separation from service" (within the meaning of Code Section 409A(a)(2)(A)(i) and Treasury Regulation Section 1.409A-1(h)) to the extent that the Administrator determines that paying such amounts at the time or times indicated in the Plan would result in a prohibited distribution under Code Section 409A(a)(2)(B)(i). If the payment of any such amounts is delayed as a result of the previous sentence, then on the first business day following the end of such six (6)-month period (or such earlier date upon which such amount can be paid under Code Section 409A without resulting in a prohibited distribution, including as a result of the Participant's death), the Participant shall receive payment of a lump-sum amount equal to the cumulative amount that would have otherwise been payable to the Participant during such six (6)-month period without interest thereon.

- 6.3 <u>Separation from Service</u>. A termination of employment shall not be deemed to have occurred for purposes of any provision of the Plan providing for the payment of any amounts or benefits that constitute "nonqualified deferred compensation" under Code Section 409A upon or following a termination of employment unless such termination is also a "separation from service" within the meaning of Code Section 409A and, for purposes of any such provision of the Plan, references to a "termination," "termination of employment" or like terms shall mean "separation from service".
- Reimbursements. To the extent that any payments or reimbursements provided to a Participant under the Plan are deemed to constitute compensation to the Participant to which Treasury Regulation Section 1.409A-3(i)(1)(iv) would apply, such amounts shall be paid or reimbursed reasonably promptly, but not later than December 31st of the year following the year in which the expense was incurred. The amount of any such payments eligible for reimbursement in one (1) year shall not affect the payments or expenses that are eligible for payment or reimbursement in any other taxable year, and the Participant's right to such payments or reimbursement of any such expenses shall not be subject to liquidation or exchange for any other benefit.
- 6.5 <u>Installments</u>. For purposes of applying the provisions of Code Section 409A to the Plan, each separately identified amount to which a Participant is entitled under the Plan shall be treated as a separate payment. In addition, to the extent permissible under Code Section 409A, the right to receive any installment payments under the Plan shall be treated as a right to receive a series of separate payments and, accordingly, each such installment payment shall at all times be considered a separate and distinct payment as permitted under Treasury Regulation Section 1.409A-2(b)(2)(iii). Whenever a payment under the Plan specifies a payment period with reference to a number of days, the actual date of payment within the specified period shall be within the sole discretion of the Company.

### 7. Limitation on Payments.

- Participant (whether pursuant to the terms of the Plan or any other plan, arrangement or agreement) (all such payments and benefits, including the Severance Benefits, being hereinafter referred to as the "Total Payments") would be subject (in whole or part), to the excise tax imposed under Code Section 4999 (the "Excise Tax"), then, after taking into account any reduction in the Total Payments provided by reason of Code Section 280G in any other plan, arrangement or agreement, the cash Severance Benefits under the Plan shall first be reduced, and any non-cash Severance Benefits hereunder shall thereafter be reduced, to the extent necessary so that no portion of the Total Payments is subject to the Excise Tax but only if (a) the net amount of such Total Payments, as so reduced (and after subtracting the net amount of federal, state and local income taxes on such reduced Total Payments) is greater than or equal to (b) the net amount of such Total Payments without such reduction (but after subtracting the net amount of federal, state and local income taxes on such Total Payments and the amount of Excise Tax to which the Participant would be subject in respect of such unreduced Total Payments and after taking into account the phase out of itemized deductions and personal exemptions attributable to such unreduced Total Payments and after taking into account the phase out of itemized deductions and personal exemptions attributable to such unreduced Total Payments).
- 7.2 <u>Certain Exclusions</u>. For purposes of determining whether and the extent to which the Total Payments will be subject to the Excise Tax, (a) no portion of the Total Payments, the receipt or retention of which the Participant has waived at such time and in such manner so as not to constitute a "payment" within

the meaning of Code Section 280G(b), will be taken into account; (b) no portion of the Total Payments will be taken into account which, in the written opinion of an independent, nationally recognized accounting firm (the "<u>Independent Advisors</u>") selected by the Company, does not constitute a "parachute payment" within the meaning of Code Section 280G(b)(2) (including by reason of Code Section 280G(b)(4)(A)) and, in calculating the Excise Tax, no portion of such Total Payments will be taken into account which, in the opinion of Independent Advisors, constitutes reasonable compensation for services actually rendered, within the meaning of Code Section 280G(b)(4)(B), in excess of the "base amount" (as defined in Code Section 280G(b)(3)) allocable to such reasonable compensation; and (c) the value of any non-cash benefit or any deferred payment or benefit included in the Total Payments shall be determined by the Independent Advisors in accordance with the principles of Code Sections 280G(d)(3) and (4).

8. **No Mitigation**. No Participant shall be required to seek other employment or attempt in any way to reduce or mitigate any Severance Benefits payable under the Plan and the amount of any such Severance Benefits shall not be reduced by any other compensation paid or provided to any Participant following such Participant's termination of service.

#### 9. Successors.

- 9.1 <u>Company Successors</u>. The Plan shall inure to the benefit of and shall be binding upon the Company and its successors and assigns. Any successor (whether direct or indirect and whether by purchase, lease, merger, consolidation, liquidation or otherwise) to all or substantially all of the Company's business and/or assets shall assume and agree to perform the obligations of the Company under the Plan.
- 9.2 <u>Participant Successors</u>. The Plan shall inure to the benefit of and be enforceable by each Participant's personal or legal representatives, executors, administrators, successors, heirs, distributees, devisees, legatees or other beneficiaries. If a Participant dies while any amount remains payable to such Participant hereunder, all such amounts shall be paid in accordance with the terms of the Plan to the executors, personal representatives or administrators of such Participant's estate.
- 10. **Notices**. All communications relating to matters arising under the Plan shall be in writing and shall be deemed to have been duly given when hand delivered, faxed, emailed or mailed by reputable overnight carrier or United States certified mail, return receipt requested, addressed, if to a Participant, to the address on file with the Company or to such other address as the Participant may have furnished to the other in writing in accordance herewith and, if to the Company, to such address as may be specified from time to time by the Administrator, except that notice of change of address shall be effective only upon actual receipt.

### 11. Claims Procedure; Arbitration.

11.1 <u>Claims</u>. Generally, Participants are not required to present a formal claim in order to receive benefits under the Plan. If, however, any person (the "<u>Claimant</u>") believes that benefits are being denied improperly, that the Plan is not being operated properly, that fiduciaries of the Plan have breached their duties, or that the Claimant's legal rights are being violated with respect to the Plan, the Claimant must file a formal claim, in writing, with the Administrator. This requirement applies to all claims that any Claimant has with respect to the Plan, including claims against fiduciaries and former fiduciaries, except to the extent the Administrator determines, in its sole discretion that it does not have the power to grant all relief reasonably being sought by the Claimant. A formal claim must be filed within ninety (90) days after the date the Claimant first knew or should have known of the facts on which the claim is based, unless the Administrator consents otherwise in writing. The Administrator shall provide a Claimant, on request, with a copy of the claims procedures established under Section 11.2 hereof.

11.2 <u>Claims Procedure</u>. The Administrator has adopted procedures for considering claims (which are set forth in <u>Exhibit C</u> attached hereto), which it may amend or modify from time to time, as it sees fit. These procedures shall comply with all applicable legal requirements. These procedures may provide that final and binding arbitration shall be the ultimate means of contesting a denied claim (even if the Administrator or its delegates have failed to follow the prescribed procedures with respect to the claim). The right to receive benefits under the Plan is contingent on a Claimant using the prescribed claims and arbitration procedures to resolve any claim.

### 12. Covenants.

- 12.1 <u>Restrictive Covenants</u>. A Participant's right to receive and/or retain the Severance Benefits payable under this Plan is conditioned upon and subject to the Participant's continued compliance with any restrictive covenants (e.g., confidentiality, non-solicitation, non-competition, non-disparagement) contained in any other written agreement between the Participant and the Company, as in effect on the date of the Participant's Qualifying Termination.
- 12.2 <u>Return of Property</u>. A Participant's right to receive and/or retain the Severance Benefits payable under the Plan is conditioned upon the Participant's return to the Company of all Company documents (and all copies thereof) and other Company property (in each case, whether physical, electronic or otherwise) in the Participant's possession or control.

#### 13. Miscellaneous.

- 13.1 Entire Plan; Relation to Other Agreements. The Plan, together with any Participation Notice issued in connection with the Plan, contains the entire understanding of the parties relating to the subject matter hereof and supersedes any prior agreement, arrangement and understanding between any Participant, on the one hand, and the Company and/or any subsidiary, on the other hand, with respect to the subject matter hereof. Severance Benefits payable under the Plan are not intended to duplicate any other severance benefits payable to a Participant by the Company. By participating in the Plan and accepting the Severance Benefits hereunder, the Participant acknowledges and agrees that any prior agreement, arrangement and understanding between any Participant, on the one hand, and the Company and/or any subsidiary, on the other hand, with respect to the subject matter hereof is hereby revoked and ineffective with respect to the Participant.
- 13.2 <u>Participation Notices</u>. The Administrator shall have the authority, in its sole discretion, to select Employees to participate in the Plan and to provide written notice to any such Employee that he or she is a Participant in, and eligible to receive Severance Benefits under, the Plan (a "<u>Participation Notice</u>") at or any time prior to his or her termination of employment.
- 13.3 <u>No Right to Continued Service</u>. Nothing contained in the Plan shall (a) confer upon any Participant any right to continue as an employee of the Company or any subsidiary, (b) constitute any contract of employment or agreement to continue employment for any particular period, or (c) interfere in any way with the right of the Company to terminate a service relationship with any Participant, with or without Cause.
- 13.4 <u>Termination and Amendment of Plan</u>. Prior to the consummation of a Change in Control, the Plan may be amended or terminated by the Administrator at any time and from time to time, in its sole discretion. For a period of nine (9) months from and after the consummation of a Change in Control, the Plan may not be amended, modified, suspended or terminated except with the express written consent of

each Participant who would be adversely affected by any such amendment, modification, suspension or termination. After the expiration of such nine (9)-month period, and subject to Section 2 hereof, the Plan may again be amended or terminated by the Administrator at any time and from time to time, in its sole discretion (provided, that no such amendment or termination shall adversely affect the rights of any Participant who has experienced a Qualifying Termination on or prior to such amendment or termination).

- 13.5 <u>Survival</u>. Section 7 (Limitation on Payments), Section 11 (Claims Procedure; Arbitration) and Section 12 (Covenants) hereof shall survive the termination or expiration of the Plan and shall continue in effect.
- 13.6 <u>Severance Benefit Obligations</u>. Notwithstanding anything contained herein, Severance Benefits paid or provided under the Plan may be paid or provided by the Company or any subsidiary employer, as applicable.
- 13.7 <u>Withholding</u>. The Company shall have the authority and the right to deduct and withhold an amount sufficient to satisfy federal, state, local and foreign taxes required by law to be withheld with respect to any Severance Benefits payable under the Plan.
- 13.8 <u>Benefits Not Assignable</u>. Except as otherwise provided herein or by law, no right or interest of any Participant under the Plan shall be assignable or transferable, in whole or in part, either directly or by operation of law or otherwise, including without limitation by execution, levy, garnishment, attachment, pledge or in any manner; no attempted assignment or transfer thereof shall be effective; and no right or interest of any Participant under the Plan shall be liable for, or subject to, any obligation or liability of such Participant. When a payment is due under the Plan to a Participant who is unable to care for his or her affairs, payment may be made directly to his or her legal guardian or personal representative.
- 13.9 <u>Applicable Law</u>. The Plan is intended to be an unfunded "top hat" pension plan within the meaning of U.S. Department of Labor Regulation Section 2520.104-23 and shall be interpreted, administered, and enforced as such in accordance with ERISA. To the extent that state law is applicable, the statutes and common law of the State of Delaware, excluding any that mandate the use of another jurisdiction's laws, will apply.
- 13.10 <u>Validity</u>. The invalidity or unenforceability of any provision of the Plan shall not affect the validity or enforceability of any other provision of the Plan, which shall remain in full force and effect.
- 13.11 <u>Captions</u>. The captions contained in the Plan are for convenience only and shall have no bearing on the meaning, construction or interpretation of the Plan's provisions.
  - 13.12 Expenses. The expenses of administering the Plan shall be borne by the Company or its successor, as applicable.
- 13.13 <u>Unfunded Plan</u>. The Plan shall be maintained in a manner to be considered "unfunded" for purposes of ERISA. The Company shall be required to make payments only as benefits become due and payable. No person shall have any right, other than the right of an unsecured general creditor against the Company, with respect to the benefits payable hereunder, or which may be payable hereunder, to any Participant, surviving spouse or beneficiary hereunder. If the Company, acting in its sole discretion, establishes a reserve or other fund associated with the Plan, no person shall have any right to or interest in any specific amount or asset of such reserve or fund by reason of amounts which may be payable to such person under the Plan, nor shall such person have any right to receive any payment under the Plan except as and to the extent expressly provided in the Plan. The assets in any such reserve or fund shall be part of the general assets of the Company, subject to the control of the Company.

\*\*\*\*

# CALCULATION OF NON-CHANGE IN CONTROL SEVERANCE AMOUNTS

Severance Classification	Cash Salary Severance	Severance Period
Tier 1	12 months of Base Salary	12 months
Tier 2	6 months of Base Salary	6 months
Tier 3	3 months of Base Salary	3 months

Exh. A-1

# CALCULATION OF CHANGE IN CONTROL SEVERANCE AMOUNTS

Severance Classification		Cash Salary Severance	Severance Period
Tier 1	1. 2.	12 months of Base Salary 100% Target Bonus	12 months
Tier 2	1. 2.	6 months of Base Salary 50% Target Bonus	6 months
Tier 3	1. 2.	3 months of Base Salary 25% Target Bonus	3 months

Exh. B-1

#### DETAILED CLAIMS PROCEDURES

- Section 1.1. <u>Claim Procedure</u>. Claims for benefits under the Plan shall be administered in accordance with Section 503 of ERISA and the Department of Labor Regulations thereunder. The Administrator shall have the right to delegate its duties under this Exhibit and all references to the Administrator shall be a reference to any such delegate, as well. The Administrator shall make all determinations as to the rights of any Participant, beneficiary, alternate payee or other person who makes a claim for benefits under the Plan (each, a "<u>Claimant</u>"). A Claimant may authorize a representative to act on his or her behalf with respect to any claim under the Plan. A Claimant who asserts a right to any benefit under the Plan he has not received, in whole or in part, must file a written claim with the Administrator. All written claims shall be submitted to Deborah Rodriguez, Chief Talent Officer; 2165 E Spring St., Long Beach, CA 90806; DRodriguez@hydrafacial.com.
- (a) Regular Claims Procedure. The claims procedure in this subsection (a) shall apply to all claims for Plan benefits.
- (1) <u>Timing of Denial</u>. If the Administrator denies a claim in whole or in part (an "<u>adverse benefit determination</u>"), then the Administrator will provide notice of the decision to the Claimant within a reasonable period of time, not to exceed ninety (90) days after the Administrator receives the claim, unless the Administrator determines that any extension of time for processing is required. In the event that the Administrator determines that such an extension is required, written notice of the extension will be furnished to the Claimant before the end of the initial ninety (90) day review period. The extension will not exceed a period of ninety (90) days from the end of the initial ninety (90) day period, and the extension notice will indicate the special circumstances requiring such extension of time and the date by which the Administrator expects to render the benefit decision.
- (2) <u>Denial Notice</u>. The Administrator shall provide every Claimant who is denied a claim for benefits with a written or electronic notice of its decision. The notice will set forth, in a manner to be understood by the Claimant:
  - i. the specific reason or reasons for the adverse benefit determination;
  - ii. reference to the specific Plan provisions on which the determination is based;
  - iii. a description of any additional material or information necessary for the Claimant to perfect the claim and an explanation as to why such information is necessary; and
  - iv. an explanation of the Plan's appeal procedure and the time limits applicable to such procedures, including a statement of the Claimant's right to bring an action under Section 502(a) of ERISA after receiving a final adverse benefit determination upon appeal.

- (3) <u>Appeal of Denial</u>. The Claimant may appeal an initial adverse benefit determination by submitting a written appeal to the Administrator within sixty (60) days of receiving notice of the denial of the claim. The Claimant:
  - i. may submit written comments, documents, records and other information relating to the claim for benefits;
  - ii. will be provided, upon request and without charge, reasonable access to and copies of all documents, records and other information relevant to the Claimant's claim for benefits; and
  - iii. will receive a review that takes into account all comments, documents, records and other information submitted by the Claimant relating to the appeal, without regard to whether such information was submitted or considered in the initial benefit determination.
- (4) <u>Decision on Appeal</u>. The Administrator will conduct a full and fair review of the claim and the initial adverse benefit determination. The Administrator holds regularly scheduled meetings at least quarterly. The Administrator shall make a benefit determination no later than the date of the regularly scheduled meeting that immediately follows the Plan's receipt of an appeal request, unless the appeal request is filed within thirty (30) days preceding the date of such meeting. In such case, a benefit determination may be made by no later than the date of the second (2nd) regularly scheduled meeting following the Plan's receipt of the appeal request. If special circumstances require a further extension of time for processing, a benefit determination shall be rendered no later than the third (3rd) regularly scheduled meeting of the Administrator following the Plan's receipt of the appeal request. If such an extension of time for review is required, the Administrator shall provide the Claimant with written notice of the extension, describing the special circumstances and the date as of which the benefit determination will be made, prior to the commencement of the extension. The Administrator generally cannot extend the review period any further unless the Claimant voluntarily agrees to a longer extension. The Administrator shall notify the Claimant of the benefit determination as soon as possible but not later than five (5) days after it has been made.
- (5) <u>Notice of Determination on Appeal</u>. The Administrator shall provide the Claimant with written or electronic notification of its benefit determination on review. In the case of an adverse benefit determination, the notice shall set forth, in a manner intended to be understood by the Claimant:
  - i. the specific reason or reasons for the adverse benefit determination;
  - ii. reference to the specific Plan provisions on which the adverse benefit determination is based;
  - iii. a statement that the Claimant is entitled to receive, upon request and without charge, reasonable access to, and copies of, all documents, records and other information relevant to the claim for benefits:

- iv. a statement describing any voluntary appeal procedures offered by the Plan and the Claimant's right to obtain the information about such procedures; and
- v. a statement of the Claimant's right to bring an action under Section 502(a) of ERISA.
- (c) Exhaustion; Judicial Proceedings. No action at law or in equity shall be brought to recover benefits under the Plan until the claim and appeal rights described in the Plan have been exercised and the Plan benefits requested in such appeal have been denied in whole or in part. If any judicial proceeding is undertaken to appeal the denial of a claim or bring any other action under ERISA other than a breach of fiduciary claim, the evidence presented may be strictly limited to the evidence timely presented to the Administrator. Any such judicial proceeding must be filed by the earlier of:
  (a) one (1) year after the Administrator's final decision regarding the claim appeal or (b) one (1) year after the Participant or other Claimant commenced payment of the Plan benefits at issue in the judicial proceeding.
- (d) Administrator's Decision is Binding. Benefits under the Plan shall be paid only if the Administrator decides in its sole discretion that a Claimant is entitled to them. In determining claims for benefits, the Administrator has the authority to interpret the Plan, to resolve ambiguities, to make factual determinations, and to resolve questions relating to eligibility for and amount of benefits. Subject to applicable law, any decision made in accordance with the above claims procedures is final and binding on all parties and shall be given the maximum possible deference allowed by law. A misstatement or other mistake of fact shall be corrected when it becomes known and the Administrator shall make such adjustment on account thereof as it considers equitable and practicable.

#### INDEMNIFICATION AGREEMENT

This Indemnification Agreement ("**Agreement**"), dated as of [●], 2021 is by and between The Beauty Health Company, Inc., a Delaware corporation (the "**Company**") and [●] (the "**Indemnitee**").

WHEREAS, Indemnitee is a [director/an officer] of the Company / the Company expects Indemnitee to join the Company as [a director/an officer];

WHEREAS, the board of directors of the Company (the "**Board**") has determined that enhancing the ability of the Company to retain and attract as directors and officers the most capable persons is in the best interests of the Company and that the Company therefore should seek to assure such persons that indemnification and insurance coverage is available; and

WHEREAS, in recognition of the need to provide Indemnitee with substantial protection against personal liability, in order to procure Indemnitee's service as a [director/officer] of the Company and to enhance Indemnitee's ability to serve the Company in an effective manner, and in order to provide such protection pursuant to express contract rights (intended to be enforceable irrespective of, among other things, any amendment to the Company's certificate of incorporation or bylaws (collectively, the "Constituent Documents"), any change in the composition of the Board or any change in control or business combination transaction relating to the Company), the Company wishes to provide in this Agreement for the indemnification of, and the advancement of Expenses (as defined in Section 1(f) below) to, Indemnitee as set forth in this Agreement and for the coverage of Indemnitee under the Company's directors' and officers' liability insurance policies.

NOW, THEREFORE, in consideration of the foregoing and the Indemnitee's agreement to provide services to the Company, the parties agree as follows:

- 1. <u>Definitions</u>. For purposes of this Agreement, the following terms shall have the following meanings:
- (a) "Beneficial Owner" has the meaning given to the term "beneficial owner" in Rule 13d-3 under the Securities Exchange Act of 1934, as amended (the "Exchange Act").
  - (b) "Change in Control" means the occurrence after the date of this Agreement of any of the following events:
  - (i) any Person is or becomes the Beneficial Owner, directly or indirectly, of securities of the Company representing 25% or more of the Company's then outstanding Voting Securities unless the change in relative Beneficial Ownership of the Company's securities by any Person results solely from a reduction in the aggregate number of outstanding shares of securities entitled to vote generally in the election of directors;
  - (ii) the consummation of a reorganization, merger or consolidation, unless immediately following such reorganization, merger or consolidation, all of the Beneficial Owners of the Voting Securities of the Company immediately prior to such transaction beneficially own, directly or indirectly, more than 50% of the combined voting power of the outstanding Voting Securities of the entity resulting from such transaction;
  - (iii) during any period of two consecutive years, not including any period prior to the execution of this Agreement, individuals who at the beginning of such period constituted the Board (including for this purpose any new directors whose election by the Board or nomination for election by the Company's stockholders was approved by a vote of at least two-thirds of the directors then still in office who either were directors at the beginning of the period or whose election or nomination for election was previously so approved) cease for any reason to constitute at least a majority of the Board; or

(iv) the stockholders of the Company approve a plan of complete liquidation or dissolution of the Company or an agreement for the sale or disposition by the Company of all or substantially all of the Company's assets.

#### (c) "Claim" means:

- (i) any threatened, pending or completed action, suit, proceeding or alternative dispute resolution mechanism, whether civil, criminal, administrative, arbitrative, investigative or other, and whether made pursuant to federal, state or other law; or
- (ii) any inquiry, hearing or investigation that the Indemnitee determines might lead to the institution of any such action, suit, proceeding or alternative dispute resolution mechanism.
- (d) "Delaware Court" shall have the meaning ascribed to it in Section 8(e) below.
- (e) "Disinterested Director" means a director of the Company who is not and was not a party to the Claim in respect of which indemnification is sought by Indemnitee.
- **(f)** "Expenses" means any and all expenses, including attorneys' and experts' fees, court costs, transcript costs, travel expenses, duplicating, printing and binding costs, telephone charges, and all other costs and expenses incurred in connection with investigating, defending, being a witness in or participating in (including on appeal), or preparing to defend, be a witness or participate in, any Claim. Expenses also shall include (i) Expenses incurred in connection with any appeal resulting from any Claim, including without limitation the premium, security for, and other costs relating to any cost bond, supersedeas bond, or other appeal bond or its equivalent, and (ii) for purposes of Section 4 only, Expenses incurred by Indemnitee in connection with the interpretation, enforcement or defense of Indemnitee's rights under this Agreement, by litigation or otherwise. Expenses, however, shall not include amounts paid in settlement by Indemnitee or the amount of judgments or fines against Indemnitee.
- (g) "Expense Advance" means any payment of Expenses advanced to Indemnitee by the Company pursuant to Section 3 or Section 4 hereof.
- (h) "Indemnifiable Event" means any event or occurrence, whether occurring on or after the date of this Agreement, related to the fact that Indemnitee is or was a director, officer, employee or agent of the Company or any subsidiary of the Company, or is or was serving at the request of the Company as a director, officer, employee, member, manager, trustee or agent of any other corporation, limited liability company, partnership, joint venture, trust or other entity or enterprise (collectively with the Company, "Enterprise") or by reason of an action or inaction by Indemnitee in any such capacity (whether or not serving in such capacity at the time any Loss is incurred for which indemnification can be provided under this Agreement).
- (i) "**Independent Counsel**" means a law firm, or a member of a law firm, that is experienced in matters of corporation law and neither presently performs, nor in the past five years has performed, services for either: (i) the Company or Indemnitee (other than in connection with matters concerning Indemnitee under this Agreement or of other indemnitees under similar

agreements) or (ii) any other party to the Claim giving rise to a claim for indemnification hereunder. Notwithstanding the foregoing, the term "Independent Counsel" shall not include any person who, under the applicable standards of professional conduct then prevailing, would have a conflict of interest in representing either the Company or Indemnitee in an action to determine Indemnitee's rights under this Agreement.

- (j) "Losses" means any and all Expenses, damages, losses, liabilities, judgments, fines, penalties (whether civil, criminal or other), ERISA excise taxes, amounts paid or payable in settlement, including any interest, assessments, and all other charges paid or payable in connection with investigating, defending, being a witness in or participating in (including on appeal), or preparing to defend, be a witness or participate in, any Claim
- (k) "**Person**" means any individual, corporation, firm, partnership, joint venture, limited liability company, estate, trust, business association, organization, governmental entity or other entity and includes the meaning set forth in Sections 13(d) and 14(d) of the Exchange Act.
  - (l) "Standard of Conduct Determination" shall have the meaning ascribed to it in Section 8(b) below.
  - (m) "Voting Securities" means any securities of the Company that vote generally in the election of directors.
- 2. <u>Indemnification</u>. Subject to <u>Section 8</u> and <u>Section 9</u> of this Agreement, the Company shall indemnify Indemnitee, to the fullest extent permitted by the laws of the State of Delaware in effect on the date hereof, or as such laws may from time to time hereafter be amended to increase the scope of such permitted indemnification, against any and all Losses if Indemnitee was or is or becomes a party to or participant in, or is threatened to be made a party to or participant in, any Claim by reason of or arising in part out of an Indemnifiable Event, including, without limitation, Claims brought by or in the right of the Company, Claims brought by third parties, and Claims in which the Indemnitee is solely a witness.
- 3. Advancement of Expenses. Indemnitee shall have the right to advancement by the Company, prior to the final disposition of any Claim by final adjudication to which there are no further rights of appeal, of any and all Expenses actually and reasonably paid or incurred by Indemnitee in connection with any Claim arising out of an Indemnifiable Event. Indemnitee's right to such advancement is not subject to the satisfaction of any standard of conduct. Without limiting the generality or effect of the foregoing, within 20 days after any request by Indemnitee, the Company shall, in accordance with such request, (a) pay such Expenses on behalf of Indemnitee, (b) advance to Indemnitee funds in an amount sufficient to pay such Expenses, or (c) reimburse Indemnitee for such Expenses. In connection with any request for Expense Advances, Indemnitee shall not be required to provide any documentation or information that is privileged or otherwise protected from disclosure. Indemnitee's obligation to reimburse the Company for Expense Advances shall be unsecured and no interest shall be charged thereon. Notwithstanding anything to the contrary contained herein, in the event that it is ultimately determined that Indemnitee is not to be entitled to indemnification, then all amounts advanced under this Section 3 shall be repaid, unless the court in such action, suit or proceeding shall determine that, despite Indemnitee's failure to establish its right to indemnification, Indemnitee is entitled to indemnity for such expenses. Indemnitee was frivolous or not made in good faith.
- 4. <u>Indemnification for Expenses in Enforcing Rights</u>. To the fullest extent allowable under applicable law, the Company shall also indemnify Indemnitee against, and, if requested by Indemnitee, shall advance to Indemnitee subject to and in accordance with <u>Section 3</u>, any Expenses actually and

reasonably paid or incurred by Indemnitee in connection with any action or proceeding by Indemnitee for (a) indemnification or reimbursement or advance payment of Expenses by the Company under any provision of this Agreement, or under any other agreement or provision of the Constituent Documents now or hereafter in effect relating to Claims relating to Indemnifiable Events, and/or (b) recovery under any directors' and officers' liability insurance policies maintained by the Company. However, in the event that Indemnitee is ultimately determined not to be entitled to such indemnification or insurance recovery, as the case may be, then all amounts advanced under this Section 4 shall be repaid, unless the court in such action, suit or proceeding shall determine that, despite Indemnitee's failure to establish its right to indemnification, Indemnitee is entitled to indemnity for such expenses. Indemnitee shall be required to reimburse the Company in the event that a final judicial determination is made that such action brought by Indemnitee was frivolous or not made in good faith.

5. <u>Partial Indemnity</u>. If Indemnitee is entitled under any provision of this Agreement to indemnification by the Company for a portion of any Losses in respect of a Claim related to an Indemnifiable Event but not for the total amount thereof, the Company shall nevertheless indemnify Indemnitee for the portion thereof to which Indemnitee is entitled.

#### 6. Notification and Defense of Claims.

- (a) Notification of Claims. Indemnitee shall notify the Company in writing as soon as practicable of any Claim which could relate to an Indemnifiable Event or for which Indemnitee could seek Expense Advances, including a brief description (based upon information then available to Indemnitee) of the nature of, and the facts underlying, such Claim. The failure by Indemnitee to timely notify the Company hereunder shall not relieve the Company from any liability hereunder unless the Company's ability to participate in the defense of such claim was materially and adversely affected by such failure, except that the Company shall not be liable to indemnify Indemnitee under this Agreement with respect to any judicial award in a Claim related to an Indemnifiable Event if the Company was not given a reasonable and timely opportunity to participate at its expense in the defense of such action. If at the time of the receipt of such notice, the Company has directors' and officers' liability insurance in effect under which coverage for Claims related to Indemnifiable Events is potentially available, the Company shall give prompt written notice to the applicable insurers in accordance with the procedures set forth in the applicable policies. The Company shall provide to Indemnitee a copy of such notice delivered to the applicable insurers, and copies of all subsequent correspondence between the Company and such insurers regarding the Claim, in each case substantially concurrently with the delivery or receipt thereof by the Company.
- (b) <u>Defense of Claims</u>. The Company shall be entitled to participate in the defense of any Claim relating to an Indemnifiable Event at its own expense and, except as otherwise provided below, to the extent the Company so wishes, it may assume the defense thereof with counsel reasonably satisfactory to Indemnitee. After notice from the Company to Indemnitee of its election to assume the defense of any such Claim, the Company shall not be liable to Indemnitee under this Agreement or otherwise for any Expenses subsequently directly incurred by Indemnitee in connection with Indemnitee's defense of such Claim other than reasonable costs of investigation or as otherwise provided below. Indemnitee shall have the right to employ its own legal counsel in such Claim, but all Expenses related to such counsel incurred after notice from the Company of its assumption of the defense shall be at Indemnitee's own expense; provided, however, that if (i) Indemnitee's employment of its own legal counsel has been authorized by the Company, (ii) Indemnitee has reasonably determined that there may be a conflict of interest between Indemnitee and the Company in the defense of such Claim, (iii) after a Change in Control, Indemnitee's employment of its own counsel has been approved by the Independent Counsel or (iv) the Company

shall not in fact have employed counsel to assume the defense of such Claim, then Indemnitee shall be entitled to retain its own separate counsel (but not more than one law firm plus, if applicable, local counsel in respect of any such Claim) and all Expenses related to such separate counsel shall be borne by the Company.

7. Procedure upon Application for Indemnification. In order to obtain indemnification pursuant to this Agreement, Indemnitee shall submit to the Company a written request therefor, including in such request such documentation and information as is reasonably available to Indemnitee and is reasonably necessary to determine whether and to what extent Indemnitee is entitled to indemnification following the final disposition of the Claim, provided that documentation and information need not be so provided if it is privileged or otherwise protected from disclosure. Indemnification shall be made insofar as the Company determines Indemnitee is entitled to indemnification in accordance with Section 8 below.

# 8. <u>Determination of Right to Indemnification</u>.

- (a) Mandatory Indemnification; Indemnification as a Witness.
  - (i) To the extent that Indemnitee shall have been successful on the merits or otherwise in defense of any Claim relating to an Indemnifiable Event or any portion thereof or in defense of any issue or matter therein, including without limitation dismissal without prejudice, Indemnitee shall be indemnified against all Losses relating to such Claim in accordance with Section 2 to the fullest extent allowable by law.
  - (ii) To the extent that Indemnitee's involvement in a Claim relating to an Indemnifiable Event is to prepare to serve and serve as a witness, and not as a party, the Indemnitee shall be indemnified against all Losses incurred in connection therewith to the fullest extent allowable by law.
- (b) <u>Standard of Conduct</u>. To the extent that the provisions of <u>Section 8(a)</u> are inapplicable to a Claim related to an Indemnifiable Event that shall have been finally disposed of, any determination of whether Indemnitee has satisfied any applicable standard of conduct under Delaware law that is a legally required condition to indemnification of Indemnitee hereunder against Losses relating to such Claim and any determination that Expense Advances must be repaid to the Company (a "**Standard of Conduct Determination**") shall be made as follows:
  - (i) if no Change in Control has occurred, (A) by a majority vote of the Disinterested Directors, even if less than a quorum of the Board, (B) by a committee of Disinterested Directors designated by a majority vote of the Disinterested Directors, even though less than a quorum or (C) if there are no such Disinterested Directors, by Independent Counsel in a written opinion addressed to the Board, a copy of which shall be delivered to Indemnitee; and
  - (ii) if a Change in Control shall have occurred, (A) if the Indemnitee so requests in writing, by a majority vote of the Disinterested Directors, even if less than a quorum of the Board or (B) otherwise, by Independent Counsel in a written opinion addressed to the Board, a copy of which shall be delivered to Indemnitee.

The Company shall indemnify and hold harmless Indemnitee against and, if requested by Indemnitee, shall reimburse Indemnitee for, or advance to Indemnitee, within 20 days of such request, any and all Expenses incurred by Indemnitee in cooperating with the person or persons making such Standard of Conduct Determination.

- (c) <u>Making the Standard of Conduct Determination</u>. The Company shall use its reasonable best efforts to cause any Standard of Conduct Determination required under Section <u>8(b)</u> to be made as promptly as practicable. If the person or persons designated to make the Standard of Conduct Determination under Section <u>8(b)</u> shall not have made a determination within 30 days after the later of (A) receipt by the Company of a written request from Indemnitee for indemnification pursuant to <u>Section 8</u> (the date of such receipt being the "**Notification Date**") and (B) the selection of an Independent Counsel, if such determination is to be made by Independent Counsel, then Indemnitee shall be deemed to have satisfied the applicable standard of conduct; provided that such 30 day period may be extended for a reasonable time, not to exceed an additional thirty 30 days, if the person or persons making such determination in good faith requires such additional time to obtain or evaluate information relating thereto. Notwithstanding anything in this Agreement to the contrary, no determination as to entitlement of Indemnitee to indemnification under this Agreement shall be required to be made prior to the final disposition of any Claim.
  - (d) Payment of Indemnification. If, in regard to any Losses:
    - (i) Indemnitee shall be entitled to indemnification pursuant to Section 8(a);
    - (ii) no Standard Conduct Determination is legally required as a condition to indemnification of Indemnitee hereunder; or
    - (iii) Indemnitee has been determined or deemed pursuant to Section 8(b) or Section 8(c) to have satisfied the Standard of Conduct Determination,
  - then the Company shall pay to Indemnitee, within 10 days after the later of (A) the Notification Date or (B) the earliest date on which the applicable criterion specified in clause (i), (ii) or (iii) is satisfied, an amount equal to such Losses.
- (e) Selection of Independent Counsel for Standard of Conduct Determination. If a Standard of Conduct Determination is to be made by Independent Counsel pursuant to Section 8(b)(i), the Independent Counsel shall be selected by the Board of Directors, and the Company shall give written notice to Indemnitee advising [him/her] of the identity of the Independent Counsel so selected. If a Standard of Conduct Determination is to be made by Independent Counsel pursuant to Section 8(b)(ii), the Independent Counsel shall be selected by Indemnitee, and Indemnitee shall give written notice to the Company advising it of the identity of the Independent Counsel so selected. In either case, Indemnitee or the Company, as applicable, may, within 10 days after receiving written notice of selection from the other, deliver to the other a written objection to such selection; provided, however, that such objection may be asserted only on the ground that the Independent Counsel so selected does not satisfy the criteria set forth in the definition of "Independent Counsel" in Section 1(i), and the objection shall set forth with particularity the factual basis of such assertion. Absent a proper and timely objection, the person or firm so selected shall act as Independent Counsel. If such written objection is properly and timely made and substantiated, (i) the Independent Counsel so selected may not serve as Independent Counsel unless and until such objection is withdrawn or a court has determined that such objection is without merit; and (ii) the non-objecting party may, at its option, select an alternative Independent Counsel and give written notice to the other party advising such other party of the identity of the alternative Independent Counsel so selected, in which case the provisions of the two immediately preceding sentences, the introductory clause of this sentence and numbered clause (i) of this sentence shall apply to such subsequent selection and notice. If applicable, the provisions of clause (ii) of the immediately preceding sentence shall apply to successive alternative selections. If no Independent Counsel that is permitted under the foregoing provisions of this Section 8(e) to make the Standard

of Conduct Determination shall have been selected within 20 days after the Company gives its initial notice pursuant to the first sentence of this Section 8(e) or Indemnitee gives its initial notice pursuant to the second sentence of this Section 8(e), as the case may be, either the Company or Indemnitee may petition the Court of Chancery of the State of Delaware ("Delaware Court") to resolve any objection which shall have been made by the Company or Indemnitee to the other's selection of Independent Counsel and/or to appoint as Independent Counsel a person to be selected by the Court or such other person as the Court shall designate, and the person or firm with respect to whom all objections are so resolved or the person or firm so appointed will act as Independent Counsel. In all events, the Company shall pay all of the reasonable fees and expenses of the Independent Counsel incurred in connection with the Independent Counsel's determination pursuant to Section 8(b).

### (f) Presumptions and Defenses.

- (i) <u>Indemnitee's Entitlement to Indemnification</u>. In making any Standard of Conduct Determination, the person or persons making such determination shall presume that Indemnitee has satisfied the applicable standard of conduct and is entitled to indemnification, and the Company shall have the burden of proof to overcome that presumption and establish that Indemnitee is not so entitled. Any Standard of Conduct Determination that is adverse to Indemnitee may be challenged by the Indemnitee in the Delaware Court. No determination by the Company (including by its directors or any Independent Counsel) that Indemnitee has not satisfied any applicable standard of conduct may be used as a defense to any legal proceedings brought by Indemnitee to secure indemnification or reimbursement or advance payment of Expenses by the Company hereunder or create a presumption that Indemnitee has not met any applicable standard of conduct.
- (ii) Reliance as a Safe Harbor. For purposes of this Agreement, and without creating any presumption as to a lack of good faith if the following circumstances do not exist, Indemnitee shall be deemed to have acted in good faith and in a manner he or she reasonably believed to be in or not opposed to the best interests of the Company if Indemnitee's actions or omissions to act are taken in good faith reliance upon the records of the Company, including its financial statements, or upon information, opinions, reports or statements furnished to Indemnitee by the officers or employees of the Company or any of its subsidiaries in the course of their duties, or by committees of the Board or by any other Person (including legal counsel, accountants and financial advisors) as to matters Indemnitee reasonably believes are within such other Person's professional or expert competence and who has been selected with reasonable care by or on behalf of the Company. In addition, the knowledge and/or actions, or failures to act, of any director, officer, agent or employee of the Company shall not be imputed to Indemnitee for purposes of determining the right to indemnity hereunder.
- (iii) No Other Presumptions. For purposes of this Agreement, the termination of any Claim by judgment, order, settlement (whether with or without court approval) or conviction, or upon a plea of nolo contendere or its equivalent, will not create a presumption that Indemnitee did not meet any applicable standard of conduct or have any particular belief, or that indemnification hereunder is otherwise not permitted.
- (iv) <u>Defense to Indemnification and Burden of Proof</u>. It shall be a defense to any action brought by Indemnitee against the Company to enforce this Agreement (other than an action brought to enforce a claim for Losses incurred in defending against a Claim

related to an Indemnifiable Event in advance of its final disposition) that it is not permissible under applicable law for the Company to indemnify Indemnitee for the amount claimed. In connection with any such action or any related Standard of Conduct Determination, the burden of proving such a defense or that the Indemnitee did not satisfy the applicable standard of conduct shall be on the Company.

- 9. Exclusions from Indemnification. Notwithstanding anything in this Agreement to the contrary, the Company shall not be obligated to:
- (a) indemnify or advance funds to Indemnitee for Expenses or Losses with respect to proceedings initiated by Indemnitee, including any proceedings against the Company or its directors, officers, employees or other indemnitees and not by way of defense, except:
  - (i) proceedings referenced in <u>Section 4</u> above (unless a court of competent jurisdiction determines that each of the material assertions made by Indemnitee in such proceeding was not made in good faith or was frivolous); or
    - (ii) where the Company has joined in or the Board has consented to the initiation of such proceedings.
- (b) indemnify Indemnitee if a final decision by a court of competent jurisdiction determines that such indemnification is prohibited by applicable law.
- (c) indemnify Indemnitee for the disgorgement of profits arising from the purchase or sale by Indemnitee of securities of the Company in violation of Section 16(b) of the Exchange Act, or any similar successor statute.
- (d) indemnify or advance funds to Indemnitee for Indemnitee's reimbursement to the Company of (i) any bonus or other incentive-based or equity-based compensation previously received by Indemnitee or payment of any profits realized by Indemnitee from the sale of securities of the Company, as required in each case under the Exchange Act (including any such reimbursements under Section 304 of the Sarbanes-Oxley Act of 2002 in connection with an accounting restatement of the Company or the payment to the Company of profits arising from the purchase or sale by Indemnitee of securities in violation of Section 306 of the Sarbanes-Oxley Act) and (ii) any amounts required to be repaid under the Company's Clawback Policy.
- 10. <u>Settlement of Claims</u>. The Company shall not be liable to Indemnitee under this Agreement for any amounts paid in settlement of any threatened or pending Claim related to an Indemnifiable Event effected without the Company's prior written consent, which shall not be unreasonably withheld. The Company shall not settle any Claim related to an Indemnifiable Event in any manner that would impose any Losses on the Indemnitee without the Indemnitee's prior written consent.
- 11. <u>Duration</u>. All agreements and obligations of the Company contained herein shall continue during the period that Indemnitee is a director or officer of the Company (or is serving at the request of the Company as a director, officer, employee, member, trustee or agent of another Enterprise) and shall continue thereafter (i) so long as Indemnitee may be subject to any possible Claim relating to an Indemnifiable Event (including any rights of appeal thereto) and (ii) throughout the pendency of any proceeding (including any rights of appeal thereto) commenced by Indemnitee to enforce or interpret his or her rights under this Agreement, even if, in either case, he or she may have ceased to serve in such capacity at the time of any such Claim or proceeding.

- 12. Non-Exclusivity. The rights of Indemnitee hereunder will be in addition to any other rights Indemnitee may have under the Constituent Documents, the General Corporation Law of the State of Delaware, any other contract or otherwise (collectively, "Other Indemnity Provisions"); provided, however, that (a) to the extent that Indemnitee otherwise would have any greater right to indemnification under any Other Indemnity Provision, Indemnitee will be deemed to have such greater right hereunder and (b) to the extent that any change is made to any Other Indemnity Provision which permits any greater right to indemnification than that provided under this Agreement as of the date hereof, Indemnitee will be deemed to have such greater right hereunder.
- 13. <u>Liability Insurance</u>. For the duration of Indemnitee's service as a [director/officer] of the Company, and thereafter for so long as Indemnitee shall be subject to any pending Claim relating to an Indemnifiable Event, the Company shall use commercially reasonable efforts (taking into account the scope and amount of coverage available relative to the cost thereof) to continue to maintain in effect policies of directors' and officers' liability insurance providing coverage that is at least substantially comparable in scope and amount to that provided by the Company's current policies of directors' and officers' liability insurance. In all policies of directors' and officers' liability insurance maintained by the Company, Indemnitee shall be named as an insured in such a manner as to provide Indemnitee the same rights and benefits as are provided to the most favorably insured of the Company's directors, if Indemnitee is a director, or of the Company's officers, if Indemnitee is an officer (and not a director) by such policy. Upon request, the Company will provide to Indemnitee copies of all directors' and officers' liability insurance applications, binders, policies, declarations, endorsements and other related materials.
- 14. <u>No Duplication of Payments</u>. The Company shall not be liable under this Agreement to make any payment to Indemnitee in respect of any Losses to the extent Indemnitee has otherwise received payment under any insurance policy, the Constituent Documents, Other Indemnity Provisions or otherwise of the amounts otherwise indemnifiable by the Company hereunder.
- 15. <u>Subrogation</u>. In the event of payment to Indemnitee under this Agreement, the Company shall be subrogated to the extent of such payment to all of the rights of recovery of Indemnitee. Indemnitee shall execute all papers required and shall do everything that may be necessary to secure such rights, including the execution of such documents necessary to enable the Company effectively to bring suit to enforce such rights.
- 16. Amendments. No supplement, modification or amendment of this Agreement shall be binding unless executed in writing by both of the parties hereto. No waiver of any of the provisions of this Agreement shall be binding unless in the form of a writing signed by the party against whom enforcement of the waiver is sought, and no such waiver shall operate as a waiver of any other provisions hereof (whether or not similar), nor shall such waiver constitute a continuing waiver. Except as specifically provided herein, no failure to exercise or any delay in exercising any right or remedy hereunder shall constitute a waiver thereof.
- 17. <u>Binding Effect</u>. This Agreement shall be binding upon and inure to the benefit of and be enforceable by the parties hereto and their respective successors (including any direct or indirect successor by purchase, merger, consolidation or otherwise to all or substantially all of the business and/or assets of the Company), assigns, spouses, heirs and personal and legal representatives. The Company shall require and cause any successor (whether direct or indirect by purchase, merger, consolidation or otherwise) to all, substantially all or a substantial part of the business and/or assets of the Company, by written agreement in form and substances satisfactory to Indemnitee, expressly to assume and agree to perform this Agreement in the same manner and to the same extent that the Company would be required to perform if no such succession had taken place.

- 18. <u>Severability</u>. The provisions of this Agreement shall be severable in the event that any of the provisions hereof (including any portion thereof) are held by a court of competent jurisdiction to be invalid, illegal, void or otherwise unenforceable, and the remaining provisions shall remain enforceable to the fullest extent permitted by law.
- 19. <u>Notices</u>. All notices, requests, demands and other communications hereunder shall be in writing and shall be deemed to have been duly given if delivered by hand, against receipt, or mailed, by postage prepaid, certified or registered mail:
  - (a) if to Indemnitee, to the address set forth on the signature page hereto.
  - (b) if to the Company, to: The Beauty Health Company

Attn: Paul Bokota 2165 Spring Street Long Beach, California Attention: General Counsel Email: pbokota@hydrafacial.com

Notice of change of address shall be effective only when given in accordance with this Section. All notices complying with this Section shall be deemed to have been received on the date of hand delivery or on the third business day after mailing.

- 20. Governing Law and Forum. This Agreement shall be governed by and construed and enforced in accordance with the laws of the State of Delaware applicable to contracts made and to be performed in such state without giving effect to its principles of conflicts of laws. The Company and Indemnitee hereby irrevocably and unconditionally: (a) agree that any action or proceeding arising out of or in connection with this Agreement shall be brought only in the Delaware Court and not in any other state or federal court in the United States, (b) consent to submit to the exclusive jurisdiction of the Delaware Court for purposes of any action or proceeding arising out of or in connection with this Agreement and (c) waive, and agree not to plead or make, any claim that the Delaware Court lacks venue or that any such action or proceeding brought in the Delaware Court has been brought in an improper or inconvenient forum.
- 21. <u>Headings</u>. The headings of the sections and paragraphs of this Agreement are inserted for convenience only and shall not be deemed to constitute part of this Agreement or to affect the construction or interpretation thereof.
- 22. <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, each of which shall for all purposes be deemed to be an original, but all of which together shall constitute one and the same Agreement.

[SIGNATURE PAGE FOLLOWS]

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the date first above written.

By: Name: Title:	
INDEM	NITEE
By: Name: Address:	

THE BEAUTY HEALTH COMPANY

Signature Page to Indemnification Agreement

#### AMENDED AND RESTATED MANAGEMENT SERVICES AGREEMENT

THIS AMENDED AND RESTATED MANAGEMENT SERVICES AGREEMENT (this "<u>Agreement</u>") is made as of May 4, 2021 (the "<u>Effective Date</u>") between Edge Systems, LLC, a California limited liability company (the "<u>Company</u>"), The Beauty Health Company (f/k/a Vesper Healthcare Acquisition Corp.), a Delaware corporation ("<u>New Parent</u>") and Linden Manager III LP, a Delaware limited partnership ("<u>Linden</u>"). Certain defined terms shall have the meanings set forth in <u>Section 11</u>.

WHEREAS, the Company and Linden entered into that certain Management Services Agreement, dated as of December 1, 2016 (the "Original Agreement") pursuant to which Linden provided certain financial and management advisory services to the Company;

WHEREAS, the Company is a wholly-owned subsidiary of LCP Edge Intermediate, Inc., a Delaware corporation

("Intermediate");

WHEREAS, Intermediate, New Parent, Hydrate Merger Sub I, Inc., a Delaware corporation, Hydrate Merger Sub II, LLC, a Delaware limited liability company and LCP Edge Holdco, LLC, a Delaware limited liability company entered into that certain Agreement and Plan of Merger, dated as of December 8, 2020 (the "Merger Agreement"), pursuant to which, upon the consummation of the transactions contemplated thereby, New Parent will acquire 100% of the equity interests of Intermediate; and

WHEREAS, the Company, New Parent and Linden desire to amend and restate the Original Agreement in its entirety to make certain revisions to the Original Agreement;

NOW, THEREFORE, in consideration of the foregoing premises, the respective agreements hereinafter set forth and the mutual benefits to be derived therefrom, Linden, New Parent and the Company hereby agree as follows:

- 1. <u>Effective Time</u>. This Agreement shall be effective as of the Second Effective Time, as defined in the Merger Agreement.
- 2. <u>Engagement</u>. The Company hereby engages Linden as a financial and management advisor, and Linden hereby agrees to provide certain financial and management advisory services to the Company, all on the terms and subject to the conditions set forth in this Agreement.
- 3. <u>Services</u>. During the Term, Linden agrees, if and as requested by New Parent from time to time, to consult with and advise the boards of managers (or equivalent) of the Company, New Parent and their respective subsidiaries and the management of New Parent and its subsidiaries in the identification, analysis, support and negotiation of acquisitions, and such other related services as may be reasonably requested from time to time and as may be reasonably agreed to by Linden (collectively, the "<u>Requested Services</u>").
- 4. <u>Personnel</u>. Linden shall provide and devote to the performance of this Agreement such of its partners and employees as Linden shall deem appropriate in its discretion for the furnishing of the services to be provided hereunder.
- 5. Transaction Fees. As compensation for the services to be provided by Linden to New Parent and its subsidiaries hereunder, the Company shall pay to Linden a fee upon the consummation of an Add-On Acquisition (as defined herein). When and as the Company, New Parent or any of their respective subsidiaries consummates a transaction or series of related transactions (i) in respect of which Linden has performed Requested Services and (ii) involving the acquisition of at least 50% of the equity or all or substantially all of the assets of a business, company, product line or enterprise, whether directly or indirectly and through any form of transaction, including, without limitation, merger, reverse merger, liquidation, stock sale, asset sale, asset swap, consolidation, amalgamation, spin-off, split-off or other transaction (such transaction or series of related transactions, an "Add-On Acquisition"), the Company shall pay to Linden and/or its designees a fee equal to 1.0% of the Acquisition Price of such Add-On Acquisition (the "Acquisition Fee"). The Acquisition Fee payable to Linden hereunder upon the consummation of an Add-On Acquisition shall be payable in full, in cash, immediately after and conditioned upon the closing of such Add-On Acquisition. Notwithstanding the foregoing, no transaction fee will be payable to Linden in connection with the consummation of any acquisition that constitutes a Triggering Event (as such term is defined in the Merger Agreement).

- 6. Expenses. The Company shall promptly reimburse Linden for such reasonable travel expenses and other reasonable fees and expenses as may be incurred by Linden in connection with the rendering of services hereunder (including reasonable fees and expenses of its counsel and any other independent experts retained by Linden). In addition, the Company shall pay and hold Linden harmless against liability for the payment of the expenses (including the fees and expenses of legal counsel or other advisors) arising in connection with (a) any proposed or completed Add-On Acquisition or other transaction involving the Company or any of its subsidiaries, (b) any amendments or waivers (whether or not the same become effective) under or in respect of this Agreement or the other agreements contemplated hereby (including, without limitation, in connection with any proposed Add-On Acquisition) and (c) the interpretation, investigation and enforcement of the rights granted under this Agreement or the other agreements contemplated hereby.
- 7. Term. This Agreement shall be in effect for a term (the "Term") commencing on the Effective Date and terminating on the earliest to occur of (a) the first anniversary of the Effective Date, (b) termination of this Agreement by Linden by providing five days' advance written notice to the Company and (c) such other time as mutually agreed by the Company and Linden in writing; provided, that, except as otherwise provided in this Section 7, no termination of this Agreement shall affect the Company's obligations with respect to the fees, costs and expenses incurred by Linden in rendering services hereunder and not reimbursed or otherwise paid by the Company as of the date of such termination, and the Company shall pay to Linden all such unpaid fees, costs and expenses prior to or in connection with any such termination. In the event of the termination of this Agreement pursuant to clauses (a) or (c) of the first sentence of this Section 7, Linden shall be entitled to the Acquisition Fee as set forth in Section 5 if, on or prior to the effective date of such termination of this Agreement, the Company, New Parent or any of their respective subsidiaries consummates, or enters into a definitive purchase agreement which subsequently results in, an Add-On Acquisition. Any such Acquisition Fee shall be payable upon the closing of any such Add-On Acquisition. In the event of the termination of this Agreement pursuant to clause (b) of the first sentence of this Section 7, Linden will have no further right to receive any Acquisition Fee hereunder effective upon such termination.
- 8. <u>Liability</u>. Neither Linden nor any of its direct or indirect affiliates, partners, members, employees or agents shall be liable to the Company or its subsidiaries or affiliates for any loss, liability, damage or expense arising out of or in connection with the performance of services contemplated by this Agreement, unless such loss, liability, damage or expense shall be proven to result directly from Linden's gross negligence or willful misconduct. Linden makes no representation or warranty, express or implied, in respect of the services to be provided by it hereunder. No person who is employed with, a member or partner of, or a consultant to any entity affiliated with Linden (other than the Company or any of its subsidiaries) shall have any duty or obligation to bring any "corporate opportunity" to the Company.
- 9. <u>Indemnification</u>. The Company agrees to defend, protect, indemnify and hold harmless Linden, its partners, members, affiliates, officers, managers, directors, employees, agents, and any other persons controlling Linden or any of its affiliates, to the fullest extent lawful, from and against any and all actions, causes of action, losses, liabilities, suits, claims, costs, damages, penalties, fees and expenses (irrespective of whether any such indemnitee is a party to the action for which indemnification hereunder is sought), including reasonable attorneys' fees and disbursements, incurred by the indemnitees or any of them as a result of, arising from or relating to (i) this Agreement and any performance hereunder or (ii) any transaction to which the Company or any of its subsidiaries is a party or any other circumstance, event or development with respect to the Company or any of its subsidiaries, in each case except as a result of Linden's gross negligence or willful misconduct (other than an action or failure to act undertaken at the request or with the consent of New Parent or the Company).
- 10. <u>Independent Contractor</u>. Linden shall perform services hereunder as an independent contractor, retaining control over and responsibility for its own operations and personnel. Neither Linden nor its directors, officers, partners, members or employees shall be considered employees or agents of the Company as a result of this Agreement, nor shall any of them have authority to contract in the name of or bind the Company, except as otherwise expressly agreed.

### 11. <u>Definitions</u>.

"Acquisition Price" means, with respect to a given Add-On Acquisition, the fair value (as jointly determined by New Parent, the Company and Linden) of the total sale proceeds and other consideration paid or to be paid to the target company and its equityholders (or otherwise paid on their behalf, including, without limitation, transaction related expenses) upon consummation of such Add-On Acquisition, including cash, securities, notes, consulting agreement payments, noncompete agreement payments, contingent payments (including earnouts) and escrow amounts, <u>plus</u> the fair value of all liabilities assumed and all funded debt refinanced or paid off.

12. <u>Notices</u>. All notices, demands and other communications to be given or delivered under or by reason of the provisions of this Agreement shall be in writing and shall be deemed to have been given (a) when personally delivered or sent by e-mail or comparable electronic transmission, (b) one business day following the day when deposited with a reputable and established overnight express courier (charges prepaid) or (c) five days following mailing by certified or registered mail, postage prepaid and return receipt requested. Unless another address is specified in writing, notices, demands and communications to the Company, New Parent and Linden shall be sent to the addresses indicated below (or at such other address as shall be given in writing by one party to the other):

If to Linden: c/o Linden LLC 150 North Riverside Plaza, Suite 5100 Chicago, Illinois 60606 Attn: Brian C. Miller and Kam Shah

Email: bmiller@lindenllc.com and kshah@lindenllc.com

If to the Company and New Parent:

2165 Spring St. Long Beach, CA, 90806 Attn: Executive Chairman

Email: Brent.saunders@vesperhealth.com

- 13. Entire Agreement; Modification. This Agreement and those documents expressly referred to herein contain the complete and entire understanding and agreement of Linden, the Company and New Parent with respect to the subject matter hereof and supersede all prior and contemporaneous understandings, conditions and agreements, oral or written, express or implied, respecting the engagement of Linden in connection with the subject matter hereof. The provisions of this Agreement may be amended, modified and/or waived only with the prior written consent of the Company, New Parent and Linden.
- 14. <u>Waiver of Breach</u>. The waiver by either party of a breach of any provision of this Agreement by the other party shall not operate or be construed as a waiver of any subsequent breach of that provision or any other provision hereof.
- 15. <u>Assignment</u>. Neither Linden, the Company nor New Parent may assign its rights or obligations under this Agreement without the express written consent of the other, except that Linden may assign its rights and obligations to any of its affiliates.
- 16. <u>Successors</u>. This Agreement and all the obligations and benefits hereunder shall bind and inure to the respective successors and permitted assigns of the parties.
- 17. <u>Counterparts</u>. This Agreement may be executed and delivered by each party hereto in separate counterparts (including by means of .pdf or other electronic form), each of which when so executed and delivered shall be deemed an original and both of which taken together shall constitute one and the same agreement.
- 18. <u>Choice of Law</u>. This Agreement shall be governed by and construed in accordance with the domestic laws of the State of Delaware, without giving effect to any choice of law or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the State of Delaware.
- 19. Delivery by E-mail or Other Electronic Transmission. This Agreement, the agreements referred to herein and each other agreement or instrument entered into in connection herewith or therewith, or contemplated hereby or thereby, and any amendments hereto or thereto, to the extent signed and delivered by means of e-mail or other electronic transmission shall be treated in all manner and respects as an original agreement or instrument and shall be considered to have the same binding legal effect as if it were the original signed version thereof delivered in person. At the request of any party hereto or to any other such agreement or instrument, each other party hereto or thereto shall reexecute original forms thereof and deliver them to all other parties. No party hereto or to any other such agreement or instrument shall raise the use of e-mail or other electronic transmission to deliver a signature or the fact that any signature or agreement or instrument was transmitted or communicated through the use of a e-mail or comparable electronic transmission as a defense to the formation or enforceability of a contract and each such party forever waives any such defense.

20. MUTUAL WAIVER OF JURY TRIAL. BECAUSE DISPUTES ARISING IN CONNECTION WITH COMPLEX TRANSACTIONS ARE MOST QUICKLY AND ECONOMICALLY RESOLVED BY AN EXPERIENCED AND EXPERT PERSON AND THE PARTIES WISH APPLICABLE STATE AND FEDERAL LAWS TO APPLY (RATHER THAN ARBITRATION RULES), THE PARTIES DESIRE THAT THEIR DISPUTES BE RESOLVED BY A JUDGE APPLYING SUCH APPLICABLE LAWS. THEREFORE, TO ACHIEVE THE BEST COMBINATION OF THE BENEFITS OF THE JUDICIAL SYSTEM AND OF ARBITRATION, EACH PARTY TO THIS AGREEMENT HEREBY WAIVES ALL RIGHTS TO TRIAL BY JURY IN ANY ACTION, SUIT OR PROCEEDING BROUGHT TO RESOLVE ANY DISPUTE BETWEEN THE PARTIES HERETO, WHETHER ARISING IN CONTRACT, TORT OR OTHERWISE, ARISING OUT OF, CONNECTED WITH, RELATED OR INCIDENTAL TO THIS AGREEMENT, THE TRANSACTIONS CONTEMPLATED HEREBY AND/OR THE RELATIONSHIP ESTABLISHED AMONG THE PARTIES HEREUNDER.

IN WITNESS WHEREOF, the undersigned have caused this Amended and Restated Management Services Agreement to be duly executed and delivered as of the date first above written.

LINDEN MANAGER III LP

By: Linden Capital III, LLC

Its: General Partner

By: /s/ Brian C. Miller

Name: Brian C. Miller Title: Authorized Signatory

Signature page to A&R Management Services Agreement

EDGE SYSTEMS, LLC

By: /s/ Kamlesh Shah

Name: Kamlesh Shah

Its: Vice President, Assistant Treasurer and Assistant Secretary

THE BEAUTY HEALTH COMPANY

By: <u>/s/ Brenton L. Saunders</u>
Name: Brenton L. Saunders

Its: Executive Chairman

Signature page to A&R Management Services Agreement

# **SUBSIDIARIES**

Below is a list of our major subsidiaries as of May 4, 2021, their jurisdictions of incorporation or formation, their ownership and the name under which they do business.

Subsidiary	Jurisdiction	Owner
LCP Edge Intermediate, LLC	Delaware	The Beauty Health Company
Edge Systems Holdings Corporation	Delaware	LCP Edge Intermediate, LLC
Edge Systems Intermediate, LLC	Delaware	Edge Systems Holdings Corporation
Edge Systems, LLC1	California	Edge Systems Intermediate, LLC
The HydraFacial Company Mexico Holdings, LLC	Delaware	Edge Systems, LLC
The HydraFacial Company MX, S. de R.L. de C.V.	Mexico	Edge Systems, LLC (99%)
		The HydraFacial Company Mexico Holdings, LLC (1%)
HydraFacial Trading (Shanghai) Co., Ltd.	China	Edge Systems, LLC
The Hydrafacial Company Iberia, S.L.U	Spain	Edge Systems, LLC
The Hydrafacial Company Japan K.K.	Japan	Edge Systems, LLC
Hydrafacial UK Limited	UK	Edge Systems, LLC

Edge Systems LLC does business as The HydraFacial Company.